

pubs.acs.org/CR



Nanoporous Gold: From Structure Evolution to Functional Properties in Catalysis and Electrochemistry

Gunther Wittstock,* Marcus Bäumer, Wilke Dononelli, Thorsten Klüner, Lukas Lührs, Christoph Mahr, Lyudmila V. Moskaleva, Mehtap Oezaslan, Thomas Risse, Andreas Rosenauer, Anne Staubitz, Jörg Weissmüller, and Arne Wittstock



nanometer-sized metallic struts and interconnected pores formed spontaneously by oxidative dissolution of the less noble element from gold alloys. The resulting material exhibits decent catalytic activity for low-temperature, aerobic total as well as partial oxidation reactions, the oxidative coupling of methanol to methyl formate being the prototypical example. This review not only provides a critical discussion of ways to tune the morphology and composition of this material and its implication for catalysis and electrocatalysis, but will also exemplarily review the current mechanistic understanding of the partial oxidation of



methanol using information from quantum chemical studies, model studies on single-crystal surfaces, gas phase catalysis, aerobic liquid phase oxidation, and electrocatalysis. In this respect, a particular focus will be on mechanistic aspects not well understood, yet. Apart from the mechanistic aspects of catalysis, best practice examples with respect to material preparation and characterization will be discussed. These can improve the reproducibility of the materials property such as the catalytic activity and selectivity as well as the scope of reactions being identified as the main challenges for a broader application of NPG in target-oriented organic synthesis.

CONTENTS

1. Introduction	6717
2. Tuning of Structure and Composition during	
Dealloying	6718
2.1. Nanoporous Gold by Dealloying	6718
2.2. Overview on Phenomenology and Pro-	
cesses during Dealloying	6719
2.3. Approaches to Dealloying	6720
2.4. Preparation Protocols for Nanoporous Gold:	
A Case Study	6721
2.4.1. Motivation	6721
2.4.2. Master Alloy Selection and Preparation	6721
2.4.3. Potentiostatic or Galvanostatic Deal-	
loying	6722
2.4.4. Choice of Electrolyte	6722
2.4.5. Dealloying Potential and Postdealloy-	
ing Conditioning	6722
2.4.6. Drying	6723
2.5. Hierarchical Nanoporous Gold	6723
2.6. Engineering the Macroscopic Shape of	
Nanoporous Gold	6724
2.6.1. Machining the Master Alloy Preform	6724
2.6.2. 3D Printing the Master Alloy	6724
2.6.3. Nanoporous Thin Layers and Nano-	
porous Nanowires	6724
2.7. Nanoporous Powders	6725
2.7.1. Making Powder from Bulk NPG	6725
2.7.2. Processing NPG Powders	6725

2.8. Dealloyed Porous Nanoparticles	6725
2.9. Other Ways to Meso- or Macroporous Gold	
Electrodes	6727
2.9.1. Direct Electrodeposition of Porous Gold	
Electrodes	6727
2.9.2. Assembly of Nanoparticles to Porous	
Gold	6727
2.9.3. Coating of Au on Other Porous Electro-	
des	6727
3. Microstructure and Metrics for its Description	6727
3.1. Overview	6727
3.2. Geometric Model for the Nanoporous Gold	
Microstructure	6728
3.3. Metrics for the Characteristic Length Scale	6729
3.4. Specific Area of Surface	6729
3.5. Comparison of Different Measures for	
Ligament Size	6731
3.6. Grain Size	6731
3.7. Phase Fractions	6732

Received: October 26, 2022 Published: May 3, 2023



3.8. Skin Layers on Macro- and Nanoscale Porous Bodies and Disconnection of Thin	
Films 3.9. Connectivity, Percolation-to-Cluster Transi-	6732
tion, and Self-Similarity 3.10. Tortuosity, Multiscale, and Hierarchical	6733
Nanoporous Gold 3.11. Surface Curvature, Faceting, and Rough-	6733
ness	6733
3.12. Mean Lattice Parameter and Microstrain	6734
3.13. Chemical Heterogeneity	6734
3.14. 3D Representations	6737
4. Gas Phase Catalysis	6737
4.1. Surface Composition, Active Sites, and Reaction Mechanisms from Computational	
Studies	6738
4.1.1. Aerobic Oxidation Reactions and O_2	
Activation	6738
4.1.2. Water as a Co-catalyst in CO and	6741
4.1.3. Detailed Mechanism of Aerobic Alcohol	0741
Oxidation from Computational Studies	6742
4.2. Insights from Single-Crystal Surfaces	6743
4.2.1. Nature of Activated Oxygen Species	6743
4.2.2. CO Oxidation	6744
4.2.3. The Role of Water in CO Oxidation	6745
4.2.4. Methanol Oxidation	6746
4.3. Gas Phase Catalysis on Nanoporous Gold	6747
4.3.1. Impact of Oxygen Heterogeneity on	
Catalytic Properties	6748
4.3.2. Macro- and Microkinetics: From Trans-	
port Properties to Reaction Mechanism	6749
4.4. Evolution of the Material during Catalysis	6/50
5. Liquid Phase Aerobic Oxidation of Methanol	6751
5.1. Challenges for Working in the Liquid Phase	6751
5.2. Comparison of Mechanistic Aspects in the	
Aerobic Oxidation of Methanol in Liquid	6750
and Gas Phases	6/52
5.3. Influence of Residual Ag	6/53
5.4. Roles of Acid and Base, and Activation	(7FA
Period of the Catalyst	6754
6.1 Comparison of Nanonorous Cold to other	0754
Cold Electrodec	6751
6.1.1 Comparison of Surface Voltammetry of	0754
Au Electrodes	6755
612 Tuning of Internal Surface Structure as	0/55
Followed by Metal Underpotential	
Deposition/Stripping	6756
6.1.3. Model Reactions that Show Differences	
to Planar Au Electrodes	6756
6.1.4. Structure Evolution of NPG during Use	
as Electrode Material	6757
6.2. Electrocatalysis of Model Reactions	6758
6.2.1. Hydrogen Evolution Reaction	6758
6.2.2. Oxygen Reduction Reaction and Other	
Oxygen Redox Chemistry	6758
6.2.3. Alcohol Oxidation Reaction	6759
6.2.4. Oxidation of Aldehydes and Carbonic	
Acids	6762
6.2.5. CO ₂ Reduction Reaction	6762
6.3. Surface Modification for Improvement of	
Electrode Properties	6763

6.4. Electrochemical Sensors based on Nano-		
porous Gold	6763	
6.4.1. Inorganic lons and Compounds	6763	
6.4.2. Nonenzymatic Glucose Sensing		
6.4.3. Detection of Other Organic Analytes	6765	
6.4.4. Nanoporous Gold as Support for Bio-		
molecules	6765	
6.4.5. Electrodes in Biomedical Devices	6765	
6.5. Nanoporous Gold in Electrochemical Energy		
Conversion Devices	6766	
6.5.1. Nanoporous Gold in Fuel Cells	6766	
6.5.2. Nanoporous Gold in Batteries	6766	
6.5.3. Nanoporous Gold in Supercapacitors	6766	
7. Catalysis by Nanoporous Gold in Organic Syn-		
thesis	6767	
7.1. Oxidation Reactions of Organic Molecules		
by Nanoporous Gold	6767	
7.2. Comproprotionation Reactions	6768	
7.3. Reduction Reactions by Nanoporous Gold	6769	
8. Conclusion	6771	
9. Outlook	6773	
9.1. Need for Establishing "Good Practice"		
Procedures for Catalytic Testing	6773	
9.2. Widening the Compositional Space of NPG	6773	
9.3. Need for Understanding the Impact of		
Dealloying on the Defect Structure	6773	
9.4. LNE-Free NPG as a Reference Material	6773	
9.5. Need for Scalable Production Protocols	6773	
9.6. Need for Appreciating Postdealloying Pro-		
cessing and Conditioning	6774	
9.7. NPG with Engineered Macroscale Morphol-		
ogies	6774	
Associated Content	6774	
Supporting Information	6774	
Author Information	6774	
Corresponding Author	6774	
Authors	6774	
Author Contributions	6774	
Notes	6775	
Biographies	6775	
Acknowledgments	6776	
Appreviations	6//6	
Keterences	6776	

Review

1. INTRODUCTION

pubs.acs.org/CR

Nanostructuring is one of the most successful strategies to prepare materials with new or improved chemical properties. Catalysis, being one of the core technologies of chemical industry and being involved in about 80-90% of all industrial processes, is closely linked to the current quest to reduce our energy and CO₂ footprint. Heterogeneous catalysis has used nanostructuring early on as seen by the large fraction of heterogeneous catalysts which are nanostructured materials. Heterogeneous gold catalysts are a prototypical example for appearance of new chemical functionality upon nanostructuring. In particular, gold nanoparticles grafted on a suitable oxide support were found to be good catalysts for a range of low temperature oxidation reactions back in the 1980s of the last century,¹⁻⁵ while macroscopic gold is inactive. The size (maximum activity for CO oxidation is found at a diameter of around 5 nm) as well as the support of the Au nanoparticles were identified as crucial parameters for the catalytic activity.

Hence, the first reports about a high catalytic activity of nanoporous gold (NPG) for CO oxidation in 2006/2007 by some of the authors⁶ as well as by Ding and co-workers⁷ came as a surprise because NPG not only lacks a support but also exhibits characteristic length scales (ligament and pore sizes) that are about an order of magnitude larger than the size of active Au nanoparticles. The interest into NPG as a catalytic material was further enhanced by the first report on its potential as a highly selective partial oxidation catalyst as exemplified by the aerobic oxidation of methanol to methyl formate.⁸

NPG is prepared by dealloying, a process resulting in the spontaneous, self-organized creation of a nano- or microscale, bicontinuous network structure (continuous in both the material and the void space) from a previously homogeneous solid by selective removal of the less noble component.⁹⁻¹² The pore sizes typically found in NPG (from about 10 to a few 100 nm) mostly fall in the range that is classified as mesopores comprising pore sizes of 2-50 nm according to IUPAC.¹³ Nonetheless, the term "nanoporous gold" is almost¹⁴ universally used for the material^{7,8,10,15–18} and sometimes extended even to other types of porous gold.¹⁹⁻²² Modern approaches to use dealloying as a material's preparation method were pioneered by Sieradzki,¹⁴ and the class of nanoporous metals gained additional widespread interest by conceptual work of Erlebacher and co-workers,¹⁰ who not only linked the dealloying process to modern concepts of pattern formation in nonequilibrium processes but also effectively popularized the visual appeal of the resulting structures. One of the most attractive properties of nanoporous materials in general (and NPG in particular) is the outstanding definition, uniformity, and reproducibility of its nanoscale structure. Moreover, the possibility to control the characteristic pore and ligament sizes, the ability to modify the surface properties, and the lack of support effects render this material particularly well suited as a model system for understanding size and interface effects at the nanoscale in several fields, including mechanics,^{9,15,23–28} plasmonics,^{29–32} or (electro-)catalysis.^{8,17,24,33–44}

Dealloying of AgAu alloys typically results in NPG, which predominantly consists of Au with only a small fraction (a few %) of the remaining less noble element (LNE; i.e., Ag in this case). As NPG is predominantly consisting of Au, it was suggested that its catalytic properties are largely based on the surface chemistry of gold.8 This notion was substantiated by comparing the oxidation chemistry on Au and Ag single-crystal surfaces studied under ultrahigh vacuum (UHV) by the Madix group $^{45-47}$ with the aerobic catalytic oxidation on NPG.⁸ The insight that the oxidation chemistry of NPG is mainly due to Au^{18,48} was the basis for variety of experiments on Au singlecrystal surfaces as well as theoretical studies. These studies have contributed significantly to the current microscopic understanding, including aspects such as the mechanism of oxidation reactions,^{49,50} the importance of the less noble element (LNE), mostly Ag, for oxidative catalysis,^{51,52} the activation of molecular oxygen,⁵³⁻⁵⁵ or the role of water.⁵⁶ With respect to catalysis, it is particularly appealing that NPG can be used for gas phase,^{57,58} liquid phase,⁵⁹⁻⁶¹ as well as electrocatalysis^{62,63} for the same reaction (e.g., partial oxidation of methanol), which allows elucidation of similarities and differences between the different scenarios.

A number of reviews are available on different aspects of NPG and its application, including structure evolution,⁶⁴ mechanical properties²⁷ and actuation,⁶⁵ plasmonics,³² sens-

ing,^{66,67} biomedical interfaces,^{28,68,69} energy conversion,⁷⁰ or catalysis.^{17,39,41,64,71,72} We will review the recent developments with respect to oxidation catalysis, which aim at identifying decisive properties influencing catalytic performance and the ability to alter them by tuning these properties. As the catalytic properties are intimately linked to the structural and chemical properties of the materials, we set out by discussing preparation strategies for NPG, the measures of its nanostructure, and the development of these properties during catalysis. Detailed knowledge of the morphological properties is important for catalysis as mass transport of the reagents is an important aspect to be considered not only in liquid but also in gas phase catalysis. Hence, we will not only briefly introduce the conceptual framework (tortuosity) to describe diffusive transport in a nanoporous network, but will also discuss experimental approaches to quantify these aspects and show their impact on the catalytic performance. The microscopic understanding of the reaction mechanism in gas phase catalysis being a crucial ingredient to understand the catalytic properties of NPG is based on UHV experiments on single-crystal surfaces and to a large part on theory. Emphasis will be put on comparing the mechanistic pictures in gas phase catalysis evolving from the theoretical calculations with various experimental approaches on a system with systematically increasing complexity ranging from single-crystalline model surfaces to NPG. The catalysis at the gas/solid interface will be compared to results in liquid phase as well as electrocatalysis, furthermore highlighting current developments which afforded microscopic insight into the properties decisive for catalysis under these conditions.

2. TUNING OF STRUCTURE AND COMPOSITION DURING DEALLOYING

2.1. Nanoporous Gold by Dealloying

The notion of "dealloying" refers to a family of processes in which one chemical element is selectively removed from a solid solution or compound. The removal of this sacrificial element can exploit various mechanisms, for instance, evaporation at elevated temperature,^{73,74} dissolution in a molten metal,^{12,75,76} or corrosive attack in acid or aqueous electrolyte. 10,14,15,77 An important commonality is that dealloying converts the uniform and massive (not porous) initial crystal into a nanoporous product crystal; the characteristic size of the pores represents a new length scale that is generated by nanoscale selforganization processes. Although experimental preparation protocols differ greatly among the dealloying variants, the atomic-scale processes that drive the nanostructure formation are quite similar.¹² In each case, they rely on a competition between the active process of dissolution of the sacrificial element and a passivating process mediated by diffusive rearrangement of the conserved element.^{78,7}

The present section focuses primarily on nanoporous gold (NPG) made by dealloying, emphasizing protocols for preparing macroscopic volumes of the material in clean and uniform quality (section 2.4). Why a focus on macroscopic volumes? First, it is a distinguishing feature of NPG that one and the same material provides—at the atomic scale—the catalytically active sites and—at the meso- or macroscale—the strong scaffold structure, which supplies and stabilizes the pore channels that bring the large area of surface and that enable access to its active sites. Second, macroscale samples of nanoporous gold have been the subject of intense studies with

respect to their mechanical behavior. This behavior is not only important for the stability of the scaffold, it is also extremely sensitive to heterogeneity. Therefore, the excellent agreement of independent studies of mechanical behavior by several groups²⁷ provides a signature of exceptional uniformity and reproducibility in NPG preparation. This distinguishes the underlying protocols as a unique basis for meaningful model studies of NPG in any field, and specifically including studies of catalysis.

For catalysis, the absence of an extrinsic scaffold material endows monolithic bulk samples of NPG with a unique conceptual simplicity. Yet, long transport pathways through a confined pore space can impair the catalytic conversion rates. This section also addresses four approaches to preparing samples that can mitigate this issue. First, monolithic NPG with a hierarchical pore structure (section 2.5) combines large pores for accelerated transport with small pores for high specific surface area and function. Second, cavity microelectrodes filled with μ m-scale particles of powdered NPG (section 2.7) combine large open pathways (between the powder particles) with small active volume for enhanced kinetics and equilibration. Third, a similar effect is achieved with substrate-supported thin films of NPG (section 2.6), which have at least one external sample dimension in the range of 1 μ m or below. Fourth and last, NPG nanoparticles (section 2.8) have an external size in the order of 100 nm or below in all three dimensions. Such particles can be employed in classic catalysis scheme, with particles attached to an extrinsic support scaffold, yet they profit from the enhanced specific surface area and activity of NPG.

2.2. Overview on Phenomenology and Processes during Dealloying

Preparation of the uniform, load bearing, monolithic macroscopic NPG that forms the focus of this section has been demonstrated with dealloying in aqueous media and near room temperature, in the form of either free or electrochemical corrosion. Here, bulk diffusion is exceptionally slow and can be neglected.⁸⁰ The nanostructure formation is then simply the result of the two elementary processes, dissolution and surface diffusion. The related phenomenology has been investigated for more than a century in metallurgy, specifically in the contexts of alloy corrosion and stress corrosion cracking. The reader is referred to related review articles for details, and the following, incomplete list may serve as a starting point: $^{9,11,81-86}$ The simplest concepts are that (i) the mole fraction x_{LNE} (also referd to as "atom fraction") of the sacrificial element (here the less noble element, LNE) needs to exceed a minimum value, the parting limit, $x_{\rm P}$, and that (ii) the dealloying potential, $E_{\rm D}$, (i.e., the electrode potential applied during electrochemical dealloying) needs to exceed the critical dealloying potential, $E_{\rm C}$. The parting limit is related to the requirement for a percolating cluster of the LNE in the master alloy,⁸⁷ and the critical potential is related to the requirement that the corrosion is driven sufficiently fast to proceed into the depth before the lateral rearrangement of the more noble element (MNE) by surface diffusion can form a closed layer of pure MNE that passivates the corrosion surface.^{88,89}

The current understanding of the atomic-scale processes behind the above phenomena rests largely on kinetic Monte Carlo (KMC) simulation studies.^{10,87-91} Working with a minimal and physically motivated set of materials parameters, this approach reproduces many of the key experimental observations, thereby validating the underlying model. Early work suggested that a spinodal decomposition process in an adatom gas of the more noble element on the surface was responsible for the characteristic length scale of the nanoporosity.¹⁰ Yet, more detailed inspection of the KMC results identifies the nucleation of vacancy islands on crystal terraces, followed by dissolution from receding step edges, as a more appropriate scenario.⁸⁹ Figure 1A illustrates the roughened



Figure 1. Porosity evolution during dealloying modeled by kinetic Monte Carlo simulation. Primary dealloying (A-C) starts with surface roughening and formation of passivated mounds (A). Small ligaments form by undercutting and the corrosion front proceeds inward, leaving a high concentration of less noble species in the ligaments' interior (B,C). Secondary dealloying coarsens the structure and reduces the residual amount of less noble element (D). Reproduced with permission from ref 91. Copyright 2022 Elsevier.

external surface in this early stage of dealloying. In principle, the just-mentioned processes link the characteristic ligament or pore size, L, to the magnitude of E_D . Yet, this link has not been understood in all detail, and a predictive theory for L as the function of E_D is missing.⁹¹

The corrosive attack on the bulk is carried by the formation of pore channels which bifurcate on a characteristic scale that

establishes the initial pore size (Figure 1B,C). That process of "primary dealloying"⁹² is followed by the coarsening of the porous structure by surface diffusion (Figure 1D). During coarsening, isolated domains (sometimes referred to as "clusters" because they represent percolating clusters of Ag atoms on the crystal lattice)^{90,91} of precursor alloy that had previously been buried in the interior of the primary dealloyed ligaments are exposed to the corrosive environment. "Secondary dealloying"92 then further dissolves the LNE, leading to an ongoing reduction in the residual LNE content. As an immediate consequence, the residual sacrificial element content systematically decreases with increasing, as-prepared L.93 A fraction of the buried Ag-rich domains survive in the asprepared material and can be imaged with element-sensitive microscopy techniques.^{90,94,95} Ligament sizes, L, in as-prepared NPG thus reflect the initial, very small structure size during primary dealloying, convoluted with substantial coarsening during secondary dealloying.

The atomic-scale processes that act during dealloying do not, as a general rule, include the nucleation of new crystals. As will be addressed in section 3.6, dealloying can thus conserve the grain size, often several tens of μ m, of the master alloy. In other words, dealloyed nanoporous metals can be single crystals when viewed at the scale of few ligament sizes. There is, however, a volume contraction that implies that the pore volume is typically smaller than the volume that was taken up, in the master alloy, by the dissolved LNE atoms. Section 3.7 addresses this issue, and it also points out that samples of NPG can exhibit a densified skin layer at their external surface, with the thickness comparable to the ligament size. The skin layer has little effect on the external dimensions or on the mean solid fraction of macroscopic samples, yet it contributes decisively to the densification of nanoporous nanoparticles.

Empirically, the structural evolution during dealloying can be, to some extent, controlled. To name a few, dissolution and interface diffusion rate are determined by temperature,^{79,96,97} electrolytic environment,^{81,98} alloy composition,^{99–101} and, in case of electrochemical dealloying, the dealloying potential.^{10,102–105} Furthermore, various postdealloying treatments have been demonstrated. They include the deposition of conformal oxide coatings along the pore surfaces by atomic layer deposition¹⁰⁶ or of metal monolayers by underpotential deposition,^{107,108} and the functionalization of the pore surfaces by electroactive polymers such as polyaniline¹⁰⁹ or polypyrrole^{110,111} or by organic molecules such as thiols,¹¹² nucleic acids, or antibodies.²⁸

An example for the control of the structural evolution during dealloying is the addition of Pt to the master alloy.^{100,113–118} Dealloying Ag–Au–Pt produces nanoporous Au–Pt with *L* as small as 4 nm, substantially less than the 20–40 nm that are typical for as-prepared NPG.^{116–119} The extremely small *L* in nanoporous Au–Pt has been attributed to the low surface diffusivity of Pt, which slows down microstructural coarsening. With attention to catalysis, it is relevant that Pt is strongly enriched at the surface of as-dealloyed nanoporous Au–Pt and that the Pt/Au atom fraction at the surface can be tuned in a wide interval by annealing.¹²⁰

The trend of nanoporous metals to coarsen by surface diffusion underlies a simple and powerful postdealloying treatment. Enhancing the diffusivity, either by thermal or by electrochemical annealing, can be exploited for tuning L in an interval ranging from the as-prepared value, which can be as small as 4 nm, all the way up to several micrometers.^{14,121,122}

The continuous tunability of its characteristic length scale by 3 orders of magnitude distinguishes NPG from practically all other nanomaterials.

A classic theory of microstructure evolution by diffusion predicts that L depends on time, t, as $L \propto t^{1/4}$ during coarsening.¹²³ Atomistic numerical simulation confirms this prediction.¹²⁴ Some experimental studies of NPG coarsening support the $t^{1/4}$ law, ^{125,126} while others report large confidence limits for the exponent.^{96,127–129} A data mining survey of the literature favors $t^{1/8}$ (emphasizing that this has no support from theory), yet ultimately qualifies the available data as inconclusive.¹³⁰ If the power law is accepted, then the ramifications of its low exponent merit a closer look. The exponent 1/4 implies that the time increment for a given relative increase in ligament size varies as L^4 . In other words, local regions in the high-L tail of the size distribution coarsen more slowly than the remaining microstructure, allowing the smaller ligaments to catch up. The scenario is similar to the diffusion-controlled Ostwald ripening of precipitates. That process exhibits a similarly low time exponent (1/3 as opposed)to the present 1/4) and, by virtue to the Lifshitz-Slyozov-Wagner (LSW) theory, a quasi-stationary size distribution even after extended coarsening.^{131,132} Quite analogously, NPG typically maintains a highly uniform microstructure, even after extended coarsening (see also section 3.9). When NPG coarsens outside the corrosive environment, then residual LNE exposed to the surface is not dissolved. The substantial relocation of atoms will then also affect the distribution of LNE within the ligaments. Their composition field may be homogenized, or an LNE-enriched layer at the surface may be formed. See section 3.13 for details.

Control over the dealloying process parameters is a prerequisite for the successful design of preparation protocols for nanoporous metals. This will be discussed in the following section.

2.3. Approaches to Dealloying

In principle, NPG can be prepared in every shape and dimension, as long as the precursor alloy and a surrounding electrolyte share a common interface. Geometries range from nanoscale particles, ¹³³ wires, ¹³⁴ films, ^{135–137} and leaves^{81,103,138–140} to μ m-thick coatings, ¹⁴¹ pillars, ^{142,143} and sheets, ^{144–147} up to mm-^{101,117,119,148–151} or cm-size¹⁵² bulk samples. The making of even larger parts from NPG, for instance in the form of macroscale scaffolds for catalysis, has been demonstrated based on 3D printing. ¹⁵³

Precursor alloys need to meet certain requirements to ensure successful dealloying. The corrosion potentials between the noble and less-noble species must differ by typically a few 100 mV. Also, the master alloy should be a single phase of uniform composition, either in the form of a solid solution, a glass, or an intermetallic compound. Both criteria can be readily met by various master alloys for NPG. Specifically, preparation protocols for NPG have been based on alloys containing Ag,^{14,77,93} Cu,^{154–158} Ni,¹⁰¹ Al,^{159,160} Sn,¹³⁶ Zn,^{141,161,162} Li,¹⁶³ and multicomponent glasses.^{164–166} However, in macroscopic dimensions, i.e., for at least mm-sized samples, preparation of NPG has so far been limited to alloys prepared by classical metallurgical methods such as Au–Cu, Au–Ni, and, as the by far most commonly used alloy system, Au–Ag.^{14,119,149–151} The parting limit for Au–Ag is 45 at. % Au.^{11,87,167} Monolithic

mm-size samples of NPG have been prepared from master



Figure 2. Preparation of nanoporous gold (NPG) by electrochemical dealloying. (a) $Au_{25}Ag_{75}$ precursor (left) and as-dealloyed NPG sample (right). (b) Microstructure of NPG imaged by scanning electron microscopy. Inset shows the cross section of a sample that was intentionally cleaved using a scalpel; the scale bar corresponds to 300 μ m. (c) Nanoporous gold disks air-dried after preparation. The samples were wetted by H_2O (left) and *n*-pentane (right) before drying. Note, capillary-induced stresses cause bending and breaking of samples wetted by H_2O which can be avoided by a final transfer to *n*-pentane. (a,b) Reproduced with permission from ref 193. Copyright 2020 The Author.

alloys with gold atom fractions between 5 at. $\%^{168}$ and 42 at. $\%^{.169}_{.}$

Dealloying can be prompted by free^{81,93,119,140,170–173} or by electrochemically^{10,14,77,93,104,119,151,174,175} controlled corrosion. Electrolytes include aqueous solutions across the whole pH range^{14,93,97,136,147,175} as well as ionic liquids.^{141,161,176} However, the majority of preparation protocols for NPG use acidic environments. For Au-Ag master alloys, free corrosion is generally carried out by concentrated $HNO_{3}^{81,93,119,140,170-173}$ while electrochemical dealloying typically uses dilute $HClO_4$ solutions.^{10,14,77,93,119,151} Both free and electrochemical dealloying can produce large monolithic volumes of NPG with comparable microstructure.¹⁷⁷ For samples with identical L and master alloy composition, similar mechanical properties are observed irrespective of which corrosion method is used.^{119,177} This suggests comparable sample quality and comparable microstructure. Yet, electrochemical dealloying offers a more direct control over the corrosion process. For example, variation of $E_{\rm D}$ is a facile method to control the final L. Here, higher $E_{\rm D}$ result in smaller L.^{103,104}

Dealloying does not dissolve the LNE entirely. The residual content of LNE can vary between <1 at. $\%^{149}$ and up to 80 at. $\%^{178}$ The retained amount of LNE is crucial for the preparation of hierarchical nanoporous structures by sequential dealloying.^{168,178} This will be explained in section 2.5. More importantly, the LNE content may be decisive for the catalytic performance of NPG. This will be inspected in sections 4–6. Here again, the electrochemical (as opposed to free) corrosion affords tuning the preparation protocol toward a direct control of the residual LNE content. The next section will elaborate in more detail on preparation protocols.

2.4. Preparation Protocols for Nanoporous Gold: A Case Study

2.4.1. Motivation. Among the many preparation protocols for NPG, few can produce the macroscale, uniform, crack-free, monolithic bodies with exceptionally clean surfaces that are the signature of high-quality NPG. The present section attempts a critical discussion of relevant process parameters for preparing NPG, and it also proposes a "best practice" guide for that

material. The protocol is based on an approach developed in the teams of Jin and of Weissmüller more than a decade ago¹⁷⁹⁻¹⁸¹ and refined subsequently in both teams.^{117-119,148,149,151} Results consistently show that macroscopic samples with a homogeneous microstructure can be obtained that are free of macro-defects such as cracks, show little dealloying-related shrinkage, and exhibit a low residual LNE content.^{117-119,149,151} The consistency of the mechanical behavior of samples prepared by the protocol^{117,119,149,151} testifies to an exceptional reproducibility. This is major asset and a motivation for describing that particular protocol here.

2.4.2. Master Alloy Selection and Preparation. In this case study, the master alloy is AuAg, the most commonly used precursor system in the preparation of NPG. The main advantage of AuAg is the continuous series of solid solutions in the equilibrium alloy phase diagram, in other words, full miscibility at any composition and at all temperatures up to the solidus.¹⁸² By contrast, AuCu and AuNi, the other binary alloy systems used for the preparation of macroscopic NPG, exhibit multiple ordered superstructure phases (AuCu)¹⁸³ or a miscibility gap (AuNi)¹⁸⁴ at room temperature. In order to avoid phase separation, careful heat treatment is required during preparation of AuCu and AuNi precursor al-loys.^{101,155,157} Even when the master alloy is prepared in the form of a uniform solid solution, AuCu-based NPG has been found to exhibit poor mechanical properties in the form of brittle deformation behavior. This is the result of microstructural heterogeneity, and specifically of crack formation along grain boundaries that were weakened during dealloying.¹⁵⁵ While macroscopic bodies of NPG dealloyed from AuNi show high strength and ductility, the ligament size tends to be large, $L \approx 120$ nm.¹⁰¹ Because many application scenarios proposed for NPG require both uniform microstructure and small L, AuAg is often the master alloy system of choice for making of NPG.

The present example uses master alloys $Au_{25}Ag_{75}$, prepared from Au and Ag wires (\geq 99.99% purity) by arc melting. The ingot is then encapsulated in evacuated fused silica and homogenized by annealing for 5 days at 850 °C. Afterward, numerous conventional metal shaping techniques, such as rolling, drawing, and cutting can be applied to prepare the precursor samples; an example of a cuboid sample before and after dealloying is shown in Figure 2a.

It is emphasized that a homogeneous precursor microstructure is important for obtaining high-quality nanoporous samples. Therefore, a final recovery anneal may be required to annihilate lattice dislocations and relax residual stress that result from the final machining. For AuAg alloys, an annealing temperature of 300 $^{\circ}$ C has been found suitable for recovery.

2.4.3. Potentiostatic or Galvanostatic Dealloying. The present protocol uses electrochemically controlled dealloying under conditions of constant applied potential, in other words, potentiostatic dealloying. In rare cases, galvanostatic control (constant net current, I) is used during electrochemical dealloying.^{185,186} This approach results, in principle, in a constant dissolution rate of the less noble species. That may seem an appealing concept for uniform dealloying. However, it may be argued that constant I does explicitly not lead to uniform dealloying conditions. This is so because the active surface area, A, as well as the alloy composition, vary as the dealloying procedes. In the early stages of dealloying, A of the smooth surfaces of the pristine master alloy sample is small. Therefore, the dissolution current density, j = I/A, is initially high. As soon as the rough corrosion front of primary dealloying is formed, A increases significantly and this decreases *j*. With progressing dealloying, dissolution by secondary dealloying becomes more important, as ions not only dissolve at the dealloying front but also in the trailing volume. It can be seen that secondary dealloying introduces a time dependence to the magnitude of A and thereby to *j*. Moreover, as soon as primary dealloying is complete and residual species are solely dissolved by secondary dealloying, the corrosion acts on a surface that is effectively quite dilute in LNE. In order to maintain the required high dissolution current, the dealloying potential then needs to be controlled to very positive values. That changes the oxidation state of the surface, and it may move the dissolution process into a different regime of the Pourbaix diagram. Clearly, those features of galvanostatic dealloying are not compatible with constant and uniform corrosion conditions. The effect of specific adsorbates on the dealloying process is discussed later in this section.

The by far most common electrochemical dealloying procedure is potentiostatic dealloying. Regarding the dealloying process, a constant $E_{\rm D}$ corresponds to an invariant driving force for the LNE dissolution. As the elemental composition changes during the dealloying process, a static potential causes a gradual reduction in *j*. Although dissolution conditions vary during both electrochemical preparation methods, potentiostatic dealloying can be considered the more uniform preparation technique because every volume element is subjected to the same dissolution progression irrespective of its location within the sample. In this case study, potentiostatically controlled dealloying is used. A three-electrode configuration is mandatory, not the least because the dissolved LNE, a large amount of matter, is deposited on the counter electrode.

2.4.4. Choice of Electrolyte. As with most of the abovementioned electrochemical dealloying protocols, the present case study uses dilute, namely 1 mol L^{-1} , HClO₄ as the electrolyte. In this environment, the surface diffusivity of Au atoms is rather slow. This results in a low coarsening rate during secondary dealloying and thereby reduces the final L.⁹⁸ Additions of halide-ions (Cl⁻, Br⁻, and I⁻) have been reported to significantly increase the Au surface diffusion rate of $HClO_4$ based electrolytes.⁹⁸ This may shorten the overall dealloying time, yet at the expense of considerably larger *L*.

While HClO₄ is unsuspicious of specific adsorption on gold,¹⁸⁷ adsorbates formed by impurities may influence the dealloying process and the catalytic performance of the nanoporous material. For example, during the long period of dealloying Cl⁻ impurities may enter the electrolyte when reference electrodes with a KCl environment are used, such as Ag/AgCl or calomel. In this case study, the issue is avoided by usage of a pseudoreference electrode in the same solution, namely a AgCl-coated Ag wire (+0.52 V (SHE) in 1 mol L⁻¹ HClO₄). This minimizes contamination, due to the extremely low solubility of AgCl. It is good practice to calibrate pseudoreference electrodes against more stable ones, such as the standard hydrogen electrode (SHE), before each dealloying run. When calibrated, other pseudo reference electrodes may also be appropriate, such as the reversible hydrogen electrodes (RHE). In order to minimize impurities, ultrapure water ($\geq 18.2 \text{ M}\Omega \text{ cm}$) is used to prepare all aqueous solutions.

It should be noted that dealloying in electrolytes that are, in pristine condition, devoid of ions of the less noble species exhibits poorly defined starting conditions because the Nernst potential is very low or ill-defined at the onset of corrosion. This issue settles once ions dissolve into the electrolyte. Still, as the dealloying critical potential depends on the Ag⁺ concentration in solution, the dealloying conditions in the just-mentioned media may vary with the electrolyte volume and with the ion transport kinetics.¹⁰² An approach that circumvents this issue is dealloying of Au–Ag in $AgNO_3$ solutions.^{147,175} However, due to the formation of passivating AgO layers, the residual Ag mole fraction, x_{Agr}^{res} , in as-dealloyed NPG was found to be very high ($x_{Agr}^{res} > 30\%$).¹⁴⁷ Addition of AgClO₄ into HClO₄ solution is another strategy to obtain welldefined dealloying conditions.¹⁰² Although this dealloying approach has been reported two decades ago,¹⁰² systematic addition of AgClO₄ into electrolytes for the purpose of making of NPG under even better controlled conditions has not been implemented so far.

2.4.5. Dealloying Potential and Postdealloying **Conditioning.** The dealloying critical potential of Au–Ag exhibits comparatively high values in pure HClO₄-based electrolytes.99 The present protocol features potentiostatic dealloying of Au₂₅Ag₇₅ precursors at $E_D = 1.27$ V (SHE), i.e., 0.75 V (pseudo Ag/AgCl). Dealloying is considered as complete when the current decays to a few tens of μ A. For cylindrical $\mathrm{Au}_{25}\mathrm{Ag}_{75}$ precursors with a diameter of 1 mm and a length of 2 mm a value of 10 μ A is used. As a result of the quite positive value of $E_{\rm D}$, oxygen species adsorb on the Au surface.¹⁸⁸ The adsorption impairs surface diffusion and slows down the dealloying rate. In alloy systems with a less noble sacrificial element, such as Au-Ni, oxidation of the Au surface can be mitigated by dealloying at lower E.¹⁰¹ Oxygen adsorption on NPG also embrittles the material, as lattice dislocation end points are pinned at adsorbate sites.¹⁸⁹ While oxygen species adsorption on Au thus seems unfavorable, it can be used to hinder coarsening at ambient conditions. Thus, deliberate oxidation may enhance the shelflife of stored samples of NPG.

After dealloying by the steps described so far, samples of NPG exhibit $L \approx 20 \text{ nm}^{116}$ and a high x_{Ag}^{res} around 20 at. %.¹¹⁷ The next step in the preparation protocol is designed to reduce x_{Ag}^{res} further. To this end, the sample is first polarized at E = 1.37



Figure 3. Making of hierarchical nanoporous gold by dealloying. (A) Schematic illustration of the preparation steps: (i) first dealloying of a Ag-rich precursor alloy leaves a nanoporous network with high amount of residual Ag; (ii) coarsening of the ligament structure by thermal annealing; (iii) second dealloying forms the lower level nanoporous microstructure. (B–D) Scanning electron micrographs of the microstructure after the first dealloying (B), coarsening (C), and second dealloying steps (D). Reproduced with permission from ref 178. Copyright 2021 American Association for the Advancement of Science.

V (SHE) until the current diminishes. Afterward, the electrolyte (1 mol L^{-1} HClO₄) is renewed and 20 potential cycles (0.12 to 1.62 V vs SHE) are applied at a scan rate of 5 mV s⁻¹. The two steps, polarization and potential cycles, are then repeated once. In each case, the potential cycling is concluded by an anodic scan to the electrode potential 0.82 V (SHE). This corresponds to an adsorbate-free Au surface state.¹⁹⁰ This preparation routine results in clean and reduced metallic Au surfaces. Note also that the dealloying and conditioning do not involve organic solvents. This is one more distinctive feature of dealloying that affords exceptionally clean surfaces.

The postdealloying treatment procedure reduces the overall Ag content, as determined by energy dispersive X-ray analysis, to values below $x_{Ag}^{res} < 1\%$. X-ray photoelectron spectroscopy (XPS) suggests that the remaining Ag preferably accumulates at the surface.^{35,191,192} With the present protocol, the XPS (near-surface) Ag fraction was determined at $x_{Ag}^{res} = 8\%$.¹⁹² Final ligament diameters, as determined by scanning electron microscopy, are around 40 nm.^{117,118,193}

2.4.6. Drying. Drying is an often a neglected topic in the preparation of nanoporous metals. Throughout the dealloying and conditioning, the samples are wetted by electrolyte. Upon drying, the surface tension of the receding liquid–vapor meniscus, γ_{LV} , introduces a pressure, *p*, in the pore fluid according to the Young–Laplace equation, namely

$$p = \frac{2\gamma_{\rm LV}}{r} \tag{1}$$

Here *r* denotes the radius of the meniscus, approximately equal to the pore radius.¹⁹⁴ Given that the pore size of NPG is only a few tens of nm, *p* takes on values of around 10 MPa for water $(\gamma_{LV} = 72 \text{ mJ/m}^2)^{194}$ in the pore space. However, significantly higher values of *p*, exceeding the yield strength, can ensue at local constrictions. As a result, improper drying procedures can cause severe deformation or spontaneous fracture of nanoporous metals, as demonstrated in Figure 2c. Although this issue has been well-known for many decades, for example, in

the fields of porous semiconductors^{194–196} or silica aerogels,¹⁹⁷ it is hardly appreciated in the context of nanoporous metal preparation. The drying technique below, adopted from protocols for mesoporous silicon,¹⁹⁴ has proved useful in the making of nanoporous metals.^{198,199}

An efficient way to minimize capillary-induced stresses is to transfer the nanoporous metals from the preparation medium first into water, then into ethanol and finally into n-pentane, whereupon the samples are deposited on glass and allowed to dry. This procedure profits from the considerably lower surface tension of *n*-pentane, which is $\gamma_{LV} = 14 \text{ mJ/m}^2$ as opposed to the 72 mJ/m² of water. The contrasting juxtaposition of NPG sheets displayed in Figure 2c clearly emphasizes the need for consideration of capillary-induced stresses when crack-free monolithic bodies of NPG are required. A drawback of the drying routine is its use of organic solvents that may alter the surface state of the porous metal. Therefore, drying in npentane is not established as a standard preparation step in the presented case study, and application should be made dependent on need. Other, more laborious techniques such as freeze- and supercritical drying avoid the organics but appear not to have been applied to nanoporous metal so far.

2.5. Hierarchical Nanoporous Gold

So far, this section 2 addressed the formation mechanisms behind the evolution of the lowest level microstructure required to form bodies of NPG. Most of the unique properties of NPG can be directly attributed to its small ligament size. Yet, the small size may also have adverse effects, specifically in slowing down mass transport through the pore space. This is of obvious relevance for the performance in catalysis. Structuring on multiple length scales may solve the dilemma by admitting both small pores for achieving a high specific surface area and large pores for transport. Nanoporous metals with multiple microstructural length scales have been the subject of considerable interest in recent years.^{200–202} The lowest structural level of multimodal NPG is typically made by dealloying, and numerous approaches toward the higher structural levels have been proposed, such as templating,^{203,204} additive manufacturing,¹⁵³ and selective dissolution of multiphase alloys.¹⁵⁹ A particularly stringent implementation of multileveled pore structures is hierarchical porous materials in which the structural organization at each of the individual hierarchy levels is geometrically similar, with identical geometry and topology at each level but distinctly different characteristic length scales. For the specific structural motive of NPG, networks nested on two separate hierarchy levels have been demonstrated and exemplify the stringent hierarchical organization.^{168,178,205-208} The microstructure of "nestednetwork nanoporous gold"¹⁶⁸ is illustrated in Figure 3. The straightforward architecture of such materials predestines them as model systems for understanding the impact of hierarchical structuring on properties and for the rational design on multiple microstructure levels.¹⁷⁸ Below, we briefly indicate how the preparation protocol of the previous sections can be expanded to yield nested-network nanoporous gold with two hierarchy levels. The making of porous metals with other, less stringent multiscale geometries is described in several recent reviews.^{200–202}

Figure 3A schematically illustrates the preparation of nestednetwork NPG by dealloying, as first reported in ref 168 and recently refined for robust macroscopic samples.¹⁷⁸ While minor preparation details may vary in other studies, 206,208 Figure 3 captures the essentials. The master alloy is dilute in Au, e.g., Au₇Ag₉₃, and the dealloying conditions are carefully selected so that x_{Ag}^{res} remains very high, affording a later, second dealloying step. The resulting, nanoporous Ag-Au (with, for instance, $x_{Ag}^{res} = 76\%)^{178}$ is then coarsened by annealing, which establishes the microstructure of the upper hierarchy level. The second dealloying then introduces porosity into the ligaments of that upper level, establishing the lower hierarchy level. The anticipated enhanced transport rate in nested-network nanoporous gold is indeed confirmed by experiment.^{168,208} Furthermore, the range of solid fractions accessible to dealloying is extended downward to as low as 0.12 while mass-specific mechanical strength and stiffness are superior to NPG with a single hierarchy level.¹⁷⁸

2.6. Engineering the Macroscopic Shape of Nanoporous Gold

2.6.1. Machining the Master Alloy Preform. A wide variety of conventional metal forming approaches can be applied to master alloys for macroscopic samples of NPG. These include, for instance, cutting, milling, rolling, or wiredrawing. In principle, dealloying is a shape-conserving process. Size and shape of NPG samples may therefore be selected by appropriate machining of the master alloy. Yet, care has to be taken to choose the conditions for minimizing sample shrinkage during dealloying. Section 3.7 discusses phenomenology and mechanisms of shrinkage. Figure 4A shows a cm-size rectangular beam of (epoxy-resin infiltrated) NPG as an example for a machined macroscale geometry.

2.6.2. 3D Printing the Master Alloy. Recently, it has been demonstrated that NPG catalyst samples with complex microscopic geometries can be produced by 3D printing. The process acts on an ink containing master alloy particles, which are dealloyed subsequent to printing the preform (Figure 4B).¹⁵³

2.6.3. Nanoporous Thin Layers and Nanoporous Nanowires. Monolithic dealloyed bodies of NPG, of extended size in all three dimensions, are often not very well suited for the integration in reactors. That is a consequence of the



Figure 4. Shaping samples of NPG. (A) A 1 mm \times 2 mm \times 26 mm size rectangular bar of epoxy-resin infiltrated NPG (center, brownish color), here subject to a four-point bending test. (B) 3D-printed, cm-size lattice of NPG. (C) NPG bumps with diameter of 1.0 μ m on planarized CMOS chip. (A) Reproduced in part from ref 152 under Creative Commons license CC BY 4.0. Copyright 2015 The Authors. (B) Reproduced in part with permission from ref 153. Copyright 2018 The Authors. (C) Reproduced with permission from ref 221. Copyright 2022 IEEE.

inefficient mass transport in the pore space, in conflict with the need for mitigation of diffusive and resistive gradients in the reactor. Many studies have solved that problem by working with thin layers of NPG. These can be made by dealloying the free-standing thin (typically around 100 nm) master alloy sheets of Ag–Au that are commercially available as white gold leaf.⁸¹ Thin films based on vapor-deposited master alloys on planar substrates are also frequently explored.²⁸ Alternatively, thin layers of NPG can be conformally deposited on current conductive support structures by galvanic deposition of an alloy that is subsequently dealloyed to NPG. Such processes can exploit the advantages of an all-wet-chemical processing of electrodes of almost any shape and size.²⁰⁹

Layers of NPG have been studied as cantilever actuators^{210,211} and sensors.²¹² Sensors with optical or electrical readout have been proposed for biomedical applications,²⁸ substrates for surface-enhanced Raman scattering,²¹³ studies of optical properties,^{29,214,215} energy-storage,²¹⁶ electrocatalysis,²¹⁷ and photoenhanced catalysis.^{30,218}

Substrate-supported thin layers of master alloy for NPG can be structured by electron-beam lithography and subsequently dealloyed to produce thin structures of NPG with a wide range of 2D shapes. That is of interest because lithographic structuring may, in principle, be seamlessly integrated into microtechnology manufacturing schemes. Wire- and diskshaped samples have been demonstrated, ^{219–221} and an example is shown in Figure 4C.²²¹

Nanoporous nanowires can also be made as freestanding objects. Several works have demonstrated that the master alloy can be electrodeposited into suitable nanoscale channels in a



10 µm

Figure 5. Preparation of porous gold by processes different than dealloying. (A) anodization in HCl solution; (B) H_2 bubble templated deposition; (C) templated electrodeposition between sacrificial nanoparticles. (A) Reproduced with permission from ref 21. Copyright 2020 Elsevier. (B) Reproduced with permission from ref 233. Copyright 2014 Elsevier. (C) Reproduced with permission from ref 234. Copyright 2012 Wiley-VCH.

substrate, for instance, anodized alumina^{222,223} or ion-tracketched channels in a polymer.²²⁴ Removing the substrate and subsequently dealloying generates the nanowires. In a less controlled way, dealloying columnar microstructures in rapidly solidified metal ribbons may also yield rod-like nanoporous particles.²²⁵

Lastly, thin layers of nanoporous gold can be obtained by electrochemically roughening an initially flat Au electrode. A typical procedures involved anodization, followed by a reduction of the created oxide layers.^{21,226–230} The resulting morphology is similar to that of NPG obtained by dealloying (Figure 5A). The thickness of such layers of NPG may be restricted to few tens of nanometers. Roughening of initially flat Au electrodes has also been demonstrated by electro-depositon/alloying/dealloying cycles,^{141,161,231–233} often using $Zn^{141,161,231}$ or Ag^{233} as the LNE. These can be applied to electrodes of nearly any size and shape, even when preassembled in an electrochemical cell of specific geometries such as microelectrodes for sensors.

2.7. Nanoporous Powders

2.7.1. Making Powder from Bulk NPG. Electrodes for electrochemical applications have to be integrated with other electrodes into an electrochemical cell optimized for the specific application, such as a parallel plate reactor. The direct integration of NPG has only rarely been demonstrated in devices such as fuel cell²³⁵ or metal–air batteries.^{236,237} Nanoporous gold has also been used for alloy-type negative electrodes in Li-ion batteries.²³⁸ However, the overall accounts of such application has remained limited to date.

In principle, powders obtained from NPG monoliths offer another pathway to shaping samples of NPG because they can be consolidated into appropriate shape. Cavity microelectrodes of NPG form an example, providing for particularly precise electrochemical characterization. Due to the limited material, the overall currents are small and thus the voltage drop $U_{drop} = IR_e$ ("IR drop") is kept small. Voltammograms recorded with cavity microelectrodes have particularly little distortion, cf., Figure 6E vs D. Furthermore, the external mass transport (outside of the pore space) is controlled by the effective radial diffusion (Figure 6A). In addition, the exchange of the solution in the pore space is efficient.

NPG monoliths can be powdered by mechanical impact, sonication, or similar processes. The powder can be filled in a cavity microelectrode (Figure 6A), while preserving the nanoscale structure, without the use of binder (Figure 6C).¹⁹² NPG can be released from cavity microelectrodes by

intensive sonification, after which the same cavity may be washed and filled by another aliquote of the same NPG powder (Figure 6B). Using powders from identical starting material tends to decreases data scatter in electrochemical experiments with NPG by avoiding small variation between monoliths in the dealloying process.¹⁰⁸

2.7.2. Processing NPG Powders. Instead of upscaling the size of the original NPG electrode, pulverized monoliths can be processed with binders to obtain porous electrodes coated on a back contact. The use of binders and the associated problems in reliably establishing electrical contact between particles removes many of the original advantages associated with the bicontinuous structure of NPG. Nevertheless, such processes may be required for the integration of NPG into devices.^{240–242}

2.8. Dealloyed Porous Nanoparticles

Use of a dealloying process is increasingly extending from macroscopic materials to nanomaterials. Dealloying of binary and ternary alloy nanoparticles has become a broadly accepted facile synthetic route toward highly active and durable catalyst materials for heterogeneous and electrochemical reactions.^{43,64,80,243} The focus of this section is on the relation of nanosizing of binary alloy nanoparticles on the (electro)-chemical dealloying and their comparison to bulk materials.

Dealloying of multimetallic nanoparticles covers a broad range of particle motifs like core—shell, hollow, donut-like to porous nanoparticles to tune their physicochemical and (electro)catalytic properties. In contrast to the macroscopic dealloyed materials, the fundamental dealloying mechanisms behind the formation of size-dependent particle morphologies are still poorly understood to date.

The critical dealloying potential is largely influenced by the initial composition of the master alloys. It is well studied that smaller monometallic nanoparticles tend to dissolve at lower anodic potentials compared to the bulk materials; this tendency is more pronounced with decreasing particle size.²⁴⁴ This shift of the dissolution potential as the function of the particle size can be related to the Gibbs–Thomson effect. With decreasing size, the fraction of low-coordinated surface atoms rises, resulting in an increase of the surface energy of these nanoparticles.

Recent KMC simulations have indicated how the sizedependent electrochemical properties of monometallic nanoparticles can be related to the critical dealloying potential of alloy nanoparticles.²⁴⁵ One of the key results is that the evolution of porosity in alloy nanoparticles takes place at



Figure 6. Cavity microelectrodes. (A) Schematic cross section; (B) SEM image of empty cavity microelectrode produced by etching a glass-sealed gold wire; (C) SEM image of a cavity microelectrodes filled with NPG powder, insets show zoomed areas with NPG ligaments; (D) cyclic voltammogram of a *macroscopic* NPG monolith in 1 mol L⁻¹ HClO₄, potential scan rate v = 10 mV s⁻¹; (E) cyclic voltammogram of the very same NPG material inside a cavity microelectrode in 1 mol L⁻¹ HClO₄, v = 10 mV s⁻¹. (B) Reproduced with permission from ref 239. Copyright 2019 Wiley-VCH. (C) Adapted from ref 192 under Creative Commons license CC 4.0. Copyright 2020 The Authors. (D,E) Reproduced with permission from ref 638. Copyright 2022 The Author.

higher critical dealloying potentials than in the corresponding bulk alloys. The increase of the critical potential in relationship to $1/r_{\rm NP}$, where $r_{\rm NP}$ is the radius of the alloy nanoparticle is based on kinetic effects. Compared to bulk Ag₇₅Au₂₅, the KMC simulations predicted an increase in critical dealloying potential by 25 mV for 17 nm and by 100 mV for 8 nm nanoparticles.²⁴⁵ As has been explained in section 2.2, the relative rates of (1) the dissolution rate of the LNE and (2) the passivation of the surface by diffusion of the remaining MNE are decisive for the nanostructure formation during dealloying. The ratio between these rates may be significantly modified in nanoparticles. According to these KMC simulations,²⁴⁵ the following time-dependent characteristics have been identified: In the initial stage of the dealloying process, Ag atoms at the top layer of the nanoparticles are dissolved very rapidly. During the intermediate time, the particle surface is entirely passivated by the very mobile Au surface atoms. The surface roughening induced by dissolution of Ag surface atoms enhances the formation of a monolayer of Au atoms as a passivation layer. Therefore, nanoparticles tend to form a passivation layer of MNE atoms very quickly. Only over longer times and at a constant potential, the fluctuations in the passivation layer might be sufficient for the dissolution of the Ag atoms from the subsurface layers. Smaller alloy nanoparticles with a high content of low-coordinated surface atoms and short-lived fluctuations in the structure of passivation layer do not tend to evolve pores. In contrast, the pore evolution is more favored if the fluctuations are long-lived and allow successive Ag dissolution, which is the case for larger particles.

Only few experimental studies have investigated the particle size effect on the morphology of dealloyed nanoparticles. One of the first systematic studies pointed out the particle size effect for electrochemical dealloying on binary alloy nanoparticles.²⁴³ In case of Ag-Au, the particle size effect on the morphology of 2-6 nm and 20-55 nm alloy has been studied by applying different constant potentials in acidic media and characterized by high-resolution microscopic techniques.⁸⁰ In agreement with the reported KMC simulations,²⁴⁵ a superficial dealloying occurs at potentials above the critical potential of 2-6 nm AgAu nanoparticles. Due to the inappropriate ratio of surface atoms to bulk atoms, the pore evolution is not favored for small nanoparticles, and therefore the core-shell formation with an Au-rich particle shell takes place. For AgAu alloy nanoparticles larger than 40 nm, the pores form only above 0.75 V (SHE), and a significant shrinkage of these dealloyed particles is observed.

A comprehensive reconstruction study of porous Au nanoparticles prepared by dealloying has been reported.²⁴⁶ Figure 7 illustrates a reconstruction of a single representative porous Au-rich particle prepared by electrochemical dealloying. Structural information like pore size, porosity, specific surface area, and tortuosity were determined for porous Aurich nanoparticles with an initial diameter of 77 \pm 22 nm, which can be used for the simulations and models of diffusion and mass transport phenomena in confined environments.²⁴⁷ Two types of dealloyed porous Au nanoparticles with different ligament size and residual Ag content could be identified: Porous nanoparticles with smaller ligaments (~14 nm) show a low residual Ag content ($x_{Ag}^{res} \sim 6\%$). In contrast, higher Ag content ($x_{Ag}^{res} = 20\%$) is observed for porous nanoparticles with larger ligament sizes (~23 nm). However, bulk nanoporous Au-rich materials prepared by dealloying show an opposite behavior. Moreover, the porosity and ligament size of the dealloyed nanoparticles were evaluated by a statistical analysis method, so-called chord length distributions.²⁴⁸⁻²⁵⁰ For these analysis a large number of arbitrary straight lines is placed within the reconstructed volume. The so-called chords can be divided whether they are within the material or pores and averaged over their length, giving statistical information on pore and ligament size. Very interestingly, the dealloyed nanoparticles published in ref 246 show less porosity compared to macroscopic materials published in ref 138. It is expected that a high porosity is associated with a large specific surface area. The bulk materials, yet possess a lower specific surface area compared to the porous nanoparticles. This discrepancy might be explained by the differences in the structure and



Figure 7. Three-dimensional STEM tomography reconstruction of porous nanoparticles prepared by dealloying. (a) particle I, (b) particle II. Reproduced with permission from ref 246 Copyright 2021 The Author(s) under License CC-BY.

interconnection between the ligaments and pores between bulk materials and nanoparticles.

2.9. Other Ways to Meso- or Macroporous Gold Electrodes

2.9.1. Direct Electrodeposition of Porous Gold Electrodes. Studies of nanoporous gold in the context of catalysis invite the comparison to other, meso- or macroporous gold electrodes. Selected procedures toward such electrodes will be briefly discussed here. Porous gold can be electrodeposited using sacrificial nanoparticles as templates.^{251,234,252} The template particles can be arranged to inverse opal structures followed by electrodeposition of gold, yielding a macroporous material that can be refined by the use of different template nanoparticles (Figure 5C).²³⁴ Instead of particle templates, H₂ bubbles can serve as template.^{19,253-256} The resulting material has a wide distribution of pore sizes (Figure 5B), but the process can easily be applied to substrates electrodes of almost any shape and size. The direct deposition of alloys is also possible.^{19,256} Sophisticated template structures have been used such as diatoms, i.e., marine microorganisms with an exoskeleton.²⁵⁷ Those exoskeletons exhibit regular nanostructures in a large variety of shapes that can be selected by choosing a specific kind of diatom as template.

Electrodeposition can also be guided by electrolyte additives and control of the current density to yield directly a flexible porous material with interwoven needles similar to the structure of paper.²⁵⁸ Powders of different metals can also be sintered to a porous network of alloy materials with micrometer-sized feature sizes followed by dealloying to form NPG.²⁰⁷

2.9.2. Assembly of Nanoparticles to Porous Gold. Some other approaches shall be mentioned for the specific potential to alter either composition or the macroscopic shape of the resulting porous solid. Presynthesized metal nanoparticles can be combined to an interconnected aerogel using either only Au nanoparticles or mixtures of different nano-particles.^{259,260} Those aerogels exhibit a much higher porosity than dealloyed NPG, and the size of the struts is related to the size of the used nanoparticles. The possible use of different nanoparticles provides means for exactly controlling the resulting overall composition. The local composition may greatly deviate from the mean composition and may be used for bifunctional catalyst materials. Presynthesized nanoparticles with a hydrophobic ligand shell can also be prearranged at the air-water interface of a Langmuir-Blodgett trough. 22,261,262 After solution exchange the initially loosely arranged particles can be connected by electroless deposition of gold yielding a mesoporous membrane with a thickness similar to that of the original nanoparticles.^{22,261,262} Nanoporous gold has also been obtained by chemical reduction of aqueous 1 mmol L^{-1} $H[AuCl_4]$ by ascorbic acid and using 100 mmol L⁻¹ of the ionic liquid tetrapropylammonium glycine to control the nucleation. The process yields a porous material of aggregated particles that was efficient for electroreduction of nitroaromatric compounds.²⁶³ This area has recently been reviewed.²⁶⁴

2.9.3. Coating of Au on Other Porous Electrodes. Further suggested procedures use hydrothermal growth on Ti²⁶⁵ or deposition of an Au monolayer on nanoporous copper by a galvanic displacement reaction.²⁶⁶ Galvanic displacements reactions have also been used to convert a nanostructured Co electrode to nanoporous Au/Pd electrodes.²⁶⁷

3. MICROSTRUCTURE AND METRICS FOR ITS DESCRIPTION

3.1. Overview

As has been pointed out in section 2, the preparation and processing conditions of dealloying afford the controlled preparation of quite different microstructures in the nanoporous material. As the materials properties can be tuned along with the microstructure, the wide options for microstructure design provide a key asset of nanoporous gold. In consequence, understanding the microstructure-property relations is an important scientific challenge. Raising to this challenge requires, in the first place, that an appropriate and complete set of microstructural parameters is identified, that the implications of those parameters for scale and geometry of the microstructure are understood, and that their values are reliably measured for the material at hand. The present section exposes the relevant state of the art, highlighting as distinct metrics as the characteristic length scales, phase fractions, residual LNE content and distribution, connectivity, and tortuosity. Emphasis is on the relevance specifically in the field of catalysis. Recent reviews with reference to the



Figure 8. 3D geometry and topology of NPG's microstructure. (A) Rendering of an experimental tomographic reconstruction; solid fraction $\varphi = 0.30$.¹²⁵ (B) Rendering of a leveled-wave model generated structure with $\varphi = 0.30$.²⁷⁰ (C) Rendering of the identical realization of the leveled-wave model, yet with $\varphi = 0.10$.²⁷⁰ Note the remarkable qualitative agreement between experimental and model structures. Note also that the model structure with $\varphi = 0.10$ represents an array of disconnected clusters, not a percolating and loadbearing network. (A) Reproduced with permission from ref 125. Copyright 2016 Taylor and Francis Ltd. (B,C) Reproduced with permission from ref 270. Copyright 2016 The Authors.



Figure 9. Selected examples for signatures of and experimental approaches to the ligament size. (A) SEM-based evaluation of a size, L_{SEM} , by analysis of the characteristic diameter of ligaments at their waist (white arrows).¹³⁰ Pore size (dark arrows) and the skeleton of the ligament network (legend) are also indicated. (B) Analysis based on Bragg reflection width (Williamson–Hall analysis) in wide-angle X-ray powder diffraction.¹⁰⁵ The Williamson–Hall size, D_{W-H} is systematically correlated to L_{SEM} validating this analysis for measuring the ligament size. (C) In situ small-angle X-ray scattering showing the characteristic interference peak of NPG and its evolution to lesser wave numbers as the structure coarsens during dealloying at 20 °C.²⁸⁴ The mean spacing between neighboring ligaments can be inferred from the peak position. (A) Reproduced with permission from ref 130. Copyright 2018 under license CC-BY-4.0. (B) Reproduced with permission from ref 105. Copyright 2017 under license CC. (C) Reproduced with permission from ref 284. Copyright 2017 under license CC 3.0.

parametrization of NPG's microstructure can be found in refs 268 and 269.

3.2. Geometric Model for the Nanoporous Gold Microstructure

In the form of the "leveled-wave model", we here introduce a toy model that provides illustration for our discussion of the microstructure metrics of NPG.²⁷⁰ Other models and approaches to 3D representation of NPG will be discussed in section 3.14. The leveled-wave model goes back to the seminal work on early stage spinodal decomposition by Cahn.²⁷¹ Its underlying notion is that, out of a random spectrum of small fluctuations, the microstructural evolution of the decomposition selects those with a single characteristic wavelength for fastest growth. The selection results from a competition between interfacial energy and transport kinetics. Those factors are equally decisive in spinodal decomposition and in dealloying, even though the driving forces and the kinetics underlying those two processes are quite different. The resulting microstructure can be constructed in two simple steps, first superimposing plane waves of identical wavelength but different wave vector orientation and with random phase shifts, and second binarizing the resulting random field by a level cut that is selected to provide the desired phase fraction (for instance, the solid volume fraction in the case of a

nanoporous solid).^{271,270} A simple strategy for selecting the wave vector directions provides the model with a 3D periodic structure, enhancing its appeal as a basis for numerical simulation.²⁷⁰ Earlier studies had generated model microstructures for NPG by following the spinodal decomposition numerically.^{272–275} The resulting geometry depends on the time at which the evolution is interrupted. The leveled-wave algorithm is free of that issue and generates random structures with unique statistical characteristics. The algorithm is also extremely facile to implement and execute.

Figure 8B,C illustrates a typical leveled-wave structure^{124,270} and sets it next to an experimental 3D tomographic reconstruction of NPG (Figure 8A).¹²⁵ The close resemblance of the microstructural geometries is apparent. What is more, leveled-wave structures have been found to accurately reproduce, with no free parameters, the dependence of Young's modulus, Poisson's number, and yield strength on the solid fraction for as-prepared NPG.^{178,270} The model also accurately reproduces a topological feature, the scaled genus (see section 3.9) of as-prepared NPG.^{125,270} As an added benefit of the leveled-wave model, many of its microstructural metrics, including characteristic size, solid fraction, volume-specific surface area, and averages of the mean and Gaussian curvatures of the surface, are interrelated by exact analytical expressions.²⁷⁰

The characteristic length-scale of the leveled-wave model is set by the underlying wavelength, which can be chosen at will. Applying the model to NPG rests on the tacit assumption that NPG microstructures of different characteristic microstructural length scale (different ligament size) are self-similar, which is discussed in section 3.9. Available studies present strong evidence for the applicability of the model to as-prepared NPG, irrespective of the specific preparation protocol or the ligament size.^{119,178,270}

3.3. Metrics for the Characteristic Length Scale

Undisputedly, one of the most important microstructural parameters of any given sample of NPG is the magnitude of its characteristic length scale. This scale determines area per volume and, thereby, the fraction of atoms exposed at the surface and available for catalysis. It also determines size effects controlling mechanical strength,^{27,142,276} optical func-tion,^{215,277,278} and electric resistance^{223,279,280} of the nanomaterial. The importance of the characteristic length scale for so diverse phenomena emphasizes that the option of tuning this scale within wide margins (section 2) is a distinguishing asset of NPG. Inspection of the microstructural geometry, for instance in Figures 8 and 9A, immediately reveals that any single notion of "size" may not adequately cover all relevant aspects of the geometry. One may focus on different features, e.g., a characteristic diameter of the struts or "ligaments" that make up the network material or, alternatively, a characteristic spacing between such features, and one may be interested in various measures for the distribution of the feature in question. In other words, there is no such thing as a unique definition of size, and one must expect that different measures for size may be relevant for different phenomena. That situation is ubiquitous in materials science, and specifically it can be relevant for the materials performance in catalysis.

The most common approach to parametrizing the characteristic length scale for NPG is evaluating "the ligament size" by measuring diameters along the waists of ligaments in scanning electron micrographs (Figure 9A).^{25,93,106,119,174,191,281–283} This is done manually for a certain number of ligaments, and the mean diameter, L_{SEM} is reported. Good practice requires that the number of ligaments counted and the variance of the diameters be specified. As macroscopic samples of NPG tend to have reduced (relative to bulk) L_{SEM} close the external surface, ^{35,93,174} good practice also requires that micrographs from the interior of the sample are examined and that their location is specified.

For nanoparticles and nanocrystalline materials, the evaluation of X-ray line broadening is a standard approach to characteristic size.²⁸⁵ The simple evaluation of the breadth of the strongest Bragg reflection in terms of the Scherrer formula must be rejected because it does not discriminate between size and microstrain broadening, thus it systematically and strongly underestimates the size. Variants of the Williamson-Hall approach avoid that error and are readily applied whenever a sufficient number of Bragg reflections are available.^{285,286} NPG exhibits substantial microstrain (section 3.12) that systematically increases with decreasing ligament size, emphasizing the need for Williamson-Hall type correction.¹⁰⁵ In NPG, an apparent complication resides in the presence of a coherent crystal lattice that extends to dimensions much larger than L. That obliterates the simple analysis of the impact of coherency on the X-ray line broadening of the standard diffraction theory. Nonetheless, empirical data for NPG show an excellent

correlation between the *L* values derived by Williamson–Hall and SEM (Figure 9B),¹⁰⁵ validating X-ray line broadening analysis as a method for measuring the ligament size of NPG.

Experimentally more demanding approaches to measuring a mean ligament size involve the analysis of 3D reconstructions.^{125,127,138,246,287–289} This may provide variously defined measures for size. An example, based on "granulometry", is the mean, L_G , over all materials points in the solid phase, of the diameter of the largest in-fitting sphere containing the respective point.^{138,290} A closer inspection of such approaches reveals various issues that require attention.

First, the sampling of space in granulometry emphasizes the wider regions of the microstructure, providing a comparatively large characteristic size.²⁹⁰ A related algorithm, combining skeletonization with a euclidean distance transform,^{146,291} provides around 30% smaller size for NPG.^{269,292} This algorithm has been applied in a data mining approach to coarsening of NPG,¹³⁰ yet its relation to other measures for *L* remains to be established.

Second, whenever there is a distribution of sizes, averaging is required for condensing the data into a single size parameter. The averaging involves a decision on the weighting function, for instance number-, area-, or volume-weighted averaging, and the resulting average values can differ strongly.²⁹³ Different materials phenomena may depend on different weighting functions, and this interrelation is poorly appreciated or understood in the field of NPG.

The characteristic spacing, \tilde{L} , between the centers of neighboring ligaments is another obvious measure for size.^{270,294} A robust signature for \tilde{L} of aperiodic structures is provided by the position of the first maximum in their autocorrelation function. For NPG and the leveled-wave model, \tilde{L} defined in this way has an immediate relation to the structure factor or interference function. As a result of the Debye formula,^{295,296} that function exhibits a distinct peak with a maximum at the wavenumber²⁷⁰

$$q_{\rm max} = 1.23 \ 2\pi/\tilde{L}$$
 (2)

Indeed, experimental small-angle X-ray scattering data of NPG exhibit a pronounced interference peak (Figure 9C). Various studies have measured \tilde{L} from the position of the peak in neutron⁹ or small X-ray angle scattering^{126,206,284,297} data, or from a maximum in the numerical Fourier transform of electron micrographs.^{94,294} In one instance,²⁹⁴ an equation similar to (eq 2) but without the prefactor was assumed ad hoc. The characteristic length evaluated in that way is smaller than the characteristic spacing \tilde{L} .

3.4. Specific Area of Surface

Another common microstructural parameter related to characteristic size is the "specific surface area", which is again linked to the number of sites available for catalysis. The specific surface area also determines in how far capillary forces induce pressure in the bulk. As will be discussed in section 3.12, that surface-induced pressure generates strain throughout the crystal lattice, which in turn may affect the catalytic activity. The surface-induced pressure is also of relevance for actuation^{101,210,298–301} or active strain sensing³⁰² with NPG and for its strength.^{112,189,303}

The specific area of surface is sometimes specified as area per mass, α_m , and then has no immediately obvious relation to size. Various scientific communities have their own distinct notions on how mass-specific surface area provides a figure of



Figure 10. Selected examples for additional microstructural signatures. (A) Cyclic voltammogram of current *I* versus electrode potential *E* for NPG in deaerated 0.1 mol L⁻¹ NaOH. Dark blue: scan to the switching potential at the Burshtein minimum (indicated by label). The shaded area under the surface oxide reduction peak scales with the surface area and can be used to measure that quantity. Light blue: scan to larger positive switching potential; note the larger reduction charge. (B) Grain size of NPG from electron backscatter diffraction in a scanning electron microscope. ¹¹⁶ Color codes crystallographic orientation, identifying individual crystal grains as regions of uniform color. Note up to 100 μ m grain size, more than thousandfold larger than the ligament size, $L_{\text{SEM}} = 18$ nm here. (C) Solid fraction, φ , of NPG versus initial Au fraction, x_{Au}^0 in the starting alloy.¹¹⁹ Strong deviation at low x_{Au}^0 shows that the starting alloy composition does not provide a good indicator of NPG solid fraction. Note also the dependency on the preparation protocol, free versus electrochemical corrosion. (B) Reproduced with permission from ref 116. Copyright 2021 from Elsevier. (C) Reproduced with permission from ref 119. Copyright 2021 The Authors.

merit. Yet, the related numbers mix a microstructure property (area of surface per volume) with a materials parameter of the solid phase, namely its mass density. Obviously, the same microstructural geometry will provide vastly different α_m , when it is realized as porous carbon as compared to porous gold. The parameter α_m is therefore not very informative when it comes to understanding the microstructure; thus, its use in studies of NPG is discouraged.

The area, $\alpha_{\rm V}$, per volume of the solid phase relates to microstructural geometry alone and should be preferred. In fact, this parameter has a particularly close relation to measures for the ligament size. That is apparent when one considers idealized cylindrical ligaments of diameter *L*, which have simply $\alpha_{\rm V} = 4/L$. Consequently, a specific-surface-area related size, L_{α} may be defined as

$$L_{\alpha} = 4/\alpha_{\rm V} \tag{3}$$

Through its product with the surface stress, *f*, along with the generalized capillary equation for solids, $\langle P \rangle = -1/3 \alpha_V f$,³⁰⁴ the specific surface area α_V determines the surface-induced mean pressure in a nanoscale solid. Thereby, α_V also parametrizes the impact of the microstructure on the size-induced change in the mean lattice parameter of NPG (see section 3.12 below).

In most experimental situations, the mass density of the solid phase is known with sufficient confidence and accuracy to afford computing the volume of the solid phase based on the mass (which, typically, is readily measured). By contrast, determining the surface area of porous electrodes is a more complex issue. The topic has been thoroughly reviewed by the IUPAC commission.³⁰⁵ There are several options, and each has advantages and disadvantages. Their application to NPG has been evaluated and compared,^{174,30} ⁶ leading to the recommendation that either the charge for surface oxidation and subsequent reduction should be evaluated or the charge consumed in the underpotential deposition (UPD) of Cu. UPD of Pb has also been used¹⁰⁷ and provides additional information on the prevailing facets in NPG (see section 6.1.2 below).

Because the UPD process ultimately changes the surface state, and it might be impossible to completely remove Cu^{2+} from the surface and the pore space, the authors of the present

review prefer the determination based on the formation of surface oxides (Figure 10A). The surface is oxidized up to the Burshtein minimum³⁰⁷ at around 1.65 V in Figure 10A. At this potential, an AuO layer is formed on all facets of Au. Its reduction is associated with a charge of 386 μ C cm^{-2,308} and dividing the net charge by that value defines the "electrochemically active surface area", A_{ECSA} . This quantity may be identified with the geometric area of surface in a description of the NPG microstructure, permitting the computation of $\alpha_{\rm V}$ and, with eq 3, of the characteristic size L_{α} . Slightly different values for the area-specific reduction charge of an AuO layer have been determined by other authors. One advantage of the oxidation/reduction approach is that, even though the peak widths may vary, it can be applied in different aqueous electrolytes. The disadvantage is a possible change of the surface structure already during one potential excursion to the Burshtein minimum, which is especially evident in acidic electrolytes.¹⁹² This is related to the high mobility of Au on the rough surface that is generated, as a transient state, when the oxygen adsorbate layers are lifted.³⁰⁹ In fact, as one instance of electrochemical annealing, repeated oxidation/reduction cycles on NPG are sometimes deliberately instigated in order to induce coarsening.¹²²

The commonly applied measurement of the double-layer capacity^{93,168,174,179,306,310,311} in the potential range below the onset of surface oxidation is plagued with several problems.³⁰⁵ It can only be recommended for a relative comparison of very similar samples within one laboratory after optimizing the procedure with respect to the used range of scan rates, the nature and the concentration of the electrolyte. An approach based on electrochemical impedance spectroscopy provided consistent values.³⁰⁶ However, those values were clearly distinct of the results obtained from UPD and surface oxide procedures.³⁰⁶

For catalysis with NPG, the numerical value of the material's specific surface area is of relevance. As NPG can be prepared with a wide range of ligament sizes, α_V can take on vastly different values. Inasmuch as α_V can be approximated as 4/L, typical *L* values of 20–40 nm (see section 2.2) translate into volume-specific surface areas of $\alpha_V \approx 0.1-0.2 \text{ nm}^{-1}$ and mass-specific areas of $\alpha_m \approx 5-10 \text{ m}^2 \text{ g}^{-1}$. Yet, smaller ligament sizes can be stabilized, for instance, by alloying with Pt (see section

2.2). For L = 4 nm, one then expects $\alpha_{\rm V}$ and $\alpha_{\rm m}$ as large as ≈ 1 nm⁻¹ and 50 m² g⁻¹, respectively.

In principle, Brunauer–Emmett–Teller (BET) adsorption isotherms provide an alternative approach to the specific surface area of NPG. Agreement with electrochemical oxide stripping data has been achieved.³¹² Yet, large quantities of the material are required for meaningful data, and this has prevented widespread use of the technique. Furthermore, NPG tends to coarsen during the heating cycle that is required for removing physisorbed monolayers of water, a prerequisite for meaningful BET data.¹⁰⁷ The adsorption will then not probe the original sample state, and the area value may emerge erroneously low. Indeed, comparative studies found the BET surface area drastically less than that obtained from the capacitance ratio³¹³ or Pb UPD¹⁰⁷ methods. So far, BET has not been established as a routine approach to surface area measurement for NPG.

The evaluation of small-angle X-ray scattering in its highwavenumber ("Porod") regime provides yet another approach to $\alpha_{\rm V}$.¹⁷² However, the required reduction of the scattering intensity to absolute units is rarely applied. Other approaches to a volume-specific surface area include the analysis of 3D representations^{125,127,138,287} and of the initial slope of the microstructure's autocorrelation function.⁹¹

Besides α_{v} , the area per total volume (solid plus pores) may be considered. That parameter depends not only on *L* but also on φ ; it can differ considerably from α_{v} . This emphasizes that it is imperative to disambiguate specific surface area data by specifying which measure is used.

3.5. Comparison of Different Measures for Ligament Size

How are the size measures L_{SEM} , L_G , \tilde{L} , and L_{α} interrelated? In general, materials can have widely different microstructural morphologies that allow for individually quite different relations between those size measures. Yet, inasmuch as the microstructure of NPG can be approximated by that of the leveled-wave model, one can find approximate conversion rules between those quantities. Figure 11 presents those rules, and the Supporting Information (SI), SI-1, shows their derivation. In essence, it is based on the following insights from the literature. First, results in ref 270 imply an exact close-form relation between \tilde{L} and L_{α} for the leveled-wave morphology. Second, Richert et al.²⁹⁰ present L_G for that morphology based on numerical evaluation. Third, there are experimental data for L_{SEM} and α_V (which imply L_{α}) for one and the same set of NPG samples.³¹¹

As motivated in SI, SI-1, $L_{\rm G}$ has been chosen as the reference quantity in Figure 11. The comparative display emphasizes that the numerical values of the individual measures for size can be quite different. Two important insights are that the difference depends rather weakly on the solid fraction and that L_{SEM} provides the smallest size values. That latter observation is not surprising in view of the fact that the measurement probes the dimension of the ligaments at their thinnest cross-section. Furthermore, the figure illustrates that \hat{L} is twice or more larger than the other size parameters. This is natural because the distance between the centers of neighboring ligaments represents a combination of ligament and pore-related sizes. An obvious consequence of the compilation in Figure 11 is that different measures highlight different aspects of the microstructural geometry; in other words, there is no such thing as "the true ligament size" of NPG.



Figure 11. Comparing different measures for the ligament size and their dependency on the solid fraction, φ . All sizes are normalized to the granulometry mean ligament size, L_G , and shown with logarithmic axis scaling. The geometry of the leveled-wave model is assumed to approximate NPG. L_{α} : size derived as $4/\alpha_V$ from α_V , the area of surface per solid volume. \tilde{L} : mean spacing between neighboring ligaments, as derived by eq 2, from the interference peak position in the structure factor. L_{SEM} : mean diameter of ligaments at their waist as determined from scanning electron micrographs. Shaded area: confidence limit. See SI, SI-1, for detailed discussion and references. As emphasized there, the data for L_{SEM} in this figure must be considered as tentative, simply illustrating that this measure for size provides particularly small values. Additional experiments are required for quantifying the relation between L_{SEM} and the remaining measures for size.

Uncertainties in the data set³¹¹ comparing L_{SEM} to L_G are responsible for the quite large confidence interval for L_{SEM} in Figure 11. In fact, and as is argued in SI, SI-1, the available experimental data in the literature provide at best weak support for the assumption of a linear scaling between L_{SEM} and L_G , as it underlies the representation in Figure 11. Additional experiments are needed as a basis for a reliable conclusion on that scaling. The result, as it stands, however, confirms the expectation, based on the definitions of the various size parameters, that L_{SEM} provides a particularly small numerical value compared to the remaining measures, possibly about twice smaller than the granulometry ligament size.

3.6. Grain Size

The various measures for ligament size relate to the microstructural length scale of primary interest for NPG. The grain size constitutes an additional length scale. Based on ion channeling¹⁰³ and electron backscatter diffraction images,^{116,180,281,283} it is well established that dealloying Ag–Au to make NPG conserves the grain structure of the master alloy. For standard preparation protocols of macroscopic NPG samples, grain sizes in the order of 50 μ m are typical (Figure 10B).¹⁸⁰ In other words, even though ligament sizes can vary between a few nm and few μ m, the crystalline coherency length is typically very much larger. In fact, dedicated protocols allow NPG to be made in the form of mm-sized single crystals,^{314,315} and micrometer-sized single crystals of the material are readily prepared by focused ion beam cutting from standard samples.^{142,143,276}

While the retention of the coarse grain structure of the master alloy during dealloying is well established for NPG, the finding is by no means obvious. The success of KMC simulations (see section 2.2) on rigid lattices, which maintain crystalline coherency by construction, verifies that the nucleation of new crystals is not inherently necessary for forming the nanoscale pore structure during dealloying. Dealloying Ag–Au is particularly close to that simulation

scenario for several reasons. First, the two constituents Ag and Au form a complete series of solid solutions down to room temperature, second, they share the same crystal lattice symmetry, and last, their lattice parameters are exceptionally close, agreeing to within 0.19%.¹⁸² This favors a continuous transition from the Ag-rich solid solution to the Au-rich product phase, on the same crystal lattice. Yet, experiments find the crystal lattice structure retained even under much less ideal conditions. Most notably, NPG prepared by dealloying Cu-Au also retains the master alloy grain size, 155,157 even though the lattice parameters differ by 12%, 60-fold more than for Ag-Au.¹⁸³ In spite of the large misfit, epitaxial layers of Au on the Cu-rich host crystal have been demonstrated by in situ diffraction during dealloying.³¹⁶ A large crystalline coherency length has also been reported in NPG made by dealloying the intermetallic compound Al₂Au,^{97,317} suggesting that a one-toone match of the crystal lattice structures of host and product is not mandatory for coarse-grained NPG. Apparently, a substantial amount of lattice parameter misfit can be accommodated by elastic strain. This is reminiscent of the well-known phenomenology of precipitation from supersaturated metallic solid solutions, where nanoscale precipitates tend to minimize their interfacial excess energy by remaining coherent, even at the expense of elastic distortion.³¹

Full solubility is also not required for coherency at the dealloying front. This is exemplified by Ag–Au–Pt, which can be dealloyed into single-phase nanoporous equiatomic Au–Pt with a crystalline coherency length much larger than the ligament size, even though Au and Pt are immiscible at room temperature and equilibrium.¹²⁰

Besides the idealized structure of typical NPG samples, with nanoscale pores in an extended coherent crystal lattice, dealloying can also produce nanocrystalline nanoporous metal. The grain size can then be comparable to the ligament size. The dealloying of Au-based metallic glasses exemplifies this situation.^{319,320} Nanocrystalline nanoporous metals have also been obtained by dealloying crystalline precursors, for instance, from Al-³²¹ and Mn-based¹⁹⁸ master alloys. The absence of simple epitaxial relations between master alloy and the product structure may favor such nanocrystalline nanoporous products. For NPG made from Ag–Au, the grain size can be reduced by working with severely plastically deformed master alloys,²⁸³ and the number density of crystal lattice defects can be strongly enhanced by dealloying at very high driving forces.¹⁰³

3.7. Phase Fractions

Next to L, the solid volume fraction, φ , is a fundamental microstructural parameter of a porous solid. This parameter adds up to unity together with the "porosity" (i.e., the pore volume fraction). It is relevant for transport cross sections in the pore space 322 and for mechanics.²⁷ As exposed in section 2, the external sample dimensions are preserved during an idealized preparation by dealloying (assuming that differences in the atomic volumes of the constituents can be neglected, which is a good assumption for Ag-Au). Inasmuch as the idealized process also removes all of the less noble element, φ would then agree with the more-noble element atom fraction in the starting alloy. Yet, φ -values estimated in this way may be considerably in error. This is due to spontaneous plastic deformation during dealloying. As a general rule, this phenomenon leads to substantial shrinkage, acting to increase φ . The volume reduction can range from 2% to 25% in

standard dealloying scenarios,^{92,93,103,119} and 40% or more reduction have been reported for starting alloys dilute in Au, that is, with gold atom fraction ≤ 0.2 .^{119,168} The retention of Ag (section 2) can further increase φ .^{323,324} These observations imply that reliable values of φ may not be inferred from the starting alloy composition; instead, φ requires a dedicated measurement. Figure 10C illustrates that φ of NPG can deviate substantially from the Au fraction in the starting alloy.

Classic Archimedes density measurements are problematic for solids with open porosity, such as NPG. Thus, the standard approach to φ is the separate measurement of sample mass and of external sample volume (typically using an optical measurement microscope).^{111,119} As an equivalent approach, the sample volume can be estimated from the starting alloy sample dimensions and the relative length change, measured by in situ dilatometry^{103,144} or from the displacement of markers on the sample surface,¹⁵¹ during dealloying. That approach presupposes isotropic dimension change.

The amount of densification by shrinkage during dealloying increases systematically with the dealloying rate.¹⁰³ This appears natural as faster dealloying brings lesser L and,^{91,105} hence, more impact of surface stress and surface tension. The densification is also strongly dependent on the dealloying protocol, with substantially more shrinkage during free corrosion as compared to electrochemical corrosion.^{93,119}

The microscopic mechanisms behind shrinkage during dealloying have not been conclusively identified.⁹² As sintering can be ruled out in view of the large grain size, plastic deformation appears the obvious process.¹⁰³ It has been suggested that the shear components of the stress within the ligaments, induced by their surface stress, prompt spontaneous plastic shear by dislocation glide.¹⁰³ Yet, it is not obvious which capillary parameter drives the shrinkage-surface stress (parametrizing a local stress state in the surface) or surface tension (parametrizing the area-specific excess energy of the surface). In fact, investigations of the impact of the surface on plasticity of NPG single out surface tension, as opposed to surface stress, as the relevant capillary parameter.^{118,303} When NPG shrinks, its ligaments become shorter and thicker, thereby reducing their surface area. Because that process also reduces the overall excess energy residing in the surface, surface tension provides a driving force for shrinkage.

3.8. Skin Layers on Macro- and Nanoscale Porous Bodies and Disconnection of Thin Films

The shrinkage and densification discussed so far arise from deformation processes that act throughout the bulk of NPG. On top of these processes, there is also a separate densification process that acts near the external sample surface and that results in a skin layer of locally enhanced solid fraction, with a thickness comparable to the ligament size.^{105,174} That skin layer has little effect on the external dimensions or on the mean solid fraction of macroscopic samples, yet it can contribute decisively to the densification of nanoporous nanoparticles (section 2.8). A volume shrinkage, during dealloying, of up to around 50% has been reported for nanoparticles.⁸⁰ Their densification is enhanced at lesser particle size, and the magnitude of that effect has been found consistent with a dense skin layer underneath the surface.³²⁵

A recent study has confirmed that the subsurface densification originates in enhanced inward MNE diffusion fluxes near the external surface. Those fluxes act during secondary dealloying or coarsening, and they originate in the enhanced convexity of the ligament surfaces near their termination at the external surface of a porous body.³²⁶ The same process also leads to an opposite effect, namely loss of density and disconnection, at the interface between NPG and support layers of massive (not porous) gold. This is of relevance because it leads to the widely observed, but rarely acknowledged, observation of a low-density, poorly connected layer at the base of thin films of NPG.³²⁶

3.9. Connectivity, Percolation-to-Cluster Transition, and Self-Similarity

A defining feature of NPG as a catalyst is that one and the same, homogeneous material provides both the active surface sites at the atomic scale and the supporting macroscale scaffold. This presupposes mechanical integrity and not too low strength and stiffness. In other words, percolating loadbearing pathways within the material are essential. For electrocatalysis, the scaffold also needs percolating paths for the conduction of electrons. It is therefore of interest to characterize a degree of connectivity within the solid phase of NPG's microstructure.

Reports of an anomalously low elastic stiffness in NPG^{117,275} first led to the suggestion that "broken" ligaments may be inherent in its microstructure^{117,272,327} and that its connectivity may deteriorate upon coarsening.^{27,324} Comparing the mechanics of the leveled-wave model to experiment confirmed that notion and revealed that the connectivity of the asprepared material systematically depends on φ .²⁷⁰

An important parameter in the above context is the topological genus, G. In essence, G counts the number of interconnected (as opposed to broken) ligaments within a sample of the material.³²⁸ Normalization to the characteristic volume of an elementary structural unit in the microstructure leads to a "scaled genus", g, that is independent of L or of the sample volume and so provides a measure for the inherent contiguity of the material's microstructure.^{125,268,272} Experimental tomographic reconstruction suggests excellent agreement between g in as-prepared NPG and in the leveled-wave model.²⁷⁰ Significantly, that model features a strong correlation between g and φ : as the solid fraction decreases, so does the connectivity. This effect is strong and it abuts in a percolationto-cluster transition at the solid fraction $\varphi_{\rm perc} = \frac{1}{2}(1 - \operatorname{erf}(2^{-1/2})) \approx 0.159,^{270}$ see exemplification by the φ = 0.1 leveled-wave structure in the right panel of Figure 8C. At even lower solid fractions, regular dealloying will not form load-bearing bodies of NPG. Furthermore, strength and stiffness of the material decrease dramatically as φ approaches $\varphi_{\rm perc}$ from above.²⁷⁰ Hierarchical structuring (section 2.5) has been shown to afford bodies of NPG at solid fractions below $arphi_{
m perc}$ that remain loadbearing, beating the percolation limit of structures with a single characteristic length scale.¹⁷⁸

The topology-changing event in the microstructural evolution of NPG is the breaking of ligaments through a Plateau–Rayleigh-like instability.³²⁸ The link between solid fraction and scaled genus has been rationalized as a consequence of this instability: low φ requires long, thin ligaments, and these will disconnect more readily by the instability.¹²⁴ That is also in keeping with the observation that NPG with $\varphi > 0.3$ (low aspect-ratio ligaments) tends to maintain g, whereas NPG with $\varphi < 0.3$ will gradually disconnect upon coarsening.^{124,324} Reconnection has been shown to partly compensate for the disconnection.¹¹⁶

The observations on the evolution of the connectivity during coarsening link to the question, will the NPG microstructure evolve self-similarly in that process?²⁶⁸ Experimental studies have been discussed as evidence for^{125,289,329} or against^{117,127,324} that notion. The more recent work highlights the dependence of the connectivity evolution on φ as a possible origin of the discrepancy,¹²⁴ the samples in the respective studies may have been partly above and partly below the threshold value of $\varphi \approx 0.3$.

To summarize, the state of the art provides good arguments for expecting NPG with $\varphi > 0.3$ to coarsen self-similarly. By contrast, NPG with lesser solid fraction must be expected to disconnect and/or densify during coarsening.

3.10. Tortuosity, Multiscale, and Hierarchical Nanoporous Gold

The geometry of NPG implies that mass transport in the pore space is slowed down as it follows sinuous pathways that may feature constrictions. The notion of "tortuosity" parametrizes that situation in an effective continuum description of transport. The precise nature of the link between microstructure and transport rate depends on the transport process, for instance, molecular diffusion or viscous flow.^{330,331} Consequently, there are different notions of tortuosity.

A tortuosity parameter for viscous flow in NPG has been inferred from imbibition experiments, and its value, $\tau = 3.2 \pm 0.2$, was found³³² to be compatible with that, $\tau = 3.5 \pm 0.3$, reported³³³ for porous Vycor glass.^{334,335} The agreement is in keeping with the notion that the Vycor microstructure, resulting from spinodal decomposition, is equally well compatible with the leveled-wave model³³⁶ as is NPG. Branch and geometric tortuosities for NPG, with values in the range $1.2 < \tau < 2.0$, have been estimated based on φ ,³²² or on tomographic reconstructions of nanoparticles.²⁴⁶ Apparently, these parameters remain yet to be linked to observations on transport in NPG. Structural hierarchy (see details in section 2) may reduce the constrictions and accelerate transport; the tortuosity of hierarchically structured NPG remains to be studied.

3.11. Surface Curvature, Faceting, and Roughness

The local value of the mean curvature, κ (the sum of the inverse values of the principal radii), at a given point on the surface is an important parameter, because gradients in κ provide the driving forces for transport of matter by surface diffusion that lead to coarsening. The curvature is linked to an excess, $\Delta \mu$, in chemical potential by the Gibbs–Thomson equation, $\Delta \mu = \kappa \gamma \Omega$, with γ the surface tension (i.e., specific surface excess free energy) and Ω the atomic volume (see, e.g., refs 102 and 337).

The specific surface excess free energy ("surface tension") of the continuum thermodynamics description is reflected, at the atomic scale, by broken bonds at low-coordinated surface atoms. The atomic coordination number is reduced at any type of crystallographic surface facet. It is reduced even further at step-edge and corner lines or on rough regions of the surface. The effective $\Delta \mu$ of a nanostructure is controlled by its size, and it is otherwise sensibly independent of whether a surface is rough or faceted.³³⁷ This underlines that low-coordinated atom positions, of central importance for catalysis, are an inherent and geometrically necessary property of nanostructures such as NPG, irrespective of whether their surfaces are rough or faceted. The presence of low-coordinated sites in

Review

NPG and their relevance for catalysis are well acknowledged. 6,17,24,94,338,339

High-resolution transmission electron microscopy reveals microfacets (at the scale of few atomic diameters) at the surface of NPG.^{94,338} Transitions in the crystallography of the microfacets can be induced by changes of the environment³³⁸ or by electrochemical cycles,^{108,192,340} see section 4.4. In view of the established relevance of surface crystallography for catalysis, it can be expected that these details of the microstructure are important for the catalytic activity of NPG.

Equilibration of the NPG surface structure at the scale of the ligament size requires a prevailing presence of the equilibrium facets of the Wulff shape at that scale (see, for instance, ref 341). The Wulff shape of gold exhibits facets at any temperature, up to the melting point.^{342,343} Indeed, NPG with faceted ligaments at the scale of *L* may be obtained by a custom annealing protocol.¹²¹ Yet, the mainstream of scanning electron micrographs of the NPG microstructure at that scale presents smoothly rounded and, hence, rough surfaces.¹²¹ Thus, the observations would seem to suggest that the kinetics of ordering by surface diffusion near room temperature permit microfaceting at a very small scale, but is too slow for the surface crystallography of NPG to reflect the Wulff shape.

Experimental 3D reconstructions find the average value, $\langle \kappa \rangle$, of κ in NPG either near zero^{287,344} or positive,^{127,138,268} with the near-zero results reported for symmetric microstructures with roughly equal volume fractions of pore and solid, $\varphi \approx \frac{1}{2}$.¹²⁷ That is also borne out by the leveled-wave model, which has $\langle \kappa \rangle = 0$ (at any *L*) when $\varphi = \frac{1}{2}$, along with a gradual increase in κ upon decreasing φ .²⁷⁰ The mainstream of studies of NPG work with solid fractions near 0.3, where convex surfaces ($\kappa > 0$) may be expected on average. Interfacial shape distributions, characterizing statistic correlations between the two principal radii of curvature, have been found to evolve in a roughly self-similar manner during coarsening. This has been discussed as the signature of self-similar microstructure evolution^{127,268,329} (yet, see also section 3.9).

3.12. Mean Lattice Parameter and Microstrain

When dealloying converts the master alloy into nearly pure (nanoporous) gold, the composition change will entail a change in the stress-free lattice parameter, a_0 . A small contraction of the crystal lattice after dealloying is indeed observed in NPG made from Ag–Au,¹⁰⁵ consistent with a_0 of Au falling below a_0 of Ag by about 0.19%.¹⁸² Furthermore, local strains in the crystal lattice of NPG are expected because the surface stress, a capillary force that arises from modified interatomic bond forces at low-coordinated surface atoms sites, requires compensating stresses in the bulk. As addressed in section 3.4, the mean magnitude of those stresses scales with the volume-specific surface area. Thus, the lattice of NPG will exhibit stress and strain that increase in magnitude with diminishing ligament size. A dependence of the mean lattice parameter, \overline{a} , on the ligament size may then arise, similar to the well-known size dependence of nanoparticle lattice parameters (e.g., refs 345 and 346).

The size dependence of \overline{a} appears as an obvious consequence of the surface-induced stresses. Yet, that effect is small: for NPG with clean surfaces and with $L_{\text{SEM}} = 20 \text{ nm}, \overline{a}$ was found reduced, relative to the bulk value, by merely 0.03%.¹⁰⁵ Stresses at the corrosion front can locally induce larger lattice strain (relative change in the lattice parameter)

during ongoing corrosion.^{172,281,347} Changes in surface stress controlled through adsorption or electrochemical charging will change \bar{a} , and this can be exploited for sensing or actuation with NPG.^{210,298,302,348,349} A detailed inspection of that topic is beyond the scope of this review.

While the surface-induced change in the mean value, \overline{a} , of the lattice parameter, a, of NPG is small, the distribution of lattice parameters around the mean can be quite wide. In other words, local regions of substantially different a abound. This phenomenon is quantified by the "microstrain", that is, the standard deviation of the distribution.²⁸⁵ The local lattice distortions may be important for catalysis. Density functional theory predicts that tensile strain in the tangent plane typically (experimental counterexamples show that the trend is by no means forceful)³⁵⁰ makes the surface of transition and noble metals more binding for the reactants and intermediates of heterogeneous catalysis, and this affects their reactivity.³⁵¹ The strain-dependent reaction rate at the surface of Au has been quantified and confirmed in situ.³⁵² This suggests that local strains at the surfaces of NPG ligaments will have an impact on the catalytic activity. X-ray diffraction shows that the microstrain of NPG increases systematically with decreasing ligament size. A microstrain of 0.61% was found for ligament sizes of ≈ 10 nm.¹⁰⁵ Along with the findings for the lattice parameter, the observation emphasizes that the variance of the lattice parameter distribution along the surface is much larger than the shift of the mean lattice parameter.

Yet another aspect of strain in NPG is a rhombohedral distortion of the crystal lattice in each ligament. Nanobeam electron diffraction in scanning transmission electron microscopy (STEM) revealed a compression of the lattice along the ligament long axis, along with an expansion in radial direction by several percent.³⁵³ That confirms theory for the stress and strain in the bulk that is induced by surface stresses in elongated nanoscale objects.^{304,354}

Lastly, high-resolution STEM revealed inward or outward relaxation of the outermost atomic layer at the surfaces of ligaments.^{338,353} This feature exemplifies a relaxation at both clean³⁵⁵ and adsorbate-covered surfaces³⁵⁶ that has been well established in studies of single crystals.

To summarize, the following statements can be made about the distribution of lattice parameters in NPG. The mean value does not significantly deviate from bulk Au, yet the standard deviation can be quite wide. That implies that regions of increased as well as decreased lattice parameter can be found in both the bulk and the surface of NPG.

Besides the strain state of pristine NPG, there are first hints that restructuring of the samples in catalysis experiments also changes the strain state.³⁵⁷ The TEM images used for that analysis are not ideal with respect to defocus and thus delocalization problems, yet they provide compelling evidence for the generation of lattice defects in operando. In principle, the link between microstructure and stress in NPG might be exploited for deliberately tailoring the strain state at the pore surfaces, thereby tuning the catalytic activity. This "strain engineering" of NPG is a field to be explored in the future.

3.13. Chemical Heterogeneity

The chemical composition is another important metric for characterization of NPG. Although the majority of the LNE is removed during dealloying, a significant atom fraction of residual LNE is always retained. As outlined in section 2, the net amount of retained LNE can be controlled in a wide range



Figure 12. (A) STEM tomography reconstruction of NPG. (B) Quantitative 3D elemental distribution. (C) Slice cut out of the reconstruction in (B), revealing a Ag mole fraction up to 70 at. %, close to the composition of the master alloy. (D,F) STEM image and elemental map of a pristine NPG sample. (E,G) STEM image and elemental map of NPG after annealing showing a homogenized distribution of elements. (H) APT

reconstruction of NPG. (I) Line profile along the line indicated in (H), showing higher Au content at the surface of a ligament. (J-N) kinetic Monte Carlo simulations of dealloying and annealing. (J) Top-view of surface of model sample. (K) Surface roughening during dealloying. (L) Initial ligament network after primary dealloying. (M) Coarsened ligament network after secondary dealloying. (N) Coarsened and homogenized network after annealing without corrosion. (A–C) Reproduced with permission from ref 94. Copyright 2017 Elsevier. (D–G, J–N) Reproduced with permission with permission from ref 90. Copyright 2017 The Authors under License CC. (H–I) Reproduced with permission from ref 95. Copyright 2017 Elsevier.

by proper choice of the dealloying conditions. That is of relevance because, as will be discussed below in sections 4.1, 5.3, and 6.1, the residual LNE has important consequences for the catalytic activity. Mechanical properties may also depend on the LNE content, either indirectly through the solid fraction (which is higher if not all LNE has been removed),³²³ or directly through the effect of the composition on the local strength of the ligaments.³⁵⁸

The impact of the residual LNE on the materials behavior, for instance for catalysis and mechanics, depends on the spatial distribution of that component. For instance, catalytic reactions or the nucleation of lattice dislocations in mechanics will depend on the local concentration of LNE at the surface. It is therefore of relevance that the LNE is distributed inhomogeneously within NPG. Typically, the overall content of the LNE is determined by energy-dispersive X-ray analysis (EDX), which probes a layer of about 1 μ m thickness, i.e., several times the diameter of typical ligaments. Within a ligament the concentration of the LNE changes by surface segregation^{191,359} during storage or during (electro)chemical reactions. XPS with a conventional Al K_{α} source probes a layer thickness of 4.4 nm limited by the inelastic mean free path length of the photoelectrons in NPG. A single number, namely the net LNE atom fraction, is then not sufficient to describe the impact of the extra component on the material's behavior. More surface sensitive determination of the LNE requires excitation with soft X-rays.¹⁹¹

Several publications have pointed toward a macroscale heterogeneity in the LNE distribution. That heterogeneity takes the form of a skin layer with a thickness <100 nm at the macroscopic outer surface of NPG samples. As compared to the bulk, the surface layer exhibits reduced pore size (section 3.8) and an enhanced LNE content.^{35,93,174}

As has been discussed in section 2, the nanostructure formation during dealloying rests inherently on a competition between the trend for corrosion to passivate the alloy surface by enriching it in the MNE and a trend for the nucleation and growth of nanoscale pores that consume the pristine master alloy below the passive layer. It is then not surprising that the nanoporous product structure of dealloying exhibits heterogeneities in composition at the nanoscale, including regions that are practically pure MNE, yet also regions that still retain the LNE-rich master alloy. Indeed, element-resolved 2D micrographs show nonuniform element distribution in NPG made from Ag-Au³⁶⁰ and from Cu-Au.^{156,282,361} Nature and origin of these features could only be unraveled with the advent of composition-resolved 3D nanotomographic reconstruction. Using analytical transmission electron microscopy or the tomographic atom probe, three independent studies of NPG made from Ag–Au revealed (Figure 12A–I) that Ag in the porous material is typically localized in mostly buried Agrich regions that are embedded in a matrix of essentially pure Au.^{90,94,95} KMC simulations naturally reproduce that microstructure, revealing that the Ag-rich regions are relics of the

master alloy that have never been exposed to the corrosive environment (Figure 12J–N).^{90,91} This is consistent with passivation, in other words, with an at least one atom thick layer of pure Au covering the Ag-rich regions. Experiment confirms that the Ag content within those latter regions is not exceeding that of the master alloy.⁹⁴

KMC simulations reveal that the characteristic size of the Ag-rich regions scale with the initial ligament size that is generated during primary dealloying, prior to coarsening.^{90,91} As the regions are buried, and as bulk diffusion in Ag–Au is negligible near room temperature, ^{80,92,362} the regions cannot grow when the microstructure coarsens during secondary dealloying. In principle, this offers the opportunity of studying their size experimentally as a basis for unraveling the initial structure size of the nanostructure formation during primary dealloying, right at the corrosion front.

The present understanding of dealloying suggests a strong trend for passivation in the form of an MNE-rich ligament surface during dealloying. This is confirmed by tomographic atom probe data^{95,363'} and by KMC simulation.⁸⁹⁻⁹¹ Yet, once the dealloying process is stopped, LNE that was originally buried in the bulk of ligaments may accumulate at their surface. The underlying processes may be surface migration during coarsening by surface diffusion or surface reconstruction driven by changes in the environment. A trend for surface segregation may contribute as a driving force. During the coarsening of NPG (see section 2.2), the surface migrates and sweeps the crystal lattice. As a consequence, LNE-rich regions that were originally buried underneath the surface are exposed. Their LNE content can then be redistributed by surface diffusion. In this way, the bulk composition is homogenized even though bulk diffusion is inactive. LNE-rich regions connected to the surface have indeed been identified in experiment,⁹⁴ bulkhomogenization during coarsening has been confirmed by experiment and surface diffusion has been identified as the active process by simulation.⁹⁰

Reconstruction is well documented for Au surfaces and is known to react to changes in the environment.³⁶⁴ Not surprisingly, the surface of NPG has been found to reorder and/or migrate during operation in catalysis^{24,58,191,360,365} or during potential cycles in electrochemistry^{108,122,340} (see also sections 3.11, 4.4, and 6.1.2).

This brief overview shows that a complete characterization of the LNE distribution requires that the average LNE content, an LNE content profile (depending on the sample geometry), and an analysis of LNE-rich regions (size and composition) are all considered. As such measurements consume time and experimental resources, reasonable trade-offs must be found for the number of samples investigated. For catalysis, immediate interest is on the composition right at the pore surfaces. Yet, the composition field throughout the material may be of importance for the stability and for the evolution of the material under operating conditions.

3.14. 3D Representations

It is readily estimated that 1 mm³ of NPG with 10 nm ligament size contains around 10¹⁴ ligaments, ¹⁷⁸ and this represents an extreme structural complexity. Most of the microstructure metrics discussed in the previous sections condense individual, relevant aspects of this complex geometry into single scalar parameters. The abstraction provided by this strategy is indispensable as a basis for identifying and understanding microstructure-property relationships and for comparing and assessing results from separate studies. Yet, some situations require more complex representations of the microstructure. This applies specifically to numerical studies exploring transport behavior,^{247,366,367} wave propagation,³⁶⁸ optical properties, 215,369 or the mechanics of porous solids and specifically NPG. Volume elements containing a representative sample of the 3D microstructure (representative volume elements, RVEs) of a stochastic material for those purposes can be generated in various ways, as recently reviewed.³⁷

Some studies in the area of NPG directly adopt experimental 3D reconstructions as their RVEs,^{272,290,371} and approaches based more generally on experimentally informed RVEs have been reviewed.²⁶⁹ The major asset of this experiment-based pathway is that it warrants a realistic geometry. Yet, there are also drawbacks—conceivable artifacts from the reconstruction need to be considered, periodic boundary conditions do not apply, and it is not typically possible to vary microstructural parameters.

Overcoming the limitations of experimental reconstructions requires models-more or less realistic abstractions matching selected aspects of the real geometry-for generating 3D representations of the NPG microstructure. A prominent example is the truss structure by Gibson and Ashby,³⁷² which successfully describes the mechanical behavior of a surprisingly wide variety of open-cell foams in nature and technology³ and which provides benchmarks for the mechanics of NPG.^{27,142,169,171} This structure takes the form of a crystalline array of nodes, connected by struts. It exemplifies periodic models for porous materials. Periodic network models based on other crystal structures, for instance simple cubic^{215,371} or diamond,³⁷⁴ have also been considered. The stochastic nature of NPG can partly be accounted for when random displacements of the nodes are incorporated,³⁷⁴ or when struts are randomly cut.³⁷⁵ Other periodic models for NPG adopt minimal surface geometries, such as the gyroid struc-ture.^{272,313,371}

Periodic models are at odds with the stochastic nature of the NPG microstructure and with its statistical isotropy. Models representing the pore structure by random arrays of spherical voids in a solid matrix^{354,369,376} remedy that problem, but their geometry conflicts with the network character of NPG. Furthermore, along with all of the aforementioned approaches, they do not naturally reflect the systematic variation of NPG's connectivity with the solid fraction. In those respects, spinodal structures are closer to NPG. Several modeling studies of NPG have generated spinodal-like RVEs by implementing spinodal decomposition numerically in Monte Carlo^{129,275} or phase field^{272-274,377} simulation. The resulting geometries appear quite close to NPG. Yet, spinodal structures can lose connectivity when allowed to enter the "late stage" (coarsening) regime, ^{378,124} and that regime is readily encountered in simulations of spinodal decomposition. Indeed, studies modeling NPG with solid fractions near 0.3 by numerically generated spinodal structures report effective

Young's moduli that vary more than 10-fold, ranging from 0.004^{272} and 0.007^{275} to $0.05^{274,377}$ of the bulk value. In view of the strong correlation between effective elasticity and connectivity (section 3.9), the discrepancy confirms that the topology of numerically generated spinodal structures is not unique.

The observations above emphasize the relevance of the leveled-wave model (section 3.2). That model approximates early stage spinodal decomposition, avoiding the extra degrees of freedom for disconnection that come with coarsening during the later stage. As compared to other strategies toward 3D representations of NPG, the leveled-wave model is distinguished by its uniqueness and by its ability to reproduce, with no free parameters, several decisive properties of NPG, specifically (1) the systematic variation of the connectivity with solid fraction, (2) the isotropic macroscopic materials properties, and (3) the presence of a well-defined dominant wavelength. Lastly, as outlined in section 3.2, the microstructure of the model results from energetics and kinetics that act in closely similar form in the real material. In that sense, the model appears particularly suitable as a basis for guiding the intuition toward new insights into the fundamentals of NPG.

4. GAS PHASE CATALYSIS

In 2006, NPG was established as an active catalyst for lowtemperature aerobic oxidation reactions as demonstrated for CO oxidation.^{6,7} While oxide-supported Au nanoparticles (NPs) had been in the focus of gold catalysis since the 1980s, starting with the work by Haruta¹ on CO oxidation and Hutchings⁵ on the hydrochlorination of acetylene, the high activity found for NPG at room temperature and even below was quite unexpected at that time. This was due to the large structural differences between NPG being a (bi)continuous metal structure with ligament sizes in the range of several 10 nm and Au nanoparticle catalysts, which were found to be catalytically active only for a narrow range of particle sizes around 5 nm and only in the presence of particular (reducible) oxide supports.

Shortly afterward, it turned out that the material is not only a good catalyst for total but also for partial oxidation reactions. The latter ability was first exemplified for aerobic methanol oxidation, yielding methyl formate at high selectivity (at not too high oxygen partial pressures).8 This finding initiated a wealth of further experimental as well as theoretical studies in the following years, which widened the knowledge regarding the catalytic scope of NPG considerably. Apart from other oxidation reactions, such as preferential CO oxidation (PROX) or the oxidative coupling of alcohols and amines to the corresponding amides, it was demonstrated that even reductions are feasible, and this oftentimes in gas and liquid phase. To this end, it is important to note that some of these reactions require modifications of NPG, e.g., by introduction of oxide particles deposited onto the ligaments of NPG or the use of NPG systems with a different LNE than Ag.^{59,379-381}

The interest in the material was also driven by the perception that NPG is structurally and chemically less complex than oxide-supported Au nanoparticles. This perception was corroborated already by the first reports of NPG's partial oxidation capabilities demonstrated at the example of methanol conversion to methyl formate, which could be explained by insights from previous surface science studies on Au single crystals.^{8,382} Hence, NPG may open a way to potentially address the catalytic properties of Au more

directly than it was possible for Au nanoparticle catalysts. The notion that NPG is a metallic catalyst (lacking the complexity arising from a metal—oxide interface) renders a combination of catalytic studies on NPG with surface science experiments on single-crystal surfaces and computationally based methodologies a particularly promising strategy to attain a mechanistic understanding of heterogeneously catalyzed processes at the atomic level. This strategy is clearly not restricted to the abovementioned partial oxidation reaction but can also be applied to total oxidation reactions such as CO oxidation. In this realm, it could, e.g., provide insights to the question, which oxygen species or oxide surface phases are responsible for the efficient catalytic turnover.^{46,56} The strategy has been successfully used over the past decade to obtain insight into several catalytically interesting reactions, and several reviews have summarized these efforts already.^{72,383–386}

Nevertheless, it is important to note that the current mechanistic understanding of the aerobic oxidation chemistry on NPG is still fragmentary as far as some important aspects are concerned. One question, which immediately came up when the first reports regarding NPG's catalytic potential were published, relates to the mechanism of O₂ activation. Notably, the binding of molecular oxygen is very weak on Au surfaces and the probability for its subsequent dissociation is negligible even at pressures of several mbar, where the catalytic experiments with NPG were carried out. Accordingly, its ability to catalyze aerobic oxidation reactions cannot be explained on the basis of the Au surface chemistry alone. In agreement with that, experimental as well as theoretical insight gained over the last years indicate that oxygen activation indeed does not occur on Au sites but that residual amounts of the less noble metal (Ag for most of the catalytic studies on NPG) are likely to be responsible for this step. At the same time, a comparison with Ag, exhibiting a distinctly different oxidation chemistry, suggests that all subsequent reaction steps take place on Au surface sites and not on Ag sites.^{46,382,387,38} As far as the chemical nature of the oxygen species, which are involved and determine NPG's selectivity for partial or total oxidation reactions is concerned, many questions, however, remained unsolved so far. The selective formation of coupling products (esters) observed for alcohol oxidation, for instance, was primarily explained in the literature on the basis of chemisorbed O atoms and their Brønstedt basicity, enabling the necessary H-abstraction steps.^{8,383} Such a perspective, however, neither takes sufficiently into account the wealth of oxygen species or surface oxide phases, theoretically predicted and also partly observed in Au single-crystal studies, nor their Lewis basicity (nucleophilicity), i.e., the ability for oxygen transfer steps and formation of total oxidation products. Even though a number of experimental findings could be interpreted on this basis, for a more differentiated mechanistic picture, a detailed view on the potential range of oxygen species and associated activation and reaction pathways on the surface is mandatory.

Aiming at complementing other reviews about NPG in this respect, we have thus chosen a corresponding focus in this chapter.^{17,39,41,72,389} The current state of knowledge will be summarized at the examples of CO and methanol oxidation, representing the most intensively studied examples of total and partial oxidation reactions catalyzed by NPG. In section 4.1, we dwell on corresponding insight from theory, whereas section 4.2 provides an overview over the most important results from UHV single-crystal studies. Section 4.3 elucidates the

connection of these fields to the catalytic behavior of NPG (in the gas phase), and finally section 4.4 addresses the structural evolution of NPG during catalytic reactions.

4.1. Surface Composition, Active Sites, and Reaction Mechanisms from Computational Studies

4.1.1. Aerobic Oxidation Reactions and O₂ Activation. The ability to catalyze aerobic oxidation makes nanoporous gold attractive in the context of green chemistry and sustainable development. Molecular oxygen (or ideally air) is an inexpensive, abundant, and environmentally friendly oxidant. The activation of a dioxygen molecule is known to be central to all aerobic reactions. Therefore, this important reaction step will also be one of the reoccurring themes throughout this subsection, where current understanding of the mechanistic aspects of the catalytic activity (including open questions and opposing viewpoints) will be reviewed.

The activation of the adsorbed O2 molecule on Au is believed to be a difficult step in aerobic oxidation reactions catalyzed by the material. Here, "activation" subsumes all chemical transformations leading to reactive oxygen species. This may either be the O-O bond cleavage, producing two adsorbed O atoms (with a formal charge state O^{2-}), or nondissociative pathways. Options of the latter kind are connected with $\tilde{O_2}^-$ or O_2^{2-} surface species which exhibit a weakened O-O bond, resulting from a charge transfer from the metal to the adsorbate. Moreover, addition/abstraction reactions between O2 and surface species formed after adsorption of the other reactant are conceivable, which then leads to adsorbed peroxide derivatives, such as OCOO²⁻ or HOO⁻. (In the following, we will omit charges because such an assignment represents only one formal way to describe the scenario, based on the idea of a complete charge transfer so that the Au surface atoms and the adsorbate carry opposite charges of the same magnitude. An alternative perspective which is widely accepted by many scholars emanates from (uncharged) radicals, keeping in mind that the unpaired electrons of the adsorbate atoms are in fact involved in the bonds with the metal atoms at the surface.) As noted in many prior reviews, $^{17,39,42,383,386,388,390-394}$ not

As noted in many prior reviews, 17,39,42,383,386,388,390-394 not only the activation barrier for O₂ dissociation is very high on extended Au surfaces, (>1 eV according to theoretical estimates)^{51,53,54,395,396} but, and more importantly, also the adsorption of O₂ is very weak. Calculated adsorption energies range between -0.28 and -0.14 eV on various stepped Au surfaces, 51,54,396,397 while no adsorption occurs on Au(111).53,395 Adsorption energies of comparable magnitude were also found for lattice defects exposing low-coordinated surface sites such as twin boundaries which exhibit 6-fold coordinated surface sites.398,399

Based on these results, O_2 activation is unlikely to take place on Au surface sites during aerobic oxidation reactions catalyzed by NPG. Instead, there has been growing evidence over the past decade that residual impurities of the LNE (typically Ag) are of key importance for this step.^{51–53,55,397,400} Other structural features of NPG which are potentially important for binding and activating O_2 are the high abundance of lowcoordinated atoms on its ligaments and the occurrence of crystallographic strain, dislocations, and preexisting atomic oxygen species.^{51,54,338,360,398,399} Once surface atomic O is formed on nanoporous gold through one of the mechanisms outlined below, it can easily react with various adsorbates, resulting in partial or total oxidation products. Notably, chemically different types of adsorbed oxygen exhibiting different reactivity appear to exist on both single-crystalline Au surfaces and NPG, as will be discussed in more detail in sections 4.2 and 4.3.

While the way how dioxygen is activated and supplied seems to be characteristic of NPG and its special features, the remaining reaction steps (determining the catalytic behavior with respect to total or partial oxidation reactions) are in general agreement with the surface chemistry observed on well-defined single-crystal Au surfaces under UHV conditions when atomic O was preadsorbed.^{45,48,401–403} Specifically, it has been frequently emphasized in the literature that the selective oxidation chemistry of NPG corresponds to gold rather than to silver,^{8,17,39,392,404} a conclusion drawn on the basis of insight gained from UHV single-crystal studies regarding the specific and fundamental differences of Au and Ag surface chemistry.^{45,47,382}

In spite of extensive research efforts, the detailed molecularlevel mechanism of O_2 activation on NPG remains a matter of debate. Although experimental studies provide evidence that NPG is able to dissociate O_2 ,^{405,406} these studies provide no insight into the reaction mechanism. In turn, it is not known if adsorbed O_2 reacts with surface species or dissociates directly into adsorbed O atoms. The dominant mechanism may depend on the specific reaction conditions. Moreover, the different resulting active oxygen species may impact the catalytic properties of NPG such as activity or the selectivity of partial oxidation reactions.

Whereas a large body of experimental literature assumed that O_2 dissociates into O atoms at the active sites of NPG (presumably Ag or bimetallic sites), ^{17,39,42,383,386,388,390–394} the relatively high activation barrier for O2 dissociation typically found in computational studies does not readily fit into this picture.^{51,53,54,395,396} In this respect, it is important to note that high activation barriers are not limited to low-index surfaces with coordinatively saturated Au atoms but are also found for stepped surfaces with low-coordinated sites. In case of bimetallic Au-Ag sites, expected to be present on the surface of NPG prepared from Au-Ag master alloys, activation energies in the range of 0.4-0.7 eV were calculated. Even though these barriers are lower than found on pure Au surfaces, these values are still too high to explain the high activity of NPG observed at room temperature and even below. 51,53,54,407 The lowest activation energies for $\rm O_2$ dissociation (0.4-0.5 eV) were calculated on bimetallic Au-Ag sites on a stepped AgAu(211) surface for which Au atoms at the steps were surrounded by ensembles of three or more Ag atoms located on the surface and at subsurface positions (Figure 13a,b).⁵⁴ In one of the considered structures, monoatomic Au chains were decorated by adsorbed O atoms so that a chain-like 1D "gold oxide" consisting of repeated squared AuO₄ structural fragments is formed. O₂ dissociation taking place at a vacancy site (Figure 13c) was connected with a barrier of only 0.4 eV. The authors also demonstrated from first-principles thermodynamics that such arrangements of Au-Ag were more favorable than others. Another study found a slightly higher barrier for O_2 dissociation of 0.56 eV on a AgAu(110) surface, where Au was occupying low-coordinated "comb" sites, again forming monatomic Au chains, surrounded by Ag in subsurface ("valley") positions.⁵³ The presence of monatomic Au chains surrounded by Ag is a structural similarity between the latter structure as shown in Figure 13a. Another similarity results from the observation that the



Figure 13. Activation energy (in eV) and transition state geometry for O_2 dissociation on a AgAu(211) stepped surface (a) with Ag atoms in the rows next to the step Au atoms, (b) with a three-atom Ag ensemble, and (c) same as (a) but with coadsorbed O, corresponding to four O vacancies. Color coding: Au, yellow; Ag, gray; O, red. Adapted from ref 54. Copyright 2016 American Chemical Society.

dissociating O atoms form a linear O–Au–O fragment. Such a fragment represents the shortest possible O–Au–O chain and is a structural element of a 1D Au oxide. One- and twodimensional (branched) –(O–Au)– chains were identified in several theoretical studies as thermodynamically most stable forms of adsorbed oxygen on various Au surfaces.^{51,54,408–410} As shown in Figure 14, it was predicted theoretically that surface O atoms on a stepped Au surface tend to self-organize into chain structures with O–Au–O structural motifs.⁴¹¹ On a bimetallic Ag–Au surface, formation of such oxygen chains is predicted to be accompanied by surface and subsurface diffusion of Ag atoms which tend to segregate near the -(O-Au)– chains, further stabilizing them.^{365,411}

In the case of CO oxidation, computational studies have explicitly considered a bimolecular mechanism, in which adsorbed O_2 (from here on adsorbed species are denoted by *, O_2^*) directly reacts with CO* on Au, Ag, or bimetallic surface sites and identified pathways with lower barriers than for the dissociation of O_2 .^{55,397–399,412,413} The set of elementary reactions involved in the two mechanisms (O_2 dissociation (M1) and bimolecular reaction (M2)) is given in Scheme 1 for oxidation of CO.

Figure 15 exemplifies the role of Ag in CO oxidation following the bimolecular reaction mechanism for Au(321), which exhibits different low-coordinated surface sites. As seen from Figure 15b, impurities strengthen the rather weak O_2 adsorption on 6-fold coordinated Au sites. However, this increase in O_2 binding energy is associated with an increase of the reaction barrier of the subsequent association step, rendering also this mechanism limited by O_2 adsorption.

Fajín et al.⁴¹⁴ performed KMC simulations of the reaction rates of CO oxidation (CO + O_2) on pure Au(110) and Ag(110) and found no reaction on Au(110), whereas the theoretically estimated TOF on Ag(110) was comparable to experimentally reported values on NPG. That allowed the authors⁴¹⁴ to propose that patches of pure Ag could be active sites for O_2 dissociation on NPG. Another theoretical study performed microkinetic modeling of CO oxidation on



14 ps

21 ps

Figure 14. Snapshots of an AIMD simulation showing -(Au-O)- chain formation from individually adsorbed O atoms on Au(321) without Ag impurities. (3×2) unit cell and O coverage of 0.17 ML. Color coding: Au, yellow; O, red. Reproduced with permission from ref 411. Copyright 2018 American Chemical Society.

Scheme 1. Dissociative and Associative (Bimolecular) Mechanisms Discussed for CO Oxidation

M1 (O ₂ dissociation)	M2 (bimolecular):
$O_2 + * \rightarrow O_2^*$	$O_2 + * \rightarrow O_2^*$
$O_2^* \rightarrow 2O^*$	2CO + * → 2CO*
$CO + * \rightarrow CO^*$	$CO^* + O_2^* \rightarrow OCOO^* + O^*$
$CO^* + O^* \rightarrow CO_2^*$	$CO^* + O^* \rightarrow CO_2^*$
$CO_2^* \rightarrow CO_2$	$2CO_2^* \rightarrow 2CO_2$

Au(321), Ag(321), as well as on AgAu(321) bimetallic surfaces and found similarly negligible activity on pure Au(321) but much higher activity on Ag(321), whereas intermediate activity was observed on bimetallic AgAu(321) surfaces with varied concentrations of Ag.⁵⁵

Somewhat different models of stepped surfaces based on Au(111) and Au(100) slabs where some metal atoms were removed to create channel-like depressions were employed in another computational work.³⁹⁷ These depressions exhibited a stepped profile exposing infinite rows of low-coordinated Au atoms. In some of the considered models, a varied amount of Ag impurity atoms was added to replace Au atoms at the reaction site. The study reported an increasing binding strength of O_2^* (more negative adsorption energy) with increasing amounts of Ag impurity atoms at the adsorption site, in agreement with the findings on Au(321).⁵⁵ Concomitantly, higher reaction rates were predicted (using a microkinetic model) on bimetallic surfaces as compared to pure gold surfaces. In an extension to the bimolecular mechanism described above, the attack of the OCOO*



Figure 15. Direct reaction of CO* with molecular oxygen. (a) Top view of reactants, products, OCOO* intermediate, and transition states on Au(321) with Ag impurities. Selected bond distances are given in Å. (b) Reaction energy profiles in eV. Energies are given with respect to the CO and O_2 molecules in the gas phase and Au(321) with or without Ag impurities. The horizontal bars denote axis breaks. The adsorption energies of CO* on pure Au(321) and bimetallic surfaces with one and two Ag atoms at the reaction site are -0.80, -0.76, and -0.79 eV, respectively. Color coding: Au, yellow; Ag, gray; O, red; C, black. Adapted with permission from ref 55. Copyright 2019 American Chemical Society.

intermediate by a second coadsorbed CO* molecule was found to lower the barrier for its dissociation.³⁹⁷ Hence, instead of one CO₂ molecule resulting from a reaction of a single CO* molecule with an O₂* molecule, two CO₂ molecules were formed with a low activation barrier. Assuming this mechanism, microkinetic modeling predicted higher rates of CO oxidation compared to a conventional bimolecular mechanism, especially on pure gold and on some of the Ag substituted models.

To summarize, theoretical studies predict the adsorption of O_2 on Ag or bimetallic sites of NPG. Adsorption on Au sites is found to be too weak to result in a reaction. Several mechanisms of O_2 activation have been discussed. Most mechanistic studies focus on a dissociative pathway with oxygen atoms as the reactive species. In the case of CO oxidation, the computational results suggest that a bimolecular

mechanism, in which CO^{*} reacts with O_2^* , is the more likely pathway, at least at low temperatures because of considerably lower activation barriers associated with the bimolecular mechanism as compared to O_2 dissociation.

4.1.2. Water as a Co-catalyst in CO and Alcohol **Oxidation.** A possible beneficial role of water in cocatalyzing CO oxidation was reported in catalytic studies on supported Au nanoparticles,^{415–417} and a similar rate enhancement by moisture was also experimentally found for CO oxidation on NPG.³⁹⁰ Even though the underlying mechanistic details are likely to be different, and related to the oxide support (formation of surface OH groups) in the former case, it is interesting to understand the effect of water on the reactivity of both types of Au catalysts. Notably, moisture could either be inadvertently present in reaction mixtures (as an impurity) or water may be a byproduct of the catalytic transformation as in the partial oxidation of organic compounds (such as methanol oxidation). It has been suggested that water could facilitate O_2^* activation by converting O_2^* to surface hydroperoxyl (OOH*). However, other ways by which water may be involved have also been proposed and investigated. The proposed roles of water in enhancing catalytic activity of supported Au nanoparticles were comprehensively summarized and classified in four categories: 418 "(i) maintain cationic state of gold $(Au^{3+} \text{ or } Au^{+})$, (ii) direct involvement of H₂O and OHgroups in CO oxidation, (iii) activation of O₂ molecules, and (iv) transformation of catalytic intermediates and inhibitors (spectators) such as carbonate species".

Computational studies played a crucial role in addressing possible mechanistic scenarios of how either water (an impurity or byproduct) or an alcohol molecule itself (in the context of partial alcohol oxidation) may help to activate O_2^* . The latter aspect is related to an important question: What nondissociative mechanistic pathways may exist for O_2 activation that are based on an involvement or assistance of the reductant molecule itself? A formation of OOH* as a result of an abstraction of a hydrogen atom from water or an α -H from the OH-group of an alcohol molecule by O_2^* (note the analogy) was proposed in a computational study about methanol oxidation on Au(111):³⁹⁵

$$H_2O^* + O_2^* \to OOH^* + OH^*$$
⁽⁴⁾

$$ROH^* + O_2^* \to RO^* + OOH^*$$
(5)

Several pathways involving these reactions were considered.³⁹⁵ For a simple hydrogen transfer from water or from methanol, quite large barriers of 0.84 and 0.91 eV, respectively, were calculated. However, the authors also found lower-energy transition states involving a mediator H₂O or methanol molecule connected to the reactants via a chain of H bonds that facilitated H transfer to O2* significantly lowering the barriers to about 0.45 eV. In the next step, OOH* is expected to dissociate to OH* and O*, but quite a large barrier of 0.79 eV was calculated for this process. Therefore, it was suggested that the OOH* intermediates themselves might be acting as oxidizing agents rather than their dissociation products OH* and O*. To this end, a surprisingly small barrier of only 0.3 eV was calculated in that work for a transfer of an α -H from the OH-group of methanol to a surface OOH* species resulting in a methoxy (CH_3O^*) species and $H_2O_2^*$. Nevertheless, this activation barrier is larger than the values of 0.13 and 0.08 eV, as calculated in the same work, for the analogous α -H abstractions by O* or OH*, respectively.

Generally, it seems that a high concentration of methoxy on the surface helps converting all formaldehyde that is formed in the subsequent dehydrogenation step to the partial oxidation product methyl formate, most of which desorbs without reacting further. This is possible thanks to low concentrations of the strongly oxidizing surface species (O* and OH*), which otherwise would drive overoxidation to the total oxidation product CO_2 . In conjunction with relatively low diffusion barriers for adsorbates on Au surfaces, these factors appear to be the basis of the high selectivity of aerobic methanol oxidation toward partial oxidation products, as discussed in the following.

To further investigate nondissociative routes of O₂ activation discussed above, a Au–Ag(111) surface model was considered, where Ag impurities were introduced as Ag dimers substituting two adjacent Au atoms.⁴¹⁹ Much lower activation barriers were found for the formation and dissociation of the OOH* species⁴¹⁹ than calculated in a previous study for the same reaction steps on pure Au(111) without additional H₂O molecules as mediators.³⁹⁵ The α -H-transfer from methanol to O₂* proceeded with an activation energy of 0.5 eV and the dissociation to O* + OH* with a barrier of 0.45 eV. The α -H abstraction from the OH-group of methanol by OOH* revealed an activation barrier of 0.32 eV, very close to the result mentioned above.³⁹⁵ The hydrogen peroxide (H₂O₂*) formed in this process was found to easily dissociate yielding two OH* groups with a negligible activation barrier.

Significantly smaller activation barriers than found on a Au(111) surface, all lying below 0.4 eV, were reported for the formation of OOH* from O_2^* and adsorbed water or methanol and its dissociation to O* and OH* on a stepped Au(310) surface.³⁹⁶ Likewise, comparatively low barriers were calculated for the H transfer to O_2^* from water and for OOH* dissociation on a stepped Au(321) surface, with or without Ag impurities.⁵⁵ Figure 16 shows a reaction network considered



Figure 16. Possible pathways for $CO^* + O_2^*$ reaction over gold including those cocatalyzed with water.⁵⁵ Reactants and products are in yellow, and water as cocatalyst is in blue. Arrow colors mark four different mechanisms: purple, dissociative mechanism; orange, associative mechanism; blue, water catalyzed mechanism via dissociation of OOH*; green, water catalyzed mechanism via direct reaction of OOH* with CO*.

including the pathways involving water as a reactant. The first step (a H transfer from adsorbed water to O_2^*) exhibits a barrier of 0.30 eV on pure Au(321) which remains almost unchanged on Ag substituted surfaces, whereas the barriers for OOH* dissociation increase from 0.26 eV on pure Au(321) to 0.35–0.49 eV on Ag alloyed Ag–Au(321). Alternatively, OOH* can react directly with CO* to give CO₂ + OH* with a



Figure 17. General mechanism for methanol oxidation over gold. The green arrows show the coupling route, while the red ones show the overoxidation route. Reproduced with permission from ref 421. Copyright 2021 The Authors.

barrier of 0.36 eV. Possibly, the lowering of activation barriers compared to the Au(111) model can be attributed to more flexible low-coordinated Au atoms at the step sites of Au(321)and Au(310) helping to stabilize the transition states. Despite rather low calculated barriers for water-assisted O2 activation on Au(321), microkinetic modeling predicted the bimolecular $CO^* + O_2^*$ mechanism via an OCOO* intermediate to be the fastest reaction route for CO oxidation. In this respect, theory could not explain conclusively (i.e., on the basis of the selected models and investigated mechanisms) why the CO oxidationas found in the experiment-is accelerated by water.⁵⁵ This inability of theory to provide evidence for the beneficial effect of water in CO oxidation catalysis on NPG may be related to the reaction network considered. To this end, an interesting idea was proposed in a recent computational work,⁴²⁰ where very fast pathway for CO oxidation on Au(100) was suggested in the context of PROX. It involves the formation of a hydrocarboxyl intermediate, OCOH*, from CO* and OH*. This species could react with O2* with a negligible barrier forming CO_2 and OOH*. In combination with OOH* + CO^* \rightarrow CO₂ + OH* this could provide a claimed very fast pathway for CO oxidation.⁴²⁰

4.1.3. Detailed Mechanism of Aerobic Alcohol **Oxidation from Computational Studies.** While reviewing the proposed role of surface hydroperoxyl (OOH*) in O_2 activation, we have already mentioned the potential importance of this intermediate in the context of alcohol oxidation. Despite the significantly lower activation barriers predicted for the methanol-assisted OOH* route (0.35-0.5 $eV^{55,396,419}$) as compared to the O₂* dissociation pathway on Au and even Ag–Au surfaces (0.45 to ~1 eV,^{51,53,54,407}), the former mechanism has not been given attention in the published experimental work on methanol oxidation so far. The standard mechanistic model used in many experimental studies emanated from the postulated O2* dissociation into (single) adsorbed O* atoms and their Brønsted basic properties.^{45,48,382,401} The corresponding scheme in Figure 17 is based on consecutive H-abstraction reactions. In this way, the conversion of methanol to methoxy and subsequently (by β -H abstraction) to formaldehyde is enabled. (The notations α and β are applied to the H atom of the OH and to the hydrogen atoms of the CH₃ group, respectively.) Thus, the

transformation sequence is $CH_3OH^* \rightarrow CH_3O^* \rightarrow CH_2O^*$. While surface O* atoms are mostly considered the key reactive intermediates, a mechanistic discussion restricted to this species might be an inappropriate simplification in view of the accumulated body of computational results reviewed above.

Along these lines, other oxygenated species, such as O_2^* , OH^* , OOH^* , or CH_3O^* , are likely to be important intermediates in the reaction network, as described in section 4.1.2 above. These species could exhibit sufficient (Brønsted) basicity as well—at least for the α -H abstraction converting methanol to methoxy—and some of them also for β -H abstraction from methoxy, yielding formaldehyde or could be otherwise involved in the corresponding elementary steps (e.g., OOH* may dissociate to strongly basic O* and OH*). Consequently, four possibilities have been considered for the formation of methoxy (CH₃O*) in computational studies:

$$CH_3OH^* + O_2^* \to CH_3O^* + OOH^*$$
(6)

$$CH_3OH^* + O^* \to CH_3O^* + OH^*$$
⁽⁷⁾

$$CH_3OH^* + OH^* \rightarrow CH_3O^* + H_2O^*$$
(8)

$$CH_3OH^* + OOH^* \rightarrow CH_3O^* + HOOH^*$$
 (9)

Reaction 6 is the most difficult one, with theoretically predicted activation barriers of 0.4–0.5 eV on extended Au surfaces (depending on the specific surface model used).^{395,396,419} Reaction 9, assuming α -H atom abstraction from methanol by OOH*, is a bit more facile with a theoretical barrier <0.4 eV.^{395,419} For the remaining two reactions, reactions 7 and 8, however, DFT calculations yielded notably lower activation energies <0.27 eV.^{395,419,421–423} Most of the respective computations have been performed on a flat Au(111) model surface (or Au(111) alloyed with Ag) with the exception of two studies carried out for a stepped Au(310) surface.³⁹⁶ and a stepped Au(997) surface.⁴²⁴

Further oxidation of methoxy to formaldehyde by O^* , OH^* , OOH^* , or CH_3O^* species can then proceed according to the following reaction steps:

$$CH_3O^* + O^* \to CH_2O^* + OH^*$$
⁽¹⁰⁾

$$CH_3O^* + OH^* \rightarrow CH_2O^* + H_2O^*$$
(11)

$$CH_3O^* + CH_3O^* \rightarrow CH_2O^* + CH_3OH^*$$
(12)

$$CH_3O^* + OOH^* \rightarrow CH_2O^* + HOOH^*$$
(13)

 O_2^* has not been considered here as a potential β -H abstracting agent due to its expected low nucleophilicity and basicity. The first two reactions $(CH_3O^* + O^*/OH^*)$ were predicted to proceed quite easily on Ag-Au(111), with activation barriers of only 0.27 and 0.45 eV, respectively.419 Even smaller values, 0.07 and 0.24 eV, were found for the same reactions on pure Au(111),³⁹⁵ whereas 0.14 eV was calculated in another study on Au(111) for $CH_3O^* + O^*$.⁴²² Notably, these kinetic barriers are lower than reported for the same reactions in three further computational studies of methanol oxidation on Au(111).^{49,421,423⁺} These latter works computed barriers of 0.43–0.49 eV and 0.48–0.70 eV for β -H abstraction from methoxy by O* and OH*, respectively. A study on a stepped Au(310) surface, on the other hand, reported an activation energy of 0.48 eV for a reaction of CH₃O* with O* but also found that the barrier increases to 0.65 eV if methoxy and O* are far away from each other in the initial state and are not interacting.³⁹⁶ The same tendency of increasing the barrier when reactants are placed far apart from each other was found in a study of methanol oxidation on Au(111).⁴²¹ Here, the activation barrier for the β -H abstraction by O* increased from 0.43 to 0.56 eV for CH_3O^* + O* and from 0.70 to 0.79 eV for $CH_3O^* + OH^*$ when calculated with respect to noninteracting reactants. Two studies also considered β -H abstraction by OOH* and reported low barriers of 0.06 and 0.02 eV on Au(111)³⁹⁵ and Ag-Au(111),⁴¹⁹ respectively.^{45,382} Nevertheless, one should keep in mind that under real catalytic conditions the concentrations of surface O*, OH*, as well as OOH* species are probably very low. Furthermore, as will be discussed below, OH* and O* species commonly assumed in the literature to be responsible for β -H abstraction from CH₃O*, can as well be responsible for undesired deep oxidation routes. The observation that the latter routes are almost not followed on gold catalysts can be attributed to too low concentration of these reactive oxygenates on the surface under the corresponding reaction conditions. In contrast, CH₃O* is likely to be the most abundant surface intermediate, which could abstract β -H from another methoxy species, acting as a base in analogy to OH* and O*. The activation barrier for this reaction was calculated to be 0.66 eV on Au(111).49 Hence, it is important to take this reaction channel for the $CH_3O^* \rightarrow CH_2O^*$ transformation into consideration under the conditions of ambient catalysis.

Once formaldehyde is formed, it can either desorb or undergo a barrierless reaction with either CH₃O*, OH*, or O*, forming CH₃OCH₂O* (hemiacetal alkoxide), HOCH₂O*, or OCH₂O* (methane diolates), respectively. In all of these reactions, oxygenated species act as nucleophiles, while formaldehyde is an electrophile. As far as the latter pathway is concerned (CH₂O* + O* \rightarrow OCH₂O*), an experimental study indicates that the formation of OCH₂O* is reversible.³⁸⁴ The hemiacetal and diolate intermediates, resulting from the three addition reactions above, can quite easily undergo β -H elimination, yielding CH₃OCHO* (methyl formate), HOCHO* (formic acid), or OCHO* (formate), respectively.^{49,419,421,422} Formate is a very stable species on the surface. According to computational studies on Au(111), a hydrogen abstraction from OCHO* by O* or OH* forming $CO_2 + OH^*/H_2O^*$ is associated with a large barrier in the range of 0.95–1.16 eV.^{421–423}

In catalytic studies regarding the aerobic oxidation of methanol over NPG and in surface-science experiments on Au surfaces precovered with small amounts of O*, however, almost exclusively methyl formate was formed.^{8,382} This high selectivity toward methyl formate can be explained by a relatively high activation barrier of the $CH_3O^* \rightarrow CH_2O^*$ step (" β -H elimination")—higher than for the preceding CH₃OH* \rightarrow CH₃O* step—and essentially barrierless subsequent coupling and deprotonation steps: $CH_3O^* + CH_2O^* \rightarrow$ $CH_3OCH_2O^* \rightarrow CH_3OCHO^*$. The same basic oxygenated species as in the case of β -H abstraction are likely to also be responsible for the "final" H abstraction step in this sequence. Formaldehyde (CH₂O*) formed in a slow step should be a minority, whereas CH₃O* a majority species on the surface. Formaldehyde can thus immediately react with the unreacted methoxy (CH_3O^*) , yielding methyl formate (CH_3OCHO^*) .^{8,382} In contrast, an analogous addition reaction between CH2O* and O*, OH*, or OOH* (which may react directly or first decompose to $O^* + OH^*$) would result in the formation of diolates OCH₂O* or HOCH₂O*, followed by their transformation to OCHO* (formate) and HOCHO* (formic acid), respectively, after subsequent β -H abstraction steps. Thus, it is the abundance of methoxy that seems to be responsible for the selectivity to partial oxidation. Total oxidation of methanol would proceed via OCHO* (formate). However, this route is limited by the concentration of O*, OH*, and/or OOH* on the surface (required to convert formaldehyde to $HOCH_2O^*$ or OCH_2O^*).

4.2. Insights from Single-Crystal Surfaces

4.2.1. Nature of Activated Oxygen Species. Experiments conducted on single-crystal surfaces under well-defined conditions significantly contributed to our understanding of some of the structural and chemical preconditions manifesting NPG's catalytic potential. Although gold surfaces, due to their inertness, exhibit only weak interactions with many molecules, including molecular oxygen and hydrogen (see above), corresponding UHV experiments carried out at sufficiently low temperatures allowed studying some of the central surface species involved in the catalytic reaction network and, in this way, tremendously helped unraveling their respective surface chemistry.

As far as oxygen is concerned, one important impediment inherent to this approach had, however, to be conquered first, namely the fact that, under UHV conditions, molecular oxygen cannot be activated on Au surfaces. The effectiveness of this restriction was validated for polycrystalline as well as for various single-crystal Au surfaces, including stepped ones, such as Au(221), for pressures up to 1 bar and temperatures up to 900 K.^{425–428} This inability is not unexpected and perfectly in line with theoretical calculations predicting very low adsorption energies and high dissociation barriers for molecular oxygen on pure gold surfaces (section 4.1).

To experimentally overcome this problem, single-crystal Au samples have to be exposed to activated oxygen species to allow studies of their subsequent reaction with other molecules, such as CO or methanol. To this end, various strategies had been taken advantage of, including reactive adsorption of NO_2 , O_3 , or atomic oxygen as generated by a plasma source or thermally by a filament or a crack-



Figure 18. (A) CO titration experiment on Au(111) at 77 K. An RF-plasma was used to adsorb reactive oxygen in the quantities given above each curve. The surface is exposed to a beam of CO molecules (flux 9×10^{13} molecules/cm^{2,434} (B) CO₂ intensity (m/z = 45) observed for CO oxidation on Au(332) during a pulsed molecular beam experiment. A constant flux of ¹³CO ($p = 2.4 \times 10^{-6}$ mbar) and pulses of oxygen atoms (duration 150 s; flux(O) = 8×10^{12} cm⁻² s⁻¹) were applied.⁴³⁵ (A) Reproduced with permission from ref 434. Copyright 2007 Springer Nature. (B) Prepared by us from data in ref 435.

er.^{401,428–432} To characterize the nature and the varying properties of the oxygen phases, produced on the surface by such methods, STM and XPS studies, mainly on Au(111), were carried out which succeeded to identify a wealth of different species forming as a function of coverage and differing in structure and reactivity.⁴⁰¹ Interestingly, aggregation leading to the growth of first small oxygen islands was already observed at small coverages. These features, however, structurally changed considerably at higher O doses, as also reflected by characteristic variations in the O 1s binding energy. In short, the results were interpreted in terms of chemisorbed oxygen atoms created first, followed by the formation of various surface and subsurface oxides at higher coverages.

Even though such oxygen surface species are most likely relevant for NPG as an oxidation catalyst as well, the situation is expected to be even more intricate here, taking the significantly higher structural complexity of the material into account. As discussed in the previous section, low-coordinated sites, such as steps or kinks, were theoretically proposed to serve as preferred adsorption sites for oxygen atoms which then tend to aggregate into chain-like structures with increasing coverage.⁴¹¹

Catalytic results with NPG (see section 4.3) are compatible with the idea that different oxygen species with deviating chemical nature are available on its surface and that these are the decisive factor regulating the activity for total and partial oxidations. It was found, for instance, that NPG samples pretreated with ozone (O_3) and strongly oxidized are readily active for methanol oxidation, while it turned out that they are inactive for CO oxidation (for which, in contrast, other activation protocols work: see below).⁴³³ These results therefore suggest that the oxygen species required for both types of reactions are of different character. Even though the picture obtained in this respect is not complete so far, we will, in the following, review the hitherto existing knowledge gathered from single-crystal studies of CO oxidation and methanol oxidation.

4.2.2. CO Oxidation. Already in the late 1980s, isothermal (300 K) CO titration experiments on an oxygen precovered Au(110) surface suggested that the reactivity of activated

oxygen species toward CO oxidation drops substantially with increasing oxygen coverage.⁴⁶ This result was corroborated by later studies depicted in Figure 18A. The CO titration experiments shown and carried out at 77 K on Au(111) evidence a significant activity toward CO oxidation at small initial oxygen coverages but reveal very little activity at higher coverages under otherwise identical conditions.⁴³⁴ Concomitantly, the changes in the shape of the titration curves point to a rather complex evolution as far as the reactivity of the formed oxygen species is concerned, when increasing the surface concentration.

More profound insight into such questions can be gained based on molecular beam experiments, offering a unique way to follow the development of such oxygen species alongside with their reactivity under isothermal and well-defined flux conditions. Corresponding results, recently obtained using a stepped Au(332) surface and two separate effusive beam sources for CO (constant) and oxygen atoms (pulsed supply), are presented in Figure 18B.⁴³⁵ Qualitative differences in CO₂ formation are readily identified when comparing the pulse series acquired for 200 and 250 K. While the amount of CO₂ formed during the oxygen pulses decreases with the number of pulses in the first case, such a behavior is not detected at the higher temperature. Yet, at 250 K, a considerable deactivation within the pulse is found instead, as deduced from the peak shapes. (In contrast, the activity toward CO₂ formation during each pulse is only slightly reduced at 200 K.) As the rate and total number of oxygen atoms impinging on the Au surface are constant during each pulse and a constant sticking coefficient was found for a wide range of oxygen exposures, the declining peak intensities during the pulse sequence at 200 K suggest that the removal of oxygen atoms by CO oxidation cannot compete with their supply by the molecular beam so that oxygen accumulates during the pulse and cannot be reacted off completely in the delay times. In consequence, unreactive oxygen islands (see above) are apparently formed which cumulatively deaden the oxidation activity as more oxygen is deposited.43

While, as proven by IR spectroscopy, CO has a sizable transient surface concentration at 200 K on Au(332), the



Figure 19. (A) (top) STM image of 0.04 ML of O/Au(110)-(1×2); (bottom) analysis of oxygen redistribution induced by 3×10^{-10} Torr water at 170 K.⁴⁴² (B) CO₂ formation upon a CO titration experiment on Au(111) at 140 K. The Au(111) surface had been precovered with 0.11 ML ¹⁶O. In (B) 0.14 ML H₂¹⁸O were additionally adsorbed. Inset shows the total amount of CO₂ produced in each case.⁵⁶ (C) CO₂ intensity (*m/z* = 45 (red); *m/z* = 47 (blue)) during CO oxidation on Au(332) during a pulsed molecular beam experiment at 250 K for "dry" (left) and "wet" conditions (right). A constant flux of ¹³CO (*p* = 2.4 × 10⁻⁶ mbar) and H₂¹⁸O (*p* = 5.1 × 10⁻⁶ mbar) and pulses of oxygen atoms (duration 150 s; flux(¹⁶O) = 8 × 10¹² cm⁻²s⁻¹) was applied. (A) Reproduced with permission from ref 442. Copyright 2018 Royal Society of Chemistry. (B) Reproduced in part with permission from ref 56. Copyright 2008 American Chemical Society. (C) Prepared by us from data in ref 435.

corresponding absorption bands disappear at 250 K indicative for a significantly reduced transient concentration.^{436,437} As a consequence, more oxygen is accumulating during the pulse, resulting in a more pronounced decrease of the reactivity. However, the rate constant for CO oxidation is significantly higher at 250 K than at 200 K, as readily inferred from the fact that CO is capable to react with the accumulated oxygen atoms in the off-periods of the pulse sequence. In essence, a comparable (reactive) surface state is reached at the beginning of each pulse so that the same shape of the CO₂ formation pulse is recorded throughout the sequence. This and the other work described above implies that NPG's catalytic activity for low-temperature CO oxidation, i.e., the CO₂ formation rate observed under (real) reaction conditions (section 4.3), should sensitively depend on the availability of oxygen entities providing the right reactivity for total oxidation on the surface which, in turn, is expected to be the result of a delicate balance of different species and phases changing with temperature and oxygen partial pressure.

In contrast, for CO, as the other reactant, the situation seems less complex. In this case, surface science studies on Au(111) clearly indicated that the CO binding energy on ideal terrace sites is too low to account for sizable CO surface coverages under catalytic conditions (300 K, 1 bar).^{46,437–440} Yet, the detailed structural characterization of NPG by HRTEM provided ample evidence for a high abundance of low-coordinated Au surface atoms located along atomic step edges (in areas of step bunching) or at kinks on the ligaments.³⁶⁰ Experiments on stepped as well as roughened surfaces showed that low coordinated sites exhibit significantly higher binding energies, which is also in line with the computational results (section 4.1).^{437,439}

Nevertheless, not only with respect to oxygen activation, but also to CO adsorption, UHV model studies were not able to disclose all the properties of NPG. Specifically, temperatureprogrammed desorption spectra recorded for NPG samples under UHV conditions provided clear evidence for additional CO species bound even stronger than at "simple" Au step and kink sites.⁴⁰⁶ Different scenarios such as Au sites in the direct neighborhood of Ag (or AgO_x species) were discussed to account for this extra stabilization. Yet, a final clarification of this topic, including the question whether these CO species play a role in the catalytic cycle, is still outstanding.

4.2.3. The Role of Water in CO Oxidation. The preceding subsection summarized what is known regarding the dependence of CO oxidation activity on the coverage and speciation of oxygen species that can form on Au. It should be stressed, however, that the catalytic behavior can also be strongly impacted by other environmental conditions, i.e., by other molecules which are present in the gas phase and can coadsorb on the surface. This topic can be especially well-illustrated at the example of water which, as a matter of fact, represents a common impurity in many (technical) feeds.

Interestingly, it was observed that water enhances the catalytic activity of NPG for CO oxidation.³⁹⁰ This is in line with findings for oxide-supported Au nanoparticles, but opposed to other catalytic reaction of NPG such as the partial oxidation of methanol. This cocatalytic effect could also be reenacted in experiments on Au(111) under UHV conditions, suggesting an increase of the reactivity of predeposited oxygen species toward CO oxidation.⁴³¹ In conjunction with further work on this subject, such model studies enabled proposing a reasonable mechanism for this effect.



Figure 20. (A) TPR spectra of CH₃OH on O/Au(111) ($\theta_0 = 0.2 \text{ ML}$). Ozone was dosed at 200 K to deposit oxygen on the Au(111) surface.³⁸² (B) Isothermal pulsed molecular beam experiments on Au(332) at 310 K pulsing atomic oxygen ($6.9 \times 10^{11} \text{ cm}^2 \text{ s}^{-1}$, 30 s on, 100 s off), while continuously supplying methyl formate ($1.2 \times 10^{15} \text{ cm}^2 \text{ s}^{-1}$; black trace) or methanol-¹³C ($4.2 \times 10^{13} \text{ cm}^2 \text{ s}^{-1}$; red trace).⁴⁴⁹ (C) TPR spectra of 0.11 L methyl formate on 0.13 ML ¹⁸O precovered Au(332) (blue); reference spectra: 0.13 ML ¹⁸O reference on Au(332) (black) and a 0.11 L methyl formate reference (gray dots).⁴⁵⁰ (A) Reproduced with permission from ref 382. Copyright 2009 Wiley-VCH. (B) Adapted with permission from ref 449. Copyright 2022 Royal Society of Chemistry. (C) Adapted with permission from ref 450. Copyright 2021 American Chemical Society.

First, experiments using isotopically labeled water $(H_2^{18}O)$ yielded evidence that oxygen atoms (pre)adsorbed on Au can efficiently interact with water by transiently forming hydroxyl groups. Specifically, CO₂ containing ¹⁸O was detected as an oxidation product of C¹⁶O and ¹⁶O_{ad} when H₂¹⁸O was co-dosed (Figure 19B), a finding which is only explainable by a scrambling reaction according to

$${}^{16}OH^* + H_2{}^{18}O \rightarrow {}^{16}OH^* + {}^{18}OH^* \rightarrow {}^{18}OH^* + H_2{}^{16}O$$
(14)

In experimental agreement with larger desorption temperatures of water from oxygen precovered vs clean Au surfaces, DFT calculations confirmed that such exchange processes are indeed feasible because they are associated with comparatively low activation barriers.^{56,441}

Another piece of information was provided by STM investigations performed on oxygen precovered Au(110) surfaces. Upon coadsorption of water, an increased mobility of oxygen atoms and, in succession, a restructuring of previously formed oxygen islands on the surface was observed (Figure 19A).⁴⁴² Combining all findings, the following scenario becomes likely: The presence of water and its coadsorption lays the foundation for a transient and reversible formation of surface hydroxyl groups. In addition to their impact on the reaction mechanism discussed in section 4.1, they have the capability to mobilize oxygen on the surface and may thus initiate (at least a partial) disaggregation of otherwise catalytically inactive oxygen phases/islands. On these grounds, water can bring about a regeneration of active oxygen species and, in consequence, can lead to higher catalytic turnovers.

This cocatalytic effect of water for low-temperature CO oxidation was also underpinned by recent isothermal molecular beam experiments depicted in Figure 19C. Here, CO oxidation was probed at 250 K by pulsing ¹⁶O atoms on a Au(332) surface while keeping a constant flux of ¹³C¹⁶O under "dry" (left) as well as under "wet" (right) conditions, using H₂¹⁸O.⁴³⁵ In line with the titration experiments shown in Figure 19B, quite efficient isotopic scrambling within the product CO₂ (¹⁶O/¹⁸O) was detected in spite of the fact that the reaction

was carried out about 60 K *above* the water desorption temperature. While corroborating the studies discussed above, the data additionally disclosed considerable changes in the transient kinetics of the system under the "wet" conditions. While significant deactivation is observed without water (Figure 18B), the presence of water obviously enables a stable steady-state reactivity. Attaining this steady state requires almost the whole pulse length (note the increasing and finally saturating CO_2 formation during the pulses), indicating that the corresponding processes to establish steady-state conditions are rather slow.

Overall, all results reviewed here point in the same direction, namely to an enhanced availability of those oxygen species which are responsible for the catalytic activity in case of CO, i.e., for total oxidation, in the presence of water. Even though a more detailed mechanistic description of the altered reaction network would be highly desirable, it currently lies beyond the scope of the knowledge which has been achieved based on UHV model studies carried out so far.

4.2.4. Methanol Oxidation. Apart from single-crystal experiments related to CO oxidation, model studies were particularly successful in proposing the reaction mechanism and further mechanistic details for the partial oxidation of alcohols over NPG.⁸ First and foremost, methanol has been the focus of numerous studies ever since the first work was published reporting the ability of NPG to efficiently catalyze both, total and partial oxidation reactions at low temperatures.

The underlying mechanism already proposed in the first reports and experimentally derived from a combination of previous TPR experiments (Figure 20A) and vibrational spectroscopy was already presented in Figure 17.³⁸² The surface chemistry determining the partial oxidation of alcohols was accordingly interpreted on the basis of acid–base surface chemistry concepts and proved to be quite predictable on these grounds. The starting point of the mechanistic considerations is the Brønsted basicity of the oxygen species adsorbed on the Au surface.^{47,443–446} It was possible to readily explain the selectivity pattern observed for methanol oxidation over NPG as soon as the first results were published on this

topic. In agreement with that, not the direct aldehydes or carboxylic acids were found as products, but solely esters, i.e., coupling products, such as methyl formate in the case of methanol oxidation. This preference for oxidative coupling reactions was not only verified for single alcohols, but the validity of the mechanistic framework was also demonstrated to hold for the cross coupling of different alcohols and alcohol aldehyde mixtures as well as for the oxidative coupling of amines and alcohols to amides.^{379,447,448}

The common notion regarding the nature of the relevant oxygen species driving the selective formation of oxidative cross coupling products is that chemisorbed O atoms and their favorable basicity are responsible for the initial and ratedetermining hydrogen abstraction steps. Within this picture, however, the less specific redox properties of the oxygen species and oxide phases, governing, as discussed above, the ability of NPG for CO oxidation, are not considered. A corresponding transition of reactivity is reflected, e.g., in temperature programmed reaction spectroscopy (TPRS) experiments on Au(111) showing a rapid decrease of selectivity toward overoxidation products at oxygen coverages above 0.1 ML. Only below, selectivities above 70% toward methyl formate were found.³⁸² In agreement, comparatively high selectivities toward methyl formate of 65% were reported recently for Au(332) at 230 K if exposing the surface to a high flux ratio of methanol to oxygen atoms (f(MeOH)/f(O) =660) in a pulsed molecular beam experiment. Notably, these were carried out under single collision conditions, meaning that desorbing educts or intermediates cannot undergo subsequent reactions. In contrast, for NPG studied under catalytic conditions molecules, be it reactants, intermediates, or even products, typically collide several times with the surface before leaving the catalyst bed. Consequently, also model studies based on molecular beam experiments, although coming closer to real catalytic conditions as compared to TPD or TPRS studies may yield selectivity patterns deviating from what is observed for the working catalyst. In the case of partial methanol oxidation, for example, formaldehyde desorption taking place before the molecule can undergo the subsequent coupling reaction to the ester will obviously result in a lower selectivity. In accord, dropping selectivities were indeed observed in isothermal molecular beam experiments upon raising the temperature.⁴⁵¹

A further aspect that may limit the comparability of such model studies with catalytic results is related to the fact that, in the former case, the product methyl formate may desorb after its formation and thus escape further potential conversion steps. On the contrary, the chance for secondary reactions, possibly leading to total oxidation, in a catalyst bed or in a larger monolithic NPG catalyst is high due to the likelihood of multiple collisions with the catalytic surface. The propensity of a partial oxidation product, in this case methyl formate, to undergo such an overoxidation process upon readsorbing on the catalyst surface can, however, be separately checked in molecular beam experiments. Notably, such information can be of value when aiming at an optimization of real catalytic processes, e.g., by adapting catalyst bed lengths or the flow conditions to minimize secondary reactions.

In this spirit, additional pulsed molecular beam experiments were performed under isothermal conditions to assess the propensity of methyl formate for overoxidation, i.e., to allow for an explanation why methanol partial oxidation is distinctly preferred over its total oxidation under working conditions. To

this end, the isothermal oxidation rate of methanol and methyl formate was compared by monitoring the water (mass spectrometrically) formed in the corresponding reactions. To mimic the scenario of an NPG catalyst working at high conversion and high selectivity the flux of methyl formate was chosen to be a factor of 35 higher than that of methanol, keeping all other parameters constant. Figure 20B shows the amount of water produced during an oxygen pulse while exposing the surface to a constant flux of either methyl formate (black) or methanol (red) during the entire period.⁴⁴⁹ Despite the significantly higher supply of methyl formate, its overoxidation rate is small as compared to that of partial oxidation. These results suggest that reaction pathways leading to overoxidation of methyl formate do not play any significant role under practical conditions, well in line with the high selectivity of NPG in the partial oxidation of methanol to methyl formate.

In general, two different circumstances could be responsible for the low extent of overoxidation (in spite of its thermodynamic preference): either no reaction channels with sufficiently low activation barriers exist or the availability of corresponding surface sites is (too) low to compete with the partial oxidation pathway. TPRS as shown in Figure 20C also allows to discriminate between the two scenarios.450 Three reaction channels resulting in CO₂ formation are identified, all belonging to minority species of the preadsorbed oxygen. Importantly, two of these are related to low desorption temperatures (135 and 185 K) in accord with low activation energies, while only one occurs at significantly higher temperatures (320 K). Because the two lower ones fall below the temperature of methyl formate formation from methanol in TPRS experiments (220 K, Figure 20A), the associated reaction channels cannot be observed in these experiments as methyl formate has not yet formed.³⁸² These results clearly show that overoxidation pathways for methyl formate exist on stepped Au surfaces, although only small amounts of formate are formed. A high selectivity toward methyl formate hence requires that the highly reactive oxygen species capable of overoxidation of methyl formate were not populated under the catalytic conditions applied. In the case of methanol oxidation, this is maintained at not too high oxygen pressures because of the rather slow formation of the oxygen species and a fast and rather unspecific reaction of methanol with different oxygen species. The difference in reactivity can be rationalized by a simple chemical picture: while the partial oxidation of methanol involves a series of hydrogen abstraction processes, i.e., requires Brønstedt basicity of the oxygen species, the overoxidation of methyl formate and other overoxidation channels as well involve an attack of the carbon atom by an oxygen species, i.e., depend on the nucleophilicity or-in the language of acid base chemistry-on the Lewis basicity of these species.

4.3. Gas Phase Catalysis on Nanoporous Gold

In sections 4.1 and 4.2, some of the key findings identifying NPG as an active and, more importantly, very selective catalyst for aerobic oxidation reactions in the gas phase have already been mentioned and correlated with insight from theory and advanced UHV model studies. Here, we will complete the picture with the results of catalytic experiments carried out under ambient conditions, again focusing on oxygen as the species apparently controlling the reactivity on the surface. In addition, we review recent efforts to quantify diffusive mass



Figure 21. Nanopore coarsening during catalytic reaction: (a) Initial state. (b) Faceting after exposure to CO/air mixture for \sim 50 s. (c) Crystallography of nanopore shown in (b). (d) Indication of the preferential growth direction normal to (100) surface plane. (e) Necking of gold ligament before rupture. Inset: magnified view shows deformation twins. (f) Coalescence of small and large nanopores after rupture. Reproduced with permission from ref 360. Copyright 2014 American Chemical Society.

transport in NPG and address its impact on limiting the catalytic performance in gas phase applications.

4.3.1. Impact of Oxygen Heterogeneity on Catalytic Properties. With respect to aerobic oxidation catalysis on NPG, the activation of molecular oxygen is one of the key steps. As discussed in section 4.1, theory has proposed a variety of mechanisms for oxygen activation. Experiments combining catalysis with spectroscopy provided evidence that residual Ag moieties are involved in the activation amending the original perception that NPG is a pure Au catalyst.⁵⁸ In agreement with that, an apparent barrier of about 21 kJ mol⁻¹ was determined for oxygen activation on ozone-treated NPG being active for selective methanol oxidation.⁴⁵² This value is close to values found for different Ag surfaces,⁴⁵³ while for Au surfaces significantly higher activation barriers are expected (section 4.1). Experimentally, a barrier of about 140 kJ mol⁻¹ was estimated for molecular oxygen dissociation on Au(110), for instance, on the basis of temperature programmed desorption results.438

Experiments with NPG in a temporal analysis of products (TAP) reactor revealed an oxygen saturation coverage of 4×10^{-3} ML at 423 K, being consistent with an activation of molecular oxygen on Ag sites. Yet, no indication for spillover of oxygen species to Au sites could be deduced from these results.⁴⁵² This is a surprising observation as the selective oxidation chemistry of NPG is characteristic of Au, hence, requiring oxygen spillover from Ag to Au when oxygen activation is not possible on Au sites alone. As far as CO oxidation is concerned, previous experiments with a similar experimental setup revealed for a NPG sample containing about 3 times as much Ag on the surface (as determined by XPS), an oxygen saturation coverage being about 5 times as large, corroborating the role of Ag as that component which lends NPG the ability to bind oxygen at the surface.⁴⁰⁵

Although it is difficult on these grounds to draw clear conclusions with respect to those fundamental structureproperty relationships governing the catalytic properties of NPG, theory as well as experiments on model systems provide indications that surface oxygen species of different reactivity play a central role, as discussed above. This is consistent with experimental observations of decreasing selectivity for the partial oxidation of methanol to methyl formate when increasing the oxygen partial pressure.8 Further evidence comes from differences noticed for the procedures which are necessary to catalytically activate NPG. While ozone treatments result in an active and selective catalyst for the esterification of methanol, this treatment is not suitable for achieving steady-state activity for CO oxidation.⁴³³ Only transiently oxygen species are created which are active in this case and also lead to total oxidation of methanol (to CO_2); these sites, however, apparently cannot be replenished sustainably under reaction conditions. In contrast to that, it was recently shown that NPG can be reliably activated for CO oxidation by short annealing treatments to 300 °C in a reaction mixture of CO and O_2 .⁴⁵⁴ It was verified that no thermal coarsening of the pore structure is induced as a side effect, as previously reported for longer annealing times at such temperatures.455

The rather complex transient kinetics associated with the activation procedures point to significant changes of the surface chemistry induced in this way. In this context, dynamic surface restructuring causing a different Ag surface composition could play a role, as observed, for instance, by in situ TEM during CO oxidation (Figure 21).³⁶⁰ The extent to which Ag concentrations at the surface are altered is probably also influenced by the fact that the Ag distribution in NPG samples can be quite heterogeneous after dealloying (section 3.13).^{90,94} It furthermore has to be considered for the restructuring dynamics that residual Ag was found to stabilize steps and



Figure 22. Plot of the catalytic efficiency (η) as a function of the characteristic length (l) associated with NPG catalysts exhibiting different macroscopic shapes. Values for NPG (ligament size 30 nm) discs (d = 5 mm; h = 200 mm) and particles (all dimensions approximately 200 mm) are marked. Reproduced with permission from ref 457. Copyright 2022 The Authors.

kinks and is thus likely to mitigate structural changes of the surface structure. $^{360}\,$

Overall, the combined insight from theory, single-crystal studies and catalytic experiments suggests that oxygen atoms with quite different properties drive CO oxidation and the esterification of methanol. While the latter requires surface oxygen with sufficient Brønsted basicity to allow for a rapid formation of adsorbed methoxy species (or, in general, alkoxy species) and subsequent β -hydrogen abstraction, such species should exhibit a low reactivity with respect to a nucleophilic attack of the reaction intermediates such as formaldehyde to avoid overoxidation products (Figure 17). In contrast to that, good CO oxidation activity requires oxygen species which readily attack the carbon atom of adsorbed CO, i.e., have a sufficiently high Lewis basicity. More far-reaching conclusions, however, are hampered in view of a complex surface chemistry as revealed, for example, by the finding that various oxygen species on NPG were found to oxidize CO readily, but that only a subset of them seems to possess the potential for stable and sustainable conversion rates under catalytic condition. This has not only been observed in case of ozone-activated NPG (s. above) but also for NPG samples activated in a mixed CO/O_2 stream.⁴⁰⁶

The heterogeneity of the system renders a clear assignment of the moieties responsible for the catalytic activity of NPG, in particular for CO oxidation, challenging. Future mechanistic models, however, have to comply with the observation that experimental kinetic studies revealed reaction orders for oxygen close to 0 and for CO between 0.8 and 1.58 These data suggest that the supply of active oxygen species for CO oxidation is rather efficient on the surface and in the first place limited by the availability of CO molecules. Because oxygen activation is ascribed to Ag sites, in line with the experimental finding that increasing amounts of residual Ag leads to increasing CO_2 formation rates, the low reaction order of O_2 suggests a high reactivity of such sites. Even though in the literature mostly metallic Ag or bimetallic Ag/Au sites have been discussed in this context, AgO_x species, likely to be formed on the surface in particular under oxygen excess conditions, cannot be ruled out as possible candidates. In essence, it has to be stated that the current mechanistic

understanding derived from theory, model studies, and catalysis is not sufficient enough to explain all aspects of the catalytic behavior of NPG, including, for instance, selectivity changes. So far it is not clear which changes regarding the nature and/or concentration of Ag-related surface sites or also Au sites are in detail responsible for the detected transition from partial (esters) to total oxidation products (CO₂) in the case of alcohol oxidation.⁸

4.3.2. Macro- and Microkinetics: From Transport Properties to Reaction Mechanism. As already pointed out, one aspect also and potentially considerably impacting the catalytic performance of NPG in practical applications relates to mass transport limitations as resulting from the nano- or, being more precise, mesoporosity of the material. If the rate at which the reactants are supplied within the pore system or the products are transported out of it is in the same range as the reaction rate or even lower, a reduction of the achievable catalytic conversion is expected. In view of NPG's pore diameters lying in the range of a few 10 nm, corresponding mass transport limitations are not unlikely.

Such contributions can be assessed based on the Thiele modulus formalism, being a well-established concept in chemical engineering. To this end, however, structural information is needed.⁴⁵⁶ On the one hand, the mean pore diameter will determine to which extent the mean free path of the diffusing gas molecules is reduced by Knudsen diffusion. On the other hand, nonstraight pores, i.e., the tortuosity of the pore system (section 3.10), will lead to elongated diffusive pathways which the molecules have to overcome within the porous material.

Recently, pulsed field gradient (PFG) NMR spectroscopy was employed to evaluate to which extent NPG's tortuosity slows down the diffusive transport. For CO and CO₂, as reactant and product molecules of CO oxidation, as well as for CH₄, included in the study as an inert gas for reference purposes, consistently, a mean pathway elongation by a factor of about 2 was derived in all cases.⁴⁵⁷ On the basis of this result, representing the first experimental assessment of NPG's (diffusive) tortuosity factor, and the mean pore diameter to take the contribution of Knudsen diffusion into account, it was possible to quantitatively describe the macrokinetics, i.e., the



Figure 23. (A) Selectivity phase diagram for the oxidation of methanol at 383 K. In each region formaldehyde, methyl formate, or CO_2 is the major product (>50% selectivity). (B) Degree of selectivity control (DOSC) for methyl formate formation. The DOSC was calculated for different elementary reactions using plug flow reactor simulations at 423 K as a function of oxygen and methanol partial pressures varying between 10^{-7} and 10^1 bar. Reproduced from ref 403. Copyright 2019 American Chemical Society.

experimentally observable and obtainable catalytic CO conversion rates, as a function of the diffusive mass transport and the microkinetics of the surface reactions.

In this context, it turned out that larger monolithic NPG catalysts, as were often applied in the literature in the form of discs exhibiting diameters in the range of several mm and thicknesses of a few 100 μ m, are indeed associated with reduced catalytic efficiencies. As shown in Figure 22, for such a monolithic NPG disc (diameter, 5 mm; thickness, 200 μ m; mean ligament size, 30 nm), the catalytic efficiency is limited to \sim 70%. When, however, breaking up the discs into smaller platelets (of the same thickness) so that their diameters decrease to just about a few 100 micometers, the efficiency can be increased to almost 100%-in agreement with the predictions of the Thiele modulus formalism and NPG's structural characteristics (see solid line in Figure 22).⁴⁵⁷ This result demonstrates that the well-defined and well-controllable features of NPG in conjunction with options to adapt the macroscopic shape of the material to a particular set of reaction conditions provides a good basis for predicting and optimizing its performance in a given catalytic application.⁴⁵⁶

An insight into the kinetics determining the processes on the surface itself was gained by surface microkinetic simulations carried out for the esterification of methanol using a reaction network in line with the knowledge about the reaction mechanism established so far (Figure 17).⁴⁰³ For these simulations, the kinetic constants entering the microkinetic simulations were largely deduced from experimental data as obtained from single-crystal experiments; however, a few were taken from DFT calculations.

In a first study, the validity of the approach was confirmed by simulating the results of a TAP reactor study.⁴⁵⁸ As the calculations were able to predict the measured transients with reasonable precision, the methodology was extended to simulations of a plug flow reactor.⁴⁰³ Based on these simulations, it was possible to predict a variety of interesting properties important for an understanding of the catalytic performance, in particular the selectivity of the system. Figure 23 illustrates this point. The simulations predict preferential formation of formaldehyde, methyl formate, and CO₂ in specific regions of the pressure space (Figure 23A). Additionally, Figure 23B shows so-called degree of selectivity control (DOSC) plots for several elementary reactions of the reaction network using the same parameter space as in Figure 23A, i.e., the pressure of methanol and oxygen. The DOSC is a parameter which ranges between -1 and 1 and approaches 1 (-1) for a specific reaction channel in case the selectivity toward a certain product is strongly enhanced (decreased) with increasing reaction rate. As seen for the three pairs of elementary reactions plotted on top of each other, the simulations predict complementary DOSC values, i.e., the plots in each row show DOSC with opposite sign in very similar regions of methanol/oxygen pressures, suggesting a rather strict interdependence of the associated reaction paths within the network. It should, however, be kept in mind that the analysis still needs to simplify the system. In this particular case, the simulation assumes a single active oxygen species, whose activation energy of formation is comparable to values found on Ag surfaces.⁴⁵²

4.4. Evolution of the Material during Catalysis

Although it was claimed that NPG is a catalyst with a high structural stability,^{8,459} there is an evolution of the material during catalysis. The most obvious change of the material is coarsening, i.e., an increase of pore and ligament sizes, which sometimes is noticeable at first glance when examining the samples, e.g., in a SEM. As has been outlined in section 2.2 above, coarsening is routinely observed in NPG; it acts during the dealloying process itself, and it can be deliberately promoted by annealing in order to tune the ligament size. Under otherwise identical annealing conditions, it has been found that the chemical environment in the pore space has a strong influence on the thermal coarsening of NPG; samples in vacuum are typically more stable than in a gas atmosphere,⁴⁶⁰ but strongly adsorbing species such as atomic oxygen can also impede the coarsening.⁴⁶¹ Obviously, a strong connection between the coarsening in NPG and the environment to which it is imposed during catalysis may be expected. Generated heat during exothermic reactions or high reaction temperatures (>70-80 °C) might suffice to overcome kinetic barriers for transport processes at the surface which are required for reducing the surface area and, thereby, the net excess surface free energy, by coarsening.⁶⁴

The growth of ligament sizes was observed for example during (electrochemical) methanol oxidation^{35,60,62} and aerobic CO oxidation.³⁶⁰ Yet, it was found in the case of disc shaped NPG that coarsening of the structure is smaller in the bulk than in a skin layer at the outer surface of the disc exhibiting a thickness of ≈ 100 nm (see section 3.8).³⁵

As coarsening is mainly a surface diffusion-dominated process, the surface of NPG must be considered as a highly dynamic system under reaction conditions.³⁶⁵ A great challenge thus is the in situ investigation under catalytic conditions. In recent years, in situ TEM measurements became technically more and more available, but such experiments still suffer from significant material and pressure gaps as compared to catalytic studies carried out at atmospheric pressure or even above. In particular, a dedicated sample preparation is necessary because TEM measurements require samples with a thickness <100 nm. Furthermore, only 2D projections can be measured in situ by TEM, and the influence of the electron beam can never be excluded fully and has to be checked carefully.^{16,338,360} Besides TEM, the surface of NPG can be investigated by electrochemical cycling, XPS or STM during catalysis to unravel its structural and chemical characteristics. Despite all challenges, important insights with respect to the prerequisites for catalytic activity were obtained. A rearrangement of atoms on NPG surfaces was observed, for instance, when samples were exposed to CO/air mixtures, which does not occur under exposure to the single gases (N_{2i}, O_{2i}, O_{2i}) or CO).^{338,360} Another study found that adsorbed oxygen can stabilize the Au surface (see above).⁴⁶¹ Further studies found pronounced changes by heating in air, N₂, or O₂ but not under an Ar atmosphere.¹²⁸ Studies in which electrochemical cycling of NPG was investigated revealed that surfaces restructure to form thermodynamically more stable facets under reaction conditions.^{35,100,192} For example, the fraction of the most stable Au (111)-facet increased by 10% after cycling 200 times in sulfuric acid. The degree of atom surface diffusion depends on the scan rate and is limited by curvature and sizes of ligaments, respectively. Again, this finding identifies curvaturedriven surface diffusion as the dominating process for the

reordering.46 Several processes were identified that can suppress the mobility of atoms on the NPG surface. It was found that the higher the concentration of LNEs on the surface, the more the mobility is suppressed. The presence of Ag on the surface, for instance, turned out to exert a stabilizing effect.^{338,365,462} Enhanced stability was also found when a third element with a higher melting point (e.g., Pt) was added, which is able to stabilize undercoordinated Au surface atoms.^{62,100,115,242} Also dealloying of ternary alloys results in lower surface mobility and hence less coarsening.⁶⁴ Defects in the crystal lattice can also have a stabilizing effect. An in situ TEM study revealed that coarsening is prevented by planar defects because diffusing atoms are pinned to these defects.³⁶⁰ However, also crystal defects are object to change under reaction conditions as pointed out in ref 357. Furthermore, different studies suggested that coarsening is less pronounced when the samples are electrochemically cycled prior to catalysis experiments^{360,462} or when oxide particles or layers are deposited on the ligament surfaces of NPG.^{106,381,460,463–466}

In addition to the structure, the distribution of LNEs remaining in the material after dealloying potentially is subject to evolution and, in particular, to segregation to the surface. At first sight, it can be expected that the surface of dealloyed NPG consists of pure Au (section 2). However, a complete removal of the sacrificial element is impossible for thermodynamic reasons (Nernst distribution law) so that x_{Ag}^{res} of a few at. % is typically detected even after long dealloying times. It is obvious that a movement of atoms on the surface, in particular at step edges, and concomitant coarsening of the structure can

eventually expose LNEs to the surface, leading to an increased concentration there during (or after) catalysis. Furthermore, also diffusive segregation of the LNE to the surface is likely under reaction conditions.²⁴ Such an effect (in most studies detected for Ag because Au-Ag starting alloys were predominantly used) was actually observed during catalytic CO oxidation^{58,360} and, e.g., proven by XPS measurements.¹⁵ Also theoretical studies investigated Ag segregation to the surface.³⁶⁵ A possible explanation is that surface-adsorbed oxygen drives the diffusion of subsurface Ag to the surface, 467 a phenomenon which became especially apparent when applying an ozone treatment to the surface. It was found that a drastic change of the surface takes place. While Au and Ag are oxidized, Ag segregation to the surface is favored. Under reducing conditions, Ag eventually recedes back into the bulk.¹⁸

An ozone treatment is one of many procedures that was tried and carried out to activate NPG for catalysis, but this approach represents no universal measure to reliably activate NPG for the oxidation of different molecules.⁴³³ Activation of NPG is discussed in more detail in sections 4.2.1 and 4.3.1. These diverging results very much suggest that the specific surface chemistry needed to catalytically drive various partial or total oxidation reactions, respectively, is of different nature. This conclusion is also supported by the wealth of oxygen species and surface oxide phases that may form and (co)exist on Au surfaces.³⁸⁸ Overall, the question which activation treatment is most successful for a certain reaction has not finally been answered yet and will require further investigations in the future.

In essence, the mesoscopic and microscopic structure of NPG as well as the chemical distribution of residual LNE in the material are less defined than it may be assumed at first sight and are likely to considerably change in a catalytic environment. Hence, when NPG is characterized in terms of its structure and chemical composition, it should clearly be specified, at which stage the samples were characterized, how they were dealloyed, how they were treated (e.g., activated before catalysis), how long they were stored in which atmosphere at which temperature before use, and whether they were investigated in situ or ex situ. All of these different conditions might explain detected differences between different samples often reported in the literature.

5. LIQUID PHASE AEROBIC OXIDATION OF METHANOL

The previously discussed findings prompted us to focus the review on liquid phase catalysis with NPG on the aerobic oxidation of methanol and its correlation to mechanistic studies in the gas phase and electrochemistry (sections 4 and 6). Within this review, the term "liquid phase catalysis" will be used to describe thermally induced catalytic reactions and explicitly excludes electrochemical or electrocatalytical processes. Working in the liquid, i.e., condensed phase is the requirement for a widespread use in chemical synthesis and, hence, an important aspect. In fact, NPG has a much wider scope for organic transformations in the liquid phase which will be addressed in section 7.

5.1. Challenges for Working in the Liquid Phase

The description and understanding of chemical processes at the interface between the solid catalyst surface and a condensed phase is inherently more complex and more
challenging as compared to the solid–gas interface. Yet reports suggest that a correlation to mechanistic studies at the gas/ solid interface is possible,⁵⁹ which is due to the noble character,⁴⁶⁸ i.e., a low propensity to chemisorb species on its surface.

There is a gap in between the degree of precision, with which solid-liquid interfaces can be characterized compared to the solid-gas counterpart. This is partially due to a lack of appropriate methodology for solid-liquid interfaces. As a result, an improved understanding of processes at the solidliquid interface are a major focus in catalytic research.⁴⁶⁹ The limitations are mainly due to three reasons: (i) the interface is spectroscopically not easily accessible,⁴⁷⁰ (ii) the much higher concentration of reagents and the presence of solvent molecules, 471,472 and (iii) the dispersion and diffusion of gaseous reagents such as oxygen within the reaction medium.⁴⁷² Further aspects such as the formation of a charged double layer on the catalyst surface can play an important role also in liquid phase catalysis.⁴⁷³ When in contact with a liquid such as water, an electric field on the solid surface evolves as a consequence of the rates of all oxidative and all reductive partial reactions. The potential changes the chemical potential and thus the reactivity of species such as H⁺. This effect is wellknown and established for electrochemical catalysis and corrosion, yet the effect is not as well documented in heterogeneous catalysis in liquids due to the difficulties in determining and predicting this potential.

In order to spectroscopically access the interface, the probes, typically photons or electrons, have to penetrate the dense liquid phase. This is already an issue when dealing with higher pressures of gases on the order of 0.1 mbar as operative in ambient-pressure XPS, yet, makes detection of electrons emerging from the solid—liquid interface very challenging.⁴⁷⁴ Spectroscopy at the solid—liquid interface is thus restricted to photons of various energies ranging from IR all the way to X-rays. Employing special reactor designs, those techniques can provide information on the catalyst surface *in situ* or *in operando*.^{470,475,476}

The second reason why mechanistic insights from gas phase experiments cannot directly be transferred to the solid-liquid interface is the inherently much higher concentration of reagent and solvent molecules. For instance, 2 vol % of methanol in the gas phase at 1 bar total pressure corresponds to to a molar concentration in the range of 10^{-3} mol L⁻¹, yet undiluted methanol in liquid phase catalysis results in a concentration of about 25 mol L^{-1} , hence, about 4–5 orders of magnitude higher, which leads to a considerable increase in coverage. At first sight, a simple solution to this problem is diluting the reagents such as methanol with a solvent. Yet, a solvent cannot necessarily be considered an innocent spectator. In fact, it was demonstrated that a solvent can significant change the reaction rate and the reaction mechanisms. As will be discussed later, alcohol oxidation in liquid phase containing water provides an example, where the first step of the reaction pathway, the deprotonation of the alcohol, is likely to proceed already in solution⁴⁷⁷ and is very dependent on the pH of water-based solvents. Much effort is devoted to the description of the solid–liquid interface by computational methods,⁴⁷¹ but its complexity remains to be a challenge.

Another challenge in aerobic oxidation catalysis in the liquid phase is the necessity to dissolve oxygen in the reaction medium. For instance, 1 bar of oxygen pressure at 298 K gives a concentration in the gas phase of about 40 mmol L^{-1} . At

equilibrium, the oxygen concentration in liquid methanol is on the order of 10 mmol L^{-1.478} Very similar concentrations and supposedly activities can be reached in the liquid phase by applying higher pressures of a few bars in an autoclave. However, the transfer rate of oxygen from the gas phase to the liquid phase strongly depends on the setup. For instance, it was demonstrated that bubbling O₂ through the reaction medium makes a significant difference in the measured activity by up to an order of magnitude as compared to the application of an O₂ atmosphere using a balloon.⁵⁹ The diffusion coefficient of O₂ in solution is about 3 orders of magnitude smaller than in the gas phase at the same temperature. Therefore, proper mixing and stirring may be very important to enhance the otherwise slow mass transport by forced convection.

5.2. Comparison of Mechanistic Aspects in the Aerobic Oxidation of Methanol in Liquid and Gas Phases

Due to its more inert nature, methanol (MeOH) has been employed as solvent by Asao and co-workers,⁴⁷⁹ e.g., in the aerobic oxidation of secondary alcohols. However, MeOH itself can undergo oxidative conversion to methyl formate in the liquid phase at oxygen partial pressures above 1 bar (Figure 24).^{60,480} Here, MeOH is both solvent and reagent. For



Figure 24. Aerobic oxidation of methanol to the corresponding ester (a) in the liquid phase at O_2 pressures of more than 1 bar (batch reactor, after 24 h at 60 °C) and (b) in the gas phase employing a tubular flow reactor (at 60 °C). While the reaction is apparently zeroorder for O_2 in both cases, the onset of CO_2 formation can be observed in the gas phase, indicating total oxidation and higher oxygen concentrations on the catalyst surface in the gas phase reaction. Reproduced with permission from ref 60. Copyright 2017 Elsevier.

mechanistic studies, it would be beneficial to avoid an additional solvent, as this minimizes the number of species in the system and thus possible side reactions and may thus allow elucidating similarities between liquid and gas phase catalysis. The liquid phase aerobic oxidation of methanol was studied in a batch reactor.⁶⁰ Similarly, as in gas phase experiments, the aerobic oxidation of methanol leads to the formation of methyl formate after an induction period in case

the oxygen pressure was above 1 bar (Figure 24a) and the temperature between 40 to 80 $^{\circ}$ C. Taking into account that undiluted methanol (60 mL) and about 100 mg of catalyst was used, the observed conversion was in the range of a few percent. Such low conversions were chosen to reduce the influence of consecutive reactions on products such as water which accumulate in batch type reactions.

The formation of the corresponding ester is in line with gas phase experiments and can be understood by the reaction of the dehydrogenation of adsorbed methoxy and a subsequent coupling of the aldehyde with another methoxy species to form the ester by subsequent hydrogen abstraction reactions. But how does it depend on the oxygen concentration, i.e., the oxygen pressure? As can be seen in Figure 24a, the reaction proceeds only slowly at pressures of 1 bar, yet the rate steeply increases and levels of at pressures above 2 bar. The pressure independence indicates a zero-order kinetics with respect to O₂ as also observed in the gas phase (Figure 24b). Hence, the catalysis is limited by the formation of reactive oxygen on the surface, i.e., the number of reactive surface sites on the surface and not on the proliferation from the gas phase. When plotting the conversion or yield as a function of time, the catalyst activity (rate of the catalyzed reaction) in a batch-type reaction can be determined. Often the rate of the catalyzed reaction is referred to the amount of active sites or more simply to surface sites. The resulting quantity is the TOF. The calculation of the TOF requires knowledge of the surface area of the catalyst and the area density of surface atoms, which can for example be determined by electrochemical methods (section 3.4).³⁰⁵ In this way, the formation of the product molecules per surface atoms and time is calculated as 380 h^{-1} at 80 °C in liquid phase. This value is indeed very comparable to the values obtained in gas phase experiments (396 h⁻¹ at 80 °C).⁶⁰ The observation of an apparent activation energy of 74 kJ mol⁻¹ at temperatures below 330 K in the liquid phase and 61.5 kJ mol^{-1} in the gas phase for the partial oxidation pathway (in the limit of no mass transport limitation) exemplifies the very comparable activity of the NPG catalyst under both conditions.⁶⁰

The most prominent difference under both conditions is the absence of the total oxidation pathway to CO₂ in the liquid phase catalysis (cf., Figure 24a vs 24b) even at pressures up to 7 bar. This can be rationalized by the much higher concentration of methanol in the liquid phase. The total oxidation pathway requires the presence of a surplus of reactive oxygen on the catalyst surface (cf., Figure 17). It is likely that, due to the high concentrations of methanol close to the solidliquid interface, any temporary surplus of oxygen reacts immediately with methanol, resulting in adsorbed methoxy as the predominant surface species which suppresses the total oxidation pathway. This situation can be circumvented by diluting simply with water and, hence, decreasing the concentration in methanol, but already concentrations of water on the order of 5 vol % reduced the observed conversion by more than 50%.⁵⁹ A possible explanation is the formation of the geminal diol when scarce surface-bonded aldehyde reacts with water, forming the geminal diol of methanol instead of its reaction with a methoxy.⁴⁸¹ This exemplifies the difficulty of using solvents to dilute reagents as they can also participate in the reaction pathways.

In gas phase oxidation reactions as well as in theoretical calculations, the mole fraction x_{Ag}^{res} of residual Ag in NPG was determined to be the decisive factor for the activation of molecular oxygen.^{365,407,411,482} The alloying of Au with Ag can change the electronic and geometric properties of the surface, also called ligand and ensemble effects, respectively.⁴⁸³ It can be anticipated that the presence of Ag also impacts the activity and selectivity for the aerobic oxidation in the liquid phase. This was studied for the partial oxidation of methanol and ethanol.⁶⁰ In this context, the corrosion of NPG was modified and adapted to generate NPG with comparable ligament sizes and, thus, comparable specific surface area, but with varying concentration of Ag by employing a galvanostatic dealloying (GD) method (cf., section 2).¹⁸⁶

In the case of methanol oxidation, the activity drastically dropped when increasing the overall Ag content to above $x_{Ag}^{res} > 1\%$ (Figure 25), indicating that the surface of NPG becomes



Figure 25. (a) Aerobic oxidation of methanol to methyl formate (MeFm) as a function of the total Ag content in the samples determined by atomic absorption spectroscopy (AAS) obtained by free corrosion, potentiostatic dealloying, and galvanostatic dealloying. (b) Partial aerobic oxidation of ethanol as a function of Ag content in NPG obtained by GD. The corresponding aldehyde and ester are formed in a ratio of 1:1 at low Ag contents of 1 at. %, yet aldehyde formation increases with increasing Ag content. Experiments were performed at 3 bar O₂ pressure at 60 °C for 24 h. Reproduced with permission from ref 186. Copyright 2017 Royal Society of Chemistry.

less active and even poisoned when larger fractions of Ag are exposed. This is quite in contrast to the CO oxidation in the gas phase.³⁹⁰ It is, however, difficult to align the total Ag content in the bulk with the concentration of Ag on the surface of the ligaments, which might be increased by a factor of 10 and more.⁹⁴ Local Ag concentrations close to those of the master alloy, e.g., $x_{Ag}^{res} = 70\%$, are possible if such Ag-rich regions are exposed to the surface during catalysis (cf., section 3.13). It is reasonable that the likelihood for this scenario

strongly increases with rising in the total Ag content.⁹⁴ The impact of increasing the Ag content at the surface of ligaments up to $x_{Ag}^{res} = 60\%$ was studied in gas phase experiments under UHV conditions by Friend and co-workers using bimetallic Au/Ag surfaces.⁴⁸⁴ The temperature of formation of methyl formate increases for mixed sites on Au sites being the lowest (200 K). Indeed, on Ag-like sites the temperature for formation of the methyl formate increased to 260 K. The formation of the aldehyde proceeded at even higher temperatures of 350 K on such sites. This is the reason why NPG becomes inactive at low temperatures for increasing content of Ag on the surface. Formation of CO₂ was observed on all types of sites only at temperatures above 330 K and at higher oxygen coverages (i.e., above 0.05 ML), which are presumably not achieved in liquid phase catalytic conditions.⁴⁸⁴

5.4. Roles of Acid and Base, and Activation Period of the Catalyst

Several reports on the aerobic liquid phase oxidation of alcohols such as benzyl alcohol⁴⁸⁵ or glucose⁶¹ indicate that the conversion is considerably increased by adding catalytic amounts of a base. One aspect is that the reaction in solution, e.g., the abstraction of the alcoholic proton by OH⁻ can speed up the oxidation on the catalyst surface. In the case of methanol oxidation with NPG catalysts, an increase of conversion by more than 100% was observed after 24 h.⁴⁸⁰ The product distribution was not changed, indicating a similar reaction pathway in neutral and alkaline solutions.⁴⁸⁰ While the rates and activity of the catalyst (slope of the lines in Figure 26) remained similar, a strong impact of the added base on the



Figure 26. Methanol oxidation at 60 °C and 3 bar O_2 as a function of reaction time. Temporal evolution of MeFm formation under neutral as well as basic (12 mmol L⁻¹ KOH) conditions. Reproduced with permission from ref 480. Copyright 2019 The Authors License CC-BY-4.0.

activation period of the catalyst was noted, that is, the time after which a conversion of methanol to methyl formate was detected (Figure 26). Such an activation period is usually ascribed to the removal of advantageous species blocking surface sites and the generation of a catalytically active state, as for example surface oxides, and the concomitant migration of Ag to the surface. In the absence of hydroxyl, reactive oxygen must be generated by adsorption of O₂ on free Ag sites on the surface. By transient adsorption of OH⁻, however, electrons are provided which then can be used to generate an O²⁻ species from O₂ also on available Au surface sites (*):⁴⁸⁰

$$2^* + 2OH^- + \frac{1}{2}O_2 \to 2OH^* + O^{2-}$$
(15)

pubs.acs.org/CR

$$2Ag + O^{2-} + 2OH^* \rightarrow Ag_2O + 2OH^-$$
(16)

This explanation is supported by the observation that the conversion of methanol at NPG commenced without delay associated to an activation period in the presence of catalytic amounts of OH^{-} .⁴⁸⁰ Simply, the entire Au- and Ag-like surface can be activated for reaction with O_2 in the presence of OH^{-} . This is a unique feature of liquid phase catalysis, in which ionic species such as OH^{-} can be facilely employed.

6. NANOPOROUS GOLD AS ELECTRODE MATERIALS

While the preparation of NPG by dealloying is inherently an electrochemical process as discussed in section 2, the applications discussed in sections 4 and 5 are not electrochemical. Section 6 deals with the use of NPG as an electrode material, at which reactions *other* than dealloying are conducted. While NPG is an active electrode during dealloying, i.e., an electrode from which material is dissolved, NPG is mainly used as inert electrode material in electrochemical application very much like conventional planar gold electrochemistry refers to an electrode that supplies/accepts electrons to/from reactants dissolved in an adjacent liquid electrolyte solution but does not change its own mass by electrolytic dissolution/deposition.

The electrode reactions are often complicated multistep reactions involving adsorbed intermediates and coupled homogeneous reaction steps in solution. Reactants that adsorb at the electrode surface compete for adsorptions sites with anions, cations, and sometimes solvent molecules of the electrolyte. The stabilization of reaction intermediates at the electrode surface may accelerate otherwise slow reactions and is often called (heterogeneous) electrocatalysis due to the similarity to heterogeneous catalytic processes covered in section 4. Consequently, the reactions involving adsorbed intermediates tend to show a specific dependence of the precise surface composition and structure of the adsorption site. Electrodes that exhibit a particularly fast rate of a specific reaction are also called "electrocatalytically active", a terminology that should not be confused with "active dissolution" as an important feature during the preparation of the NPG electrode itself.

Homogenous electrocatalysis as the analogue to homogeneous catalysis exploits a molecularly defined reaction center (transition metal complex, enzyme,...) with which reactants interact. The electrode then supplies electrons to or accepts electrons from such a reactions center rather than directly to/ from the adsorbed reactant. Both forms of electrocatalysis can benefit from the tunable properties of NPG. Next to a dedicated chapter in the book on NPG,³⁷ the application of NPG electrodes in the sense described above has recently been covered in several specialized reviews.^{26,64,70,486,487}

6.1. Comparison of Nanoporous Gold to other Gold Electrodes

NPG shares all properties with planar Au electrodes that are directly linked to the properties of metallic gold as the main constituent of NPG. This includes, for instance, the applicable potential ranges in different electrolyte solutions as well as specific adsorption of HSO_4^- , halide, and pseudohalide anions



Figure 27. (A) Schematic representation of the internal surface of NPG by free corrosion in the presence of surfactants. (B) voltammetric scans for the stripping of a Pb UPD layer in 10^{-3} mol L Pb(NO₃)₂ + 0.1 mol L⁻¹ NaOH, $\nu = 50$ mV s⁻¹ with fits of the UPD signals for low-index crystal facets: (B1) from Ag₆₅Au₃₅ in 58 wt % HNO₃ at room temperature for 3 h; (B2) from Ag₆₅Au₃₅ in 58 wt % HNO₃ + 0.2 mg mL⁻¹ pyrogallol at room temperature for 20 h; (B3) from Ag₇₅Au₂₅ in 48% HNO₃ + 2.0 mg mL⁻¹ sodium citrate at room temperature for 6 h. (C) Comparison of the proportions of the different facets as determined from the area of the Pb UPD peaks in (B1–B3). Reproduced with permission from ref 340. Copyright 2017 Springer Nature.

(I⁻, Br⁻, Cl⁻, CN⁻, OCN⁻, SCN⁻) and chemisorption of sulfur-containing compounds.^{488–490} F⁻ and ClO₄⁻ are anions that do not chemisorb on Au and are therefore preferred anions for the characterization of the electrical double layer in the absence of specific adsorption.⁴⁸⁸ Commonly used electrolytes with $\text{HSO}_4^-/\text{SO}_4^{2-}$ form complicated potential-dependent structures of adsorbed ions on gold.^{491,492} Many cations such as Pb²⁺, Bi³⁺, Ag⁺, and Cu²⁺ interact specifically with NPG and planar Au by a process called underpotential deposition/stripping (UPD).^{493,494} Such ions form a mono-layer of (partially) reduced metal ions at a foreign substrate (e.g., Pb²⁺ on Au) at potentials more positive than the thermodynamic potential of the electrolytic deposition of this ion on its own metallic substrate (e.g., Pb²⁺ on Pb). The UPD process is very sensitive for specific sites, and we will consider it in detail in section 6.1.3.

Besides similar material properties, there are important differences between NPG and planar gold electrodes such as the specific surface area, the density of low-coordinated surface atoms as well as the occurrence of low amounts of LNE in NPG that have not been removed completely in the dealloying process (sections 2 and 3). Mass transport is very often a limiting factor in electrochemical reactions because diffusion in liquid electrolytes is orders of magnitude slower than in gas phase. For a porous electrode with a bicontinuous distribution of metal and electrolyte phases, mass transport limitation may arise from the transport of the reagents to the outer parts of the porous electrode, similar to the situation at a planar electrode. This transport mode is called external mass transport. In order to utilize the huge surface area of NPG, it is important that reagents reaching the outer parts of an NPG electrodes are distributed within the pore network to the inner surface area of the electrode by a process termed internal mass transport. This general consideration applies to diffusive mass transport as well as to the transport of ions by migration.

6.1.1. Comparison of Surface Voltammetry of Au Electrodes. Besides the determination of surface area covered in section 3.4, surface voltammetry of noble metal electrodes can reveal details about surface facets, defects, and specific adsorption of ions. The signals in surface voltammograms of gold result from the chemisorption of OH⁻, the formation of a surface oxide layer, whose completion can be clearly identified by a minimum in current-voltage curves, called the Burshtein minimum, which is well identified at around 1.65 V (RHE) in Figure 10a.³⁰⁷ At the potential of this minimum, exactly a two atom-thick "Au^{II}O" layer is formed. 495,496 Other processes, such as exchange or reorientation of ions in the double layer, are associated with capacitive currents. They can be of diagnostic value if very clean electrodes and electrolyte solutions are used.³⁶⁴ However, quite often the exact nature of a specific signal is not clear because the structure-signal relationship are derived from combinations of voltammetry and structure-sensitive methods at planar single-crystal electrodes^{491,492} that lack the multitude of low-coordinated surface atoms that are characteristic of NPG. For nanoporous gold, the voltammetric signals from different facets with their individually distinct defect types overlap. An assignment to structural features of the surface is typically made by a "fingerprint" method, in which the voltammogram at NPG is considered as a superposition of contributions from single-crystal faces. Thus, characteristic voltammetric features are compared to voltammograms obtained at single-crystal electrodes under the same

conditions (especially in the same electrolyte solution). An authoritative source of single-crystal voltammograms in acidic and neutral electrolytes is available from the work of Hamelin.^{308,497} This source also contains an in-depth discussion of experimental problems. For alkaline electrolytes, the data of Feliu and co-workers are recommended for a comparison.⁴⁹⁸ For convenience, recalculation of the potential

of common reference electrodes is detailed in SI, SI-2, examples of single-crystal voltammograms from refs 308 and 498 are reproduced in the SI, SI-3, Figures S4 and S5. Further sources of single-crystal voltammetry are listed in SI, SI-3, Table S4.

6.1.2. Tuning of Internal Surface Structure as Followed by Metal Underpotential Deposition/Stripping. Besides surface voltammetry in common electrolyte, the observation of UPD and stripping of metal adlayers is very sensitive to specific features on surfaces.⁴⁹³ Pb²⁺ yields characteristic UPD signals, both on Ag and Au electrodes and has been particularly often applied as structure-sensitive probe reaction. The scientific basis for using this system rests in the observation of an electrosorption valency close to the charge of Pb²⁺, the absence of surface alloying under electrochemical conditions and the relatively small effect exerted by water and anions on the particular UPD process of Pb²⁺ on Au facets.⁴⁹⁹ The UPD signals occur at different potential for the different single-crystal faces due to electronic interaction between the Au substrate and Pb²⁺ at the specific sites and possible superstructures of the UPD layer on a specific face of the Au substrate. Generally, more than one signal is observed even for well-prepared single-crystal electrodes. The signals correspond to the transitions between different superstructures of the adatom on the substrate.499 With this in mind, it is clear that the commonly used fitting of Pb UPD signals on NPG with a low number of peaks clearly ascribed to one low-index facet (as in Figure 27B1, B2, and B3) is a very coarse approximation. For convenience and to illustrate this point, voltammograms of Pb UPD on singlecrystal Au are compiled in SI, SI-4, Figures S6-S12. The charge corresponding to the different peaks is proportional to the surface area of the specific facet within the NPG structure.

This method can be applied both in acidic as well as in alkaline solution.^{499,500} Pb²⁺ ions added, e.g., as Pb(NO₃)₂, in precisely weighted amount to the alkaline electrolyte solution are immediately converted to plumbites $[Pb^{II}(OH)_m]^{(2-m)-}$ (m > 2), which dissolve in alkaline solution up to millimolar concentrations. Pb²⁺ stripped off the surface by the oxidation of a Pb UPD layer dissolves in acidic electrolytes. The stripping of an Pb UPD layer in alkaline solution forms $[Pb^{II}(OH)_m]^{(2-m)-}$ that remain at the surface as specifically adsorbed anion. This fact can be used for selectively blocking surface structures for electrocatalysis.¹⁰⁸

Using Pb UPD signals to characterize the internal surface structure, it became evident that tuning of this internal surface structure is possible by adding surfactants while forming NPG by free corrosion in HNO₃ solution (Figure 27A).³⁴⁰ Pyrogallol formed an inner surface enriched in {111} facets, and addition of citrate promoted the formation of {100} facets (Figure 27B1, B2, B3, and C). The different ratios of facets in NPG materials strongly correlated with different activities for methanol oxidation reaction (MOR) in 0.5 mol L⁻¹ KOH with the highest activity found for the {111}-rich internal surface formed in the presence of pyrogallol.³⁴⁰

The internal surface structure of NPG can also be tuned by a post-dealloying method. Cycling NPG in 0.1 mol L⁻¹ H₂SO₄ in the potential range of surface oxide formation and reduction lead to a restructuring of an initially amorphous surface to a {111}-rich internal surface, when the cycling was conducted at a low potential scan rate of v = 5 mV s⁻¹, whereas a {100}-rich internal surface was obtained by cycling at 50 mV s⁻¹ in the same electrolyte.⁴⁶²

It is key that the cycling is conducted in acidic solution because residual Ag, that tends to enrich at the surface during storage in air or during cycling in alkaline solution, may impede the restructuring because this process does not occur in 0.1 mol L⁻¹ KOH solution.¹⁹² However, conducting one cycle in 0.1 mol L⁻¹ H₂SO₄ removes the surface-enriched Ag (and Ag oxides) and allows restructuring. The restructuring can be confirmed by the observation of the signals from lowindex crystal facets by surface oxidation¹⁹² and Pb UPD.¹⁰⁸ After stripping of UPD-Pb, $[Pb^{II}(OH)_m]^{(2-m)-}$ (m > 2) remains at the surface as specifically adsorbed anions or as PbO₂-like species at potentials more positive than 0.25 V (Hgl HgO).¹⁰⁸ Using a small amount of NPG in a cavity microelectrode (Figure 6) and adjusting the bulk concentration of plumbites in 0.1 mol L^{-1} KOH facilitated a very controlled partial coverage of the internal facets in NPG. By increasing the plumbite concentration in the bulk solution by adding more $Pb(NO_3)_2$ to the alkaline electrolyte, the facets were sequentially filled up in the sequence of their energetics for the Pb UPD (Figure 28A).¹⁰⁸ Also in this case, the {111} facets were crucial for MOR: When the {111} facets were covered by Pb species, the current maximum at 0.25 V (Hgl HgO) dropped significantly (Figure 28B), whereas the onset of MOR shifted to a more positive potential already with the coverage of defects and {100} and {110} facets (Figure 28B, inset).

Pb UPD also showed that NPG obtained by roughening pure Au electrodes by cycling in Cl⁻-containing electrolytes exhibits a morphology very similar to that obtained by dealloying (Figure 5C). In both cases, {111} facets dominate the internal structure of the material.²¹ Such intentional roughening procedures for Au electrodes are commonly applied and result in specific electrocatalytic reactivities.^{501–504}

Ag (and Cu) also show structure-sensitive UPD processes onto NPG.³¹⁴ However, the UPD process of Ag (and Cu) on Au is complicated by the tendency of Ag to alloy with Au.³¹⁴ Such surface alloys form at room temperature in diluted acids on a time scale of 5–45 min.³¹⁴ This has been used to modify the amount of Ag on the surface after dealloying *without* affecting the bulk Ag content in the ligaments (Figure 29A).⁵⁰⁵ Ag UPD layers on NPG have only a modest effect on the MOR peak current, corresponding to the 4e⁻ oxidation of methanol to formate (Figure 29B). The onset of the peak is shifted to less positive potentials with higher Ag surface concentration.⁵⁰⁵ However, the peak for the direct 6e⁻ oxidation of methanol to CO₂ at more positive potentials of E > 1.5 V (RHE) significantly grows with a higher Ag surface concentration (Figure 29C). In this potential range, Ag is present as AgO-like species as detected by XPS.⁵⁰⁵

6.1.3. Model Reactions that Show Differences to Planar Au Electrodes. Differences in the reactivity of clean planar Au electrodes and NPG have been described for a number of electrocatalytic reactions. Some could be related to specific surface structures as described in section 6.1.3 for the MOR. Another structure-sensitive reaction is the CO_2





Figure 28. (A) Cyclic voltammograms of NPG *partially* covered by a Pb UPD layer (5th cycle, $\nu = 10 \text{ mV s}^{-1}$) in 0.1 mol L⁻¹ KNO₃ + 0.1 mol L⁻¹ KOH with addition of different concentrations of Pb(NO₃)₂ (converted to $[Pb^{(II)}(OH)_m]^{(2-m)-}$). (B) Cyclic voltammograms of methanol electrooxidation at NPG partially covered by Pb UPD or adsorbed $[Pb^{(II)}(OH)_m]^{(2-m)-}$ ($\nu = 10 \text{ mV s}^{-1}$) in 1 mol L⁻¹ MeOH + 0.1 mol L⁻¹ KNO₃ + 0.1 mol L⁻¹ KOH with different concentration of $[Pb^{(II)}(OH)_m]^{(2-m)-}$. The surface is covered by specifically adsorbed $[Pb^{(II)}(OH)_m]^{(2-m)-}$ anions in the potential range of MOR. The bulk concentrations of $[Pb^{(II)}(OH)_m]^{(2-m)-}$ in mol L⁻¹ are (1) 0, (2) 1 × 10⁻⁵, (3) 5 × 10⁻⁵, (4) 1 × 10⁻⁴, (5) 5 × 10⁻⁴, (6) 1 × 10⁻³ in (A) and (B). Reproduced and adapted with permission from ref 108. Copyright 2021 The Authors under license CC-BY.

reduction reaction (CO_2RR) .⁵⁰³ This particular reaction does not only depend of the available facets that influence the energetics and geometries of adsorption sites for reaction intermediates but also on the mesostructure of the material.²⁵¹ For instance, the selectivity of CO_2RR can be favored vs the hydrogen evolution reaction (HER) by exploiting limitations of the internal mass transport (Figure 6A) in thicker NPG layers where the increased alkalinity helps to suppress the HER.²⁵¹

The oxidation of ascorbic acid to dehydroascobic acid proceeds at solid electrodes by an EC_1 mechanism, i.e., an electron transfer step (E) is followed by a homogeneous first-order decomposition reaction (C_1) of the initially formed

product to dehydroascorbic acid. The relative rates of the E and C₁ steps depend on the internal mass transport in the pore system, which could be modified by changing parameters in the preparation of the porous Au electrodes using the H₂ bubble templating method.⁵⁰⁶

6.1.4. Structure Evolution of NPG during Use as **Electrode Material.** Besides the intended modification of the internal surface structure (section 6.1.3), the surface structure may also change unintentionally during application of the electrode. As the NPG structure is a metastable state, a trend toward formation of larger ligaments and pores is expected. Most often, this is noted by a decrease of the electrochemical surface area A_{ECSA} over time. However, the available reports arrive a quite different conclusion about the morphological stability of NPG during potential cycling or during electrocatalytic reactions. First reports showed a decrease of A_{ECSA} during continuous cycling for MOR in alkaline solution. Others reported stability of electrodes for MOR in alkaline electrolyte for NPG obtained by alloying/dealloying in an ZnCl₂-benzylacohol electrolyte.⁶³ As has been described in sections 2.1 and 4.4, the coarsening can be prevented or significantly retarded by the use of ternary alloys that contain a small fraction of an element with lower surface mobility, e.g., Pt,^{100,113–115} or by decorating NPG with Pt.⁶² A layer of Pt on NPG can also be obtained after dealloying by first forming a Cu UPD layer and then using a galvanic replacement reaction of Cu for Pt. The resulting material was structurally stable and had a high oxygen reduction reaction activity due to the Pt layer.⁵⁰⁷ Galvanic replacement of Pt for Cu has also been used to decorate Cu-containing porous materials obtained by bubble-templated electrodeposition.¹⁹ Here again, enriching the surface in Pt significantly increases the stability compared to pure porous Au.¹⁹ A similar process has also been demonstrated by first forming a Pb UPD layer and then using galvanic replacement of Pb for Pt to form a thin catalyst layer.

There have also been reports that NPG can be obtained by pulsed electrochemical dealloying by which the ligament size can be tuned. Materials with an average ligament size of 8 nm showed the highest activity for glucose oxidation in neutral buffer but also the highest stability.²¹⁷ During storage, the surface composition was changing. For NPG produced from AuAg alloys, Ag is enriched at the surface (determined by Xray photoelectron spectroscopy (XPS)) relative to the bulk ligaments (determined by EDX).¹⁹¹ The enrichment factor $x_{Ag, XPS}^{res}/x_{Ag, EDX}^{res}$, is decreasing with the overall content of residual Ag.¹⁹¹ The stability of those materials during electrochemical cycling depends very much on the details of the cycling condition.¹⁹² If the material is cycled in alkaline solution only after an intermediate storage in air, the material is stable and shows the characteristic anodic peak A2 at E = 1.5 V (RHE) in Figure 29A. However, a short cycling in H_2SO_4 removes surface Ag as shown by XPS.¹⁹² Afterward, coarsening sets in, as observed by a decrease in A_{ECSA} . A transfer back to an alkaline solution after short cycling in H₂SO₄ shows that the surface oxidation process A2 in Figure 29A is lost and does not recover during short cycling experiments.¹⁹²

Material obtained by dealloying an amorphous starting material $Au_{40}Cu_{28}Ag_7Pd_5Si_{20}$ becomes not only nanoporous but also nanocrystalline.⁵⁰⁸ In other words, it develops a polycrystalline microstructure with a grain size of nanometers, comparable to the ligament size. The intercrystalline regions are amorphous. The material is very active for MOR but loses



Figure 29. Effect of UPD-Ag at the surface of NPG. (A) Cyclic voltammograms in 0.1 mol L⁻¹ NaOH at $\nu = 10$ mV s⁻¹ with variable amounts of UPD-Ag. The legend indicates x_{Ag}^{res} as determined by XPS. (B) Cyclic voltammograms of the same electrodes in 1 mol L⁻¹ MeOH + 0.1 mol L⁻¹ NaOH at $\nu = 10$ mV s⁻¹. (C) Maximum current densities of signals MA1 and MA2 in (B) as a function of x_{Ag}^{res} (as determined by XPS). Reproduced with permission from ref 505. Copyright 2022 American Chemical Society.

the activity rapidly.⁵⁰⁸ NPG obtained by dealloying from the same metallic glass in short time (10 min to 1 h) also showed a rapid decay of catalytic activity for MOR in alkaline solution.⁵⁰⁹ NPG, which was obtained by first assembling a layer of Au nanoparticles in a Langmuir trough and subsequently interconnecting them by electroless deposition of gold, yielded a freestanding membrane.²² However, the voltammetric signals for MOR in alkaline solution rapidly decayed, in agreement with the absence of less mobile atoms that stabilize the high energy structures.²²

6.2. Electrocatalysis of Model Reactions

NPG has been tested as an electrocatalyst for a variety of reactions that include the reaction for which platinum group metals are typically used in practical application. Besides costs, potential advantages of NPG are the ease by which a porous structure with tuned mass transport conditions can be formed without the need of a support material, such as carbon, that is susceptible to corrosion itself. Gold is chemically stable in a wide range of aqueous solutions. However, mass-related and area-related activity remain below that of Pt and Pt alloys. Furthermore, the stability of the NPG morphology as a metastable structure must be sustained over long periods. This has turned out to be challenging and, therefore, the use of NPG as electrode material for sensors has received more attention. The requirement for operation stability over very long times is relaxed for sensors, and aspects of system integration are of higher importance than mass-related or arearelated activity.

6.2.1. Hydrogen Evolution Reaction. Hydrogen evolution reaction (HER) is one of the simplest electrocatalytic reactions that is often used to compare different electrode materials. Reaction rates, represented by exchange current densities or by heterogeneous standard reaction rate constants of hydrogen evolution on different metals vary by several orders of magnitude and are often represented as volcano plots.^{510,511} Besides the scientific interest in HER, it is an important reaction in water electrolysis that currently receives revived attention. Due to the position of Au in the volcano plots for HER, early approaches to utilize NPG for HER resorted to modification of the surface of NPG by a monolaver of Pt using galvanic exchange reactions.⁵¹² Interestingly, some recent works used porous Au electrodes without Pt-group elements for HER: Islands of porous Au obtained by bubbletemplated electrodeposition and thus without remaining LNE showed stable operation and impressive area-related HER

currents.²⁵⁵ Similar performance was obtained by electrochemically roughening of a planar gold electrode.⁵¹³ The procedure involves oxidation of the surface at 2.0 V (RHE) in 0.5 mol L^{-1} H₂SO₄ followed by electroreduction of the optically visible gold oxide film.⁵¹³ The control and efficacy of NPG preparation by anodization of massive Au electrodes can be refined by exploiting the effects of specifically adsorbed anions in the anodization solution that increased the ratio of $\{111\}$ vs $\{100\}$ facets in the resulting material.²¹ The increased fraction of {111} facets was given as the main reason controlling the high activity $(3 \text{ mA cm}^{-2} \text{ at } -0.049 \text{ V} \text{ (RHE)}$ in 0.5 mol L⁻¹ H₂SO₄),²¹ because experiments on planar Au single-crystal electrodes had shown a 2.3 times higher activity for the $\{111\}$ facet compared to the $\{100\}$ facet.⁵¹⁴ Efficient catalysts have also been prepared by forming an aerogel of interconnected Au nanoparticles on carbonitride membranes that sustained an HER current density of 10 mA cm⁻² at -0.185 V (RHE).⁵¹⁵

6.2.2. Oxygen Reduction Reaction and Other Oxygen Redox Chemistry. Oxygen reduction reaction (ORR) at planar gold electrodes proceeds much slower than at Pt and with a considerable formation of hydrogen peroxide H_2O_2 .⁵¹⁶ H_2O_2 causes subsequent material degradation in electrochemical cells. Although its formation may be intended, it is mostly an undesired byproduct in fuel cell or energy applications or when ORR is used as anodic reaction for a synthetic reduction reaction. The best catalysts for oxygen evolution reaction (OER) are IrO_2 and RuO_2 .⁵¹⁷ As the availability of Ir and Ru is even more critical than that of Pt, there is an intensive search for alternative electrode materials for OER.

Plating Pt on NPG is an obvious application scenario for NPG in the context of ORR electrocatalysis and has been demonstrated very early.⁵¹⁸ In this work, Pt was deposited at different loading between coverages of a submonolayer, a wetting monolayer, and as nanoparticles on a monolayer.⁵¹⁸ Pt-containing NPG has also been obtained by dealloying ternary alloys such as $Pt_1Au_{0.5}Cu_{98.5}^{-225}$ (see section 2.2) or by dealloying a AuAg alloy and modifying the resulting NPG by Cu UPD followed by a galvanic exchange reaction of Pt for Cu.⁵⁰⁷ By repeating Cu UPD on the first Pt layer followed by another galvanic exchange reaction, the amount of deposited Pt could be increased in a defined way above a monolayer coverage.⁵⁰⁷ Those electrodes showed very good performance and stability even when compared to commercial Pt/C

catalysts. A key feature is the low amount of formed H_2O_2 and the better adhesion of Pt nanostructures to Au compared to the adhesion of Pt nanoparticles to conventional carbon supports.⁵⁰⁷ Oxygen evolution has been demonstrated with Pt-modified and Ir-modified anodized porous Au.⁵¹⁹

While the results on Pt-modified NPG may not be entirely unexpected, it was reported rather early that Pt-free NPG also shows a remarkable ORR activity and that H₂O₂ formation known from planar Au electrodes is effectively suppressed.³³⁹ The increased catalytic activity is due to the higher fraction of low-coordinated surface sites in NPG relative to planar Au. The reaction at NPG as well as at Au occurs by two sequential 2e⁻ steps (as opposed to the 4e⁻ route on Pt). At NPG, the second step, namely H₂O₂ reduction, is very efficient, probably further supported by the confinement of H_2O_2 in the pore space of NPG.^{339'} NPG obtained from metal-induced crystallization and thus without LNE showed stable ORR activity in alkaline solution, which was higher than the activity of NPG the authors obtained from dealloying.520 Other preparation methods that are more suitable for sensors like anodization at 2.0 V (AglAgCl) and subsequent electroreduction of gold oxides yielded linear calibration curves for ORR currents over a wide range of O2 content in neutral solution.²²⁶ Aerogels of interconnected Au nanoparticles on carbonitride membranes showed high onset potentials for ORR in alkaline and acidic electrolytes.⁵¹⁵ The performance was dramatically increased by forming aerogels from AuCu alloy nanoparticles directly during the synthesis.⁵²¹ Those aerogels not only outperformed Au aerogels in ORR (0.906 mA cm⁻² at 0.85 V (RHE) for the optimum composition Au₅₂Cu₄₈) but also Pt/C catalysts $(0.203 \text{ mA cm}^{-2})$.⁵²¹ Another advantage is the high methanol tolerance of the AuCu aerogels compared to Pt/C catalysts.⁵²¹

The use of NPG for ORR in *nonaqueous* metal—air batteries is covered in section 6.5.2.

6.2.3. Alcohol Oxidation Reaction. As discussed in section 6.1.3, alcohol oxidation, especially methanol oxidation, is very sensitive to surface structures. Its therefore often used as a model reaction to gain insights in structure–reactivity relations at electrocatalytically active electrodes.⁵²² The oxidation of methanol can be conducted in acidic and alkaline solutions. Koper and co-workers⁵²³ attributed the higher activity of the alcohol oxidation reaction in alkaline media to a homogeneous deprotonation of the alcohol preceding its adsorption although some interactions with the (planar) gold surface is necessary for the reaction to occur.

$$R-CH_2OH + OH = R-CH_2O + H_2O$$
(17)

The oxidation proceeds at the bare metallic surface in the potential range 0.9-1.35 V (RHE) (Figure 29B, signal MA1). At more anodic potentials, the current drops when the surface oxide layer is formed at the electrode surface. At potentials more positive than the Burshtein minimum, the oxidation commences again and may overlap with oxygen evolution (Figure 29B, signal MA2). In principle, several oxidation reactions are possible, each composed of several elementary reactions.⁵²²

$$R-CH_2OH + 2OH^- \rightarrow R-CHO + 2e^- + H_2O$$
(18)

 $\text{R-CH}_2\text{OH} + 5\text{OH}^- \rightarrow \text{R-COO}^- + 4\text{e}^- + 4\text{H}_2\text{O}$ (19)

$$HCH_2OH + 8OH^- \rightarrow CO_3^{2-} + 6e^- + 6H_2O$$
 (20)

The 2e⁻-oxidation leads to the aldehyde RCHO. In the case of MeOH as reagent, the product formaldehyde enters the Cannizzaro reaction 21 in alkaline solution, in which the primary oxidation product disproportionates to MeOH and formate.

$$\text{H-CHO} + \text{H-CHO} + \text{OH}^- \rightarrow \text{CH}_3\text{OH} + \text{HCOO}^-$$
(21)

The Cannizzaro reaction is preferred only if there is no Hatom in α -position to the carbonylic C-atom; thus, it does not occur for ethanol. However, the aldehydes of higher alcohols are also unstable in alkaline solution, especially in the presence of dissolved oxygen.⁵²³ Thus, product analysis may underestimate the amount of aldehyde formed in the electrode reaction.

The reactivity in the different potential regions is influenced by the number of low coordinated sites, the crystal facets exposed to solution and the content of LNE that remains from the dealloying process or is purposefully deposited on the surface of the ligaments. However, the influence of those factors is very different in the different potential regions. The onset of the methanol oxidation current is shifted to less positive potentials in agreement with the role of Ag (Figure 29B, signal MA1).¹⁰⁸ However, the peak current at 1.2 V (RHE) decreases when the Ag content increases above an optimum value. At potentials more positive of the Burshtein minimum (1.35-1.45 V (RHE) in Figure 29B), no pronounced peak for methanol oxidation is observed for NPG with $x_{Ag}^{res} < 1\%$, whereas there is a clear signal MA2 if Ag is present at the surface of the ligaments at $x_{Ag}^{res} > 4\%$, either as remaining from the dealloying process¹⁰⁸ or intentionally added by UPD to the surface.⁵⁰⁵ There is also a very clear dependence on the crystal facets that dominate in the material. By selective poisoning the surface, it was found that especially the $\{110\}$ facet is most active close to the onset potential,¹⁰⁸ in agreement with findings for ethanol oxidation on planar Au single-crystal electrodes.524

Disentangling the effects of surface sites and elemental composition is not trivial because morphological and compositional parameters are correlated to each other and both may influence electrocatalysis. As has been pointed out in section 2, the content of residual Ag in NPG obtained by dealloying Ag-Au correlates systematically with the ligament size. Larger size implies a lower overall Ag content due to longer and thus more complete dealloying. In a comparison of the catalytic activities in NPG with different ligament sizes, it was found that the fine ligaments showed the lowest current density related to the internal surface area.³⁵ However, the samples with the smallest ligaments also had the highest Ag content, and thus it is difficult to relate the effect to either of the two factors.³⁵ Interestingly, the selectivity for formate production vs formaldehyde changed with ligament size and Ag content when electrolyzing methanol at a constant potential (selected as the peak potential of respective cyclic voltammograms).³⁵ At this potential, the surface is slowly covered by an oxide layer, which may change the activity and selectivity. As shown in Figure 29A, the metallic character of the surface is preserved to higher potentials when more Ag is present at the surface and promoted the oxidation formate. In another study, formaldehyde was not detected as a product by ¹H NMR after prolonged electrolysis in alkaline solution, probably due to fast further oxidation to formate and consumption by the homogeneous reaction 21 in alkaline solution.⁵

pubs.acs.org/CR

Table 1. Performance Data of Selected NPG Electrodes for Methanol Oxidation Reaction

electrode, preparation route	electrolyte	$\begin{bmatrix} v \\ mV \\ s^{-1} \end{bmatrix}$	E _p [V vs RHE]	$[\mathrm{mA}^{j_\mathrm{p}}\mathrm{cm}^{-2}]$	ref
potentiostatic dealloying $Ag_{77}Au_{22}Pt_1$ in 0.77 mol L^{-1} HClO ₄ passing a charge of 5 C cm ⁻² , subsequent annealing in air at 425 $^\circ$ C for 2 h	1 mol L^{-1} MeOH + 5 mol L^{-1} KOH	10	0.69	0.92	525
free corrosion of $Au_{16}Cu_{84}$ nanoparticles in 3 mol L^{-1} HNO3, final composition $x_{\rm res}^{\rm res}$ = 2% (EDX)	1 mol L^{-1} MeOH + 0.5 mol L^{-1} KOH	5	1.30	0.045	241
potentiostatic dealloying from Ag ₇₅ Au ₂₅ , final $x_{Ag}^{res} < 1\%$ (EDX), UPD deposition of Ag to surface content of $x_{Ag}^{res} = 9\%$ (XPS)	1 mol L ⁻¹ MeOH +0.1 mol L ⁻¹ NaOH	10	1.25	0.084	505
potentiostatic dealloying from $Ag_{75}Au_{257}$ final $x_{Ag}^{res} < 1\%$ (EDX)	1 mol L ⁻¹ MeOH + 0.1 mol L ⁻¹ NaOH	10	1.20	0.042	505
porous gold nano bowls, $x_{Ag}^{res} = 4\%$, from reduction of HAuCl ₄ by hydroquinone in presence of poly(vinylpyrrolidone) and AgCl nanocubes, removal of AgCl by NH ₄ OH	$\begin{array}{l} 2 \text{ mol } L^{-1} \text{ MeOH} \\ + 0.5 \text{ mol } L^{-1} \text{ KOH} \end{array}$	20	1.18	0.133	527
planar polycrystalline Au	$1.5 \text{ mol } L^{-1} \text{ MeOH} + 0.01 \text{ mol } L^{-1} \text{ KOH}$	10	1.25	0.076	528



Figure 30. MOR studied at NPG obtained from $Ag_{77}Au_{22}Pt_1$ at 298 K, current densities are calculated with respect to the true surface area. (a) Variation of scan rate; 1 mol L⁻¹ MeOH + 5 mol L⁻¹ KOH in quiescent solution. (b) Variation of solution agitation by an external stirrer, $\nu = 10$ mV s⁻¹, 1 mol L⁻¹ MeOH + 5 mol L⁻¹ KOH. (c) Variation of the positive vertex potential, $\nu = 10$ mV s⁻¹, 1 mol L⁻¹ MeOH + 5 mol L⁻¹ KOH. (c) Variation of the positive vertex potential, $\nu = 10$ mV s⁻¹, 1 mol L⁻¹ MeOH + 5 mol L⁻¹ KOH, quiescent solution. (d) Influence of MeOH concentration, $\nu = 10$ mV s⁻¹, MeOH + 5 mol L⁻¹ KOH, quiescent solution. Reproduced with permission from ref 525. Copyright 2016 Springer.

Ag was found to play a crucial role for the activity of porous nanoparticles obtained from dealloying an Au–Ag–Cu ternary alloy.²⁴¹ Ag enhanced the dissolution rate of Cu and stabilized the porous structure and substantially enhanced the catalytic activity related to the mass of Au in the electrode material compared to porous nanoparticles prepared from Au–Cu alloys.²⁴¹ This effect was found for methanol, ethanol, 2-propanol, and glycol.

The presence of Ag also changes the product distribution.⁵⁰⁵ By investigation the oxidation of methanol and formate in alkaline solution at NPG obtained by dealloying to $x_{Ag}^{res} < 1\%$ and then reintroducing Ag by UPD, it could be shown that increasing Ag surface concentration leads to an optimum of x_{Ag}^{res} for the 4e⁻ oxidation at 1.2 V (RHE). A further oxidation of methanol and formate to CO₃²⁻ occurs only at an oxide-covered surface at 1.55 V (RHE) when Ag oxide is present at

the surface. The current density of this process increases with x_{Ag}^{res} (Figure 29B,C).⁵⁰⁵

High current densities at lower potentials (starting at -0.4 V HglHgO or 0.53 V (RHE)) were obtained with NPG obtained from dealloying Al–Au–Pt and Al–Au–Pt–Pd alloys in NaOH compared to materials obtained from a Al–Au alloy.²⁴² The material was processed to a powder, from which a catalyst ink was formed with Nafion as binder. The current densities were reported as normalized to the catalyst mass and geometric area of the electrode rather than to $A_{\rm ECSA}$ (Table 1). The much higher activity of the Pt- and Pd-containing alloys is very likely an effect of the intrinsic catalytic activity of Pt and Pd contained in the alloy. However, their effect may not only originate from their intrinsic catalytic activity alone but also from their lower surface mobility. Slowing down the rates of metal atom surface diffusions allows obtaining material with



Figure 31. MOR studied at NPG obtained from $Ag_{77}Au_{23-x}Pt_x$ ($x = \{1, 2, 3\}$) at 298 K with and without surface segregation of Pt during treatment at 425 °C for 2 h. (A) Variation of ligament size and surface Pt content. (B) Comparison of peak current densities before and after Pt surface segregation. Peak current densities are calculated with respect to the true surface area. v = 10 mV s¹, 1 mol L¹ MeOH + 5 mol L¹ KOH, quiescent solution. Reproduced in part with permission from ref 525. Copyright 2016 Springer.

smaller feature sizes and thus a higher mass-related surface area. $^{\rm 242}$

The effectiveness of such catalysts with respect to the use of Pt as the most valuable component can be optimized by dealloying Au–Ag–Pt alloys with only 1–3% Pt.⁵²⁵ In this case, the alloy was prepared as ribbon, which was subsequently dealloyed partially by passing a fixed charge under potentio-static conditions. This causes the dealloying front to advance 7–10 μ m into the 200 μ m thick ribbon. There are significant differences to the dealloying of binary Au–Ag alloys: as has been noted above, the addition of Pt yields smaller ligament size (see section 2.2). Due to the correlation between ligament size and residual Ag, the residual silver content in nanoporous Au–Pt can be quite high, e.g., 45–55%, ¹¹⁵ depending on the exact alloy and the depth in the dealloyed layer. Dealloying protocols that bring down x_{Ag}^{res} to 10% even in bulk samples of nanoporous Au–Pt have been reported.¹²⁰

Those electrodes achieved higher current densities of 210 μ A cm⁻² under comparable conditions than those found by dealloying Ag–Au alloys (Table 1).⁵²⁵ The peak current density is observed at considerable less positive potentials than for NPG obtained from Ag–Au alloys. The study also provided a careful evaluation of other limiting factors.⁵²⁵ Mass transport limitations are evident from the dependence on the scan rate (Figure 30a) and the possibility to increase the observed current density by enhancing the external mass transport by stirring (Figure 30b).⁵²⁵ In this study, potential excursions were generally restricted to avoid the formation of surface oxides because this may impact the MOR in the negative-going half-cycle. Finally, the role of the MeOH concentration was demonstrated (Figure 30d).⁵²⁵

With the addition of Pt, CO_3^{2-} was found as one oxidation product besides formate, whereas in Pt-free samples, the oxidation proceeded to formate only.⁵²⁵ The amount of CO_3^{2-} formed increased with the Pt content.⁵²⁵

Interestingly, the alloy with the lowest Pt content resulted in the highest activity. However, this activity could be substantially increased to 920 μ A cm⁻² by a heat treatment of the NPG sample in O₂-atmosphere. This causes the oxophilic Pt to segregate to the surface (Figure 31A). Although the current density increased dramatically with surface segregation of Pt, the sample with the overall smallest amount of Pt at the surface (Figure 31A, as determined by hydrogen UPD) gave the largest current density (Figure 31B, related to the true surface area). This can be seen as an indication that it is not the Pt content alone, but the synergy of Pt well dispersed in an Au matrix, that gives rise to the high MOR activity.⁵²⁵

The role of Pt in those NPG samples is also evident from the MOR electroxidation in 0.5 mol L⁻¹ HClO₄. In this electrolyte, NPG obtained by dealloying Au–Ag shows almost no activity before the onset of surface oxidation, whereas, NPG obtained from Au–Ag–Pt alloys provided a current density of about 0.1 mA cm⁻² at 0.23 V (HglHg₂SO₄|K₂SO₄) or 0.91 V (RHE). In contrast to alkaline solution, the sample with the highest surface concentration of Pt provided the highest current density. Surface segregation shifted the onset of the MOR peak in negative direction but slightly decreased the peak current density. However, the effect was much lower than observed in alkaline solution. This can be seen as a hint toward the passive role of Au in MOR in acidic solution, whereas it exhibits some electrocatalytic activity in alkaline media in the sense of stabilizing adsorbed intermediates,

Special NPG architectures have been synthesized.⁵²⁶ They allowed measurement of the binding strength of CO to Pt embedded in an NPG structure vs Pt metal. The results highlighted that Pt-modified NPG is not only advantageous because of the high degree of dispersion of the more valuable Pt component, but also because of the electronic interaction of Pt and Au that favorably modulates the binding strength of intermediates of MOR.⁵²⁶ Overall, NPG as a matrix or support for a Pt-based or Pd-based electrocatalyst seems to come closest to technical requirements compared to other NPG materials that are free of Pt-group metals (Table 1). A more extensive list with more details is provided in the SI, Table S5.

Next to Pt, Pd has received attention as the third, catalytically more active metal. For instance, Pd-containing electrocatalysts are broadly explored for direct ethanol fuel cells (Table 1 of ref 529). Impressive performance was obtained by dealloying Au–Pd–Ni alloys, in which the resulting ratio between Au and Pd could be easily tuned by the ratio of the two noble elements in the precursor alloy.⁵³⁰ The catalytic activity for ethanol oxidation to acetate increased with the final Pd content and reached up to 6 mA cm⁻² for 75% Pd in the nanoporous metal. The paper does not specify the method of area determination and most likely the current was normalized to the geometric surface area. In this system, the morphology is not very much affected by the Pd–Au ratio and thus facilitates studies of the intrinsic catalytic properties

of the metals within an alloy without superimposed effects of morphology.⁵³⁰ Nanoporous Au–Pd materials have also been obtained by sputtering an Au–Pd–Mg film and dissolving Mg.⁵²⁹ Currents of up to 250 mA cm⁻² (geometric area) were reported.⁵²⁹

Au–Pd alloys have also been produced on anodically roughened stainless steel by first depositing Cu and then performing a galvanic replacement reaction with Au, or Pd, or Au and Pd. The electrodes were used for the oxidation of glycerol.⁵³¹ The best performance with current densities of up to 140 mA cm⁻² were reported for the electrodes containing Au and Pd compared to the electrodes that contained only one of the two metals.⁵³¹

6.2.4. Oxidation of Aldehydes and Carbonic Acids. Formic acid is another C1 building block which can be of interest for oxidation in a direct formic acid fuel cells or for a complete oxidation of methanol to $\text{CO}_2/\text{CO}_3^{2-}$. At Au, this reaction is slow and requires other elements to proceed with appreciable rate. Ag in NPG can enable this reaction at relatively high potentials in alkaline solution.⁵⁰⁵ However, technically interesting current densities are only approached with Pd and Pt as minor element.⁵³² Different procedures have been used to arrive at nanoporous Au–Pt or Au–Pd materials. Nanoporous alloy nanoparticles were formed starting from Ag nanoparticles and galvanic replacement with Au and Pt followed by a dealloying step to remove more Ag.^{256,532,533} Two reaction pathways were observed dominating at different potentials in 0.5 mol L⁻¹ H₂SO₄.

$$0.3 \text{ V (AglAgCl): HCOOH} \rightarrow \text{CO}_2 + 2\text{H}^+ + 2\text{e}^- \qquad (22)$$

$$0.7 \text{ V (AglAgCl): HCOOH}$$

$$\rightarrow \text{CO} + \text{H}_2\text{O} \rightarrow \text{CO}_2 + 2\text{H}^+ + 2\text{e}^- \qquad (23)$$

Similarly, active materials have been obtained by bubbleassisted galvanic codeposition of Au and Pt.²⁵⁶ Pt thin layers have also been formed by decorating NPG obtained from dealloying by Pt and a subsequent UPD of Bi.⁵³³ Formic acid oxidation in 0.5 mol L⁻¹ H₂SO₄ at this material reached 100 A mg^{-1}_{Pt} at +0.6 V (RHE).⁵³³

Au–Pd systems have been obtained by modifying NPG from dealloying by reducing Pd in the inner surface of NPG using hydrazine as chemical reductant.⁵³⁴ Current densities of 11 mA cm⁻² at -0.3 V (with respect to geometric area) were reported.⁵³⁴

6.2.5. CO₂ Reduction Reaction. Electrochemical carbon dioxide reduction reaction (CO₂RR) at planar Au electrodes yields CO but occurs concurrently with the hydrogen evolution reaction (HER). This is mainly a result of the low adsorption energy of CO on Au.⁵³⁵ The preference for CO₂RR can in principle be increased by a design of the surface structure of the electrocatalyst or by manipulating the accessibility of the electrode for the reactants CO₂ and H⁺. For relatively small nanoparticles and nanowires (2-8 nm diameter), DFT calculation predicted different CO₂RR and HER activities that indeed correlated with experimental data for those nanostructures and were related to different ratios of low coordinated Au surface atoms.⁵³⁶ It was also shown that grain boundaries increase the specific activity for CO₂RR on planar Au electrodes⁵³⁷ and in massive nanoparticles on carbon nanotubes.⁵³⁸ Residual copper from dealloying Au₂₀Cu₈₀ resulted in a nanoporous Au₃Cu alloy in which electron

Review

The effect of morphology control, i.e., the control over feature sizes on much larger length scale, on the CO₂RR has been successfully demonstrated by electrodeposited NPG in the shape of inverse opal structure.²⁵¹ The formation of a a pH gradient in the pore space as a typical feature of porous electrodes and the role of grain boundaries have also been confirmed for NPG obtained from deposited AuAg layers of 1 μ m that were subjected to chemical dealloying.³³⁹ A partial current density of CO₂RR of 6 mA cm⁻² with an Faradaic efficiency of 99% was reported in CO₂-saturated 50 mmol L⁻¹ K₂CO₃.

Interestingly, the efficiency of NPG obtained by dealloying can be tuned by the dealloying procedure itself.⁴⁰ Potentiostatic dealloying at high positive potentials leads to an oxide film that reduces surface mobility of Au and thus causes smaller ligaments as compared to NPG obtained from the same alloy by free corrosion. If NPG obtained by these two routes is used for CO₂RR, the material derived from the NPG with a thick oxide layer yielded a much higher Faradaic efficiency at low overpotentials (90% at -0.3 V (RHE)). As both materials were obtained from the same alloy with low density of grain boundaries, this is unlikely the reason for higher activity. The smaller ligaments and pores may enhance the depletion of the reactants for the competing HER. The thick oxide layer may leave some subsurface oxide when reduced only partially. This in turn may cause dislocations at the surface that may be a further reason for the observed enhanced CO₂RR reduction activities.⁴⁰ The authors also noted an analogy to the role of intermediate thick oxide layers in the gas phase oxidation of alcohols.¹⁸ With the alkalization of the solution close to the interface, not only Cu but also low amounts of other reduction products like formate and methanol were detected by recording them with a microelectrode positioned above the NPG electrode and by NMR.⁵⁰³ The product distribution could be varied by the roughness of the NPG electrodes that influenced the pH within the porous material.⁵⁰³

For technical application, the mass transport of the reactants must be improved beyond the limits that are possible with bulk NPG. Hierarchically structured electrodes obtained by templated electrodeposition of an AuAg alloy into a photoresist template followed by removal of the template and dealloying yielded a material with high surface area due to the nanometer-sized pores in the dealloyed material and fast mass transport into the depth of the electrode via the micrometersized channels templated by the photoresist.³⁸ It can also be potentially attractive to produce CO with a less than maximum Faradaic efficiency. The resulting CO-H₂ mixture can be used directly as syngas, provided that the exact ratio CO:H₂ can be adjusted. This has been demonstrated with NPG obtained by pulsed laser deposition.⁵⁴⁰ Ionic liquids and ionic liquid-water mixtures have also been used to enrich CO₂ and suppress HER.⁴⁴ Due to the high viscosity of ionic liquids, all mass transport processes are generally slowed down strongly. A combination of engineering pore sizes and variation of the ionic liquid-water ratio facilitated current densities of about 5 mA cm^{-2} at -0.5 V (RHE) with Faradaic efficiencies of approximately 90% for CO.44 The NPG material with strictly defined pore sizes was obtained by replicating twice an anodic aluminum oxide template, first by poly(methyl methacrylate) (PMMA) and dissolution of aluminum oxide, followed Au

electrodeposition into the PMMA template. The final electrode was obtained after dissolution of PMMA.⁴⁴

6.3. Surface Modification for Improvement of Electrode Properties

Section 6.2 has already highlighted the importance of dealloying ternary alloys to introduce a third element for enhancing catalytic activity or to stabilize the morphology. Pt can also be electrodeposited using the limited amount of precursor contained in the solution-filled pores after transfer of the NPG electrode to a precursor-free solution.⁵⁴¹ It has been applied to fabricate Pt nanoparticles on NPG as support.⁵¹⁸ Pd has been electrodeposited by electroless deposition.⁵³ Probably, more defined surface architectures are obtained by UPD of Cu to prepare the surface for a subsequent galvanic replacement reaction (e.g., Cu UPD, exchange for Pt⁵⁰⁷ or Pd^{531,542}). Another variation has been termed atomic layer electrodeposition. Pt is electrodeposited on NPG, and the concomitant UPD process of hydrogen on Pt was said to prevent the deposition of more than one Pt layer.⁵⁴³ Removing the capping hydrogen layer by application of potentials positive of the hydrogen UPD zone opens the surface for deposition of another Pt layer during the subsequent deposition pulse. The resulting Pt layers were used for pH sensing.⁵⁴

Some modification with other material classes shall be reviewed below that open doors for hybrid materials as catalysts. The combination of NPG with thin layers of transition metal dichalcogenides as hydrogen evolution catalysts is attractive to combine the advantages of both materials. Dichalcogenides such as MoS2 can be electrodeposited on NPG by potential cycling from solution containing precursors (such as MOS_4^{2-} sometimes formed upon mixing the deposition solution).⁵⁴⁴ This procedure allows the variation of the thickness of the material on the surface of the ligaments.⁵⁴⁴ In a recent study, the plasmonic excitation of NPG was used to initiate and guide the growth of a thin film of MoS_x on the ligament surface.⁵⁴⁵ The resulting materials showed a less negative onset potential for HER and a smaller Tafel slope than MoS₂ alone. The advantages of the hybrid material were attributed to a large number of sulfur sites and the better conductivity within the catalyst due to the 3D NPG structure.⁵⁴⁵ Electrodeposition on NPG has also been used to coat the ligaments of NPG by a composite of WS₂ embedded in poly(3,4-ethylenedioxythiophene) (PEDOT) by electropolymerization of 3,4-ethylenedioxythiophene in an aged solution of $(NH_4)_2WS_4$, polyethylene glycol, and LiClO₄ in water.⁵⁴⁶ The material was tested for HER and showed a low onset potential and a low Tafel slope. Here, NPG is used as a 3D support material of high conductivity. The details of the preparation of NPG are not crucial, and NPG from different preparation routes has been used. Even the replacement by a less costly 3D metal is conceivable for this application.⁵⁴⁶ However, the relative ease of the wet chemical procedures and the high control over a bulk 3D morphology for NPG is certainly the main advantage in these developments.

NPG is also an interesting support material for coordination network compounds such as metal–organic frameworks and metal hexacyanometallates. Many compounds of these material classes show limited ionic or electronic conductivity that severely hamper the exploitation of their interesting functional properties for electrocatalysis, electrochromism, or photoelectrochemistry.^{547,548} The potential advantages of a combination of the metal–organic framework compound ZIF-8 (zeolitic imidazolate framework of ordinal number 8) has recently been highlighted.⁵⁴⁹ Nanoporous gold nanoparticles (ca. 340 nm diameter with ligaments of ca. 37 nm diameter) were encapsulated by a 200 nm layer of ZIF-8. An electrode prepared from this hybrid material showed good activity and Faradaic efficiency of 44% for nitrogen reduction to ammonia in neutral aqueous solution at -0.6 V (RHE).⁵⁴⁹ The superior performance compared to other materials was attributed to the intrinsic catalytic activity of NPG (with about 3% of Ag). Usually, this electrocatalytic ability of NPG cannot be harvested because of the competing HER that leads to very low current efficiencies for NH₃ formation. In the hybrid material HER was suppressed by the ZIF-8 encapsulation.⁵⁴⁹

Iron hexacyanoferrate (Prussian Blue), a known catalyst for selective H_2O_2 reduction in glucose biosensors, has been electrodeposited on NPG obtained by anodizing and reduction of pure gold. The large surface area increased the linear working range of the resulting sensor.⁵⁵⁰ In this application, NPG has the role of a highly conductive and inert support material. The high conductivity of the NPG support was also used to construct an OER catalyst working at high positive potential in alkaline solution. It consists of CoMoN_x nanosheets electrodeposited on NPG wires.⁵⁵¹

6.4. Electrochemical Sensors based on Nanoporous Gold

Electrocatalytic sensors are a main field of application for NPG and NPG-derived electrode materials. While electrocatalytic activity is certainly required, catalytic activities or current densities do not need to be extremely high because the substance under study (the analyte) is usually available in μ mol L^{-1} to mmol L^{-1} concentration only, whereas 1 mol L^{-1} is often used in MOR for energy conversion. More important for chemical sensors is the selectivity, that is, the ability to distinguish a signal coming from the analyte vs signals caused by other compounds contained in the sample (the matrix). An important aspect for sensors is the possibility to integrate the NPG electrode in sensors of miniaturized cells. Here, as pointed out in section 2.6, NPG is distinguished by the fact that its preparation by thin film deposition and chemical or electrochemical corrosion affords precise lateral structuring in lithographic processes.^{219,220} This promises a seamless integration into established microdevice production lines. 3D printing approaches provide further opportunities for structuring NPG.¹⁵³ This suggests opportunities for particularly purposeful structuring in technological applications. Yet, porous gold obtained by other methods (section 2.9) than dealloying has so far found more attention for these applications. Another important aspect is biocompatibility, especially if the sensor surface shall be used in direct contact with biological tissue. This situation occurs not only for sensors but also for electrodes in biomedical devices. Electrocatalytic sensors and biomedical devices made from NPG have recently been reviewed in detail.^{28,66,67,552}

6.4.1. Inorganic lons and Compounds. NPG film electrodes have been applied for the coulometric reduction of U(VI) as a possible replacement of the traditionally used Hg electrodes.²³⁰ The procedure takes advantage of the high surface area and the relatively large overpotential for the competing HER.

Hydrogen peroxide H_2O_2 is an interesting analyte, as it is released as part of inflammatory response in biological organisms. It is one of the reactive oxygen species that are of concern in biology and in energy conversion devices, and thus, there is a need for analyzing its concentration. It is also a redox-active product of the enzymatic glucose oxidation according to eq 24.

glucose +
$$O_2 \xrightarrow{GO_x}$$
 gluconolactone + H_2O_2 (24)

Detection of $\rm H_2O_2$ is exploited in some glucose sensors, although most current commercial devices exploit mediated glucose oxidation. 553

 H_2O_2 can be detected in neutral solution such as phosphate buffer saline (PBS) by *reduction* at -0.4 V (vs SCE).⁵⁵⁴ This potential is less negative than the potential required for H_2O_2 reduction at planar Au electrodes. The use of a reduction (instead of oxidation) avoids interferences with the electrooxidation of alcohols, glucose, ascorbic acid,⁵⁵⁴ dopamine, and uric acid²²⁷ that are common interfering compound in amperometric biosensors. The samples have to be deaerated to avoid interference with the reduction of dissolved O_2 . The performance has been achieved with dealloyed material⁵⁵⁴ but also with porous Au obtained by electrochemical roughening that does not contain other elements that modulate catalytic activity.²²⁷

Besides detection of H_2O_2 at the bare nanoporous electrode, surface modifications are used in order to enhance the selectivity or to increase the sensitivity. For instance, iron hexacyanometallate was deposited on NPG as an catalyst that enabled H_2O_2 reduction at much less negative potentials (-0.05 V (AglAgCl)) than at bare NPG.⁵⁵⁰ The inner surface of NPG electrodes has also been used for modification with a layer of hemoglobin, a redox protein able of direct electron transfer with roughened gold surface and the ability to catalyze H_2O_2 reduction at -0.3 V (AglAgCl).²⁶⁵ Of course, the surface of NPG can also be modified with a monolayer of Pt that is more active for H_2O_2 reduction than Au and provided detection limits below 1 μ mol L^{-1.555} Such an electrode was used to detect H_2O_2 released by stimulated PC12 cells in a cell culture medium.⁵⁵⁵

 $\rm H_2S$ is a toxic gas; however, at trace levels in the parts-perbillion (ppb) concentration range, it is a biomarker formed in the walls of arteries and is informative about peripheral artery disease and small vessel disease. The detection of trace $\rm H_2S$ levels emitted through the skin has been achieved with a gasdiffusion electrode at which NPG is in direct contact with skin and linked by a Nafion polymer electrolyte to the reference and counter electrode. The efficient capture of almost all molecules transdermally released enabled the very low limits of detection required for this application. ⁵⁵⁶

Electrochemical detection on nitrite NO_2^- was accomplished by anodic oxidation at nanoporous gold that was obtained by bubble-templated electrodeposition. The morphology could be tuned by the deposition time and the potential applied for the reduction of the gold precursor and the concomitant formation of H₂ bubbles.²⁰

An interesting application is the use of NPG in all-liquid fuel cells with hydrazine $N_2H_4^{557}$ or $NaBH_4^{558}$ as fuels and H_2O_2 as oxidant. In case of $NaBH_4$ as fuel, Pt-coated NPG showed a higher open circuit voltage in agreement with the higher catalytic activity for the BH_4^- oxidation at Pt.⁵⁵⁸ However, the cell operation under load was much better with unmodified NPG. The authors related this to the tendency of Pt to catalyze the disproportionation of H_2O_2 that does not inject electrons into the external circuit but may lead to gas bubble formation

that hinders mass and charge transport in the electrolyte. The lower activity of NPG compared to Pt-coated NPG for the disproportionation was thus an advantage for this application. 558

6.4.2. Nonenzymatic Glucose Sensing. Besides detection of blood glucose as part of managing diabetes, for which enzymatic glucose sensors are the dominating approach,⁵⁵³ there is a need for the detection of glucose and other carbohydrates in other applications, for example, in electrochemical detectors after a separation by anion-exchange high performance liquid chromatography^{559'} or in urine.²⁵³ Nonenzymatic detection of glucose in neutral solution is possible at planar gold electrodes when using pulsed amperometric detection to avoid blocking of the electrode by intermediates or matrix components.⁵⁶⁰ Similar to the oxidation of alcohols, the high number of undercoordinated surface atoms and the high surface area can improve the performance. Apart from sensing, there is also a current interest in the conversion of carbohydrate and other natural compounds as sustainable feedstock for the synthesis of versatile building blocks for a wide array of chemicals and materials.^{561–563}

In this context, glucose oxidation has frequently been studied at NPG as a model reaction.^{217,232,234,266} A key challenge for the use of NPG electrodes as sensors is the presence of Cl⁻ in most sample matrices.⁵⁶⁴⁻⁵⁶⁷ Cl⁻ specifically adsorbs to Au surfaces and may inhibit electrocatalytic reactions and thus change the sensitivity for glucose and interfering compounds such as ascorbic acid. The porous Au electrodes can also be prepared by electrochemical treatment other than dealloying (section 2.9) and work without the presence of other elements.^{253,254,566} The effect of residual Ag in NPG obtained by pulsed dealloying has been studied in acidic,²¹⁷ neutral, and alkaline solutions.⁵⁶⁸ Coating with a cation exchange membrane such as Nafion, that is impermeable for anions but permeable for neutral compounds and cations, was found to be an effective mean for enabling glucose detection in this matrices.^{567,569} Nanoporous copper obtained from dealloying Cu30Mn70 was coated with a thin gold film and provided similar performance as NPG with a much lower Au content.²⁶⁶

NPG electrodes have been further modified in order to enhance the performance. For instance, films of cobalt oxides were deposited under hydrothermal conditions in the presence of surfactants to coat the ligament surface of NPG with a film of cobalt oxide.⁵⁷⁰ In this case, the catalytic reaction proceeds at the metal oxides and NPG provides a conductive support with high surface area to the metal oxides, which have a low intrinsic conductivity and thus cannot be used as porous electrode material by themselves. In turn, the metal oxide coating inhibits the coarsening of the NPG ligaments. Coating with Pt showed a rather minor effect⁵⁴¹ compared to the effect Pt coating has on MOR (section 6.2.3). An optimum coverage was identified.⁵⁴¹ Pt-containing material was also obtained by dealloying Pt₄Au₁₆Cu₈₀.⁵⁶⁹ The obtained material with a Nafion coating was active for glucose oxidation and H₂O₂ oxidation in PBS.⁵⁶⁹

Occasionally, the use of NPG for glucose biofuel cells has been envisioned.²⁶¹ Despite high current densities that have been obtained, the high potential of 1.2 V (vs RHE) leaves little room for the development of a usable voltage output when combined with an ORR as the cathodic reaction, which yields usable current densities at around 0.9 V (vs RHE).

6.4.3. Detection of Other Organic Analytes. The detection of dopamine, a neurotransmitter, and ascorbic acid have been studied on porous Au electrodes obtained by dynamic hydrogen bubble templating, i.e., a material that does not contain a LNE.³⁶ The two analytes responded differently to structural features in neutral solution. Dopamine adsorbs to Au and benefits from the larger surface area which allows preconcentration, whereas the irreversible ascorbic acid oxidation requires surface defects for enhanced oxidation currents.³⁶ The detection of ascorbic acid has been investigated at NPG and could be enhanced by coating the ligaments with small amounts of Pd.⁵⁴² Reactions of biomolecules typically involve heterogeneous electron transfer steps coupled to homogeneous proton transfer. The nanopore environment enhances the probability that reactants encounter each other during the homogeneous reaction steps, independent of the chemical nature of the electrode material.⁵⁰⁶ This confinement mechanism may contribute to the enhanced oxidation current and the observed lower overpotentials in dopamine oxidation and ascorbic acid oxidation.³⁶ Exploiting the reduction of the oxidized dopamine by an a excess of ascorbic acid could also be used to boost the sensitivity for dopamine without too large signals from ascorbic acid itself.⁵⁷¹ Adaption of the electrochemical detection for the detemination of ascorbic acid in acidic environments⁵⁷² and in the presence of a strongly absorbing matrix with uric acid as an interferent⁵⁷³ were demonstrated as well.

The confinement effect was explicitly studied for 1,4benzoquinone electrochemistry at nanoporous Pt and Au electrodes, in which the nanoporous materials showed an enhanced reversibility of the electrode reactions.⁵⁷⁴ A much faster reduction kinetics was also reported for the pharmaceutical drug metronidazole, 1-(2-hydroxyethyl)-2-methyl-5nitroimidazole, on NPG electrodes produced by pulsing the electrode potential applied to a planar Au electrode.⁵⁷⁵ *p*-Nitrophenol, a compound that is of concern when released to the environment, has been detected at porous Au electrodes obtained by an anodization process in oxalic acid.²²⁸

Epinephrine, a redox-active neurotransmitter, was determined at porous Au with good detection limits.⁵⁷⁶ The material was obtained by electrodeposition of Au inside the pores of anodic aluminum oxide templates.⁵⁷⁶ Similar results were also obtained by dealloyed NPG by the same authors,⁵⁷⁷ indicating that, in this case, it is indeed the confinement effect in the pores that accelerates the overall reaction rather than the surface chemistry of the ligaments.

6.4.4. Nanoporous Gold as Support for Biomolecules. Au surfaces in general lend themselves to surface functionalization by thiol chemistry under mild conditions by which a large variety of structural motifs for further coupling chemistry can be anchored to the metal surface.⁵⁷⁸ The transfer of those concepts has been demonstrated early for template-deposited mesoporous Au, in which a monolayer of the redox mediator (4-carboxy-2,5,7-trinitro-9-fluorenylidene)malononitrile (TNF) was linked to a layer of the cofactor nicotinamide adenine dinucleotide (NAD⁺) and the enzyme glucose dehydrogenase to realize an artificial electron transfer chain.²⁵² The enzyme horse radish peroxidase (HRP) was immobilized in NPG and used to realize a direct electron transfer, which was claimed to be improved by a higher packing of HRP in the porous material.⁵⁷⁹ Direct electron transfer was also reported for the enzyme fructose dehydrogenase (FDH).⁵⁸⁰ In a detailed study, the effect of different thiol

linker molecules was studied, and a comparison between planar Au(111) electrodes and NPG was conducted, which showed similar trends with respect to catalytic efficiency of FDH and the linker chemistry. Enzyme electrodes based on NPG showed higher currents and prolonged stability, which are attractive features for application in biosensors and biofuel cells.⁵⁸⁰

Immobilization of glucose oxidase (GOx), which catalyzes reaction 24, and H_2O_2 detection at the NPG electrode were used for measuring of the glucose concentrations.⁵⁸¹ Apart from demonstrating the surface functionalization scheme, this model sensor cannot compete with commercial glucose biosensing devices especially with respect to sustainability and price.⁵⁸¹

Other biosensing approaches use a support electrode to immobilize a biorecognition element, for instance, an antibody or a single-strand oligonucleotide, to capture the analyte, which in those cases is the antigen or a specific nucleotide sequence, respectively. Along this line, NPG was used for a DNA assay with detection by electrogenerated chemiluminescence (ECL).⁵⁸² The analyte oligonucleotide was first captured by an oligonucleotide strand immobilized on NPG. The CdTe quantum dot as ECL emitter was also modified by singlestrand oligonucleotides and could recognize the other end of the analyte DNA sequence. The use of NPG enhanced the available surface area for the determination.⁵⁸² The advantage of a larger surface area to hold more probe molecules is offset for thicker layers or narrow pores by transport hindrance of the analyte oligonucleotides in the pore space. Consequently, an optimum for the thickness and pore morphology was found for assays with a redox-active label that undergoes electron transfer with NPG.⁵⁸³ Aptamers are artificially selected antibodies consisting of self-hybridized single DNA strands that are able to recognize non-DNA analytes. Such aptamers against bisphenol A have been immobilized on the ligament surface of NPG.⁵⁸⁴ Bisphenol A is a monomer used in the production of polymers that are used in food packaging. Its release in trace quantities raises health concerns due to its hormone-like properties, and thus, it must be monitored. The immobilized aptamers enriched bisphenol from the sample during an exposure time of 30 min. Afterward, bisphenol A was electrooxidized at NPG, which showed a slightly lower overpotential for this electrooxidation than planar gold or glassy carbon.⁵⁸⁴ The sensor benefited from the high surface area to accommodate more aptamers and the electrocatalytic activity of the ligament surfaces in NPG.

6.4.5. Electrodes in Biomedical Devices. The noble character of gold, the high surface area, and thus high capacity and the absence of particulate matter that may have harmful biological effects are advantageous properties for the use of NPG as contact electrodes in biomedical advices. Macro-/ mesoporous gold has been used as electrodes for extracellular recording in which the biological cells couple capacitively to the recording electrode.⁶⁸ Another example is the use of two flexible NPG thin film electrodes on the inside of a contact lens for an integrated enzymatic biofuel cell.⁵⁸⁵ One electrode, acting as anode, was modified by lactate oxidase to utilize lactate from the tear liquid as fuel. The cathode was modified with bilirubine oxidase that is able to reduce O₂ smoothly to H₂O. In this application the inertness and high surface area of NPG was beneficial.

6.5. Nanoporous Gold in Electrochemical Energy Conversion Devices

The use of NPG in electrochemical energy conversion devices has received a lot of attention due to the urgency to develop drastically improved devices for the decarbonization of the economy and their widespread deployment. Recently, the area has been reviewed.^{26,70,486,487} Electrochemical energy conversion plays a central role in almost all sustainable energy supply scenarios. However, the range of conductive materials that withstand high positive or high negative potentials in acidic solution is small. Nanoporous gold for energy applications and the required widespread use of such devices will meet supply issues, especially for the most active metals such as Ir, Ru, and Pt. Although Au is a noble metal, the supply is much larger than that of the aforementioned metals, and it has been investigated as an alternative because it meets the chemical stability requirements.

6.5.1. Nanoporous Gold in Fuel Cells. The oxygen electrode is the more critical electrode in a fuel cell and water electrolyzer. Unfortunately, the obtained catalytic activity for NPG falls short of technical requirements in those devices unless the surface is modified with other precious metals (section 6.2). As cathode for the ORR, Au has the disadvantage that it favors the $2e^{-1}$ reduction to H_2O_2 .⁵¹⁶ Although the second 2e⁻ reduction to H₂O is promoted in NPG (compared to planar Au) due to the confinement effect, it does not compete with Pt that promotes almost exclusively the 4e⁻ reduction. However, Pt nanoparticles supported on carbon tend to oxidize the carbon support at high positive potentials during start-stop cycles. Here, NPG as support offers improved stability⁵⁰⁷ compared to carbon supports although being associated with substantially higher costs. The use for fuel cell anodes is most advanced for the oxidation of small organic molecules at NPG decorated with a low amount of Pt brought to the material as a component of the original alloy^{115,225,525} or by postdealloying modification.^{19,507,533}

6.5.2. Nanoporous Gold in Batteries. NPG has found application in research batteries that have not yet entered the commercial market. The use of new negative electrode materials with very high Li storage capability (e.g., lithiated $Li_{3.75}$ Si 3500 mA h g⁻¹, $Li_{4.4}$ Ge 1600 mA h g⁻¹, $Li_{4.4}$ Sn 990 mA h g^{-1} vs LiC₆ 373 mA h g^{-1}) is hampered by their poor stability upon cycling, which is routed in their large volume expansion upon lithiation (over 200%) that causes enormous stress and ultimately pulverzation of the materials when applied in macroscopic shape.^{586–588} However, successful applications of those materials have been demonstrated by building three-dimensional nanostructured electrodes, allowing for accommodation of the volume change.586 The accommodation of the volume changes of Sn upon lithiation/ delithation were demonstrated with a NPG electrode as a three-dimensional current collector onto which Sn clusters were deposited as active material by currentless deposition.⁵⁸⁸ Further advantages were demonstrated for Ge as active electrode material coated onto NPG by a gas phase process as a continuous conformal film.⁵⁸⁷ This electrode showed an excellent rate capability (47.5% capacity at 60 C compared to 1 C) and capacity retention of 90% after 100 cycles. These achievements were due to the use of an amorphous Ge layer, the exploitation of the interconnecting networks of a phase with high electronic conductivity (Au), and another phase with high ionic conductivity in the electrolyte-filled pores, short iontransport distances within the thin layer of the active material

on the surface of the ligaments, and the accommodation of the volume change of the lithiated Sn in the pore space.⁵⁸⁷ In a similar direction, NPG has been used as a support and current collector for transition metal oxides (e.g., TiO₂) that can intercalate Li⁺ at negative potentials but have low electronic conductivities. Conventionally, they are used with a conductive carbon agent processed in complicated formulation steps that must balance a high content of the active material, good conductivity, and mechanical stability of the composite electrode. In the case of NPG, films of TiO₂ have been coated around the ligaments of NPG so that access of electrons to all parts of the active material is easily ensured without binder and composite formation.⁵⁸⁹ Electrodeposition of MnO₂ with low electronic conductivity on NPG enabled the use of this material as intercalation electrode without a binder.⁵⁹⁰

Most notable is the use as positive electrodes in rechargeable $\text{Li}-\text{O}_2^{236,591,592}$ and $\text{Na}-\text{O}_2$ cells⁵⁹³ in organic electrolyte solution. In this application, O_2 is reduced to Li_2O_2 , thus the promotion of the 4e⁻ reduction of O_2 is not desired. Instead, high positive potentials during charging must be tolerated. NPG offers excellent ways of designing separately a high surface area for the ORR and at the same time transport pathways for O_2 by using a bimodal pore size distribution²⁰⁸ that improved the performance over NPG with a monomodal pore size distribution.

6.5.3. Nanoporous Gold in Supercapacitors. Supercapacitors store energy in the electric double layer and/or in a thin redox-active film around the actual electrode material. The key feature of this storage device is the fast uptake and release of the charge (high power) and the very high stability over several 10 000 cycles. Usually, the capacity is inferior to batteries. The energy stored in the electric double-layer scales with the surface area and the potential that can be applied to it without electrolyzing the electrode material or the electrolyte. Thus, it is very important to avoid electrocatalytically active electrode materials that would be able to decompose the solvent or other electrolyte components. In this respect, nonaqueous electrolytes can be of high interest.⁵⁹⁴ However, often their ionic transport properties are inferior compared to concentrated aqueous electrolyte solutions or gels. The energy stored in a thin redox-active film depends on the density of the redox centers and the total amount of the redox-active material. Making redox-active layers thicker is often limited by the low electronic conductivity of the redox-active material itself. A low conductivity would prevent a fast charging and discharging under full utilization of the active material and thus give away one of the central features of a supercapacitor. Here the high electronic conductivity of NPG in combination with a tunable pore size for ionic transport offers the possibility to realize supercapacitors using very thin films of redox-active material such as MnO2⁵⁹⁵ and to exceed the performance characteristics that can be achieved with the more conventional porous carbons as conductive support.⁵⁹⁶ This is especially true when the metrics are made with respect to volume instead of mass.

The high surface area of NPG together with the inertness of this noble metal is an attractive feature for supercapacitors. Accordingly, the modification of the ligament surface with redox-active layers has been explored in order to enhance the performance beyond a pure double-layer capacitance (similar to their common use in combination with porous carbons).⁵⁹⁷ Successful modification with redox-active transition metal oxides/hydroxides include MnO₂, ⁵⁹⁴, ⁵⁹⁵ RuO₂, ⁵⁹⁸ CuO, ⁵⁹⁹ and

 $Ni(OH)_2$.⁶⁰⁰ Intrinsically conducting polymer films, such as poly(pyrrole)^{110,601} and poly(aniline),^{602,603} have also been tested. The combination of different redox-active materials has been demonstrated with MnO_2 and poly(pyrrole).⁶⁰⁴ This even applies to combination of NPG with further electro-deposited nanomaterials such as reduced graphene oxide that was assembled with poly(pyrrole) to a planar device with outstanding characteristics.⁶⁰⁵ As exemplified in ref 605, electrochemical procedures are often used in various ways for the preparation of the redox-active films offering high control over the amount of deposited material. This is needed to adjust the capacity and the transport characteristics under operating conditions. A recent review provides a detailed comparison of the different approaches.⁷⁰

Overall, the use of NPG in energy conversion devices has seldom advanced to experiments on the device level despite the promising results that have been collected in half-cell experiments and miniaturized laboratory cells. Beside the high cost of Au, this is also caused by procedural problems to integrate NPG in larger devices of layered architecture. Therefore, the prospective for real-world application is seen in niche applications such as electroorganic synthesis of highvalue chemicals, high-cost medical devices, and sensors that can exploit the high surface area and use planar integration.^{599,604,605} Those areas may be more receptible to the material cost of NPG than commodity application such as conventional batteries, fuel cells, or blood glucose sensing.

7. CATALYSIS BY NANOPOROUS GOLD IN ORGANIC SYNTHESIS

Many studies have been conducted in the gas phase^{39,72,606} because of faster mass transport, lower variety of products, and avoidance of homogeneous background reaction. 607,469 However, working in the liquid, i.e., condensed phase is a crucial requirement for a widespread use of NPG in organic chemical synthesis.⁶⁰⁸ Considering the maturity of the field in the gas phase, there are relatively few reports on reactions in solution. One can only speculate as to the reasons for this, but there seems to be very little predictability of reactivity and selectivity and relatively little mechanistic understanding. On the work that has been done, several reviews have appeared that describe the synthetic value of NPG in the liquid phase for substrates beyond MeOH that was treated in this paper in section 5. In 2013^{609} and 2014,^{39,610} the first reviews that dealt with NPG in organic synthesis listed already the most important transformations, to which few fundamental additions have been reported subsequently.

Perhaps this illustrates the difficulties of this area. The review paper by Jin et al.⁶¹¹ from 2019 provides a summary of synthetically useful oxidation and reduction reactions with the perspective on general catalysis by nanoporous metals, which builds on the foundations of an earlier review on the same topic.⁶¹² A further review deals with transformations that form carbon–heteroatom linkages that are either catalyzed by Au nanoparticles or NPG, in which the different reaction outcomes illustrate the influence of the support on the reaction.⁶¹³ The latest review in the field focuses on new insights into structure–property relationships in the context of synthetically useful reactions catalyzed by NPG.⁴¹

The following section will attempt to describe the reasons for current limitations of NPG in organic synthesis and highlight more recent work that has not been covered in previous reviews. We will exclude reports in which the NPG was used as a high-surface area support for the immobilization of active materials such as photocatalysts^{614,615} or enzymes.⁶⁹

The reactions with NPG are all either oxidation reactions or reduction reactions, although some may be classed as comproportionation reactions. Oftentimes, the reaction does not halt at the direct oxidation product, but further products such as condensation products may be observed as exemplified in detail in section 5 for MeOH oxidation in liquid phase.

7.1. Oxidation Reactions of Organic Molecules by Nanoporous Gold

Oxidation reactions with oxygen are environmentally benign, cheap, and do not lead to residual reagent. As of the time of this review, there are only a few classes of starting materials that have been used for this transformation (Scheme 2).

The best researched liquid phase oxidation with NPG is the oxidation of methanol to give methyl formate (Scheme 2A). Other primary alcohols react in a similar manner, giving either the ester or the corresponding aldehyde in mixtures.⁶⁰ This reaction is therefore not competitive in preparative organic chemistry, as many, much more selective, and cheaper options exist, many of them based on supported Au nanoparticles.⁶¹⁶ Secondary alcohols react without such complications to the corresponding ketones with oxygen as the oxidant (Scheme 2B).⁴⁷⁹ The solvent in these reactions was MeOH, which appeared not to influence the yield of the desired alcohol. However, when the primary alcohol benzylalcohol was used under these conditions, methylbenzoate was formed with MeOH. Only when MeOH was excluded and 1,4-dioxane was used, benzaldehyde was the exclusive product (Scheme 2B).⁴⁷⁹

Considering the complex postulated mechanism for the oxidation of primary alcohols (section 5), it was perhaps a logical extension to present additional nucleophiles other than MeOH. Indeed, when an oxidation of MeOH with oxygen is performed with primary or secondary amines, which are much superior nucleophiles, the corresponding amides result in excellent yields (Scheme 2A).⁶¹⁷

Aldehydes can also be oxidized to the carbonic acids: One of the earlier examples in which oxidations have been examined for larger, synthetically relevant molecules was the oxidation of D-glucose to D-gluconic acid in water with oxygen as an oxidant (Scheme 2C).⁶¹ The focus of this work was on the mechanism rather than on the scope: The reaction mechanism was postulated to proceed in three steps: the adsorption of glucose on NPG, the surface oxidation reaction, and the desorption of the product. A detailed molecular reaction mechanism was not provided, but because the reaction proceeded best in slightly alkaline reaction conditions, the desorption step was thought to be rate-determining. Smaller ligament sizes (as low a 6 nm diameter) and slightly elevated temperatures of 50 °C provided shorter reaction times.

The oxidation of nitrogen-containing organic molecules is often difficult to perform in a selective fashion because a variety of products and intermediates may be involved starting from the amine to the highest oxidation state: hydroxylamines, nitrones, oximes, nitroso compounds, and nitro compounds as well as N–N coupled products such as hydrazines, diazocompounds, and azoxycompounds. Therefore, the selective oxidation of hydroxylamines to the nitrones can be regarded as a breakthrough in this regard (Scheme 2D).⁶¹⁸

A more unusual oxidation reaction that is one of the very few applied reactions is the oxidative dealkylation or cyclization of lidocaine, which was performed to synthesize a





drug metabolite (Scheme 2E).⁶¹⁹ This reaction can yield two products, depending entirely on the reaction conditions, but in a selective fashion. It needs pointing out that this is one of the very few reactions, in which a peralcohol, *tert*-butylperoxide, can also be used as the oxidant, which may be more easily administered than O_2 in a synthetic organic chemistry laboratory.

NPG has also been used in oxidative dehydrocoupling reactions between terminal alkynes and silanes (Scheme 2F).⁶²⁰ This is of synthetic relevance because terminal alkynes are C–H acidic, which may interfere with other reactions, and thus, silylgroups serve as protecting groups. The reaction may also become of synthetic relevance as it tolerates hydroxy groups, which in other silylation protocols (with silyl chlorides and bases) typically react first and thus require additional protection steps, although other catalytic alternatives exist.⁶²¹

The oxidation of olefins is also of potential interest, but selectivity is often a problem. Oxidation products may be epoxides, diols, diketones, and further oxidation products that may also involve C–C bond breaking. Thus, it is perhaps unsurprising that this avenue has not yielded much success in oxidative processes with NPG. However, there is one report on the oxidation of cyclohexene, in which different surface treatments were carried out prior to the reaction.⁶²² Selectivity

is low and products are 2-cyclohexen-1-one 2-cyclohexen-1-ol cyclohexene oxide cyclohexenyl hydroperoxide.

7.2. Comproprotionation Reactions

In silanes, the hydrogen atom attached to Si has a slightly hydridic character. It can therefore react relatively easily with a proton, thus forming dihydrogen. Although this reaction may be viewed as an acid-base reaction, if one regards the formal oxidation states of the hydrogen atoms before and after the reaction, it can also be seen as a comproportionation. Because protons are typically easy to provide, silanes often oxidize easily to the corresponding silanols. With NPG as a catalyst, this reaction can be facilitated with water as the formal oxidant, while dihydrogen is being released (Scheme 3A).⁶²³ In a seemingly similar vein, silanes are able to react with alcohols, forming the corresponding silanol ethers (Scheme 3B).⁶²⁴ Even reactions with cyclic ethers are possible, in which the ring is opened; however, with a reduced selectivity (Scheme 3C).⁶²⁴ If one compares this reaction with the oxidative dehydrocoupling described above (Scheme 2F), one might reasonably wonder why in that case,⁶²⁰ oxygen was required whereas here, it is not. The literature does not offer any mechanistic explanation but merely observations: First, the reaction between the alkynes and silanes does not proceed if

Scheme 3. Oxidations with Water, Alcohols, or Ethers

Oxidation of silanes

A Oxidations with water

$$R_{(4-n)}SiH_n \xrightarrow{H_2O, npAu} R_{(4-n)}Si(OH)_n + H_2$$

B Oxidations with alcohols

 $R_{2}^{1}SiH_{2} + R^{2}OH \xrightarrow[toluene, 50 °C]{} R_{2}^{1}SiH(OR^{2}) + H_{2}$

C Oxidations with ethers

 $\underset{(\bigtriangledown)_{n}}{\overset{O}{\bigtriangledown}_{n}} + tBu_{2}SiH_{2} \xrightarrow{npAu} \xrightarrow{H(tBu)_{2}Si}_{O} \xrightarrow{(\frown)_{n}} + H(tBu)_{2}Si_{O} \xrightarrow{(\frown)_{N}}_{O}Si(tBu)_{2}H$

D Oxidation of conjugated aldehydes



there is no oxygen present. Instead, a hydrosilylation reaction takes place, which is reductive (cf., section 7.3). Second, if the catalyst was prepared from a dealloyed AuAl alloy, this reaction was also the dominant one.⁶²⁴ As residual Ag is well-documented to be necessary for oxygen activation, the mechanism for the silylation of alkynes does not appear to follow a simple acid—base reaction mechanism.⁶²⁰ Indeed, the reactions of silanes with alcohols are only formally acid—base reactions.⁶²⁴ Electron spin resonance spectroscopy (EPR) revealed that the silane reacts with the NPG by forming

Scheme 4. Reduction Reactions with Nanoporous Gold





radicals. It was suggested that the abstracted hydrogen atoms recombine on the Au surface to form H_{2j} which was indeed observed.⁶²⁴ Ring opening of cyclic ethers can relatively easily be envisioned following a radical pathway, but the silyl ether formation with the alcohols remains mechanistically unclear. All reactions described in this section are very sensitive to solvents, temperature, and even residual less-noble atoms in NPG⁶²⁴ that mechanistic generalizations should not be made at this stage. Last, silanes can also be employed to perform selective 1,2-reductions of conjugated aldehydes (Scheme 3D).⁶²⁵

7.3. Reduction Reactions by Nanoporous Gold

Reduction reactions (Scheme 4) would be ideal if they could be performed with hydrogen, H_2 . However, H_2 does not dissociate as easily on Au surfaces as it does on other metals, such as Pt or Pd, which renders it ineffective.⁴⁶⁸ Therefore, transfer hydrogenation reagents are often employed, which are typically hydride donors in the combination with water.

One type of such hydride transfer reagent are silanes, which are relatively easily oxidized. For example, alkynes can be reduced exclusively to the Z-alkenes with PhMe₂SiH and water in the presence of amines (Scheme 4A).⁶²⁶ As avoiding overoxidation and Z-selectivity are not easy to achieve, this may become a useful reaction. A Z-selective reduction of alkynes has also been performed with formic acid and an amine, which may be considered "greener" chemistry than the use of silanes (Scheme 4B).⁶²⁷

PhMe₂SiH and water were also used as hydrogen transfer reagents for the reduction of quinolines to 1,2,3,4-tetrahydroquinolines with NPG, which did not proceed at all with hydrogen.⁶²⁸ The proposed reaction mechanism is an indirect hydrogen transfer, in which a hydride is abstracted from the silane, which is simultaneously attacked by water. Deuteration experiments with D₂O showed a preferential incorporation in the 3-position. Based on the data presented, it is possible to



pubs.acs.org/CR

Scheme 5. Reductions under a Hydrogen Atmosphere



Scheme 6. Reduction Reactions Yielding Organometallic Reagents

Reduction reactions with organometallic reagents



propose a subsequent hydration/protonation reaction mechanism, in which hydrogen bonding may play a role in arranging the reagents on the gold surface. Even amides, which are relatively difficult to reduce and are often treated with the aggressive LiAlH₄ can be reduced with silanes, using a NPG catalyst.⁶²⁹ The reduction of imines and enimines under similar conditions is perhaps to be expected; however, the high selectivity that allowed aldehydes to remain intact without protection is notable.⁶³⁰

Silanes can also be activated electrocatalytically and ketone and aldehyde reduction can take place with high selectivity, leaving the normally extremely sensitive nitro group untouched (Scheme 4C).⁶³¹ The high yields and selectivity of this reaction projects a future application in organic synthesis.

As previously mentioned, H_2 used to be considered a poor reductant. However, Yamamoto and Bao and co-workers⁶³² reported in 2016 that H_2 could be used at a relatively low pressure of 8 atm for the reduction of aldehydes, imines, and

alkynes (Scheme 5A), with triethylamine as a solvent, provided that the residual Ag content in NPG was particularly low (ca. 1 mol %). Less reactive substrates like ketones, alkenes, quinolines, and aromatic nitro groups required ethanol as a solvent (Scheme 5B). A bit later, the same groups reported the hydrodebromination of aromatic bromides by NPG with H₂ (Scheme 5C).⁶³³ This reaction is useful in that there are occasions in which the Br substituent is used to block a certain otherwise reactive site. If it can be removed easily, it becomes a valuable protection group. As impressive as these results are, the activation of hydrogen remains relatively unexplained. XPS data did not convincingly demonstrate hydrogen activation, and the authors speculated that in fact, the altered structure of nanoporous Au₉₉Al₁ compared to nanoporous Au₉₀Al₁₀ caused the difference.⁶³² However, all of these reactions are most likely more complex than merely involving a H₂ dissociation. The hydrodebromination, for instance, was performed in MeOH, and a substantial, quite similar turnover was detected

even in the absence of H₂.⁶³³ It is also notable that there was no reaction when no base was added,⁶³³ making it very likely that MeOH is being oxidized to formaldehyde, thus MeOH is the reductant. Moreover, the reaction is not very tolerant of other functional groups: aldehydes, carbamates, and nitro groups get reduced and esters hydrolyze.⁶³³ In the other cases, it is not clear whether the solvent diethylamine was entirely innocent or, indeed, why triethylamine was used in the optimization reactions, whereas diethylamine was used in reactions in which the scope was explored.⁶³³ Like MeOH, amines can be oxidized, although this reaction has been shown only for Au nanoparticles⁶³⁴ but not for NPG.

There are other formal reduction reactions, in which not (only) hydrogen is transferred, but in which an organometallic reagent is generated that is commonly used as reactive intermediates in syntheses (Scheme 6). Notable reactions are, for example, hydrosilylation reactions of alkynes, which proceed much more smoothly with NPG prepared from an AuAl than from AuAg alloys.⁶³⁵ The diborylation of alkynes (Scheme 6A) serves as another example.⁶³⁶ Although this reaction can be performed slightly more efficiently with nanoporous Pt (dealloyed from PtCu), the use of NPG avoids leaching of ions into the reaction mixture, which can be difficult to remove. Instead of alkynes, strained rings such as methylene cyclopropanes can also serve as substrates (Scheme 6B).⁶³⁷

The use of NPG as a catalyst in oxidation or reduction reactions has not yet become commonplace in organic chemistry. Thus far, literature reports are mainly concerned with testing the scope of a reaction and starting to elucidate mechanisms, but no study has been performed in which a particular synthetic goal was reached with such a reaction as the key synthetic step. This is likely due to both the scarcity of transformations possible at this present time and the availability of easily reproducible catalyst formation protocols that can be performed in an organic laboratory. Moreover, the scalability of the procedures is yet to be demonstrated.

8. CONCLUSION

The bicontinuous pore structure featuring a comparatively narrow size distribution of ligaments and pores in the range of a few 10 nm is a defining and outstanding feature of nanoporous gold (NPG). Its formation by a corrosive chemical or electrochemical dealloying process can be mechanistically understood in terms of a kinetically controlled selforganization process, in which the nanoscale pore structure results from the competition of two concomitantly occurring surface processes. While the less noble element (LNE) of a binary Au alloy used as a starting material is continuously dissolved under oxidizing wet-chemical conditions, surface diffusion of the remaining Au atoms provides for the generation of Au layers, eventually passivating the surface. Further curvature-driven surface diffusion minimizes the net interfacial excess energy and leads to coarsening of the structure accompanied by the formation of voids and finally pores. These processes and thus the morphology of the nanoporous Au material obtained in the end can be tuned by a multitude of adjustable parameters, such as the composition of the master alloy, the temperature, the electrochemical settings, potential thermal treatments subsequently applied, and the storage conditions before use (section 2). The possibility to tailor important structural parameters, such as the mean ligament and pore sizes and the porosity or solid fraction,

respectively, distinguishes NPG from other mesoporous materials and qualifies it as a model system for studies aiming at exploring systematically structure—property relationships in areas for which Au nanostructures hold promising application perspectives as being the case in catalysis and electrocatalysis, for instance.

To fully exploit this potential, it is necessary to strive for ensuring the highest possible level of reproducibility when fabricating NPG. When furthermore aiming at tuning its structural features, stringent and reliable preparation protocols are mandatory and much effort was invested in the past to develop suitable procedures, ranging from the manufacturing of the starting alloy to various elaborated dealloying techniques. Section 2.4 describes such a protocol that has been proven to reproducibly yield NPG materials in different laboratories, exhibiting comparable structural and functional properties.

When trying to correlate catalytic properties with structural features to reveal, e.g., microkinetic rate constants or turnover frequencies, it does, however, not suffice to control ligament or pore sizes or the porosity alone. Equally important are quantities, such as the specific surface area and the tortuosity of the pore system, determining the catalytic capacity (per mass), on the one hand, and the diffusive mass transport properties of the porous material, on the other hand. How to derive these characteristics in case of NPG, is discussed in sections 3.3, 3.4, and 3.10. As became obvious there, such numbers can differ up to a factor of 2 when applying different conceptional approaches, even for one and the same sample.

One must then ask which of the conceptional approaches is relevant for catalysis. As far as the mass transport properties and their influence on the catalytic turnover are concerned, emerging techniques to directly measure the diffusive mass transport in porous matter, such as PFG NMR, can provide a remedy here (section 4.3). The diffusion-related tortuosity of the pore system in NPG is found to be well correlated with structural tortuosity values extracted from TEM tomography. This demonstrates how the comprehensive knowledge about NPG's pore system meanwhile allows disentangling the observable macrokinetics from the microkinetics of the surface reaction.

NPG always contains an unavoidable residual content of the LNE that, in the case of catalytic applications, turned out to be a crucial ingredient. For instance, molecular oxygen can only be activated in the presence of such a LNE but not on pure gold surfaces. Even though the overall amount of the remaining LNE can be adjusted by dealloying procedures optimized in this respect, its distribution within the material is less easily controllable and can depend on various parameters. As exemplified for NPG obtained from AuAg master alloys in particular, it can change during storage and use of the material. Specifically, it was observed that Ag segregates to the surface in oxygen-containing atmospheres (section 4.1) or during electrochemical cycling in alkaline solutions (section 6.1). Moreover, it can be removed from the surface by electrochemical surface oxidation under acidic conditions. Surprisingly and unexpectedly in the first instance, it was shown by advanced TEM methods (section 3.13) that NPG may even contain nanometer-sized volumes within the ligaments that exhibit the composition of the master alloy and are surrounded by regions with Ag concentrations corresponding to the nominal overall bulk residual content. As revealed by high resolution micrographs, these Ag clusters had been protected

by a continuous gold film during the dealloying step, preventing the oxidative dissolution of the LNE. Due to the surface mobility of gold atoms, however, it cannot be excluded that such regions might later be exposed to species adsorbed on the surface under catalytic turnover conditions. Given the importance of the LNE, in particular for the activation of molecular oxygen, such structural irregularities may thus have a considerable impact on the catalytic properties, such as activities and selectivities. Coarsening outside the corrosive environment has been demonstrated as a strategy for homogenization. It is desirable to routinely characterize the degree of compositional heterogeneity in catalysis studies and to explore in more detail how this aspect of the NPG microstructure affects its performance in various scenarios of catalysis.

In spite of such uncertainties, it is undisputable though that the research on NPG not only established a reliable knowledge base to control catalytically important structural and chemical characteristics, but also allowed obtaining a deeper mechanistic insight into its complex surface chemistry. The partial oxidation of alcohols is a good example in this respect. The generic reaction mechanism initially proposed for methanol oxidation to methyl formate emanated from the Brønsted basicity of oxygen atoms adsorbed on the Au surface as the main underlying driving force (section 4.1). As known from single-crystal studies carried out under UHV conditions, such species are indeed able to abstract hydrogen atoms from methanol and various other reaction intermediates playing a role in the catalytic cycle. While this mechanistic picture can readily explain the observed selectivity, including the formation of oxidative cross coupling between different alcohols as well as the formation of amides if reacting alcohols with amines, growing evidence revealed in the last years that additional aspects are important to reconcile all experimental observations with mechanistic explanations and theoretical predictions. One of the key issues in this respect concerns the question which of the various pathways, theoretically conceivable for the activation of molecular oxygen, is actually effective and how these pathways differ for total and partial oxidation reactions (sections 4.1 and 4.3). The generic mechanistic picture mentioned above leaves, for instance, open how oxygen species created at Ag sites are transferred and become available at Au sites where then the Au-specific oxidation chemistry takes place (section 4.1). Moreover, theory has identified reaction pathways other than a direct dissociative adsorption of molecular oxygen which may exhibit evenly low barriers (section 4.1). The situation is further complicated by the fact that activated oxygen species tend to aggregate on Au surfaces into islands. This does not only influence the activity of the species, but does also modulate the relative importance of their Brønsted and Lewis basicity and thus the propensity for partial or total oxidation reactions. Such differences are also reflected by quite different activation protocols necessary to prepare catalytically active samples for CO and methanol oxidation (section 4.3).

A further degree of complexity arises in the presence of water, which, for example, is a byproduct of partial oxidation reactions and was experimentally found to modify the surface reactivity significantly (section 4.2). While CO oxidation benefits from co-dosed water, the selectivity toward methyl formate suffers from its presence. From a mechanistic point of view, it was shown that adsorbed water is able to enhance the mobility of oxygen species bound to oxygen islands. In

particular, theory proved that water may react with activated oxygen and other coadsorbed species to form reactive intermediates, which further on open new reaction channels for the partial oxidation so that a different product distribution is observed.

The catalytic versatility of NPG is documented by the fact that it not only can be utilized for gas phase reactions but also for applications in the liquid phase. As far as methanol oxidation is concerned, methyl formate is preferentially formed also here with a similar activation energy and comparable turnover numbers (section 5). Despite these analogies and the possibility to mechanistically explain effects specific to the liquid phase (such as the beneficial influence of an added base), the level of understanding regarding NPG's catalytic properties is less advanced here and needs further research. Even though NPG has meanwhile been applied in a number of synthetically interesting organic reactions going beyond what has been tried in the gas phase, the experimental results obtained so far do not provide a consistent picture and oftentimes suffer from low yields and/or low selectivities (section 7). It should be kept in mind, though, that NPG was rarely characterized to an extent in the corresponding studies, as described in section 3. This leaves a margin of uncertainty with respect to the chemical properties of the material used in such work, rendering a more detailed discussion of possible reasons impossible at present.

In contrast, the electrochemical and electrocatalytic properties of NPG have been investigated and characterized in much more detail. The application of the material in this area is particularly attractive in view of the large surface area available for adsorbates and reactants or functional layers (section 6). The electrocatalytic properties depend on the surface composition, as in case of gas and liquid phase catalysis, as well as on the electrolyte used. This was exemplified for the methanol oxidation reaction in alkaline solution, revealing that Ag as the LNE plays a different role for the various reaction steps involved (section 6.1.2). While the 4-electron oxidation of methanol to formate is somewhat enhanced by small amounts of Ag at the surface, the 6-electron oxidation to $CO_2/$ CO_3^{2-} occurring at higher positive potentials is significantly promoted by the presence of Ag. In contrast to that, the latter reaction channel is suppressed in the case of LNE-free NPG within the stability range of water. The effect of Ag can be associated with the presence of AgO, being the stable species at the potential where the 6-electron oxidation occurs. In acidic solution, on the contrary, the effect of Ag is small because of its dissolution from the surface in these electrolytes. This process is accompanied by a restructuring of the surface, which provides a means to tune the distribution of various surface sites (section 6.1) and thus the reactivity of the surface. However, the current densities observed, and potentials used for methanol oxidation reaction on NPG in alkaline solution are incompatible with a technical application of this material. Sufficiently high current densities at low potentials require the presence of Pt or Pd on the surface of NPG surface (section 6.2.3).

With respect to CO_2 reduction, the activity of NPG benefits from an increased alkalinity in the pore space during electrolysis. This effect, which suppresses the competing hydrogen evolution reaction, originates from the porosity and mass transport conditions in the system (section 6.2.5).

Review

9. OUTLOOK

Despite the significant progress that has been made in recent years resulting in an increased understanding and command of the structural and compositional features of NPG as well as in atomistic insights into the mechanism of a variety of oxidation reactions, harnessing those achievements in terms of active and stable catalysts is still challenging. In the following, a number of areas are discussed, where ongoing research is expected to overcome existing hurdles for future applications.

9.1. Need for Establishing "Good Practice" Procedures for Catalytic Testing

The previous section revealed that NPG represents a catalytic material with well-defined structural and chemical characteristics, which can either be synthetically adjusted (such as pore and ligament sizes or the porosity) or are predetermined by the corrosive dealloying process (such as the tortuosity of the pore system for instance). In contrast to the various applications of the material being studied in materials science for a long time already and depending on the bulk properties, for catalytic applications the surface and its chemistry is essential. Previous research revealed that reaching control over the surface structure and composition is much more complicated. To accomplish this task and to establish standardized testing protocols which eventually allow identifying robust structureproperty relationships for various reactions and fields of application, further research effort is necessary from two directions:

(a) In the interaction of theory and experiment further mechanistic insight must be gained, e.g., with respect to the surface sites and species which are crucial for an understanding of the reaction network determining the catalytic process. In case of oxidation reactions, it is clear so far that undercoordinated Au sites (i.e., step and kink sites) are important and that the LNE is likely to play an important role for the activation of molecular oxygen. However, it is unclear in which chemical form—as isolated metal sites, corresponding islands, or rather in the form of oxidic species-the LNE participates in the catalysis and to which degree nondissociative (with respect to molecular oxygen) reaction pathways contribute to the overall mechanistic scenario. Here, experiments with suitable model systems, e.g., studied under UHV conditions, as well as further theory can help. Moreover, investigations under catalytic working conditions carried out with modern operando techniques are necessary, taking into account that structure and composition at the surface, as is known, dynamically change as a function of the surrounding gas atmosphere.

(b) From a *practical* point of view, a better and more advanced understanding of the correlation between bulk properties which can be well-controlled already, and the surface properties must be established. To this end, more systematic studies are needed, elucidating to which extent the LNE segregates to the surface as a function of the bulk content and the composition of the gas atmosphere. Further research is necessary to clarify whether and in which way the presence of LNE-rich, buried clusters in NPG can be avoided or purposefully taken advantage of. It may be speculated that the clusters may otherwise cause unreproducible differences between nominally similar samples. Apart from that, the development of reliable activation protocols needs more attention, in particular for liquid phase applications. So far, corresponding recipes have been reported in the literature only for methanol oxidation and CO oxidation in the gas phase.

Given progress in these directions, NPG can emerge as catalytic material serving, for instance, as a reference catalyst in Au catalysis and as an example of good practice in testing heterogeneous catalysts with well-defined properties. As far as the preparation and a broadening of the material's property space is concerned, further research regarding the following topics will support this goal:

9.2. Widening the Compositional Space of NPG

The structural and compositional properties of NPG ultimately determine the range of catalytic applications. Already in the past, a lot of effort has been undertaken to widen the compositional space of NPG, using, for instance, different LNEs or functionalizing the inner surfaces of the material with oxide deposits or organic molecules. This not only enlarged the scope of applications, but also allowed to gain further mechanistic insights, e.g., in the activation of oxygen or the use of water as an oxygen delivering agent (e.g., for the water gas shift reaction). Further research in this direction thus appears as an important field for future developments.

9.3. Need for Understanding the Impact of Dealloying on the Defect Structure

It is important to note that the dealloying process results in materials with a complex microstructure, details of which may critically depend on small variations in the preparation and processing protocols. While a detailed quantitative understanding of central aspects determining the dealloying process has been reached for AuAg master alloys, a similarly systematic understanding remains to be established for binary master alloys containing other LNEs and for ternary master alloys. In this regard, it would be important to better understand three separate issues: First, what is the correlation between, on the one hand, the crystal structure of the master alloy, and, on the other hand, the final structure of the material on the meso- and microscopic scale? Second, how is the lattice parameter misfit accommodated at the corrosion front? Third, how is the defect structure, grain size and the density of crystallographic defects such as dislocations, twin boundaries, and stacking faults, affected by the choice of master alloy and by the corrosion conditions?

9.4. LNE-Free NPG as a Reference Material

While we emphasize the essential role of the LNE for catalytic and electrocatalytic applications, it would also be desirable to have nanoporous gold materials with a comparable and similarly well-defined morphology available that are entirely free of LNEs. Such systems would be highly welcome as reference materials, to which the catalytic behavior of NPG containing LNEs can be compared.

9.5. Need for Scalable Production Protocols

From a practical point of view, approaches toward a scalable production of NPG are desirable. Although electrochemically controlled dealloying procedures currently offer better control and thus may yield materials of choice for sensors and microreactor technology, a scalable production of NPG is easier to achieve by using free corrosion protocols. Free corrosion processes can be varied by the choice of the corrosion medium and temperature in order to improve the controllability of structural (ligament and pore sizes) as well as chemical characteristics (residual LNE content). It is desirable to explore this approach with the aim of eventually achieving

Review

similar tuning options as given in the case of electrochemical dealloying.

9.6. Need for Appreciating Postdealloying Processing and Conditioning

Between the dealloying process and the catalytical application, NPG may undergo changes. These are either intended, such as coarsening to optimize mass transport conditions, or they occur spontaneously under catalytic working conditions. A fundamentally improved understanding of changes to the morphology as well as to the chemical and geometric structure upon interaction of NPG with adsorbates and neighboring phases (such as metal oxides) would open another turn key for controlling the reactivity and the stability of NPG-based catalysts.

9.7. NPG with Engineered Macroscale Morphologies

Other shapes of NPG than monoliths, foils, and thin films should find more attention in the future. Such engineered bodies from NPG may be flexibly fitted into integrated reactors. Examples are NPG powders processed to gasdiffusion electrodes combining macroscopically large area, flexible shapes, and direct connection to, e.g., membranes with optimized transport conditions for gaseous reactants and products. 3D printed and then dealloyed scaffolds have been demonstrated and might be explored further. Issues of recycling materials in different states of the life cycle (LNE in dealloying, usage of the very same NPG lot in multiple organic synthesis batches) may become more important as the application of NPG becomes more widespread.

Finally, it should be mentioned that there are a few applications, in which NPG serves as a mere support or back contact of other materials (metal oxides for catalysis and energy storage, biomolecules) or as a plasmonic absorber with excellent chemical stability and an optimum pore structure for efficient transport. For most of these applications, (electro)catalytic reactions are not required and are even detrimental to the stability of the technical device. Those fields may greatly benefit from the material as it is, and, consequently, these areas may be closest to technical applications at present.

ASSOCIATED CONTENT

Supporting Information

The Supporting Information is available free of charge at https://pubs.acs.org/doi/10.1021/acs.chemrev.2c00751.

Relations between different measures for the ligament size; conversion of electrode potentials between different reference electrodes; reference data for surface voltammetry of single crystal of Au electrodes; reference data for Pb UPD on Au single-crystal electrodes; performance data of porous gold electrodes for methanol oxidation reaction(PDF)

AUTHOR INFORMATION

Corresponding Author

Gunther Wittstock – Carl von Ossietzky University of Oldenburg, School of Mathematics and Science, Institute of Chemistry, D-26111 Oldenburg, Germany; oorcid.org/ 0000-0002-6884-5515; Email: wittstock@uol.de

Authors

Marcus Bäumer – University of Bremen, MAPEX Center for Materials and Processes and Institute for Applied and Physical Chemistry, 28359 Bremen, Germany; Octid.org/0000-0002-8620-1764

- Wilke Dononelli University of Bremen, MAPEX Center for Materials and Processes, 28359 Bremen, Germany; University of Bremen, Bremen Center for Computational Materials Science, Hybrid Materials Interfaces Group, Bremen 28359, Germany; Orcid.org/0000-0001-7577-421X
- Thorsten Klüner Carl von Ossietzky University of Oldenburg, School of Mathematics and Science, Institute of Chemistry, D-26111 Oldenburg, Germany; orcid.org/ 0000-0003-1389-6013
- Lukas Lührs Hamburg University of Technology, Institute of Materials Physics and Technology, 21703 Hamburg, Germany; orcid.org/0000-0001-9004-4389
- Christoph Mahr University of Bremen, MAPEX Center for Materials and Processes, 28359 Bremen, Germany; University of Bremen, Institute of Solid State Physics, 28359 Bremen, Germany; © orcid.org/0000-0003-1872-8887
- Lyudmila V. Moskaleva University of the Free State, Department of Chemistry, Bloemfontein 9300, South Africa; orcid.org/0000-0003-0168-7126
- Mehtap Oezaslan Technical University of Braunschweig, Institute of Technical Chemistry, Technical Electrocatalysis Laboratory, 38106 Braunschweig, Germany; © orcid.org/ 0000-0001-8545-7576
- Thomas Risse Freie Universität Berlin, Institute of Chemistry and Biochemistry, 14195 Berlin, Germany; orcid.org/0000-0003-0228-9189
- Andreas Rosenauer University of Bremen, MAPEX Center for Materials and Processes, 28359 Bremen, Germany; University of Bremen, Institute of Solid State Physics, 28359 Bremen, Germany; © orcid.org/0000-0003-4742-0451
- Anne Staubitz University of Bremen, MAPEX Center for Materials and Processes, 28359 Bremen, Germany; University of Bremen, Institute for Organic and Analytical Chemistry, D-28359 Bremen, Germany; Orcid.org/0000-0002-9040-3297
- Jörg Weissmüller Hamburg University of Technology, Institute of Materials Physics and Technology, 21703 Hamburg, Germany; Helmholtz-Zentrum Hereon, Institute of Materials Mechanics, 21502 Geesthacht, Germany; orcid.org/0000-0002-8958-4414
- Arne Wittstock University of Bremen, MAPEX Center for Materials and Processes, 28359 Bremen, Germany; University of Bremen, Institute for Organic and Analytical Chemistry, D-28359 Bremen, Germany; orcid.org/0000-0002-0139-3710

Complete contact information is available at: https://pubs.acs.org/10.1021/acs.chemrev.2c00751

Author Contributions

The manuscript was written through contributions of all authors. First drafts were supplied by T.R., J.W., and G.W. for section 1; L.L. for sections 2.1–2.6; G.W. and J.W. for section 2.7; M.Oe. for section 2.8; J.W. and G.W. for section 2.9; J.W., C.M., and A.R. for section 3; L.V.M., W.D., and T.K. for section 4.1; T.R. for section 4.2; M.B. for section 4.3; C.M. and A.R. for section 4.4; A.W. for section 5; G.W. for section 6; A.S. and A.W. for section 7; T.R., J.W., and G.W. for section 8; T.R., J.W., M.B., and G.W. for section 9. All authors have edited and refined the manuscript and have given approval to

the final version of the manuscript. CRediT: Gunther Wittstock conceptualization, funding acquisition.

Notes

The authors declare no competing financial interest.

Biographies

Gunther Wittstock studied Chemistry at the University of Leipzig 1986–1991 followed by a Ph.D. in Analytical Chemistry in 1994. Following research stays at the University of Cincinnati (W. R. Heineman) and the Technical University of Munich (H. L. Schmidt, W. Schuhmann), he received a habilitation in Physical Chemistry working on surface analysis of modified electrodes with R. Szargan. He has been a full professor of Physical Chemistry at the Carl von Ossietzky University Oldenburg since 2001. His research is focused on analysis of localized electrochemical reactions, modified electrodes, electrocatalysis, and spectroelectrochemistry at solid–liquid interfaces. He has served as member of the Executive Committee and Treasurer of the International Society of Electrochemistry 2017– 2022.

Marcus Bäumer studied Chemistry at the Ruhr-University Bochum and received his diploma in 1990. In the following, he did his Ph.D. under the supervision of H.-J. Freund there, which he finished in 1994. Interrupted by a postdoc stay at Stanfort University (Bob Madix), he received his habilitation in 2000 (Technical University Berlin) on grounds of work on model catalytic studies at the Fritz-Haber-Institute in Berlin. In 2002, M.B. got an appointment at Bremen university, where he ever since holds a position as full professor. His research focuses on heterogeneous gas phase catalysis, nanomaterials (nanoporous metals and oxides) and model studies under UHV conditions. He is author or coauthor of over 250 publications.

Wilke Dononelli is Senior Scientist at the University of Bremen in Germany. He received his Ph.D. in Theoretical Chemistry at the University of Oldenburg in 2018. After a stay at Aarhus University in the Physics Department from 2018 until 2020, Wilke started working at the interdisciplinary MAPEX Center for Materials and Processes at University of Bremen in May 2020. In his research he focuses on atomistic modelling of catalytic processes using density functional theory and high level post-HF *ab initio* wave function based methods. In addition, he works with new global optimization methods to determine the structure of functional materials.

Thorsten Klüner studied Chemistry at the Ruhr University Bochum in Germany and received his Ph.D. in 1997 in the field of Theoretical Chemistry. From 1997 to 2004, he was group leader at the Fritz-Haber-Institute of the Max-Planck Society in Berlin as well postdoctoral fellow at UCLA in 2000. Since 2004, he is Professor of Theoretical Chemistry at the Carl-von-Ossietzky University Oldenburg, Germany. His main research interest is quantum chemistry (wave-function based and DFT) with a focus on multireference problems and excited states in surface chemistry as well as dynamics and control of open quantum systems.

Lukas Lührs is a postdoctoral researcher at the Institute of Materials Physics and Technology at Hamburg University of Technology, Germany, where he studied Mechanical Engineering and received his Ph.D. in engineering in the year 2020. The research activities of him and his team focus on nanoscale networks with large surface areas and their functionalization by controlled manipulation of interfaces. In particular, his research includes preparation concepts and the modification of mechanical properties by electrochemical surface modulation. Lyudmila V. Moskaleva is an Associate Professor at the University of the Free State, South Africa, in the Department of Chemistry. She received her Ph.D. in Physical Chemistry in 2001 from Emory University, USA, and her MS in Chemistry in 1997 from the Higher Chemical College of the Russian Academy of Sciences, Moscow, Russia. In 2002–2006, she was an Alexander von Humboldt Research Fellow and a postdoctoral fellow at Technische Universität München, Germany. From 2009 to 2018, she was a principal investigator funded by DFG at Universität Bremen, Germany. Moskaleva's group investigates surface reactivity of solids at the atomic level using first-principles quantum-chemical methods, molecular dynamics, statistical theory, microkinetic modeling, and thermodynamics.

Mehtap Oesazlan obtained her Ph.D. degree (2012) in Electrochemistry from Technische Universität Berlin, Germany. After a postdoctoral stay at the Electrochemistry Laboratory Group, Paul Scherrer Institute, Switzerland, she was a Junior professor for Electrochemistry at Carl von Ossietzky University of Oldenburg from 2014–2019. Since June 2019, she holds a Full Professorship at the Institute of Technical Chemistry, Technical University of Braunschweig. Her research is focused on the development and characterization of electrocatalysts for PEM fuel cells and electrolysers. Furthermore, she was part of the FOR 2213 research unit as Prinicipal Investigator of Subproject 9 during the second funding period.

Thomas Risse studied chemistry at the university of Cologne and received his doctorate in Physical Chemistry from Ruhr-Universität Bochum in 1996. From 1997 until 2010, he was a staff scientist at the Fritz Haber Institute interluded by one year as a postdoc at the Jules Stein Eye institute of UCLA in the group of Prof. Hubbell. In 2010, he became professor of Physical Chemistry at Freie Universität Berlin. He is interested in chemical processes at surfaces as well as the structural and dynamic properties of proteins primarily using spectroscopic techniques such as IR- or ESR-spectroscopy.

Andreas Rosenauer received his Ph.D. from the University of Regensburg in 1996. In 1999, he received a Humboldt postdoctoral fellowship, which he spent at the EMAT-Research Center of the University of Antwerp. In 2002, he habilitated at the University of Karlsruhe on "Composition Evaluation and Strain State Analysis of Semiconductor Nanostructures by Transmission Electron Microscopy". In 2002–2003 he was a visiting professor at the EMAT-Center, where he was appointed full professor of physics in 2003. Since 2004, he has been Professor of Experimental Physics at the University of Bremen.

Anne Staubitz started her independent academic career as an assistant professor at Kiel University in 2010. In 2015, she received an appointment at the University of Bremen where she heads the group "Organic Functional Materials". Her interests are the synthesis of new molecular switches, switchable polymeric systems, and their application in materials science. Furthermore, the group is also developing organo-tin chemistry, in particular heterocycles, and group 13/group 15 element combinations in organic scaffolds. In the DFG Research Unit FOR2213 "Nanoporous Gold", she is working on new oxidative transformations with NPG and peroxides and peralcohols. Jörg Weissmüller is a full professor and head of the Institute of Materials Physics and Technology at Hamburg University of Technology. He is also the leader of the Hybrid Materials Systems Department at Helmholtz-Zentrum Hereon. He studied materials science at Saarland University, Germany, and the University of Dundee, UK. He received his Ph.D. degree in Engineering and the Habilitation in Experimental Physics from Saarland University. He has worked at the Institute of New Materials, Germany, the National Institute of Standards and Technology, Gaithersburg, USA and the Institute of Nanotechnology at Karlsruhe Institute of Technology, Germany. His current research focuses on interfaces and on nanomaterials with interface-controlled behavior, with emphasis on the coupling between mechanics and chemical or electrochemical processes.

Arne Wittstock enrolled at the University of Bremen for his Ph.D. studies under the supervision of Prof. Marcus Bäumer. His research focused on the synthesis of nanoporous gold materials and their application for gas phase catalytic reactions. He received his Ph.D. in May, 2010. Afterwards, Dr. Wittstock joined to the Lawrence Livermore Nationzl Laboratory as a directorate postdoc. His research was focused on graphene type materials for energy related applications. In 2012, he returned to the University Bremen as leader of a junior research group on novel nanoporous hybrid materials for (photo)-catalytic reactions. In 2021, he joined the research group of Prof. Anne Staubitz as a senior research assistant.

ACKNOWLEDGMENTS

We acknowledge funding of Deutsche Forschungsgemeinschaft (DFG) through the Research unit FOR 2213 "Nanoporous Gold-A Prototype for a Rational Catalyst Design", within the subprojects WI 1617/21-2 (G.W.), BA 1710/29-2 (M.B.), KL 1175/14-1 (T.K.), MO 1863/5-1 (L.V.M.), OE610/1-1 (M.Oe.), RI 1025/3-2 (T.R.), RO-2057/12-2 (A.R.), Wi 4497/1-2 (A.S.), WE1424/17-2 (J.W.), WI 4497/1 (A.W.); the DFG Research Training Group 2226 "Activation of Chemical Bonds" (G.W.); the APF project "Materials on Demand" within the "Humans on Mars" Initiative funded by the Federal State of Bremen and the University of Bremen (W.D.). L.L. gratefully acknowledges support by DFG through the Collaborative Research Center SFB 986, project 192346071. L.V.M. thanks the North German Association for High Performance Computing (HLRN) and Center for High-Performance Computing (CHPC) in South Africa for providing computational resources. T.K. thanks the North German Association for High Performance Computing (HLRN) and the Alexander von Humboldt foundation (AvH) for a Humboldt Research Fellowship (CHN/ 1120376). W.D. acknowledges funding by the APF project "Materials on Demand" within the "Humans on Mars" Initiative funded by the Federal State of Bremen and the University of Bremen. Within this Unit, the experimental and computational contribution and discussion with Ph.D. students and research associates Lenard Carroll, Alexandra Dworzak, Christoph Feldt, Matthias Graf, Mareike Haensch, Sandra Hoppe, Anastasia Lackmann, Yong Li, Eric Meyer, Raphael Moeira, Wiebke Riedel, Marco Schowalter, Okikiola Olaniyan, Alex Ricardo Silva Olaya, Jorge Adian Tapias Burgos, Jan Thayssen, Niklas Thoben, Gabriele Tomaschun, Stefan Wild, and Xinyan Wu are highly appreciated. The Unit also gratefully acknowledges research contributions by the late colleague Stefan Müller from TUHH.

ABBREVIATIONS

AAS = atomic absorption spectroscopy BET = Brunauer-Emmett-Teller CO_2RR = electrochemical carbon dioxide reduction reaction DFT = density functional theory DOSC = degree of selectivity control ECL = electrogenerated chemiluminescence EDX = energy-dispersive X-ray analysis FDH = fructose dehydrogenase FC = free corrosion GOx = glucose oxidaseGD = galvanostatic dealloying GC-MS = gas chromatography-mass spectrometry HAADF-STEM = high-angle annular dark-field scanning transmission electron microscopy HER = hydrogen evolution reaction KMC = kinetic Monte Carlo (simulations) LEED = low-energy electron diffraction LNE = less noble element MARI = most abundant reaction intermediate MNE = more noble element $MSE = mercury - mercury sulfate electrode Hg|Hg_2SO_4|$ KHSO₄ $MeOH = methanol, CH_3OH$ MeFm = methyl formate HCOOCH₃ ML = monolayers MOR = methanol oxidation reaction NAD⁺ = nicotinamide adenine dinucleotide NMR = nuclear magnetic resonance NPG = nanoporous gold OER = oxygen evolution reaction ORR = oxygen reduction reaction PBS = phosphate buffer saline PD = potentiostatic dealloying PEDOT = poly(3,4-ethylenedioxythiophene) PFG-NMR = pulsed field gradient nuclear magnetic resonance PMMA = poly(methyl methacrylate) PROX = preferential CO oxidation RHE = reversible hydrogen electrode RVE = representative volume element SHE = standard hydrogen electrode SCE = saturated calomel electrode $Hg|Hg_2Cl_2|KCl_{sa}$ STEM = scanning transmission electron microscopy TAP = temporal analysis of product TEM = transmission electron microscopy TOF = turnover frequency TPR(S) = temperature-programmed reaction (spectroscopy) UHV = ultrahigh vacuum UPD = underpotential deposition/stripping XPS = X-ray photoelectron spectroscopy. REFERENCES

(1) Haruta, M.; Kobayashi, T.; Sano, H.; Yamada, N. Novel Gold Catalysts for the Oxidation of Carbon Monoxide at a Temperature far Below 0 °C. *Chem. Lett.* **1987**, *16*, 405–408.

(2) Hashmi, A. S. K.; Hutchings, G. J. Gold Catalysis. Angew. Chem., Int. Ed. 2006, 45, 7896–7936.

(3) Corma, A.; Garcia, H. Supported Gold Nanoparticles as Catalysts for Organic Reactions. *Chem. Soc. Rev.* 2008, 37, 2096–2126.

(4) Ciriminna, R.; Falletta, E.; Della Pina, C.; Teles, J. H.; Pagliaro, M. Industrial Applications of Gold Catalysis. *Angew. Chem., Int. Ed.* **2016**, 55, 14210–14217.

(5) Hutchings, G. Vapor Phase Hydrochlorination of Acetylene: Correlation of Catalytic Activity of Supported Metal Chloride Catalysts. J. Catal. **1985**, *96*, 292–295.

(6) Zielasek, V.; Juergens, B.; Schulz, C.; Biener, J.; Biener, M. M.; Hamza, A. V.; Bäumer, M. Gold Catalysts: Nanoporous Gold Foams. *Angew. Chem., Int. Ed.* **2006**, *45*, 8241–8244.

(7) Xu, C.; Su, J.; Xu, X.; Liu, P.; Zhao, H.; Tian, F.; Ding, Y. Low Temperature CO Oxidation over Unsupported Nanoporous Gold. *J. Am. Chem. Soc.* **2007**, *129*, 42–43.

(8) Wittstock, A.; Zielasek, V.; Biener, J.; Friend, C. M.; Bäumer, M. Nanoporous Gold Catalysts for Selective Gas-Phase Oxidative Coupling of Methanol at Low Temperature. *Science* **2010**, 327, 319–322.

(9) Newman, R. C.; Corcoran, S. G.; Erlebacher, J.; Aziz, M. J.; Sieradzki, K. Alloy Corrosion. MRS Bull. **1999**, 24, 24–28.

(10) Erlebacher, J.; Aziz, M. J.; Karma, A.; Dimitrov, N.; Sieradzki, K. Evolution of Nanoporosity in Dealloying. *Nature* **2001**, *410*, 450–453.

(11) Weissmüller, J.; Newman, R. C.; Jin, H.-J.; Hodge, A. M.; Kysar, J. W. Nanoporous Metals by Alloy Corrosion: Formation and Mechanical Properties. *MRS Bull.* **2009**, *34*, 577–586.

(12) McCue, I.; Karma, A.; Erlebacher, J. Pattern Formation during Electrochemical and Liquid Metal Dealloying. *MRS Bull.* **2018**, *43*, 27–34.

(13) Everett, D. H. Manual of Symbols and Terminology for Physicochemical Quantities and Units, Appendix II: Definitions, Terminology and Symbols in Colloid and Surface Chemistry. *Pure Appl. Chem.* **1972**, *31*, 577–638.

(14) Li, R.; Sieradzki, K. Ductile-Brittle Transition in Random Porous Au. *Phys. Rev. Lett.* **1992**, *68*, 1168–1171.

(15) Weissmüller, J.; Sieradzki, K. Dealloyed Nanoporous Materials with Interface-Controlled Behavior. *MRS Bull.* **2018**, *43*, 14–19.

(16) Liu, P.; Guan, P.; Hirata, A.; Zhang, L.; Chen, L.; Wen, Y.; Ding, Y.; Fujita, T.; Erlebacher, J.; Chen, M. Visualizing Under-Coordinated Surface Atoms on 3D Nanoporous Gold Catalysts. *Adv. Mater.* **2016**, *28*, 1753–1759.

(17) Biener, J.; Biener, M. M.; Madix, R. J.; Friend, C. M. Nanoporous Gold: Understanding the Origin of the Reactivity of a 21st Century Catalyst Made by Pre-Columbian Technology. *ACS Catal.* **2015**, *5*, 6263–6270.

(18) Zugic, B.; Wang, L.; Heine, C.; Zakharov, D. N.; Lechner, B. A. J.; Stach, E. A.; Biener, J.; Salmeron, M.; Madix, R. J.; Friend, C. M. Dynamic Restructuring Drives Catalytic Activity on Nanoporous Gold-Silver Alloy Catalysts. *Nat. Mater.* **2017**, *16*, 558–564.

(19) Cherevko, S.; Kulyk, N.; Chung, C.-H. Nanoporous Pt@ Au_xCu_{100-x} by Hydrogen Evolution Assisted Electrodeposition of Au_xCu_{100-x} and Galvanic Replacement of Cu with Pt: Electrocatalytic Properties. *Langmuir* **2012**, *28*, 3306–3315.

(20) Kumar, A.; Gonçalves, J. M.; Sukeri, A.; Araki, K.; Bertotti, M. Correlating Surface Growth of Nanoporous Gold with Electrodeposition Parameters to Optimize Amperometric Sensing of Nitrite. *Sens. Actuators, B* **2018**, *263*, 237–247.

(21) Mie, Y.; Takayama, H.; Hirano, Y. Facile Control of Surface Crystallographic Orientation of Anodized Nanoporous Gold Catalyst and its Application for Highly Efficient Hydrogen Evolution Reaction. *J. Catal.* **2020**, *389*, 476–482.

(22) Xia, H.; Ran, Y.; Li, H.; Tao, X.; Wang, D. Freestanding Monolayered Nanoporous Gold Films with High Electrocatalytic Activity via Interfacial Self-Assembly and Overgrowth. *J. Mater. Chem. A* **2013**, *1*, 4678.

(23) Biener, J.; Hamza, A. V.; Hodge, A. M. Deformation Behavior of Nanoporous Metals. In *Micro and Nano Mechanical Testing of Materials and Devices*; Springer: Boston, MA, 2008; pp 118–135. DOI: 10.1007/978-0-387-78701-5 6.

(24) Ding, Y.; Chen, M. Nanoporous Metals for Catalytic and Optical Applications. *MRS Bull.* **2009**, *34*, 569–576.

(25) McCue, I.; Benn, E.; Gaskey, B.; Erlebacher, J. Dealloying and Dealloyed Materials. *Annu. Rev. Mater. Res.* **2016**, *46*, 263–286.

(26) Chen, Q.; Ding, Y.; Chen, M. Nanoporous Metal by Dealloying for Electrochemical Energy Conversion and Storage. *MRS Bull.* **2018**, 43, 43–48.

(27) Jin, H.-J.; Weissmüller, J.; Farkas, D. Mechanical Response of Nanoporous Metals: A Story of Size, Surface Stress, and Severed Struts. *MRS Bull.* **2018**, *43*, 35–42.

(28) Şeker, E.; Shih, W.-C.; Stine, K. J. Nanoporous Metals by Alloy Corrosion: Bioanalytical and Biomedical Applications. *MRS Bull.* **2018**, 43, 49–56.

(29) Ahl, S.; Cameron, P. J.; Liu, J.; Knoll, W.; Erlebacher, J.; Yu, F. A Comparative Plasmonic Study of Nanoporous and Evaporated Gold Films. *Plasmonics* **2008**, *3*, 13–20.

(30) Pröschel, A.; Chacko, J.; Whitaker, R.; Chen, M. A. U.; Detsi, E. Visible Light Plasmonic Heating-Enhanced Electrochemical Current in Nanoporous Gold Cathodes. *J. Electrochem. Soc.* **2019**, *166*, H146–H150.

(31) Wang, Z.; Du, J.; Zhang, Y.; Han, J.; Huang, S.; Hirata, A.; Chen, M. Free-standing Nanoporous Gold for Direct Plasmon Enhanced Electro-Oxidation of Alcohol Molecules. *Nano Energy* **2019**, *56*, 286–293.

(32) Koya, A. N.; Zhu, X.; Ohannesian, N.; Yanik, A. A.; Alabastri, A.; Proietti Zaccaria, R.; Krahne, R.; Shih, W.-C.; Garoli, D. Nanoporous Metals: From Plasmonic Properties to Applications in Enhanced Spectroscopy and Photocatalysis. *ACS Nano* **2021**, *15*, 6038–6060.

(33) Ding, Y.; Chen, M.; Erlebacher, J. Metallic Mesoporous Nanocomposites for Electrocatalysis. J. Am. Chem. Soc. 2004, 126, 6876–6877.

(34) Guo, D.-J.; Ding, Y. Porous Nanostructured Metals for Electrocatalysis. *Electrocanalysis* **2012**, *24*, 2035–2043.

(35) Graf, M.; Haensch, M.; Carstens, J.; Wittstock, G.; Weissmüller, J. Electrocatalytic Methanol Oxidation with Nanoporous Gold: Microstructure and Selectivity. *Nanoscale* **2017**, *9*, 17839.

(36) Kumar, A.; Gonçalves, J. M.; Selva, J. S. G.; Araki, K.; Bertotti, M. Correlating Selective Electrocatalysis of Dopamine and Ascorbic Acid Electrooxidation at Nanoporous Gold Surfaces with Structural-Defects. J. Electrochem. Soc. **2019**, *166*, H704–H711.

(37) Ma, H.; Ding, Y. Electrocatalytical Properties of Nanoporous Gold. In *Nanoporous Gold: From an Ancient Technology to a High-Tech Material*; RSC Nanoscience & Nanotechnology; Wittstock, A., Biener, J., Erlebacher, J., Bäumer, M., Eds.; Royal Society of Chemistry: Cambridge, UK, 2012; Vol. 22, pp 199–223.

(38) Hyun, G.; Song, J. T.; Ahn, C.; Ham, Y.; Cho, D.; Oh, J.; Jeon, S. Hierarchically Porous Au Nanostructures with Interconnected Channels for Efficient Mass Transport in Electrocatalytic CO₂ Reduction. *Proc. Natl. Acad. Sci. U.S.A.* **2020**, *117*, 5680–5685.

(39) Wittstock, A.; Bäumer, M. Catalysis by Unsupported Skeletal Gold Catalysts. Acc. Chem. Res. 2014, 47, 731-739.

(40) Qi, Z.; Biener, J.; Biener, M. Surface Oxide-Derived Nanoporous Gold Catalysts for Electrochemical CO_2 -to-CO Reduction. ACS Appl. Energy Mater. **2019**, 2, 7717–7721.

(41) Gao, Y.; Ding, Y. Nanoporous Metals for Heterogeneous Catalysis: Following the Success of Raney Nickel. *Chem.—Eur. J.* **2020**, *26*, 8845–8856.

(42) Wittstock, A.; Wichmann, A.; Biener, J.; Bäumer, M. Nanoporous Gold: A New Gold Catalyst with Tunable Properties. *Faraday Discuss.* **2011**, *152*, 87–98.

(43) Rurainsky, C.; Manjón, A. G.; Hiege, F.; Chen, Y.-T.; Scheu, C.; Tschulik, K. Electrochemical Dealloying as a Tool to Tune the Porosity, Composition and Catalytic Activity of Nanomaterials. *J. Mater. Chem. A* **2020**, *8*, 19405–19413.

(44) Zeng, M.; Liu, Y.; Hu, Y.; Zhang, X. High-Efficient CO₂ Electrocatalysis over Nanoporous Au Film Enabled by a Combined Pore Engineering and Ionic Liquid-Mediated Approach. *Chem. Eng. J.* **2021**, 425, No. 131663. (45) Outka, D. A.; Madix, R. J. Broensted Basicity of Atomic Oxygen on the Gold(110) Surface: Reactions with Methanol, Acetylene, Water, and Ethylene. J. Am. Chem. Soc. **1987**, 109, 1708–1714.

(46) Outka, D. A.; Madix, R. J. The Oxidation of Carbon Monoxide on the Au(110) Surface. *Surf. Sci.* **1987**, *179*, 351–360.

(47) Wachs, I. E.; Madix, R. J. The Oxidation of Methanol on a Silver(110) Catalyst. *Surf. Sci.* **1978**, *76*, 531–558.

(48) Stowers, K. J.; Madix, R. J.; Friend, C. M. From Model Studies on Au(1 1 1) to Working Conditions with Unsupported Nanoporous Gold Catalysts:Oxygen-assisted coupling reactions. *J. Catal.* **2013**, 308, 131–141.

(49) Xu, B.; Haubrich, J.; Baker, T. A.; Kaxiras, E.; Friend, C. M. Theoretical Study of O-Assisted Selective Coupling of Methanol on Au(111). *J. Phys. Chem. C* **2011**, *115*, 3703–3708.

(50) Moskaleva, L. V.; Zielasek, V.; Klüner, T.; Neyman, K. M.; Bäumer, M. CO Oxidation by Co-Adsorbed Atomic O on the Au(321) Surface with Ag Impurities: A Mechanistic Study from First-Principles Calculations. *Chem. Phys. Lett.* **2012**, *525–526*, 87–91.

(51) Moskaleva, L. V.; Weiss, T.; Kluener, T.; Bäumer, M. Chemisorbed Oxygen on the Au(321) Surface Alloyed with Silver:A First-Principles Investigation. *J. Phys. Chem. C* **2015**, *119*, 9215–9226.

(52) Wang, L. C.; Zhong, Y.; Widmann, D.; Weissmüller, J.; Behm, R. J. On the Role of Residual Ag in Nanoporous Au Catalysts for CO Oxidation: A Combined Microreactor and TAP Reactor Study. *ChemCatChem.* **2012**, *4*, 251–259.

(53) Fajín, J. L. C.; Cordeiro, M. N. D. S.; Gomes, J. R. B. On the Theoretical Understanding of the Unexpected O_2 Activation by Nanoporous Gold. *Chem. Commun.* **2011**, 47, 8403–8405.

(54) Montemore, M. M.; Madix, R. J.; Kaxiras, E. How Does Nanoporous Gold Dissociate Molecular Oxygen? J. Phys. Chem. C 2016, 120, 16636–16640.

(55) Dononelli, W.; Tomaschun, G.; Klüner, T.; Moskaleva, L. V. Understanding Oxygen Activation on Nanoporous Gold. *ACS Catal.* **2019**, *9*, 5204–5216.

(56) Ojifinni, R. A.; Froemming, N. S.; Gong, J.; Pan, M.; Kim, T. S.; White, J. M.; Henkelman, G.; Mullins, C. B. Water-Enhanced Low-Temperature CO Oxidation and Isotope Effects on Atomic Oxygen-Covered Au(111). *J. Am. Chem. Soc.* **2008**, *130*, 6801–6812.

(57) Han, D.; Xu, T.; Su, J.; Xu, X.; Ding, Y. Gas-Phase Selective Oxidation of Benzyl Alcohol to Benzaldehyde with Molecular Oxygen over Unsupported Nanoporous Gold. *ChemCatChem.* **2010**, *2*, 383–386.

(58) Wittstock, A.; Neumann, B.; Schaefer, A.; Dumbuya, K.; Kuebel, C.; Biener, M. M.; Zielasek, V.; Steinrueck, H.-P.; Gottfried, J. M.; Biener, J.; Hamza, A.; Bäumer, M. Nanoporous Au: An Unsupported Pure Gold Catalyst? *J. Phys. Chem. C* **2009**, *113*, 5593–5600.

(59) Wichmann, A.; Bäumer, M.; Wittstock, A. Oxidative Coupling of Alcohols and Amines over Bimetallic Unsupported Nanoporous Gold: Tailored Activity through Mechanistic Predictability. *ChemCatChem.* **2015**, *7*, 70–74.

(60) Lackmann, A.; Mahr, C.; Schowalter, M.; Fitzek, L.; Weissmüller, J.; Rosenauer, A.; Wittstock, A. A Comparative Study of Alcohol Oxidation over Nanoporous Gold in Gas and Liquid Phase. *J. Catal.* **2017**, *353*, 99–106.

(61) Yin, H.; Zhou, C.; Xu, C.; Liu, P.; Xu, X.; Ding, Y. Aerobic Oxidation of D-Glucose on Support-Free Nanoporous Gold. *J. Phys. Chem. C* **2008**, *112*, 9673–9678.

(62) Zhang, J.; Liu, P.; Ma, H.; Ding, Y. Nanostructured Porous Gold for Methanol Electro-Oxidation. J. Phys. Chem. C 2007, 111, 10382–10388.

(63) Yu, C.; Jia, F.; Ai, Z.; Zhang, L. Direct Oxidation of Methanol on Self-Supported Nanoporous Gold Film Electrodes with High Catalytic Activity and Stability. *Chem. Mater.* **2007**, *19*, 6065–6067.

(64) Li, G. G.; Wang, H. Dealloyed Nanoporous Gold Catalysts: From Macroscopic Foams to Nanoparticulate Architectures. *Chem-NanoMat* **2018**, *4*, 897–908.

(65) Shao, L. H.; Jin, H. J.; Weissmüller, J. Actuation with High-Surface-Area Materials. In Nanoporous Gold: From an Ancient *Technology to a High-Tech Material*; RSC Nanoscience & Nanotechnology; Wittstock, A., Biener, J., Erlebacher, J., Bäumer, M., Eds.; Royal Society of Chemistry: Cambridge, UK, 2012; Vol. 22, pp 137– 166.

(66) van der Zalm, J.; Chen, S.; Huang, W.; Chen, A. Review— Recent Advances in the Development of Nanoporous Au for Sensing Applications. J. Electrochem. Soc. **2020**, 167, 037532.

(67) Xiao, X.; Si, P.; Magner, E. An Overview of Dealloyed Nanoporous Gold in Bioelectrochemistry. *Bioelectrochemistry* **2016**, 109, 117–126.

(68) Heim, M.; Rousseau, L.; Reculusa, S.; Urbanova, V.; Mazzocco, C.; Joucla, S.; Bouffier, L.; Vytras, K.; Bartlett, P.; Kuhn, A.; Yvert, B. Combined Macro-/Mesoporous Microelectrode Arrays for Low-Noise Extracellular Recording of Neural Networks. *J. Neurophysiol.* **2012**, *108*, 1793–1803.

(69) Xiao, S.; Wang, S.; Wang, X.; Xu, P. Nanoporous Gold: A Review and Potentials in Biotechnological and Biomedical Applications. *Nano Select* **2021**, *2*, 1437–1458.

(70) Kim, S. H. Nanoporous Gold for Energy Applications. *Chem. Rec.* **2021**, *21*, 1199.

(71) Wittstock, A.; Biener, J.; Bäumer, M. Surface Chemistry and Catalysis. In *Nanoporous Gold: From an Ancient Technology to a High-Tech Material*; Wittstock, A., Biener, J., Erlebacher, J., Bäumer, M., Eds.; RSC Nanoscience & Nanotechnology, Royal Society of Chemistry: Cambridge, 2012; Vol. 22, pp 167–198.

(72) Lee, J. D.; Miller, J. B.; Shneidman, A. V.; Sun, L.; Weaver, J. F.; Aizenberg, J.; Biener, J.; Boscoboinik, J. A.; Foucher, A. C.; Frenkel, A. I.; van der Hoeven, J. E. S.; Kozinsky, B.; Marcella, N.; Montemore, M. M.; Ngan, H. T.; O'Connor, C. R.; Owen, C. J.; Stacchiola, D. J.; Stach, E. A.; Madix, R. J.; Sautet, P.; Friend, C. M. Dilute Alloys Based on Au, Ag, or Cu for Efficient Catalysis: From Synthesis to Active Sites. *Chem. Rev.* **2022**, *122*, 8758–8808.

(73) Lu, Z.; Li, C.; Han, J.; Zhang, F.; Liu, P.; Wang, H.; Wang, Z.; Cheng, C.; Chen, L.; Hirata, A.; Fujita, T.; Erlebacher, J.; Chen, M. Three-dimensional Bicontinuous Nanoporous Materials by Vapor Phase Dealloying. *Nat. Commun.* **2018**, *9*, 276.

(74) Han, J.; Li, C.; Lu, Z.; Wang, H.; Wang, Z.; Watanabe, K.; Chen, M. Vapor Phase Dealloying: A Versatile Approach for Fabricating 3D Porous Materials. *Acta Mater.* **2019**, *163*, 161–172.

(75) Wada, T.; Yubuta, K.; Inoue, A.; Kato, H. Dealloying by Metallic Melt. *Mater. Lett.* **2011**, *65*, 1076–1078.

(76) Wada, T.; Setyawan, A. D.; Yubuta, K.; Kato, H. Nano- to Submicro-porous β -Ti Alloy Prepared from Dealloying in a Metallic Melt. *Scr. Mater.* **2011**, *65*, 532–535.

(77) Senior, N. A.; Newman, R. C. Synthesis of Tough Nanoporous Metals by Controlled Electrolytic Dealloying. *Nanotechnology* **2006**, *17*, 2311–2316.

(78) Wagner, K.; Brankovic, S. R.; Dimitrov, N.; Sieradzki, K. Dealloying below the Critical Potential. *J. Electrochem. Soc.* **1997**, 144, 3545–3555.

(79) Rugolo, J.; Erlebacher, J.; Sieradzki, K. Length Scales in Alloy Dissolution and Measurement of Absolute Interfacial Free Energy. *Nat. Mater.* **2006**, *5*, 946–949.

(80) Li, X.; Chen, Q.; McCue, I.; Snyder, J.; Crozier, P.; Erlebacher, J.; Sieradzki, K. Dealloying of Noble-Metal Alloy Nanoparticles. *Nano Lett.* **2014**, *14*, 2569–2577.

(81) Ding, Y.; Kim, Y.-J.; Erlebacher, J. Nanoporous Gold Leaf. "Ancient Technology"/Advanced Material. *Adv. Mater.* **2004**, *16*, 1897–1900.

(82) Sieradzki, K.; Newman, R. C. Stress-Corrosion Cracking. J. Phys. Chem. Solids 1987, 48, 1101–1113.

(83) Tammann, G. Die chemischen und galvanischen Eigenschaften von Mischkristallreihen und ihre Atomverteilung: Ein Beitrag zur Kenntnis der Legierungen [The Chemical and Galvanic Characteristics of Compound Crystals and their Atomic Distribution: An Article on the Understanding of Alloys]. Z. Anorg. Allg. Chem. 1919, 107, 1–239.

(84) Masing, G. Zur Theorie der Resistenzgrenzen in Mischkristallen. Z. Anorg. Allg. Chem. **1921**, 118, 293-308. (85) Forty, A. J. Corrosion Micromorphology of Noble Metal Alloys and Depletion Gilding. *Nature* **1979**, *282*, 597.

(86) Badwe, N.; Chen, X.; Schreiber, D. K.; Olszta, M. J.; Overman, N. R.; Karasz, E. K.; Tse, A. Y.; Bruemmer, S. M.; Sieradzki, K. Decoupling the Role of Stress and Corrosion in the Intergranular Cracking of Noble-Metal Alloys. *Nat. Mater.* **2018**, *17*, 887–893.

(87) Artymowicz, D. M.; Erlebacher, J.; Newman, R. C. Relationship Between the Parting Limit for De-Alloying and a Particular Geometric High-Density Site Percolation Threshold. *Philos. Mag.* **2009**, *89*, 1663–1693.

(88) Sieradzki, K.; Corderman, R. R.; Shukla, K.; Newman, R. C. Computer Simulations of Corrosion: Selective Dissolution of Binary Alloys. *Philos. Mag. A* **1989**, *59*, 713–746.

(89) Erlebacher, J. An Atomistic Description of Dealloying: Porosity Evolution, the Critical Potential, and Rate-Limiting Behavior. J. Electrochem. Soc. 2004, 151, C614–C626.

(90) Krekeler, T.; Straßer, A. V.; Graf, M.; Wang, K.; Hartig, C.; Ritter, M.; Weissmüller, J. Silver-Rich Clusters in Nanoporous Gold. *Mater. Res. Lett.* **2017**, *5*, 314–321.

(91) Li, Y.; Ngo-Dinh, B.-N.; Markmann, J.; Weissmüller, J. Evolution of Length Scales and of Chemical Heterogeneity during Primary and Secondary Dealloying. *Acta Mater.* **2022**, *222*, 117424.

(92) Ye, X.-L.; Lu, N.; Li, X.-J.; Du, K.; Tan, J.; Jin, H.-J. Primary and Secondary Dealloying of Au(Pt)-Ag: Structural and Compositional Evolutions, and Volume Shrinkage. *J. Electrochem. Soc.* **2014**, *161*, C517-C526.

(93) Graf, M.; Roschning, B.; Weissmüller, J. Nanoporous Gold by Alloy Corrosion:Method-Structure-Property Relationships. J. Electrochem. Soc. 2017, 164, C194–C200.

(94) Mahr, C.; Kundu, P.; Lackmann, A.; Zanaga, D.; Thiel, K.; Schowalter, M.; Schwan, M.; Bals, S.; Wittstock, A.; Rosenauer, A. Quantitative Determination of Residual Silver Distribution in Nanoporous Gold and its Influence on Structure and Catalytic Performance. J. Catal. 2017, 352, 52–58.

(95) El-Zoka, A. A.; Langelier, B.; Botton, G. A.; Newman, R. C. Enhanced Analysis of Nanoporous Gold by Atom Probe Tomography. *Mater. Charact.* **2017**, *128*, 269–277.

(96) Qian, L. H.; Chen, M. W. Ultrafine Nanoporous Gold by Lowtemperature Dealloying and Kinetics of Nanopore Formation. *Appl. Phys. Lett.* **2007**, *91*, 083105.

(97) Zhang, Z.; Wang, Y.; Wang, Y.; Wang, X.; Qi, Z.; Ji, H.; Zhao, C. Formation of Ultrafine Nanoporous Gold Related to Surface Diffusion of Gold Adatoms during Dealloying of Al₂Au in an Alkaline Solution. *Scr. Mater.* **2010**, *62*, 137–140.

(98) Dursun, A.; Pugh, D. V.; Corcoran, S. G. Dealloying of Ag-Au Alloys in Halide-Containing Electrolytes. *J. Electrochem. Soc.* **2003**, *150*, B355.

(99) Dursun, A.; Pugh, D. V.; Corcoran, S. G. Probing the Dealloying Critical Potential. J. Electrochem. Soc. 2005, 152, B65.

(100) Snyder, J.; Asanithi, P.; Dalton, A. B.; Erlebacher, J. Stabilized Nanoporous Metals by Dealloying Ternary Alloy Precursors. *Adv. Mater.* **2008**, *20*, 4883–4886.

(101) Cheng, C.; Lührs, L. Robust Metallic Actuators Based on Nanoporous Gold Rapidly Dealloyed from Gold–Nickel Precursors. *Adv. Funct. Mater.* **2021**, *31*, 2107241.

(102) Sieradzki, K.; Dimitrov, N.; Movrin, D.; McCall, C.; Vasiljevic, N.; Erlebacher, J. The Dealloying Critical Potential. *J. Electrochem. Soc.* **2002**, *149*, B370–B377.

(103) Parida, S.; Kramer, D.; Volkert, C. A.; Rösner, H.; Erlebacher, J.; Weissmüller, J. Volume Change during the Formation of Nanoporous Gold by Dealloying. *Phys. Rev. Lett.* **2006**, *97*, No. 035504.

(104) Detsi, E.; van de Schootbrugge, M.; Punzhin, S.; Onck, P. R.; De Hosson, J.T.M. On Tuning the Morphology of Nanoporous Gold. *Scr. Mater.* **2011**, *64*, 319–322.

(105) Graf, M.; Ngô, B.-N. D.; Weissmüller, J.; Markmann, J. X-Ray Studies of Nanoporous Gold: Powder Diffraction by Large Crystals with Small Holes. *Phys. Rev. Mater.* **2017**, *1*, 98. (106) Biener, M. M.; Biener, J.; Wichmann, A.; Wittstock, A.; Baumann, T. F.; Bäumer, M.; Hamza, A. V. ALD Functionalized Nanoporous Gold: Thermal Stability, Mechanical Properties, and Catalytic Activity. *Nano Lett.* **2011**, *11*, 3085–3090.

(107) Liu, Y.; Bliznakov, S.; Dimitrov, N. Comprehensive Study of the Application of a Pb Underpotential Deposition-Assisted Method for Surface Area Measurement of Metallic Nanoporous Materials. *J. Phys. Chem. C* **2009**, *113*, 12362–12372.

(108) Silva Olaya, A. R.; Zandersons, B.; Wittstock, G. Effect of the Residual Silver and Adsorbed Lead Anions towards the Electrocatalytic Methanol Oxidation on Nanoporous Gold in Alkaline Media. *Electrochim. Acta* **2021**, *383*, 138348.

(109) Detsi, E.; Onck, P.; De Hosson, J. T. M. Metallic Muscles at Work: High Rate Actuation in Nanoporous Gold/Polyaniline Composites. *ACS Nano* **2013**, *7*, 4299–4306.

(110) Meng, F.; Ding, Y. Sub-Micrometer-Thick All-Solid-State Supercapacitors with High Power and Energy Densities. *Adv. Mater.* **2011**, 23, 4098–4102.

(111) Roschning, B.; Weissmüller, J. Nanoporous-Gold-Polypyrrole Hybrid Materials for Millimeter-Sized Free Standing Actuators. *Adv. Mater. Interfaces* **2020**, *7*, 2001415.

(112) Mameka, N.; Lührs, L.; Heissler, S.; Gliemann, H.; Wöll, C. Tailoring the Strength of Nanoporous Gold by Self-Assembled Monolayers of Alkanethiols. *ACS Appl. Nano Mater.* **2018**, *1*, 6613–6621.

(113) Xu, C.; Wang, R.; Chen, M.; Zhang, Y.; Ding, Y. Dealloying to Nanoporous Au/Pt Alloys and Their Structure Sensitive Electrocatalytic Properties. *Phys. Chem. Chem. Phys.* **2010**, *12*, 239–246.

(114) Jin, H.-J.; Wang, X.-L.; Parida, S.; Wang, K.; Seo, M.; Weissmüller, J. Nanoporous Au-Pt Alloys as Large Strain Electrochemical Actuators. *Nano Lett.* **2010**, *10*, 187–194.

(115) Vega, A. A.; Newman, R. C. Nanoporous Metals Fabricated through Electrochemical Dealloying of Ag-Au-Pt with Systematic Variation of Au:Pt Ratio. *J. Electrochem. Soc.* **2014**, *161*, C1.

(116) Xie, H.; Guan, H.; Liu, L.-Z.; Jin, H.-J. A Critical Relative Density and a Break-and-Reconnect Model for Annealing-Induced Densification in Nanoporous Gold. *Acta Mater.* **2021**, *209*, No. 116806.

(117) Liu, L.-Z.; Ye, X.-L.; Jin, H.-J. Interpreting Anomalous Lowstrength and Low-stiffness of Nanoporous Gold: Quantification of Network Connectivity. *Acta Mater.* **2016**, *118*, 77–87.

(118) Lührs, L.; Zandersons, B.; Huber, N.; Weissmüller, J. Plastic Poisson's Ratio of Nanoporous Metals: A Macroscopic Signature of Tension-Compression Asymmetry at the Nanoscale. *Nano Lett.* **2017**, *17*, 6258–6266.

(119) Zandersons, B.; Lührs, L.; Li, Y.; Weissmüller, J. On Factors Defining the Mechanical Behavior of Nanoporous Gold. *Acta Mater.* **2021**, *215*, No. 116979.

(120) Liu, M.; Weissmüller, J. Phase Decomposition in Nanoporous Au-Pt. *Acta Mater.* **2022**, *241*, No. 118419.

(121) Kertis, F.; Snyder, J.; Govada, L.; Khurshid, S.; Chayen, N.; Erlebacher, J. Structure/Processing Relationships in the Fabrication of Nanoporous Gold. *JOM* **2010**, *62*, 50–56.

(122) Sharma, A.; Bhattarai, J. K.; Alla, A. J.; Demchenko, A. V.; Stine, K. J. Electrochemical Annealing of Nanoporous Gold by Application of Cyclic Potential Sweeps. *Nanotechnology* **2015**, *26*, 085602.

(123) Herring, C. Effect of Change of Scale on Sintering Phenomena. J. Appl. Phys. 1950, 21, 301-303.

(124) Li, Y.; Dinh Ngô, B.-N.; Markmann, J.; Weissmüller, J. Topology Evolution during Coarsening of Nanoscale Metal Network Structures. *Phys. Rev. Mater.* **2019**, *3*, 076001.

(125) Hu, K.; Ziehmer, M.; Wang, K.; Lilleodden, E. T. Nanoporous Gold: 3D Structural Analyses of Representative Volumes and their Implications on Scaling Relations of Mechanical Behaviour. *Philos. Mag.* **2016**, *96*, 3322.

(126) Ng, A. K.; Welborn, S. S.; Detsi, E. Time-dependent Power Law Function for the Post-dealloying Chemical Coarsening of Nanoporous Gold Derived using Small-angle X-ray Scattering. Scr. Mater. 2022, 206, No. 114215.

(127) Chen-Wiegart, Y.-C. K.; Wang, S.; Chu, Y. S.; Liu, W.; McNulty, I.; Voorhees, P. W.; Dunand, D. C. Structural Evolution of Nanoporous Gold during Thermal Coarsening. *Acta Mater.* **2012**, *60*, 4972–4981.

(128) Kuwano-Nakatani, S.; Fujita, T.; Uchisawa, K.; Umetsu, D.; Kase, Y.; Kowata, Y.; Chiba, K.; Tokunaga, T.; Arai, S.; Yamamoto, Y.; Tanaka, N.; Chen, M. Environment-Sensitive Thermal Coarsening of Nanoporous Gold. *Mater. Trans.* **2015**, *56*, 468–472.

(129) Son, G.; Son, Y.; Jeon, H.; Kim, J.-Y.; Lee, S. A Threedimensional Monte Carlo Model for Coarsening Kinetics of the Bicontinuous System via Surface Diffusion and its Application to Nanoporous Gold. *Scr. Mater.* **2020**, *174*, 33–38.

(130) McCue, I.; Stuckner, J.; Murayama, M.; Demkowicz, M. J. Gaining New Insights into Nanoporous Gold by Mining and Analysis of Published Images. *Sci. Rep.* **2018**, *8*, 6761.

(131) Lifshitz, İ. M.; Slyozov, V. V. The Kinetics of Precipitation from Supersaturated Solid Solutions. J. Phys. Chem. Solids 1961, 19, 35–50.

(132) Wagner, C. Theorie der Alterung von Niederschlägen durch Umlösen (Ostwald-Reifung). Zeitschrift für Elektrochemie, Berichte der Bunsengesellschaft für physikalische Chemie **1961**, 65, 581–591.

(133) Wang, D.; Schaaf, P. Nanoporous Gold Nanoparticles. J. Mater. Chem. 2012, 22, 5344.

(134) Ji, C.; Searson, P. C. Fabrication of Nanoporous Gold Nanowires. *Appl. Phys. Lett.* **2002**, *81*, 4437–4439.

(135) Sun, Y.; Ye, J.; Shan, Z.; Minor, A. M.; Balk, T. J. The Mechanical Behavior of Nanoporous Gold Thin Films. *JOM* **2007**, *59*, 54–58.

(136) Xu, Y.; Ke, X.; Yu, C.; Liu, S.; Zhao, J.; Cui, G.; Higgins, D.; Chen, Z.; Li, Q.; Wu, G. A Strategy for Fabricating Nanoporous Gold Films Through Chemical Dealloying of Electrochemically Deposited Au-Sn Alloys. *Nanotechnology* **2014**, *25*, No. 445602.

(137) Chapman, C. A. R.; Chen, H.; Stamou, M.; Biener, J.; Biener, M. M.; Lein, P. J.; Seker, E. Nanoporous Gold as a Neural Interface Coating: Effects of Topography, Surface Chemistry, and Feature Size. *ACS Appl. Mater. Interfaces* **2015**, *7*, 7093–7100.

(138) Rösner, H.; Parida, S.; Kramer, D.; Volkert, C. A.; Weissmüller, J. Reconstructing a Nanoporous Metal in Three Dimensions:An Electron Tomography Study of Dealloyed Gold Leaf. *Adv. Eng. Mater.* **2007**, *9*, 535–541.

(139) Hou, Y.; Chen, L.; Hirata, A.; Fujita, T.; Chen, M. Non-aqueous Nanoporous Gold Based Supercapacitors with High Specific Energy. *Scr. Mater.* **2016**, *116*, 76–81.

(140) Rout, S.; Qi, Z.; Biener, M. M.; Courtwright, D.; Adrien, J. C.; Mills, E.; Shahabuddin, M.; Noginova, N.; Noginov, M. A. Nanoporous Gold Nanoleaf as Tunable Metamaterial. *Sci. Rep.* **2021**, *11*, 1795.

(141) Huang, J.-F.; Sun, I.-W. Fabrication and Surface Functionalization of Nanoporous Gold by Electrochemical Alloying/Dealloying of Au-Zn in an Ionic Liquid, and the Self-Assembly of L-Cysteine Monolayers. *Adv. Funct. Mater.* **2005**, *15*, 989–994.

(142) Volkert, C. A.; Lilleodden, E. T.; Kramer, D.; Weissmüller, J. Approaching the Theoretical Strength in Nanoporous Au. *Appl. Phys. Lett.* **2006**, *89*, 061920.

(143) Wu, Y.; Markmann, J.; Lilleodden, E. T. Electro-chemomechanical Coupling of Nanoporous Gold at the Microscale. *Appl. Phys. Lett.* **2019**, *115*, 251602.

(144) Ma, K.; Corsi, J. S.; Fu, J.; Detsi, E. Origin of the Volume Contraction during Nanoporous Gold Formation by Dealloying for High-Performance Electrochemical Applications. *ACS Appl. Nano Mater.* **2018**, *1*, 541–546.

(145) Hakamada, M.; Mabuchi, M. Mechanical Strength of Nanoporous Gold Fabricated by Dealloying. *Scr. Mater.* 2007, *56*, 1003–1006.

(146) Badwe, N.; Chen, X.; Sieradzki, K. Mechanical Properties of Nanoporous Gold in Tension. *Acta Mater.* **2017**, *129*, 251–258.

(147) Steyskal, E.-M.; Seidl, M.; Simic, S.; Würschum, R. Enhanced Charging-Induced Resistance Variations of Nanoporous Gold by Dealloying in Neutral Silver Nitrate Solution. *Langmuir* **2018**, *34*, 13110–13115.

(148) Mameka, N.; Markmann, J.; Jin, H.-J.; Weissmüller, J. Electrical Stiffness Modulation—Confirming the Impact of Surface Excess Elasticity on the Mechanics of Nanomaterials. *Acta Mater.* **2014**, *76*, 272–280.

(149) Lührs, L.; Soyarslan, C.; Markmann, J.; Bargmann, S.; Weissmüller, J. Elastic and Plastic Poisson's Ratios of Nanoporous Gold. *Scr. Mater.* **2016**, *110*, 65–69.

(150) Bürckert, M.; Briot, N. J.; Balk, T. J. Uniaxial Compression Testing of Bulk Nanoporous Gold. *Philos. Mag.* **2017**, *97*, 1157–1178. (151) Liu, L.-Z.; Zhang, Y.-Y.; Xie, H.; Jin, H.-J. Transition from Homogeneous to Localized Deformation in Nanoporous Gold. *Phys. Rev. Lett.* **2021**, *127*, 95501.

(152) Wang, K.; Kobler, A.; Kübel, C.; Jelitto, H.; Schneider, G.; Weissmüller, J. Nanoporous-Gold-Based Composites: Toward Tensile Ductility. *NPG Asia Mater.* **2015**, *7*, e187–e187.

(153) Zhu, C.; Qi, T.; Beck, V. A.; Luneau, M.; Lattimer, J.; Chen, W.; Worsley, M. A.; Ye, J.; Duoss, E. B.; Spadaccini, C. M.; Friend, C. M.; Biener, J. Toward Digitally Controlled Catalyst Architectures: Hierarchical Nanoporous Gold via 3D Printing. *Sci. Adv.* **2018**, *4*, eaas9459.

(154) Fritz, J. D.; Pickering, H. W. Selective Anodic Dissolution of Cu-Au Alloys: TEM and Current Transient Study. *J. Electrochem. Soc.* **1991**, *138*, 3209–3218.

(155) Zhong, Y.; Markmann, J.; Jin, H.-J.; Ivanisenko, Y.; Kurmanaeva, L.; Weissmüller, J. Crack Mitigation during Dealloying of Au₂₅Cu₇₅. *Adv. Eng. Mater.* **2014**, *16*, 389–398.

(156) Mahr, C.; Schowalter, M.; Mitterbauer, C.; Lackmann, A.; Fitzek, L.; Mehrtens, T.; Wittstock, A.; Rosenauer, A. Nanoporous Gold Dealloyed from AuAg and AuCu: Comparison of Structure and Chemical Composition. *Materialia* **2018**, *2*, 131–137.

(157) Tse, A. Y.; Karasz, E. K.; Sieradzki, K. Dealloying and Morphology Evolution of Ordered and Disordered Cu_3Au . *Scr. Mater.* **2020**, 176, 112–116.

(158) Welborn, S. S.; Simafranca, A.; Wang, Z.; Wei, H.; Detsi, E. Chelation-mediated Synthesis of Nanoporous Gold at Near-neutral pH and Room Temperature by Free corrosion Dealloying of Gold-Copper Alloy Driven by Oxygen Reduction. *Scr. Mater.* **2021**, *200*, No. 113901.

(159) Zhang, Q.; Wang, X.; Qi, Z.; Wang, Y.; Zhang, Z. A Benign Route to Fabricate Nanoporous Gold through Electrochemical Dealloying of Al–Au Alloys in a Neutral Solution. *Electrochim. Acta* **2009**, *54*, 6190–6198.

(160) Wang, X.; Zhang, Z.; Ji, H.; Xu, J.; Huang, X.; Ma, Y. Dealloying of Single-phase Al_2Au to Nanoporous Gold Ribbon/Film with Tunable Morphology in Inorganic and Organic Acidic Media. *Appl. Surf. Sci.* **2012**, *258*, 9073–9079.

(161) Jiang, J.; Wang, X.; Zhang, L. Nanoporous Gold Microelectrode Prepared from Potential Modulated Electrochemical Alloying-Dealloying in Ionic Liquid. *Electrochim. Acta* **2013**, *111*, 114–119.

(162) Rong, K.; Huang, L.; Zhang, H.; Zhai, J.; Fang, Y.; Dong, S. Electrochemical Fabrication of Nanoporous Gold Electrodes in a Deep Eutectic Solvent for Electrochemical Setections. *Chem. Commun.* **2018**, *54*, 8853–8856.

(163) Nishio, K.; Yuda, K.; Masuda, H. Fabrication of Nanoporous Gold Film by Electrochemical Alloying and Dealloying with Li at Room Temperature. *ECS Electrochem. Lett.* **2013**, *2*, C1–C3.

(164) Yu, J.; Ding, Y.; Xu, C.; Inoue, A.; Sakurai, T.; Chen, M. Nanoporous Metals by Dealloying Multicomponent Metallic Glasses. *Chem. Mater.* **2008**, *20*, 4548–4550.

(165) Rizzi, P.; Scaglione, F.; Battezzati, L. Nanoporous Gold by Dealloying of an Amorphous Precursor. *J. Alloys Compd.* **2014**, 586, S117–S120.

(166) Paschalidou, E. M.; Celegato, F.; Scaglione, F.; Rizzi, P.; Battezzati, L.; Gebert, A.; Oswald, S.; Wolff, U.; Mihaylov, L.; Spassov, T. The Mechanism of Generating Nanoporous Au by Dealloying Amorphous Alloys. *Acta Mater.* **2016**, *119*, 177–183.

(167) Lu, X.; Bischoff, E.; Spolenak, R.; Balk, T. J. Investigation of Dealloying in Au–Ag Thin Films by Quantitative Electron Probe Microanalysis. *Scr. Mater.* **2007**, *56*, 557–560.

(168) Qi, Z.; Weissmüller, J. Hierarchical Nested-Network Nanostructure by Dealloying. ACS Nano 2013, 7, 5948–5954.

(169) Biener, J.; Hodge, A. M.; Hamza, A. V.; Hsiung, L. M.; Satcher, J. H. Nanoporous Au: A High Yield Strength Material. *J. Appl. Phys.* **2005**, *97*, 024301.

(170) Hodge, A. M.; Hayes, J. R.; Caro, J. A.; Biener, J.; Hamza, A. V. Characterization and Mechanical Behavior of Nanoporous Gold. *Adv. Eng. Mater.* **2006**, *8*, 853–857.

(171) Balk, T. J.; Eberl, C.; Sun, Y.; Hemker, K. J.; Gianola, D. S. Tensile and Compressive Microspecimen Testing of Bulk Nanoporous Gold. *JOM* **2009**, *61*, 26–31.

(172) Dotzler, C. J.; Ingham, B.; Illy, B. N.; Wallwork, K.; Ryan, M. P.; Toney, M. F. In Situ Observation of Strain Development and Porosity Evolution in Nanoporous Gold Foils. *Adv. Funct. Mater.* **2011**, *21*, 3938–3946.

(173) Lee, D.; Wei, X.; Chen, X.; Zhao, M.; Jun, S. C.; Hone, J.; Herbert, E. G.; Oliver, W. C.; Kysar, J. W. Microfabrication and Mechanical Properties of Nanoporous Gold at the Nanoscale. *Scr. Mater.* **2007**, *56*, 437–440.

(174) Cattarin, S.; Kramer, D.; Lui, A.; Musiani, M. M. Preparation and Characterization of Gold Nanostructures of Controlled Dimension by Electrochemical Techniques. *J. Phys. Chem. C* 2007, *111*, 12643–12649.

(175) Snyder, J.; Livi, K.; Erlebacher, J. Dealloying Silver/Gold Alloys in Neutral Silver Nitrate Solution:Porosity Evolution, Surface Composition, and Surface Oxides. *J. Electrochem. Soc.* **2008**, *155*, C464–C473.

(176) Snyder, J.; Erlebacher, J. Kinetics of Crystal Etching Limited by Terrace Dissolution. *J. Electrochem. Soc.* **2010**, *157*, C125.

(177) Mangipudi, K. R.; Epler, E.; Volkert, C. A. Morphological Similarity and Structure-dependent Scaling Laws of Nanoporous Gold from Different Synthesis Methods. *Acta Mater.* **2017**, *140*, 337–343.

(178) Shi, S.; Li, Y.; Ngo-Dinh, B.-N.; Markmann, J.; Weissmüller, J. Scaling Behavior of Stiffness and Strength of Hierarchical Network Nanomaterials. *Science* **2021**, *371*, 1026–1033.

(179) Jin, H.-J.; Parida, S.; Kramer, D.; Weissmüller, J. Sign-Inverted Surface Stress-Charge Response in Nanoporous Gold. *Surf. Sci.* **2008**, *602*, 3588–3594.

(180) Jin, H. J.; Kurmanaeva, L.; Schmauch, J.; Rösner, H.; Ivanisenko, Y.; Weissmüller, J. Deforming Nanoporous Metal: Role of Lattice Coherency. *Acta Mater.* **2009**, *57*, 2665–2672.

(181) Jin, H.-J.; Weissmüller, J. Bulk Nanoporous Metal for Actuation. *Adv. Eng. Mater.* **2010**, *12*, 714–723.

(182) Okamoto, H.; Massalski, T. B. The Ag–Au (Silver-Gold) System. Bull. Alloy Phase Diagrams 1983, 4, 30–38.

(183) Okamoto, H.; Chakrabarti, D. J.; Laughlin, D. E.; Massalski, T. B. The Au-Cu (Gold-Copper) System. J. Phase Equilib. 1987, 8, 454.

(184) Wang, J.; Lu, X.-G.; Sundman, B.; Su, X. Thermodynamic Assessment of the Au–Ni System. *Calphad* **2005**, *29*, 263–268.

(185) Okman, O.; Kysar, J. W. Fabrication of Crack-free Blanket Nanoporous Gold Thin Films by Galvanostatic Dealloying. *J. Alloys Compd.* **2011**, *509*, 6374–6381.

(186) Lackmann, A.; Bäumer, M.; Wittstock, G.; Wittstock, A. Independent Control over Residual Silver Content of Nanoporous Gold by Galvanodynamically Controlled Dealloying. *Nanoscale* **2018**, *10*, 17166–17173.

(187) Borkowska, Z.; Slimming, U. Perchlorate Adsorption on Polycrystalline Gold Electrodes in Aqueous Perchloric Acid. J. Electroanal. Chem. Interfacial Electrochem. **1991**, 312, 237–244.

(188) Burke, L. D.; Buckley, D. T.; Morrissey, J. A. Novel View of the Electrochemistry of Gold. *Analyst* **1994**, *119*, 841.

(189) Jin, H. J.; Weissmüller, J. A Material with Electrically Tunable Strength and Flow Stress. *Science* **2011**, *332*, 1179–1182.

(190) Angerstein-Kozlowska, H.; Conway, B. E.; Hamelin, A.; Stoicoviciu, L. Elementary Steps of Electrochemical Oxidation of Single-crystal Planes of Au—I. Chemical Basis of Processes Involving Geometry of Anions and the Electrode Surfaces. *Electrochim. Acta* **1986**, *31*, 1051–1061.

(191) Haensch, M.; Graf, M.; Wang, W.; Nefedov, A.; Wöll, C.; Weissmüller, J.; Wittstock, G. Thermally Driven Ag–Au Compositional Changes at the Ligament Surface in Nanoporous Gold: Implications for Electrocatalytic Applications. *ACS Appl. Nano Mater.* **2020**, *3*, 2197–2206.

(192) Silva Olaya, A. R.; Zandersons, B.; Wittstock, G. Restructuring of Nanoporous Gold Surface During Electrochemical Cycling in Acidic and Alkaline Media. *ChemElectroChem.* **2020**, *7*, 3670–3678.

(193) Lührs, L. Mechanical Properties of Nanoporous Metals: Model Experiments and Technology-Relevant Materials; Technische Universität Hamburg, 2020.

(194) Bellet, D.; Canham, L. Controlled Drying: The Key to Better Quality Porous Semiconductors. *Adv. Mater.* **1998**, *10*, 487–490.

(195) Bisi, O.; Ossicini, S.; Pavesi, L. Porous Silicon: A Quantum Sponge Structure for Silicon Based Optoelectronics. *Surf. Sci. Rep.* **2000**, *38*, 1–126.

(196) Korotcenkov, G.; Cho, B. K. Silicon Porosification: State of the Art. *Crit. Rev. Solid State Mater. Sci.* **2010**, *35*, 153–260.

(197) Gurav, J. L.; Jung, I.-K.; Park, H.-H.; Kang, E. S.; Nadargi, D. Y. Silica Aerogel: Synthesis and Applications. *J. Nanomater.* **2010**, 2010, 1–11.

(198) Lührs, L.; Weissmüller, J. Nanoporous Copper-Nickel – Macroscopic Bodies of a Strong and Deformable Nanoporous Base Metal by Dealloying. *Scr. Mater.* **2018**, *155*, 119–123.

(199) Yue, X.; Hu, R.; Qi, J.; He, Y.; Meng, Q.; Wei, F.; Ren, Y.; Sui, Y. Fabrication and Degradation Properties of Nanoporous Copper with Tunable Pores by Dealloying Amorphous Ti-Cu Alloys with Minor Co Addition. J. Mater. Eng. Perform. **2021**, *30*, 1759–1767.

(200) Juarez, T.; Biener, J.; Weissmüller, J.; Hodge, A. M. Nanoporous Metals with Structural Hierarchy: A Review. *Adv. Eng. Mater.* **2017**, *19*, 1700389.

(201) Fujita, T. Hierarchical Nanoporous Metals as a Path Toward the Ultimate Three-dimensional Functionality. *Sci. Technol. Adv. Mater.* **2017**, *18*, 724–740.

(202) Sondhi, P.; Stine, K. J. Methods to Generate Structurally Hierarchical Architectures in Nanoporous Coinage Metals. *Coatings* **2021**, *11*, 1440.

(203) Nyce, G. W.; Hayes, J. R.; Hamza, A. V.; Satcher, J. H. Synthesis and Characterization of Hierarchical Porous Gold Materials. *Chem. Mater.* **2007**, *19*, 344–346.

(204) Lee, M. N.; Santiago-Cordoba, M. A.; Hamilton, C. E.; Subbaiyan, N. K.; Duque, J. G.; Obrey, K. A. D. Developing Monolithic Nanoporous Gold with Hierarchical Bicontinuity Using Colloidal Bijels. J. Phys. Chem. Lett. **2014**, *5*, 809–812.

(205) Ding, Y.; Erlebacher, J. Nanoporous Metals with Controlled Multimodal Pore Size Distribution. J. Am. Chem. Soc. 2003, 125, 7772–7773.

(206) Qi, Z.; Vainio, U.; Kornowski, A.; Ritter, M.; Weller, H.; Jin, H.; Weissmüller, J. Porous Gold with a Nested-Network Architecture and Ultrafine Structure. *Adv. Funct. Mater.* **2015**, *25*, 2530–2536.

(207) Fujita, T.; Kanoko, Y.; Ito, Y.; Chen, L.; Hirata, A.; Kashani, H.; Iwatsu, O.; Chen, M. Nanoporous Metal Papers for Scalable Hierarchical Electrode. *Adv. Sci.* **2015**, *2*, 1500086.

(208) Guo, X.; Han, J.; Liu, P.; Chen, L.; Ito, Y.; Jian, Z.; Jin, T.; Hirata, A.; Li, F.; Fujita, T.; Asao, N.; Zhou, H.; Chen, M. Hierarchical Nanoporosity Enhanced Reversible Capacity of Bicontinuous Nanoporous Metal Based Li-O₂ Battery. *Sci. Rep.* **2016**, *6*, 33466.

(209) McCurry, D. A.; Kamundi, M.; Fayette, M.; Wafula, F.; Dimitrov, N. All Electrochemical Fabrication of a Platinized Nanoporous Au Thin-Film Catalyst. *ACS Appl. Mater. Interfaces* **2011**, *3*, 4459–4468. (210) Kramer, D.; Viswanath, R. N.; Weissmüller, J. Surface-Stress Induced Macroscopic Bending of Nanoporous Gold Cantilevers. *Nano Lett.* **2004**, *4*, 793–796.

(211) Li, X.; Liu, M.; Huang, B.; Liu, H.; Hu, W.; Shao, L.-H.; Wang, Z. L. Nanoporous-Gold-Based Hybrid Cantilevered Actuator Dealloyed and Driven by A Modified Rotary Triboelectric Nanogenerator. *Sci. Rep.* **2016**, *6*, 24092.

(212) Lavrik, N. V.; Tipple, C. A.; Sepaniak, M. J.; Datskos, P. G. Enhanced Chemi-Mechanical Transduction at Nanostructured Interfaces. *Chem. Phys. Lett.* **2001**, 336, 371–376.

(213) Zhang, L.; Lang, X.; Hirata, A.; Chen, M. Wrinkled Nanoporous Gold Films with Ultrahigh Surface-Enhanced Raman Scattering Enhancement. *ACS Nano* **2011**, *5*, 4407–4413.

(214) Jalas, D.; Shao, L.-H.; Canchi, R.; Okuma, T.; Lang, S.; Petrov, A.; Weissmüller, J.; Eich, M. Electrochemical Tuning of the Optical Properties of Nanoporous Gold. *Sci. Rep.* **2017**, *7*, 44139.

(215) Jalas, D.; Canchi, R.; Petrov, A. Y.; Lang, S.; Shao, L.; Weissmüller, J.; Eich, M. Effective Medium Model for the Spectral Properties of Nanoporous Gold in the Visible. *Appl. Phys. Lett.* **2014**, *105*, 241906.

(216) Zeng, Z.; Long, X.; Zhou, H.; Guo, E.; Wang, X.; Hu, Z. On-Chip Interdigitated Supercapacitor Based on Nano-porous Gold/ Manganese Oxide Nanowires Hybrid Electrode. *Electrochim. Acta* 2015, *163*, 107–115.

(217) Chen, A. Y.; Wang, J. W.; Wang, Y.; Jia, Y. Q.; Gu, J. F.; Xie, X. F.; Pan, D. Effects of Pore Size and Residual Ag on Electrocatalytic Properties of Nanoporous Gold Films Prepared by Pulse Electrochemical Dealloying. *Electrochim. Acta* **2015**, *153*, 552–558.

(218) Graf, M.; Jalas, D.; Weissmüller, J.; Petrov, A. Y.; Eich, M. Surface-to-Volume Ratio Drives Photoelelectron Injection from Nanoscale Gold into Electrolyte. *ACS Catal.* **2019**, *9*, 3366–3374.

(219) Chauvin, A.; Stephant, N.; Du, K.; Ding, J.; Wathuthanthri, I.; Choi, C.-H.; Tessier, P.-Y.; El Mel, A.-A. Large-Scale Fabrication of Porous Gold Nanowires via Laser Interference Lithography and Dealloying of Gold–Silver Nano-Alloys. *Micromachines* **2017**, *8*, 168.

(220) Arnob, M. M. P.; Zhao, F.; Li, J.; Shih, W.-C. EBL-Based Fabrication and Different Modeling Approaches for Nanoporous Gold Nanodisks. ACS Photonics 2017, 4, 1870–1878.

(221) Dietrich, L.; Oppermann, H.; Lopper, C.; Mackowiak, P. Fabrication and Characterization of Nanoporous Gold (NPG) Interconnects for Wafer Level Packaging. In 2022 IEEE 72nd Electronic Components and Technology Conference (ECTC); IEEE, 2022; pp 873–882.

(222) Ji, C.; Searson, P. C. Synthesis and Characterization of Nanoporous Gold Nanowires. J. Phys. Chem. B 2003, 107, 4494–4499.

(223) Liu, Z.; Searson, P. C. Single Nanoporous Gold Nanowire Sensors. J. Phys. Chem. B 2006, 110, 4318-4322.

(224) Burr, L.; Schubert, I.; Sigle, W.; Trautmann, C.; Toimil-Molares, M. E. Surface Enrichment in Au–Ag Alloy Nanowires and Investigation of the Dealloying Process. J. Phys. Chem. C 2015, 119, 20949–20956.

(225) Qiu, H.-J.; Gao, J. J.; Chiang, F.-K.; Wen, Y. R.; Yao, A. Y.; Du, P.; Fang, G.; Wang, J. Q.; Liu, X. J. A General and Scalable Approach to Produce Nanoporous Alloy Nanowires with Rugged Ligaments for Enhanced Electrocatalysis. *J. Mater. Chem. A* **2018**, *6*, 12541–12550. (226) Sukeri, A.; Hernández Saravia, L. P.; Bertotti, M. A Facile Electrochemical Approach to Fabricate a Nanoporous Gold Film Electrode and its Electrocatalytic Activity towards Dissolved Oxygen Reduction. *Phys. Chem. Chem. Phys.* **2015**, *17*, 28510–28514.

(227) Sukeri, A.; Lima, A. S.; Bertotti, M. Development of Non-Enzymatic and Highly Selective Hydrogen Peroxide Sensor Based on Nanoporous Gold Prepared by a Simple Unusual Electrochemical Approach. *Microchem. J.* **2017**, *133*, 149–154.

(228) Xu, J.; Kou, T.; Zhang, Z. Anodization Strategy to Fabricate Nanoporous Gold for High-Sensitivity Detection of p-Nitrophenol. *CrystEngComm* **2013**, *15*, 7856.

(229) Xu, S.; Yao, Y.; Wang, P.; Yang, Y.; Xia, Y.; Liu, J.; Li, Z.; Huang, W. Anodic Fabrication of Nanoporous Gold Films from Pure Gold in Oxalic Acid Solution and Their Applications in Electrocatalysis and SERS. *Int. J. Electrochem. Sci.* **2013**, *8*, 1863–1870.

(230) Sharma, M. K.; Aggarwal, S. K. Electrochemical Reduction of U(VI) in H_2SO_4 at Gold Nanoporous Film Electrode. *Radiochim. Acta* **2013**, *101*, 253–258.

(231) Jiang, J.; Wang, X. Fabrication of High-Surface Nanoporous Gold Microelectrode. *Electrochem. Commun.* **2012**, *20*, 157–159.

(232) Lu, L.; Wu, G.; Dong, Y.; Wang, J.; Bai, G. Green and Facile Preparation of Self-Supporting Nanoporous Gold Electrode and Effect of Ionic Liquids on its Electrocatalytic Oxidation toward Glucose. *J. Porous Mater.* **2016**, *23*, 671–678.

(233) Jiang, J. Fabrication of Uniform Nanoparticulate Gold through Potential-Modulated Electrochemical Deposition and Dissolution of Silver in Ionic Liquids. J. Electrochem. Soc. **2019**, 166, E521–E525.

(234) Heim, M.; Reculusa, S.; Ravaine, S.; Kuhn, A. Engineering of Complex Macroporous Materials Through Controlled Electrodeposition in Colloidal Superstructures. *Adv. Funct. Mater.* **2012**, 22, 538–545.

(235) Erlebacher, J.; Snyder, J. Dealloyed Nanoporous Metals for PEM Fuel Cell Catalysis. *ECS Trans.* 2009, 25, 603–612.

(236) Peng, Z.; Freunberger, S. A.; Chen, Y.; Bruce, P. G. A Reversible and Higher-Rate Li-O₂ Battery. *Science* **2012**, 337, 563–566.

(237) Wang, M.; Meng, A. C.; Fu, J.; Foucher, A. C.; Serra-Maia, R.; Stach, E. A.; Detsi, E.; Pikul, J. H. Surface Facet Engineering in Nanoporous Gold for Low-Loading Catalysts in Aluminum-Air Batteries. *ACS Appl. Mater. Interfaces* **2021**, *13*, 13097–13105.

(238) Corsi, J. S.; Welborn, S. S.; Stach, E. A.; Detsi, E. Insights into the Degradation Mechanism of Nanoporous Alloy-Type Li-Ion Battery Anodes. *ACS Energy Lett.* **2021**, *6*, 1749–1756.

(239) Behnken, J.; Yu, M.; Deng, X.; Tüysüz, H.; Harms, C.; Dyck, A.; Wittstock, G. Oxygen Reduction Reaction Activity of Mesostructured Cobalt-Based Metal Oxides Studied with the Cavity-Microelectrode Technique. *ChemElectroChem.* **2019**, *6*, 3460–3467.

(240) Wang, H.; Zhu, L.; Yang, L.; Liao, F.; Sheng, M.; Jiang, B.; Shao, M. Prominent Electrocatalytic Methanol Oxidation from Cauli-Flower Shape Gold with High-Index Facets. *Mater. Chem. Phys.* **2017**, *186*, 301–304.

(241) Li, G. G.; Lin, Y.; Wang, H. Residual Silver Remarkably Enhances Electrocatalytic Activity and Durability of Dealloyed Gold Nanosponge Particles. *Nano Lett.* **2016**, *16*, 7248–7253.

(242) Wang, X.; Frenzel, J.; Wang, W.; Ji, H.; Qi, Z.; Zhang, Z.; Eggeler, G. Length-Scale Modulated and Electrocatalytic Activity Enhanced Nanoporous Gold by Doping. *J. Phys. Chem. C* 2011, *115*, 4456–4465.

(243) Oezaslan, M.; Heggen, M.; Strasser, P. Size-Dependent Morphology of Dealloyed Bimetallic Catalysts: Linking the Nano to the Macro Scale. J. Am. Chem. Soc. **2012**, 134, 514–524.

(244) Mikhlin, Y. L.; Vishnyakova, E. A.; Romanchenko, A. S.; Saikova, S. V.; Likhatski, M. N.; Larichev, Y. V.; Tuzikov, F. V.; Zaikovskii, V. I.; Zharkov, S. M. Oxidation of Ag Nanoparticles in Aqueous Media: Effect of Particle Size and Capping. *Appl. Surf. Sci.* **2014**, 297, 75–83.

(245) McCue, I.; Snyder, J.; Li, X.; Chen, Q.; Sieradzki, K.; Erlebacher, J. Apparent Inverse Gibbs-Thomson Effect in Dealloyed Nanoporous Nanoparticles. *Phys. Rev. Lett.* **2012**, *108*, No. 225503.

(246) Mahr, C.; Dworzak, A.; Schowalter, M.; Oezaslan, M.; Rosenauer, A. Quantitative 3D Characterization of Nanoporous Gold Nanoparticles by Transmission Electron Microscopy. *Microsc. Microanal.* **2021**, *27*, 678–686.

(247) Tallarek, U.; Hlushkou, D.; Rybka, J.; Höltzel, A. Multiscale Simulation of Diffusion in Porous Media: From Interfacial Dynamics to Hierarchical Porosity. *J. Phys. Chem. C* **2019**, *123*, 15099–15112.

(248) Courtois, J.; Szumski, M.; Georgsson, F.; Irgum, K. Assessing the Macroporous Structure of Monolithic Columns by Transmission Electron Microscopy. *Anal. Chem.* **2007**, *79*, 335–344.

(249) Lu, B.; Torquato, S. Chord-length and Free-path Distribution Functions for Many-body Systems. J. Chem. Phys. **1993**, 98, 6472–6482.

(250) Torquato, S. Random Heterogeneous Materials: Microstructure and Macroscopic Properties; Interdisciplinary Applied Mathematics; Springer: New York, 2002; Vol. 16.

(251) Hall, A. S.; Yoon, Y.; Wuttig, A.; Surendranath, Y. Mesostructure-Induced Selectivity in CO_2 Reduction Catalysis. J. Am. Chem. Soc. 2015, 137, 14834–14837.

(252) Szamocki, R.; Reculusa, S.; Ravaine, S.; Bartlett, P. N.; Kuhn, A.; Hempelmann, R. Tailored Mesostructuring and Biofunctionalization of Gold for Increased Electroactivity. *Angew. Chem., Int. Ed.* **2006**, 45, 1317–1321.

(253) Du Toit, H.; Di Lorenzo, M. Electrodeposited Highly Porous Gold Microelectrodes for the Direct Electrocatalytic Oxidation of Aqueous Glucose. *Sens. Actuators, B* **2014**, *192*, 725–729.

(254) He, F.; Qiao, Z.; Qin, X.; Chao, L.; Tan, Y.; Xie, Q.; Yao, S. Dynamic Gas Bubble Template Electrodeposition Mechanisms and Amperometric Glucose Sensing Performance of Three Kinds of Three-Dimensional Honeycomb-Like Porous Nano-Golds. *Sens. Actuators, B* **2019**, *296*, No. 126679.

(255) Hernández-Saravia, L. P.; Sukeri, A.; Bertotti, M. Fabrication of Nanoporous Gold-Islands via Hydrogen Bubble Template: An Efficient Electrocatalyst for Oxygen Reduction and Hydrogen Evolution Reactions. *Int. J. Hydrog. Energy* **2019**, *44*, 15001–15008.

(256) Liu, J.; Cao, L.; Huang, W.; Li, Z. Preparation of AuPt Alloy Foam Films and their Superior Electrocatalytic Activity for the Oxidation of Formic Acid. *ACS Appl. Mater. Interfaces* **2011**, *3*, 3552– 3558.

(257) Leonardo, S.; Garibo, D.; Fernández-Tejedor, M.; O'Sullivan, C. K.; Campàs, M. Addressed Immobilization of Biofunctionalized Diatoms on Electrodes by Gold Electrodeposition. *Biofabrication* **2017**, *9*, 015027.

(258) Feng, J.-J.; Lv, Z.-Y.; Qin, S.-F.; Li, A.-Q.; Fei, Y.; Wang, A.-J. N-Methylimidazole-Assisted Electrodeposition of Au Porous Textile-Like Sheet Arrays and its Application to Electrocatalysis. *Electrochim. Acta* **2013**, *102*, 312–318.

(259) Du, R.; Joswig, J.-O.; Hübner, R.; Zhou, L.; Wei, W.; Hu, Y.; Eychmüller, A. Freeze-Thaw-Promoted Fabrication of Clean and Hierarchically Structured Noble-Metal Aerogels for Electrocatalysis and Photoelectrocatalysis. *Angew. Chem., Int. Ed.* **2020**, *59*, 8293– 8300.

(260) Herrmann, A.-K.; Formanek, P.; Borchardt, L.; Klose, M.; Giebeler, L.; Eckert, J.; Kaskel, S.; Gaponik, N.; Eychmüller, A. Multimetallic Aerogels by Template-Free Self-Assembly of Au, Ag, Pt, and Pd Nanoparticles. *Chem. Mater.* **2014**, *26*, 1074–1083.

(261) Xu, M.; Sui, Y.; Xiao, G.; Yang, X.; Wei, Y.; Zou, B. Kinetically Controlled Synthesis of Nanoporous Au and its Enhanced Electrocatalytic Activity for Glucose-Based Biofuel Cells. *Nanoscale* **2017**, *9*, 2514–2520.

(262) Christiansen, M. U.-B.; Seselj, N.; Engelbrekt, C.; Wagner, M.; Stappen, F. N.; Zhang, J. Chemically Controlled Interfacial Nanoparticle Assembly into Nanoporous Gold Films for Electrochemical Applications. J. Mater. Chem. A 2018, 6, 556–564.

(263) Yao, K.; Wang, N.; Li, Z.; Lu, W.; Wang, J. Ionic Liquid-Modulated Synthesis of Porous Worm-Like Gold with Strong SERS Response and Superior Catalytic Activities. *Nanomaterials* **2019**, *9*, 1772.

(264) Du, R.; Jin, X.; Hübner, R.; Fan, X.; Hu, Y.; Eychmüller, A. Engineering Self-Supported Noble Metal Foams Toward Electrocatalysis and Beyond. *Adv. Energy Mater.* **2020**, *10*, 1901945.

(265) Kafi, A. K. M.; Ahmadalinezhad, A.; Wang, J.; Thomas, D. F.; Chen, A. Direct Growth of Nanoporous Au and its Application in Electrochemical Biosensing. *Biosens. Bioelectron.* **2010**, *25*, 2458– 2463.

(266) Chen, L. Y.; Fujita, T.; Ding, Y.; Chen, M. W. A Three-Dimensional Gold-Decorated Nanoporous Copper Core-Shell Composite for Electrocatalysis and Nonenzymatic Biosensing. *Adv. Funct. Mater.* **2010**, *20*, 2279–2285.

(267) Son, J.; Cho, S.; Lee, C.; Lee, Y.; Shim, J. H. Spongelike Nanoporous Pd and Pd/Au Structures: Facile Synthesis and Enhanced Electrocatalytic Activity. *Langmuir* **2014**, *30*, 3579–3588. (268) Lilleodden, E. T.; Voorhees, P. W. On the Topological, Morphological, and Microstructural Characterization of Nanoporous Metals. *MRS Bull.* **2018**, *43*, 20–26.

(269) Richert, C.; Huber, N. A Review of Experimentally Informed Micromechanical Modeling of Nanoporous Metals: From Structural Descriptors to Predictive Structure-Property Relationships. *Materials* **2020**, *13*, 3307.

(270) Soyarslan, C.; Bargmann, S.; Pradas, M.; Weissmüller, J. 3D Stochastic Bicontinuous Microstructures: Generation. *Topology and Elasticity. Acta Mater.* **2018**, *149*, 326–340.

(271) Cahn, J. W. Phase Separation by Spinodal Decomposition in Isotropic Systems. J. Chem. Phys. **1965**, 42, 93–99.

(272) Mangipudi, K. R.; Epler, E.; Volkert, C. A. Topology-Dependent Scaling Laws for the Stiffness and Strength of Nanoporous Gold. *Acta Mater.* **2016**, *119*, 115–122.

(273) Farkas, D.; Crowson, D. A.; Corcoran, S. G. Geometric Relaxation of Nanoporous Metals: The Role of Surface Relaxation. *Scr. Mater.* **2007**, *56*, 919–922.

(274) Sun, X.-Y.; Xu, G.-K.; Li, X.; Feng, X.-Q.; Gao, H. Mechanical Properties and Scaling Laws of Nanoporous Gold. *J. Appl. Phys.* 2013, 113, 023505.

(275) Ngô, B.-N. D.; Stukowski, A.; Mameka, N.; Markmann, J.; Albe, K.; Weissmüller, J. Anomalous Compliance and Early Yielding of Nanoporous Gold. *Acta Mater.* **2015**, *93*, 144–155.

(276) Biener, J.; Hodge, A. M.; Hayes, J. R.; Volkert, C. A.; Zepeda-Ruiz, L. A.; Hamza, A. V.; Abraham, F. F. Size Effects on the Mechanical Behavior of Nanoporous Au. *Nano Lett.* **2006**, *6*, 2379– 2382.

(277) Lang, X. Y.; Qian, L. H.; Guan, P. F.; Zi, J.; Chen, M. W. Localized Surface Plasmon Resonance of Nanoporous Gold. *Appl. Phys. Lett.* **2011**, *98*, 093701.

(278) Detsi, E.; Salverda, M.; Onck, P. R.; De Hosson, J. T. M. On the Localized Surface Plasmon Resonance Modes in Nanoporous Gold Films. J. Appl. Phys. 2014, 115, 044308.

(279) Wahl, P.; Traussnig, T.; Landgraf, S.; Jin, H.-J.; Weissmüller, J.; Wuerschum, R. Adsorption-Driven Tuning of the Electrical Resistance of Nanoporous Gold. *J. Appl. Phys.* **2010**, *108*, 073706.

(280) Hakamada, M.; Kato, N.; Miyazawa, N.; Mabuchi, M. Water-Adsorption Effect on Electrical Resistivity of Nanoporous Gold. *Scr. Mater.* **2016**, *123*, 30–33.

(281) van Petegem, S.; Brandstetter, S.; Maass, R.; Hodge, A. M.; El-Dasher, B. S.; Biener, J.; Schmitt, B.; Borca, C.; van Swygenhoven, H. On the Microstructure of Nanoporous Gold: An X-ray Diffraction Study. *Nano Lett.* **2009**, *9*, 1158–1163.

(282) Chauvin, A.; Delacôte, C.; Molina-Luna, L.; Duerrschnabel, M.; Boujtita, M.; Thiry, D.; Du, K.; Ding, J.; Choi, C.-H.; Tessier, P.-Y.; El Mel, A.-A. Planar Arrays of Nanoporous Gold Nanowires: When Electrochemical Dealloying Meets Nanopatterning. *ACS Appl. Mater. Interfaces* **2016**, *8*, 6611–6620.

(283) Gwak, E.-J.; Kim, J.-Y. Weakened Flexural Strength of Nanocrystalline Nanoporous Gold by Grain Refinement. *Nano Lett.* **2016**, *16*, 2497–2502.

(284) Lin, B.; Kong, L.; Hodgson, P. D.; Mudie, S.; Hawley, A.; Dumée, L. F. Controlled Porosity and Pore Size of Nano-porous Gold by Thermally Assisted Chemical Dealloying – A SAXS Study. *RSC Adv.* **2017**, *7*, 10821–10830.

(285) Klug, H. P.; Alexander, L. E. X-Ray Diffraction Procedures; Wiley, New York, 1974.

(286) Markmann, J.; Yamakov, V.; Weissmüller, J. Validating Grain Size Analysis from X-ray Line Broadening: A virtual experiment. *Scr. Mater.* **2008**, *59*, 15–18.

(287) Fujita, T.; Qian, L.-H.; Inoke, K.; Erlebacher, J.; Chen, M.-W. Three-Dimensional Morphology of Nanoporous Gold. *Appl. Phys. Lett.* **2008**, *92*, 251902.

(288) Chen-Wiegart, Y.-C. K.; DeMike, R.; Erdonmez, C.; Thornton, K.; Barnett, S. A.; Wang, J. Tortuosity Characterization of 3D Microstructure at Nano-scale for Energy Storage and Conversion Materials. J. Power Sources **2014**, 249, 349–356. (289) Jeon, H.; Kang, N.-R.; Gwak, E.-J.; Jang, J.-i.; Han, H. N.; Hwang, J. Y.; Lee, S.; Kim, J.-Y. Self-similarity in the Structure of Coarsened Nanoporous Gold. *Scr. Mater.* **201**7, *137*, 46–49.

(290) Richert, C.; Huber, N. Skeletonization, Geometrical Analysis, and Finite Element Modeling of Nanoporous Gold Based on 3D Tomography Data. *Metals* **2018**, *8*, 282.

(291) Stuckner, J.; Frei, K.; McCue, I.; Demkowicz, M. J.; Murayama, M. AQUAMI: An Open Source Python package and GUI for the Automatic Quantitative Analysis of Morphologically Complex Multiphase Materials. *Comput. Mater. Sci.* **201**7, *139*, 320– 329.

(292) Richert, C.; Odermatt, A.; Huber, N. Computation of Thickness and Mechanical Properties of Interconnected Structures: Accuracy, Deviations, and Approaches for Correction. *Front. Mater.* **2019**, *6*, 327.

(293) Krill, C. E.; Birringer, R. Estimating Grain-Size Distributions in Nanocrystalline Materials from X-ray Diffraction Profile Analysis. *Philos. Mag. A* **1998**, *77*, 621–640.

(294) Fujita, T.; Chen, M. W. Characteristic Length Scale of Bicontinuous Nanoporous Structure by Fast Fourier Transform. *Jpn. J. Appl. Phys.* **2008**, *47*, 1161–1163.

(295) Guinier, A. X-ray Diffraction in Crystals, Imperfect Crystals, and Amorphous Bodies; Dover: New York, 1994.

(296) Michels, A.; Weissmüller, J. Magnetic-Field-Dependent Small-Angle Neutron Scattering on Random Anisotropy Ferromagnets. *Rep. Prog. Phys.* **2008**, *71*, 066501.

(297) Welborn, S. S.; Detsi, E. Small-angle X-ray Sattering of Nanoporous Materials. *Nanoscale Horiz.* **2020**, *5*, 12.

(298) Biener, J.; Wittstock, A.; Zepeda-Ruiz, L. A.; Biener, M. M.; Zielasek, V.; Kramer, D.; Viswanath, R. N.; Weissmüller, J.; Bäumer, M.; Hamza, A. V. Surface-Chemistry-Driven Actuation in Nanoporous Gold. *Nat. Mater.* **2009**, *8*, 47–51.

(299) Detsi, E.; Onck, P. R.; De Hosson, J. T. M. Electrochromic Artificial Muscles Based on Nanoporous Metal-Polymer Composites. *Appl. Phys. Lett.* **2013**, *103*, 193101.

(300) Ye, X.-L.; Jin, H.-J. Sealing-free Fast-response Paraffin/ Nanoporous Gold Hybrid Actuator. *Nanotechnology* **2017**, *28*, 385501.

(301) Han, Z.; Qi, Z.; Wei, Q.; Deng, Q.; Wang, K. The Mechanical Effect of MnO₂ Layers on Electrochemical Actuation Performance of Nanoporous Gold. *Nanomaterials* **2020**, *10*, 2056.

(302) Stenner, C.; Shao, L.-H.; Mameka, N.; Weissmüller, J. Piezoelectric Gold: Charge-Load Response in a Metal-Based Hybrid Nanomaterial. *Adv. Funct. Mater.* **2016**, *26*, 5174–5181.

(303) Mameka, N.; Markmann, J.; Weissmüller, J. On the Impact of Capillarity for Strength at the Nanoscale. *Nat. Commun.* **2017**, *8*, 1976.

(304) Weissmüller, J.; Cahn, J. W. Mean Sresses in Microstructures due to Interface Stresses: A Generalization of a Capillary Equation for Solids. *Acta Mater.* **1997**, *45*, 1899–1906.

(305) Trasatti, S.; Petrii, O. A. Real Surface Area Measurements in Electrochemistry. *Pure Appl. Chem.* **1991**, *63*, 711–734.

(306) Rouya, E.; Cattarin, S.; Reed, M. L.; Kelly, R. G.; Zangari, G. Electrochemical Characterization of the Surface Area of Nanoporous Gold Films. *J. Electrochem. Soc.* **2012**, *159*, K97–K102.

(307) Mitchri, A. A.; Pshchenichnikov, A. G.; Burshtein, R. K. Determination of the True Surface of Smooth Gold Electrodes. *Elektrokhimiya* **1972**, *8*, 364–366.

(308) Hamelin, A. Cyclic Voltammetry at Gold Single-Crystal Surfaces. Part 1. Behaviour at Low-Index Faces. *J. Electroanal. Chem.* **1996**, 407, 1–11.

(309) Schneeweiss, M. A.; Kolb, D. M. Oxide Formation on Au(111): An In Situ STM Study. *Solid State Ionics* **1997**, *94*, 171–179.

(310) Lakshmanan, C.; Viswanath, R. N.; Polaki, S. R.; Rajaraman, R.; Dash, S.; Tyagi, A. K. Surface Area of Nanoporous Gold: Effect on Temperature. *Electrochim. Acta* **2015**, *182*, 565–572.

6784

(311) Mameka, N. Surface-Controlled Mechanical Properties of Bulk Nanoporous Gold. Ph.D. Thesis. Hamburg University of Technology, Hamburg, 2016.

(312) Tan, Y. H.; Davis, J. A.; Fujikawa, K.; Ganesh, N. V.; Demchenko, A. V.; Stine, K. J. Surface Area and Pore Size Characteristics of Nanoporous Gold Subjected to Thermal, Mechanical, or Surface Modification Studied using Gas Adsorption Isotherms, Cyclic Voltammetry, Thermogravimetric Analysis, and Scanning Electron Microscopy. J. Mater. Chem. 2012, 22, 6733–6745.

(313) Detsi, E.; De Jong, E.; Zinchenko, A.; Vuković, Z.; Vuković, I.; Punzhin, S.; Loos, K.; ten Brinke, G.; De Raedt, H. A.; Onck, P. R.; De Hosson, J. T. M. On the Specific Surface Area of Nanoporous Materials. *Acta Mater.* **2011**, *59*, 7488–7497.

(314) Snyder, J. D.; Erlebacher, J. D. Electrochemical Measurement of the Surface Alloying Kinetics of Underpotentially Deposited Ag on Au(111). *Langmuir* **2009**, *25*, 9596–9604.

(315) Briot, N. J.; Kennerknecht, T.; Eberl, C.; Balk, T. J. Mechanical Properties of Bulk Single Crystalline Nanoporous Gold Investigated by Millimetre-scale Tension and Compression Testing. *Philos. Mag.* **2014**, *94*, 847–866.

(316) Renner, F. U.; Gründer, Y.; Lyman, P. F.; Zegenhagen, J. Insitu X-ray Diffraction Study of the Initial Dealloying of $Cu_3Au(001)$ and $Cu_{0.83}Pd_{0.17}(001)$. *Thin Solid Films* **2007**, *515*, 5574–5580.

(317) Zhang, Z.; Wang, Y.; Qi, Z.; Zhang, W.; Qin, J.; Frenzel, J. Generalized Fabrication of Nanoporous Metals (Au, Pd, Pt, Ag, and Cu) through Chemical Dealloying. *J. Phys. Chem. C* 2009, *113*, 12629–12636.

(318) Hillert, M. Phase Equilibria, Phase Diagrams and Phase Transformations: Their Thermodynamic Basis; Cambridge University Press, 2007.

(319) Scaglione, F.; Celegato, F.; Rizzi, P.; Battezzati, L. A Comparison of De-alloying Crystalline and Amorphous Multicomponent Au Alloys. *Intermetallics* **2015**, *66*, 82–87.

(320) Xue, Y.; Scaglione, F.; Rizzi, P.; Battezzati, L. High Performance SERS on Nanoporous Gold Substrates Synthesized by Chemical De-alloying a Au-based Metallic Glass. *Appl. Surf. Sci.* 2017, 426, 1113–1120.

(321) Qi, Z.; Geng, H.; Wang, X.; Zhao, C.; Ji, H.; Zhang, C.; Xu, J.; Zhang, Z. Novel Nanocrystalline PdNi Alloy Catalyst for Methanol and Ethanol Electro-oxidation in Alkaline Media. *J. Power Sources* **2011**, *196*, 5823–5828.

(322) Haensch, M.; Balboa, L.; Graf, M.; Silva Olaya, A. R.; Weissmüller, J.; Wittstock, G. Mass Transport in Porous Electrodes Studied by Scanning Electrochemical Microscopy – Example of Nanoporous Gold. *ChemElectroChem.* **2019**, *6*, 3160–3166.

(323) Hodge, A. M.; Doucette, R. T.; Biener, M. M.; Biener, J.; Cervantes, O.; Hamza, A. V. Ag Effects on the Elastic Modulus Values of Nanoporous Au Foams. *J. Mater. Res.* **2009**, *24*, 1600–1606.

(324) Liu, L.-Z.; Jin, H.-J. Scaling Equation for the Elastic Modulus of Nanoporous Gold with "Fixed" Network Connectivity. *Appl. Phys. Lett.* **2017**, *110*, 211902.

(325) Liu, P.; Chen, Q.; Ito, Y.; Han, J.; Chu, S.; Wang, X.; Reddy, K. M.; Song, S.; Hirata, A.; Chen, M. Dealloying Kinetics of AgAu Nanoparticles by In Situ Liquid-Cell Scanning Transmission Electron Microscopy. *Nano Lett.* **2020**, *20*, 1944–1951.

(326) Henkelmann, G.; Waldow, D.; Liu, M.; Lührs, L.; Li, Y.; Weissmüller, J. Self-Detachment and Subsurface Densification of Dealloyed Nanoporous Thin Films. *Nano Lett.* **2022**, *22*, 6787.

(327) Mameka, N.; Wang, K.; Markmann, J.; Lilleodden, E. T.; Weissmüller, J. Nanoporous Gold—Testing Macro-scale Samples to Probe Small-scale Mechanical Behavior. *Mater. Res. Lett.* **2016**, *4*, 27– 36.

(328) Erlebacher, J. Mechanism of Coarsening and Bubble Formation in High-Genus Nanoporous Metals. *Phys. Rev. Lett.* **2011**, *106*, No. 225504.

(329) Ziehmer, M.; Hu, K.; Wang, K.; Lilleodden, E. T. A Principle Curvatures Analysis of the Isothermal Evolution of Nanoporous Gold: Quantifying the Characteristic Length-Scales. *Acta Mater.* **2016**, *120*, 24–31. (330) Ghanbarian, B.; Hunt, A. G.; Ewing, R. P.; Sahimi, M. Tortuosity in Porous Media: A Critical Review. *Soil Sci. Soc. Am. J.* **2013**, 77, 1461–1477.

(331) Tjaden, B.; Brett, D. J. L.; Shearing, P. R. Tortuosity in Electrochemical Devices: A Review of Calculation Approaches. *Int. Mater. Rev.* 2018, 63, 47–67.

(332) Xue, Y.; Markmann, J.; Duan, H.; Weissmüller, J.; Huber, P. Switchable Imbibition in Nanoporous Gold. *Nat. Commun.* **2014**, *5*, 4237.

(333) Gruener, S.; Huber, P. Imbibition in Mesoporous Silica: Rheological Concepts and Experiments on Water and a Liquid Crystal. *J. Phys.: Condens. Matter* **2011**, *23*, 184109.

(334) Wiltzius, P.; Bates, F. S.; Dierker, S. B.; Wignall, G. D. Structure of Porous Vycor Glass. *Phys. Rev. A* **1987**, *36*, 2991–2994. (335) Levitz, P.; Ehret, G.; Sinha, S. K.; Drake, J. M. Porous Vycor Glass: The Microstructure as Probed by Electron Microscopy, Direct Energy Transfer, Small-Angle Scattering, and Molecular Adsorption. *J. Chem. Phys.* **1991**, *95*, 6151–6161.

(336) Li, J.-C.; Ross, D. K. Dynamical Scaling for Spinodal Decomposition - A Small-angle Neutron Scattering Study of Porous Vycor Glass with Fractal Properties. *J. Phys.: Condens. Matter* **1994**, *6*, 351–362.

(337) Johnson, C. A. Generalization of the Gibbs-Thomson Equation. Surf. Sci. 1965, 3, 429-444.

(338) Fujita, T.; Guan, P.; McKenna, K.; Lang, X.; Hirata, A.; Zhang, L.; Tokunaga, T.; Arai, S.; Yamamoto, Y.; Tanaka, N.; Ishikawa, Y.; Asao, N.; Yamamoto, Y.; Erlebacher, J.; Chen, M. Atomic Origins of the High Catalytic Activity of Nanoporous Gold. *Nat. Mater.* **2012**, *11*, 775–780.

(339) Zeis, R.; Lei, T.; Sieradzki, K.; Snyder, J.; Erlebacher, J. Catalytic Reduction of Oxygen and Hydrogen Peroxide by Nanoporous Gold. *J. Catal.* **2008**, *253*, 132–138.

(340) Wang, Z.; Liu, P.; Han, J.; Cheng, C.; Ning, S.; Hirata, A.; Fujita, T.; Chen, M. Engineering the Internal Surfaces of Three-Dimensional Nanoporous Catalysts by Surfactant-Modified Dealloying. *Nat. Commun.* **2017**, *8*, 1066.

(341) Ibach, H. Physics of Surfaces and Interfaces; Springer: Berlin, 2006.

(342) Heyraud, J. C.; Metois, J. J. Equilibrium Shape of Gold Crystallites on a Graphite Cleavage Surface: Surface Eergies and Interfacial Energy. *Acta Metall.* **1980**, *28*, 1789–1797.

(343) Sadan, H.; Kaplan, W. D. Au-Sapphire (0001) Solid-solid Interfacial Energy. J. Mater. Sci. 2006, 41, 5099-5107.

(344) Chen, Y.-c. K.; Chu, Y. S.; Yi, J.; McNulty, I.; Shen, Q.; Voorhees, P. W.; Dunand, D. C. Morphological and Topological Analysis of Coarsened Nanoporous Gold by X-Ray Nanotomography. *Appl. Phys. Lett.* **2010**, *96*, 043122.

(345) Solliard, C.; Flueli, M. Surface Stress and Size Effect on the Lattice Parameter in Small Particles of Gold and Platinum. *Surf. Sci.* **1985**, *156*, 487–494.

(346) Lamber, R.; Wetjen, S.; Jaeger, N. I. Size Dependence of the Lattice Parameter of Small Palladium Particles. *Phys. Rev. B* **1995**, *51*, 10968–10971.

(347) Schofield, E. J.; Ingham, B.; Turnbull, A.; Toney, M. F.; Ryan, M. P. Strain Development in Nanoporous Metallic Foils Formed by Dealloying. *Appl. Phys. Lett.* **2008**, *92*, 043118.

(348) Detsi, E.; Chen, Z. G.; Vellinga, W. P.; Onck, P. R.; De Hosson, J. T. M. Reversible Strain by Physisorption in Nanoporous Gold. *Appl. Phys. Lett.* **2011**, *99*, 083104.

(349) Zhang, S.-M.; Jin, H.-J. Multilayer-structured Gold/Nanoporous Gold Composite for High Performance Linear Actuation. *Appl. Phys. Lett.* **2014**, *104*, 101905.

(350) Weissmüller, J. Mechanochemistry Breaks with Expectations. *Nat. Catal.* **2018**, *1*, 238–239.

(351) Mavrikakis, M.; Hammer, B.; Norskov, J. K. Effect of Strain on the Reactivity of Metal Surfaces. *Phys. Rev. Lett.* **1998**, *81*, 2819–2822. (352) Deng, Q.; Smetanin, M.; Weissmüller, J. Mechanical Modulation of Reaction Rates in Electrocatalysis. *J. Catal.* **2014**, 309, 351–361. (353) Mahr, C.; Müller-Caspary, K.; Graf, M.; Lackmann, A.; Grieb, T.; Schowalter, M.; Krause, F. F.; Mehrtens, T.; Wittstock, A.; Weissmüller, J.; Rosenauer, A. Measurement of Local Crystal Lattice Strain Variations in Dealloyed Nanoporous Gold. *Mater. Res. Lett.* **2018**, *6*, 84–92.

(354) Weissmüller, J.; Duan, H. L.; Farkas, D. Deformation of Solids with Nanoscale Pores by the Action of Capillary Forces. *Acta Mater.* **2010**, *58*, 1–13.

(355) Bohnen, K. P.; Ho, K. M. Structure and Dynamics at Metal Surfaces. *Surf. Sci. Rep.* **1993**, *19*, 99–120.

(356) Somorjai, G. A.; van Hove, M. A. Adsorbate-Induced Restructuring of Surfaces. *Prog. Surf. Sci.* **1989**, *30*, 201–231.

(357) Deng, Z.; Zhu, Y.; Yuan, W.; Xi, W. Strain Evolution in Nanoporous Gold during Catalytic CH₄ Pyrolysis by In Situ Gas-Phase Transmission Electron Microscopy. *Scr. Mater.* **2021**, *204*, No. 114146.

(358) Beets, N.; Farkas, D.; Albe, K. The Mechanical Response of Nanoporous Gold and Silver Foams with Varying Composition and Surface Segregation. *Acta Mater.* **2021**, *203*, 116445.

(359) Hoppe, S.; Müller, S. A First Principles Study on the Electronic Origins of Silver Segregation at the Ag-Au (111) Surface. *J. Appl. Phys.* **2017**, *122*, 235303.

(360) Fujita, T.; Tokunaga, T.; Zhang, L.; Li, D.; Chen, L.; Arai, S.; Yamamoto, Y.; Hirata, A.; Tanaka, N.; Ding, Y.; Chen, M. Atomic Observation of Catalysis-Induced Nanopore Coarsening of Nanoporous Gold. *Nano Lett.* **2014**, *14*, 1172–1177.

(361) Mahr, C.; Tapia Burgos, J. A.; Schowalter, M.; Wittstock, A.; Rosenauer, A. Investigation of the Dealloying Front in Partially Corroded Alloys. *Mater. Res. Lett.* **2022**, *10*, 824–831.

(362) Dick, K.; Dhanasekaran, T.; Zhang, Z.; Meisel, D. Size-Dependent Melting of Silica-Encapsulated Gold Nanoparticles. *J. Am. Chem. Soc.* **2002**, *124*, 2312–2317.

(363) Ebrahimy, A. F.; Langelier, B.; Newman, R. C. Atom Probe Tomography of Nanoporous Gold Formed by Dealloying Lean Noble Alloys. *Mater. Today Commun.* **2020**, *25*, No. 101371.

(364) Kolb, D. M. Reconstruction Phenomena at Metal-Electrolyte Interfaces. *Prog. Surf. Sci.* **1996**, *51*, 109–173.

(365) Hoppe, S.; Li, Y.; Moskaleva, L. V.; Mueller, S. How Silver Segregation Stabilizes 1D Surface Gold Oxide: A Cluster Expansion Study Combined with Ab Initio MD Simulations. *Phys. Chem. Chem. Phys.* **2017**, *19*, 14845–14853.

(366) Schiller, U. D.; Wang, F. Multiscale simulation of transport phenomena in porous media: from toy models to materials models. *MRS Commun.* **2018**, *8*, 358–371.

(367) Tu, V.; Larsson, F.; Runesson, K.; Jänicke, R. Variationally Consistent Homogenization of Electrochemical Ion Transport in a Porous Structural Battery Electrolyte. *Eur. J. Mechan. A Solids* **2023**, 98, No. 104901.

(368) Zheng, Q.; Tian, Y.; Shen, X.; Sokolowski-Tinten, K.; Li, R. K.; Chen, Z.; Mo, M. Z.; Wang, Z. L.; Liu, P.; Fujita, T.; Weathersby, S. P.; Yang, J.; Wang, X. J.; Chen, M. W. Fast Attenuation of High-frequency Acoustic Waves in Bicontinuous Nanoporous Gold. *Appl. Phys. Lett.* **2021**, *119*, 063101.

(369) Ramesh, R.; Niauzorau, S.; Ni, Q.; Azeredo, B. P.; Wang, L. Optical Characterization and Modeling of Nanoporous Gold Absorbers Fabricated by Thin-film Dealloying. *Nanotechnology* **2020**, *31*, 405706.

(370) Bargmann, S.; Klusemann, B.; Markmann, J.; Schnabel, J. E.; Schneider, K.; Soyarslan, C.; Wilmers, J. Generation of 3D Representative Volume Elements for Heterogeneous Materials: A Review. *Prog. Mater. Sci.* **2018**, *96*, 322–384.

(371) Saane, S. S. R.; Mangipudi, K. R.; Loos, K. U.; De Hosson, J. T. M.; Onck, P. R. Multiscale Modeling of Charge-induced Deformation of Nanoporous Gold Structures. *J. Mechan. Phys. Solids* **2014**, *66*, 1–15.

(372) Gibson, L. J.; Ashby, M. F. The Mechanics of Threedimensional Cellular Materials. *Proc. R. Soc. London, Ser. A* **1982**, 382, 43-59. (373) Gibson, L. J.; Ashby, M. F. Cellular Solids; Pergamon Press: Oxford, UK, 1999.

(374) Huber, N.; Viswanath, R. N.; Mameka, N.; Markmann, J.; Weißmüller, J. Scaling Laws of Nanoporous Metals under Uniaxial Compression. *Acta Mater.* **2014**, *67*, 252–265.

(375) Huber, N. Connections Between Topology and Macroscopic Mechanical Properties of Three-Dimensional Open-Pore Materials. *Front. Mater.* **2018**, *5*, 5801.

(376) Rodriguez-Nieva, J. F.; Ruestes, C. J.; Tang, Y.; Bringa, E. M. Atomistic Simulation of the Mechanical Properties of Nanoporous Gold. *Acta Mater.* **2014**, *80*, 67–76.

(377) Farkas, D.; Caro, A.; Bringa, E.; Crowson, D. Mechanical Response of Nanoporous Gold. *Acta Mater.* **2013**, *61*, 3249–3256.

(378) Kwon, Y.; Thornton, K.; Voorhees, P. W. The Topology and Morphology of Bicontinuous Interfaces during Coarsening. *Europhys. Lett.* **2009**, *86*, 46005.

(379) Xu, B.; Zhou, L.; Madix, R. J.; Friend, C. M. Highly Selective Acylation of Dimethylamine Mediated by Oxygen Atoms on Metallic Gold Surfaces. *Angew. Chem., Int. Ed.* **2010**, *49*, 394–398.

(380) Déronzier, T.; Morfin, F.; Lomello, M.; Rousset, J.-L. Catalysis on Nanoporous Gold–Silver Systems: Synergistic Effects Toward Oxidation Reactions and Influence of the Surface Composition. *J. Catal.* **2014**, *311*, 221–229.

(381) Shi, J.; Mahr, C.; Murshed, M. M.; Zielasek, V.; Rosenauer, A.; Gesing, T. M.; Bäumer, M.; Wittstock, A. A Versatile Sol-Gel Coating for Mixed Oxides on Nanoporous Gold and their Application in the Water Gas Shift Reaction. *Catal. Sci. Technol.* **2016**, *6*, 5311–5319.

(382) Xu, B.; Liu, X.; Haubrich, J.; Madix, R. J.; Friend, C. M. Selectivity Control in Gold-Mediated Esterification of Methanol. *Angew. Chem., Int. Ed.* **2009**, *48*, 4206–4209.

(383) Xu, B.; Madix, R. J.; Friend, C. M. Predicting Gold-Mediated Catalytic Oxidative-Coupling Reactions from Single Crystal Studies. *Acc. Chem. Res.* **2014**, *47*, 761–772.

(384) Xu, B.; Friend, C. M. Oxidative Coupling of Alcohols on Gold: Insights from Experiments and Theory. *Faraday Discuss.* **2011**, *152*, 307–320 discussion pp 393–413.

(385) Hoppe, S.; Moskaleva, L. V. Origins of the High Reactivity of Au Nanostructures Deduced from the Structure and Properties of Model Surfaces. In *Noble and Precious Metals—Properties, Nanoscale Effects and Applications*; Bristow, A. D., Seehra, M. S., Eds.; IntechOpen, 2018; pp 65–98.

(386) Pan, M.; Gong, J.; Dong, G.; Mullins, C. B. Model Studies with Gold: A Versatile Oxidation and Hydrogenation Catalyst. *Acc. Chem. Res.* **2014**, 47, 750–760.

(387) Barteau, M. A.; Madix, R. J. The Surface Reactivity of Silver: Oxidation Reactions. In *The Chemical Physics of Solid Surfaces and Heterogeneous Catalysis*; King, D. A.M., Woodruff, D. P., Eds.; Elsevier: Amsterdam, 1988; Vol. 5, p 95.

(388) Montemore, M. M.; van Spronsen, M. A.; Madix, R. J.; Friend, C. M. O₂ Activation by Metal Surfaces: Implications for Bonding and Reactivity on Heterogeneous Catalysts. *Chem. Rev.* **2018**, *118*, 2816–2862.

(389) Wittstock, A.; Biener, J.; Erlebacher, J.; Bäumer, M., Eds.; Nanoporous Gold: From an Ancient Technology to a High-Tech Material; RSC Nanoscience & Nanotechnology; Royal Society of Chemistry: Cambridge, UK, 2012. Vol. 22.

(390) Wittstock, A.; Biener, J.; Bäumer, M. Nanoporous gold: A New Material for Catalytic and Sensor Applications. *Phys. Chem. Chem. Phys.* **2010**, *12*, 12919–12930.

(391) Gong, J.; Mullins, C. B. Surface Science Investigations of Oxidative Chemistry on Gold. Acc. Chem. Res. 2009, 42, 1063–1073.

(392) Personick, M. L.; Madix, R. J.; Friend, C. M. Selective Oxygen-Assisted Reactions of Alcohols and Amines Catalyzed by Metallic Gold: Paradigms for the Design of Catalytic Processes. *ACS Catal.* **2017**, 7, 965–985.

(393) Wittstock, A.; Wichmann, A.; Bäumer, M. Nanoporous Gold as a Platform for a Building Block Catalyst. *ACS Catal.* **2012**, *2*, 2199–2215.

(394) Baker, T. A.; Liu, X.; Friend, C. M. The Mystery of Gold's Chemical Activity: Local Bonding, Morphology and Reactivity of Atomic Oxygen. *Phys. Chem. Chem. Phys.* **2011**, *13*, 34–46.

(395) Chang, C.-R.; Yang, X.-F.; Long, B.; Li, J. A Water-Promoted Mechanism of Alcohol Oxidation on a Au(111) Surface: Understanding the Catalytic Behavior of Bulk Gold. *ACS Catal.* **2013**, *3*, 1693–1699.

(396) Tomaschun, G.; Dononelli, W.; Li, Y.; Bäumer, M.; Klüner, T.; Moskaleva, L. V. Methanol Oxidation on the Au(3 1 0) Surface: A Theoretical Study. *J. Catal.* **2018**, *364*, 216–227.

(397) Wang, P.; Tang, X.; Tang, J.; Pei, Y. Density Functional Theory (DFT) Studies of CO Oxidation over Nanoporous Gold: Effects of Residual Ag and CO Self-Promoting Oxidation. *J. Phys. Chem. C* 2015, *119*, 10345–10354.

(398) Krajčí, M.; Kameoka, S.; Tsai, A.-P. Twinning in fcc Lattice Creates Low-coordinated Catalytically Active Sites in Porous Gold. *J. Chem. Phys.* **2016**, *145*, 084703.

(399) Krajčí, M.; Kameoka, S.; Tsai, A.-P. Understanding the Catalytic Activity of Nanoporous Gold: Role of Twinning in fcc Lattice. *J. Chem. Phys.* **2017**, *147*, 044713.

(400) Wang, L.-C.; Zhong, Y.; Jin, H.; Widmann, D.; Weissmüller, J.; Behm, R. J. Catalytic Activity of Nanostructured Au: Scale Effects versus Bimetallic/Bifunctional Effects in Low-Temperature CO Oxidation on Nanoporous Au. *Beilstein J. Nanotechnol.* **2013**, *4*, 111–128.

(401) Min, B. K.; Alemozafar, A. R.; Pinnaduwage, D.; Deng, X.; Friend, C. M. Efficient CO Oxidation at Low Temperature on Au(111). J. Phys. Chem. B 2006, 110, 19833–19838.

(402) Liu, X.; Xu, B.; Haubrich, J.; Madix, R. J.; Friend, C. M. Surface-Mediated Self-Coupling of Ethanol on Gold. *J. Am. Chem. Soc.* **2009**, *131*, 5757–5759.

(403) Reece, C.; Luneau, M.; Madix, R. J. Dissecting the Performance of Nanoporous Gold Catalysts for Oxygen-Assisted Coupling of Methanol with Fundamental Mechanistic and Kinetic Information. *ACS Catal.* **2019**, *9*, 4477–4487.

(404) Liu, X.; Madix, R. J.; Friend, C. M. Unraveling Molecular Transformations on Surfaces: A Critical Comparison of Oxidation Reactions on Coinage Metals. *Chem. Soc. Rev.* **2008**, *37*, 2243–2261.

(405) Wang, L. C.; Jin, H. J.; Widmann, D.; Weissmüller, J.; Behm, R. J. Dynamic Studies of CO Oxidation on Nanoporous Au Using a TAP Reactor. *J. Catal.* **2011**, *278*, 219–227.

(406) Röhe, S.; Frank, K.; Schaefer, A.; Wittstock, A.; Zielasek, V.; Rosenauer, A.; Bäumer, M. CO Oxidation on Nanoporous Gold: A Combined TPD and XPS Study of Active Catalysts. *Surf. Sci.* **2013**, *609*, 106–112.

(407) Moskaleva, L. V.; Röhe, S.; Wittstock, A.; Zielasek, V.; Klüner, T.; Neyman, K. M.; Bäumer, M. Silver Residues as a Possible Key to a Remarkable Oxidative Catalytic Activity of Nanoporous Gold. *Phys. Chem. Chem. Phys.* **2011**, *13*, 4529–4539.

(408) Landmann, M.; Rauls, E.; Schmidt, W. G. Chainlike Au–O Structures on Au(110)-(1 \times r) Surfaces Calculated from First Principles. J. Phys. Chem. C **2009**, 113, 5690–5699.

(409) Fajín, J. L. C.; Cordeiro, M. N. D. S.; Gomes, J. R. B. Adsorption of Atomic and Molecular Oxygen on the Au(321) Surface: DFT Study. J. Phys. Chem. C 2007, 111, 17311–17321.

(410) Hiebel, F.; Montemore, M. M.; Kaxiras, E.; Friend, C. M. Direct Visualization of Quasi-ordered Oxygen Chain Structures on $Au(110)-(1 \times 2)$. Surf. Sci. 2016, 650, 5–10.

(411) Li, Y.; Dononelli, W.; Moreira, R.; Risse, T.; Bäumer, M.; Klüner, T.; Moskaleva, L. V. Oxygen-Driven Surface Evolution of Nanoporous Gold: Insights from Ab Initio Molecular Dynamics and Auger Electron Spectroscopy. J. Phys. Chem. C 2018, 122, 5349– 5357.

(412) Fajín, J. L. C.; Cordeiro, M. N. D. S.; Gomes, J. R. B. DFT Study of the CO Oxidation on the Au(321) Surface. *J. Phys. Chem. C* **2008**, *112*, 17291–17302.

(413) Dononelli, W.; Moskaleva, L. V.; Klüner, T. CO Oxidation over Unsupported Group 11 Metal Catalysts: New Mechanistic Insight from First-Principles. J. Phys. Chem. C 2019, 123, 7818–7830. (414) Fajín, J. L. C.; Moura, A. S.; Cordeiro, M. N. D. S. Firstprinciples-based Kinetic Monte Carlo Simulations of CO Oxidation on Catalytic Au(110) and Ag(110) Surfaces. *Phys. Chem. Chem. Phys.* **2021**, 23, 14037–14050.

(415) Daté, M.; Okumura, M.; Tsubota, S.; Haruta, M. Vital Role of Moisture in the Catalytic Activity of Supported Gold Nanoparticles. *Angew. Chem., Int. Ed.* **2004**, 43, 2129–2132.

(416) Ojeda, M.; Iglesia, E. Catalytic Epoxidation of Propene with H_2O-O_2 reactants on Au/TiO₂. Chem. Commun. 2009, 352–354.

(417) Ojeda, M.; Zhan, B.-Z.; Iglesia, E. Mechanistic Interpretation of CO Oxidation Turnover Rates on Supported Au Clusters. *J. Catal.* **2012**, 285, 92–102.

(418) Fujitani, T.; Nakamura, I.; Haruta, M. Role of Water in CO Oxidation on Gold Catalysts. *Catal. Lett.* **2014**, *144*, 1475–1486.

(419) Zhong, W.; Liang, J.; Hu, W.; Cao, X.; Jia, C.; Jiang, J. The O, OH and OOH-assisted Selective Coupling of Methanol on Au-Ag(111). *Phys. Chem. Chem. Phys.* **2016**, *18*, 9969–9978.

(420) Sun, K.; Kohyama, M.; Tanaka, S.; Takeda, S. Roles of Water and H_2 in CO Oxidation Reaction on Gold Catalysts. *J. Phys. Chem. C* **2018**, *122*, 9523–9530.

(421) Réocreux, R.; Fampiou, I.; Stamatakis, M. The Role of Oxygenated Species in the Catalytic Self-coupling of MeOH on O Pre-covered Au(111). *Faraday Discuss.* **2021**, *229*, 251–266.

(422) Liu, S.; Jin, P.; Zhang, D.; Hao, C.; Yang, X. Reaction Mechanism for Methanol Oxidation on Au(111): A Density Functional Theory Study. *Appl. Surf. Sci.* **2013**, *265*, 443–451.

(423) Wang, L.; He, C.; Zhang, W.; Li, Z.; Yang, J. Methanol-Selective Oxidation Pathways on Au Surfaces: A First-Principles Study. *J. Phys. Chem. C* **2014**, *118*, 17511–17520.

(424) Sun, G.; Wang, F.; Jin, Y.; Chen, X.; Chai, P.; Wu, L.; Teng, B.-T.; Huang, W. Oxidative Coupling of Methanol with Molecularly Adsorbed Oxygen on Au Surface to Methyl Formate. *J. Phys. Chem. Lett.* **2021**, *12*, 6941–6945.

(425) Eley, D. D.; Moore, P. B. Adsorption of Oxygen on Gold. Surf. Sci. 1978, 76, L599–L602.

(426) Légaré, P.; Hilaire, L.; Sotto, M.; Maire, G. Interaction of Oxygen with Au Surfaces: A LEED. AES and ELS Study. Surf. Sci. 1980, 91, 175–186.

(427) Sault, A. G.; Madix, R. J.; Campbell, C. T. Adsorption of Oxygen and Hydrogen on Au(110)-(1 \times 2). Surf. Sci. **1986**, 169, 347–356.

(428) Kim, J.; Samano, E.; Koel, B. E. Oxygen Adsorption and Oxidation Reactions on Au(211) Surfaces: Exposures Using O_2 at High Pressures and Ozone (O_3) in UHV. Surf. Sci. **2006**, 600, 4622–4632.

(429) Canning, N. D. S.; Outka, D. A.; Madix, R. J. The Adsorption of Oxygen on Gold. *Surf. Sci.* **1984**, *141*, 240–254.

(430) Deng, X.; Min, B. K.; Guloy, A.; Friend, C. M. Enhancement of O_2 Dissociation on Au(111) by Adsorbed Oxygen: Implications for Oxidation Catalysis. *J. Am. Chem. Soc.* **2005**, *127*, 9267–9270.

(431) Kim, T. S.; Gong, J.; Ojifinni, R. A.; White, J. M.; Mullins, C. B. Water Activated by Atomic Oxygen on Au(111) to Oxidize CO at Low Temperatures. *J. Am. Chem. Soc.* **2006**, *128*, 6282–6283.

(432) Feldt, C. D.; Gimm, T.; Moreira, R.; Riedel, W.; Risse, T. Methanol Oxidation on Au(332): An Isothermal Pulsed Molecular Beam Study. *Phys. Chem. Chem. Phys.* **2021**, *23*, 21599–21605.

(433) Personick, M. L.; Zugic, B.; Biener, M. M.; Biener, J.; Madix, R. J.; Friend, C. M. Ozone-Activated Nanoporous Gold: A Stable and Storable Material for Catalytic Oxidation. *ACS Catal.* **2015**, *5*, 4237–4241.

(434) Gong, J.; Ojifinni, R. A.; Kim, T. S.; Stiehl, J. D.; McClure, S. M.; White, J. M.; Mullins, C. B. Low Temperature CO Oxidation on Au(111) and the Role of Adsorbed Water. *Top. Catal.* **2007**, *44*, 57–63.

(435) Moreira, R. Setup of a Molecular Beam Apparatus to Study the Reactivity of Single Crystal Surfaces and Its Application to CO Oxidation on Au(332). Ph.D. Thesis. Freie Universität Berlin: Berlin, 2018. (436) Ruggiero, C.; Hollins, P. Interaction of CO Molecules with the Au(332) Surface. *Surf. Sci.* **1997**, 377–379, 583–586.

(437) Ruggiero, C.; Hollins, P. Adsorption of Carbon Monoxide on the Gold(332) Surface. *Faraday Trans.* **1996**, *92*, 4829.

(438) Gottfried, J. M.; Schmidt, K. J.; Schroeder, S. L. M.; Christmann, K. Oxygen Chemisorption on Au(110)- (1×2) I. Thermal Desorption Measurements. Surf. Sci. 2003, 525, 184–196.

(439) Kim, J.; Samano, E.; Koel, B. E. CO Adsorption and Reaction on Clean and Oxygen-Covered Au(211) Surfaces. J. Phys. Chem. B 2006, 110, 17512–17517.

(440) Yim, W.-L.; Nowitzki, T.; Necke, M.; Schnars, H.; Nickut, P.; Biener, J.; Biener, M. M.; Zielasek, V.; Al-Shamery, K.; Kluener, T.; Bäumer, M. Universal Phenomena of CO Adsorption on Gold Surfaces with Low-Coordinated Sites. *J. Phys. Chem. C* **2007**, *111*, 445–451.

(441) Quiller, R. G.; Baker, T. A.; Deng, X.; Colling, M. E.; Min, B. K.; Friend, C. M. Transient Hydroxyl Formation from Water on Oxygen-Covered Au(111). *J. Chem. Phys.* **2008**, *129*, 064702.

(442) Xu, F.; Fampiou, I.; O'Connor, C. R.; Karakalos, S.; Hiebel, F.; Kaxiras, E.; Madix, R. J.; Friend, C. M. Water Facilitates Oxygen Migration on Gold Surfaces. *Phys. Chem. Chem. Phys.* **2018**, *20*, 2196–2204.

(443) Wachs, I. E.; Madix, R. J. The Oxidation of Ethanol on Cu(110) and Ag(110) Catalysts. *Appl. Surf. Sci.* **1978**, *1*, 303–328.

(444) Barteau, M. A.; Bowker, M.; Madix, R. J. Acid-Base Reactions on Solid Surfaces: The Reactions of HCOOH, H_2CO , and HCOOCH₃ with Oxygen on Ag(110). Surf. Sci. **1980**, 94, 303–322. (445) Outka, D. A.; Madix, R. J. Acid-Base and Nucleophilic

Chemistry of Atomic Oxygen on the Au(110) Surface: Reactions with Formic Acid and Formaldehyde. *Surf. Sci.* **1987**, *179*, 361–376.

(446) Outka, D. A.; Madix, R. J. Broensted Basicity of Atomic Oxygen on the Gold(110) Surface: Reactions with Methanol, Acetylene, Water, and Ethylene. *J. Am. Chem. Soc.* **1987**, *109*, 1708–1714.

(447) Xu, B.; Haubrich, J.; Freyschlag, C. G.; Madix, R. J.; Friend, C. M. Oxygen-Assisted Cross-Coupling of Methanol with Alkyl Alcohols on Metallic Gold. *Chem. Sci.* **2010**, *1*, 310.

(448) Xu, B.; Liu, X.; Haubrich, J.; Friend, C. M. Vapour-Phase Gold-Surface-Mediated Coupling of Aldehydes with Methanol. *Nat. Chem.* **2010**, *2*, 61–65.

(449) Feldt, C. D.; Albrecht, P. A.; Eltayeb, S.; Riedel, W.; Risse, T. Heterogeneity of Oxygen Reactivity: Key for Selectivity of Partial Methanol Oxidation on Gold Surfaces. *Chem. Commun.* **2022**, *58*, 4336–4339.

(450) Feldt, C. D.; Low, J. L.; Albrecht, P. A.; Tang, K.; Riedel, W.; Risse, T. Low-Temperature Oxidation of Methyl Formate on Au(332). J. Phys. Chem. C 2021, 125, 26522–26529.

(451) Feldt, C. D.; Kirschbaum, T.; Low, J. L.; Riedel, W.; Risse, T. Methanol Oxidation on Au(332): Methyl Formate Selectivity and Surface Deactivation under Isothermal Conditions. *Catal. Sci. Technol.* **2022**, *12*, 1418–1428.

(452) Wang, L.-C.; Personick, M. L.; Karakalos, S.; Fushimi, R.; Friend, C. M.; Madix, R. J. Active Sites for Methanol Partial Oxidation on Nanoporous Gold Catalysts. *J. Catal.* **2016**, *344*, 778–783.

(453) Campbell, C. T. Atomic and molecular oxygen adsorption on Ag(111). Surf. Sci. 1985, 157, 43-60.

(454) Wild, S.; Bäumer, M.; Risse, T. Thermal Activation of Nanoporous Gold for Carbon Monoxide Oxidation. *J. Phys. Chem. C* **2022**, *126*, 1770–1777.

(455) Viswanath, R. N.; Chirayath, V. A.; Rajaraman, R.; Amarendra, G.; Sundar, C. S. Ligament coarsening in nanoporous gold: Insights from positron annihilation study. *Appl. Phys. Lett.* **2013**, *102*, 253101.

(456) Wild, S.; Mahr, C.; Rosenauer, A.; Risse, T.; Vasenkov, S.; Bäumer, M. New Perspectives for Evaluating the Mass Transport in Porous Catalysts and Unfolding Macro- and Microkinetics. *Catal. Lett.* **2022**, 7, 1667–1684.

(457) Baniani, A.; Wild, S.; Forman, E. M.; Risse, T.; Vasenkov, S.; Bäumer, M. Disentangling Catalysis and Mass Transport: Using Diffusion Measurements by Pulsed Field Gradient NMR to Reveal the
Microkinetics of CO Oxidation over Nanoporous Gold. J. Catal. 2022, 413, 1123-1131.

(458) Reece, C.; Redekop, E. A.; Karakalos, S.; Friend, C. M.; Madix, R. J. Crossing the Great Divide between Single-Crystal Reactivity and Actual Catalyst Selectivity with Pressure Transients. *Nat. Catal.* **2018**, *1*, 852–859.

(459) Jürgens, B.; Kübel, C.; Schulz, C.; Nowitzki, T.; Zielasek, V.; Biener, J.; Biener, M. M.; Hamza, A. V.; Bäumer, M. New Gold and Silver-Gold Catalysts in the Shape of Sponges and Sieves. *Gold Bull.* **2007**, 40, 142–149.

(460) Baier, S.; Wittstock, A.; Damsgaard, C. D.; Diaz, A.; Reinhardt, J.; Benzi, F.; Shi, J.; Scherer, T.; Wang, Di; Kübel, C.; Schroer, C. G.; Grunwaldt, J.-D. Influence of Gas Atmospheres and Ceria on the Stability of Nanoporous Gold Studied by Environmental Electron Microscopy and in situ Ptychography. *RSC Adv.* **2016**, *6*, 83031–83043.

(461) Biener, J.; Wittstock, A.; Biener, M. M.; Nowitzki, T.; Hamza, A. V.; Bäumer, M. Effect of Surface Chemistry on the Stability of Gold Nanostructures. *Langmuir* **2010**, *26*, 13736–13740.

(462) Wang, Z.; Ning, S.; Liu, P.; Ding, Y.; Hirata, A.; Fujita, T.; Chen, M. Tuning Surface Structure of 3D Nanoporous Gold by Surfactant-Free Electrochemical Potential Cycling. *Adv. Mater.* **2017**, *29*, 1703601.

(463) Wichmann, A.; Wittstock, A.; Frank, K.; Biener, M. M.; Neumann, B.; Maedler, L.; Biener, J.; Rosenauer, A.; Bäumer, M. Maximizing Activity and Stability by Turning Gold Catalysis Upside Down: Oxide Particles on Nanoporous Gold. *ChemCatChem.* **2013**, *5*, 2037–2043.

(464) Rumancev, C.; von Gundlach, A. R.; Baier, S.; Wittstock, A.; Shi, J.; Benzi, F.; Senkbeil, T.; Stuhr, S.; Garamusx, V. M.; Grunwaldt, J.-D.; Rosenhahn, A. Morphological Analysis of Cerium Oxide Stabilized Nanoporous Gold Catalysts by Soft X-ray ASAXS. *RSC Adv.* **2017**, *7*, 45344–45350.

(465) Shi, J.; Schaefer, A.; Wichmann, A.; Murshed, M. M.; Gesing, T. M.; Wittstock, A.; Bäumer, M. Nanoporous Gold-Supported Ceria for the Water–Gas Shift Reaction: UHV Inspired Design for Applied Catalysis. *J. Phys. Chem. C* **2014**, *118*, 29270–29277.

(466) Bagge-Hansen, M.; Wichmann, A.; Wittstock, A.; Lee, J. R. I.; Ye, J.; Willey, T. M.; Kuntz, J. D.; van Buuren, T.; Biener, J.; Bäumer, M.; Biener, M. M. Quantitative Phase Composition of TiO_2 -Coated Nanoporous Au Monoliths by X-ray Absorption Spectroscopy and Correlations to Catalytic Behavior. J. Phys. Chem. C **2014**, 118, 4078– 4084.

(467) Schaefer, A.; Ragazzon, D.; Wittstock, A.; Walle, L. E.; Borg, A.; Bäumer, M.; Sandell, A. Toward Controlled Modification of Nanoporous Gold. A Detailed Surface Science Study on Cleaning and Oxidation. *J. Phys. Chem. C* **2012**, *116*, 4564–4571.

(468) Hammer, B.; Norskov, J. K. Why gold is the noblest of all the metals. *Nature* **1995**, 376, 238–240.

(469) Sievers, C.; Noda, Y.; Qi, L.; Albuquerque, E. M.; Rioux, R. M.; Scott, S. L. Phenomena Affecting Catalytic Reactions at Solid–Liquid Interfaces. *ACS Catal.* **2016**, *6*, 8286–8307.

(470) Negahdar, L.; Parlett, C. M. A.; Isaacs, M. A.; Beale, A. M.; Wilson, K.; Lee, A. F. Shining Light on the Solid–liquid Interface: In Situ/Operando Monitoring of Surface Catalysis. *Catal. Sci. Technol.* **2020**, *10*, 5362–5385.

(471) Saleheen, M.; Heyden, A. Liquid-Phase Modeling in Heterogeneous Catalysis. *ACS Catal.* **2018**, *8*, 2188–2194.

(472) Li, G.; Wang, B.; Resasco, D. E. Solvent Effects on Catalytic Reactions and Related Phenomena at Liquid-solid Interfaces. *Surf. Sci. Rep.* **2021**, *76*, No. 100541.

(473) Wesley, T. S.; Román-Leshkov, Y.; Surendranath, Y. Spontaneous Electric Fields Play a Key Role in Thermochemical Catalysis at Metal–Liquid Interfaces. *ACS Cent. Sci.* **2021**, *7*, 1045–1055.

(474) Knop-Gericke, A.; Pfeifer, V.; Velasco-Velez, J.-J.; Jones, T.; Arrigo, R.; Hävecker, M.; Schlögl, R. In Situ X-ray Photoelectron Spectroscopy of Electrochemically Active Solid-gas and Solid-liquid Interfaces. J. Electron Spectrosc. Relat. Phenom. **2017**, 221, 10–17. (475) Bentrup, U. Combining In Situ Characterization Methods in One Set-up: Looking with more Eyes into the Intricate Chemistry of the Synthesis and Working of Heterogeneous Catalysts. *Chem. Soc. Rev.* **2010**, *39*, 4718–4730.

(476) Stavitski, E.; Weckhuysen, B. M. Infrared and Raman Imaging of Heterogeneous Catalysts. *Chem. Soc. Rev.* **2010**, *39*, 4615–4625.

(477) Zope, B. N.; Hibbitts, D. D.; Neurock, M.; Davis, R. J. Reactivity of the Gold/Water Interface During Selective Oxidation Catalysis. *Science* **2010**, 330, 74–78.

(478) Sato, T.; Hamada, Y.; Sumikawa, M.; Araki, S.; Yamamoto, H. Solubility of Oxygen in Organic Solvents and Calculation of the Hansen Solubility Parameters of Oxygen. *Ind. Eng. Chem. Res.* **2014**, *53*, 19331–19337.

(479) Asao, N.; Hatakeyama, N.; Menggenbateer; Minato, T.; Ito, E.; Hara, M.; Kim, Y.; Yamamoto, Y.; Chen, M.; Zhang, W.; Inoue, A. Aerobic Oxidation of Alcohols in the Liquid Phase with Nanoporous Gold Catalysts. *Chem. Commun.* **2012**, *48*, 4540–4542.

(480) Lackmann, A.; Mahr, C.; Rosenauer, A.; Bäumer, M.; Wittstock, A. Aerobic Methanol Oxidation over Unsupported Nanoporous Gold: The Influence of an Added Base. *Catalysts* **2019**, *9*, 416.

(481) Zhu, C.; Kleimeier, N. F.; Turner, A. M.; Singh, S. K.; Fortenberry, R. C.; Kaiser, R. I. Synthesis of Methanediol CH₂(OH)₂: The Simplest Geminal Diol. *Proc. Natl. Acad. Sci. U.S.A.* **2022**, *119*, e2111938119.

(482) Moskaleva, L.; Chiu, C.-C.; Genest, A.; Rösch, N. Transformations of Organic Molecules over Metal Surfaces: Insights from Computational Catalysis. *Chem. Rec.* **2016**, *16*, 2388–2404.

(483) Liu, P.; Norskov, J. K. Ligand and Ensemble Effects in Adsorption on Alloy Surfaces. *Phys. Chem. Chem. Phys.* 2001, 3, 3814–3818.

(484) Xu, B.; Siler, C. G. F.; Madix, R. J.; Friend, C. M. Ag/Au Mixed Sites Promote Oxidative Coupling of Methanol on the Alloy Surface. *Chem.—Eur. J.* **2014**, *20*, 4646–4652.

(485) Klitgaard, S. K.; DeLa Riva, A. T.; Helveg, S.; Werchmeister, R. M.; Christensen, C. H. Aerobic Oxidation of Alcohols over Gold Catalysts: Role of Acid and Base. *Catal. Lett.* **2008**, *126*, 213–217.

(486) Qiu, H. J.; Xu, H.-T.; Liu, L.; Wang, Y. Correlation of the Structure and Applications of Dealloyed Nanoporous Metals in Catalysis and Energy Conversion/Storage. *Nanoscale* 2015, 7, 386–400.

(487) Gonçalves, J. M.; Kumar, A.; da Silva, M. I.; Toma, H. E.; Martins, P. R.; Araki, K.; Bertotti, M.; Angnes, L. Nanoporous Gold-Based Materials for Electrochemical Energy Storage and Conversion. *Energy Technol.* **2021**, *9*, 2000927.

(488) Hillier, A. C.; Kim, S.; Bard, A. J. Measurement of Double-Layer Forces at the Electrode/Electrolyte Interface Using the Atomic Force Microscope: Potential and Anion Dependent Interactions. *J. Phys. Chem.* **1996**, *100*, 18808–18817.

(489) Lipkowski, J.; Shi, Z.; Chen, A.; Pettinger, B.; Bilger, C. Ionic Adsorption at the Au(111) Electrode. *Electrochim. Acta* **1998**, 43, 2875–2888.

(490) Magnussen, O. M. Ordered Anion Adlayers on Metal Electrode Surfaces. *Chem. Rev.* 2002, *102*, 679–725.

(491) Edens, G. J.; Gao, X.; Weaver, M. J. The Adsorption of Sulfate on Gold(111) in Acidic Aqueous Media: Adlayer Structural Inferences from Infrared Spectroscopy and Scanning Tunneling Microscopy. J. Electroanal. Chem. **1994**, 375, 357–366.

(492) Dakkouri, A. S.; Kolb, D. M. Reconstruction of Gold Surface. In *Interfacial Electrochemistry: Theory, Experiment and Application;* Wieckowski, A., Ed.; M. Dekker: New York, 1999; pp 151–173.

(493) Oviedo, O. A.; Reinaudi, L.; Garcia, S.; Leiva, E. P. M. Underpotential Deposition: From Fundamentals and Theory to Applications at the Nanoscale; Monographs in Electrochemistry; Springer. Cham 2016.

(494) Herrero, E.; Buller, L. J.; Abruña, H. D. Underpotential Deposition at Single Crystal Surfaces of Au, Pt, Ag and Other Materials. *Chem. Rev.* **2001**, *101*, 1897–1930.

(495) Kondo, T.; Morita, J.; Hanaoka, K.; Takakusagi, S.; Tamura, K.; Takahasi, M.; Mizuki, J.-i.; Uosaki, K. Structure of Au(111) and Au(100) Single-Crystal Electrode Surfaces at Various Potentials in Sulfuric Acid Solution Determined by In Situ Surface X-ray Scattering. *J. Phys. Chem. C* 2007, *111*, 13197–13204.

(496) Conway, B. E. Electrochemical Oxide Film Formation at Noble Metals as a Surface-chemical Process. *Prog. Surf. Sci.* **1995**, *49*, 331–452.

(497) Hamelin, A.; Martins, A. M. Cyclic Voltammetry at Gold Single-Crystal Surfaces. Part 2. Behaviour of High-Index Faces. J. Electroanal. Chem. **1996**, 407, 13–21.

(498) Hernández, J.; Solla-Gullón, J.; Herrero, E.; Aldaz, A.; Feliu, J. M. Methanol Oxidation on Gold Nanoparticles in Alkaline Media: Unusual Electrocatalytic Activity. *Electrochim. Acta* **2006**, *52*, 1662–1669.

(499) Hamelin, A.; Lipkowski, J. Underpotential Deposition of Lead on Gold Single Crystal Faces: Part II: General Discussion. *J. Electroanal. Chem.* **1984**, *171*, 317–330.

(500) Hernández, J.; Solla-Gullón, J.; Herrero, E.; Feliu, J. M.; Aldaz, A. In Situ Surface Characterization and Oxygen Reduction Reaction on Shape-Controlled Gold Nanoparticles. *J. Nanosci. Nanotechnol.* **2009**, *9*, 2256–2273.

(501) Ahrens, P.; Zander, M.; Hasse, U.; Wulff, H.; Jeyabharathi, C.; Kruth, A.; Scholz, F. Electrochemical Formation of Gold Nanoparticles on Polycrystalline Gold Electrodes during Prolonged Potential Cycling. *ChemElectroChem.* **2018**, *5*, 943–957.

(502) Ahrens, P.; Zander, M.; Hirsch, D.; Hasse, U.; Wulff, H.; Frost, F.; Scholz, F. Influence of argon ion beam etching and thermal treatment on polycrystalline and single crystal gold electrodes Au(100) and Au(111). *J. Electroanal. Chem.* **2019**, 832, 233–240.

(503) Narayanaru, S.; Chinnaiah, J.; Phani, K. L.; Scholz, F. pH Dependent CO Adsorption and Roughness-Induced Selectivity of CO_2 Electroreduction on Gold Surfaces. *Electrochim. Acta* **2018**, *264*, 269–274.

(504) Jeyabharathi, C.; Ahrens, P.; Hasse, U.; Scholz, F. Identification of Low-Index Crystal Planes of Polycrystalline Gold on the Basis of Electrochemical Oxide Layer Formation. *J. Solid State Electrochem.* **2016**, *20*, 3025–3031.

(505) Silva Olaya, A. R.; Kühling, F.; Mahr, C.; Zandersons, B.; Rosenauer, A.; Weissmüller, J.; Wittstock, G. Promoting Effect of the Residual Silver on the Electrocatalytic Oxidation of Methanol and Its Intermediates on Nanoporous Gold. *ACS Catal.* **2022**, *12*, 4415– 4429.

(506) Kumar, A.; Gonçalves, J. M.; Furtado, V. L.; Araki, K.; Angnes, L.; Bouvet, M.; Bertotti, M.; Meunier-Prest, R. Mass Transport in Nanoporous Gold and Correlation with Surface Pores for EC_1 Mechanism: Case of Ascorbic Acid. *ChemElectroChem.* **2021**, *8*, 2129–2136.

(507) Yu, Q.; Yin, S.; Zhang, J.; Yin, H. Structure Dependent Activity and Durability towards Oxygen Reduction Reaction on Pt Modified Nanoporous Gold. *Electrochim. Acta* **2019**, *298*, 599–608. (508) Paschalidou, E. M.; Scaglione, F.; Gebert, A.; Oswald, S.; Rizzi, P.; Battezzati, L. Partially and Fully De-Alloyed Glassy Ribbons Based on Au: Application in Methanol Electro-Oxidation Studies. J. *Alloys Compd.* **2016**, *667*, 302–309.

(509) Denis, P.; Fecht, H.-J.; Xue, Y.; Paschalidou, E. M.; Rizzi, P.; Battezzati, L. Microstructure and Electrochemical Properties of Nanoporous Gold Produced by Dealloying Au-Based Thin Film Nanoglass. J. Mater. Res. 2018, 33, 2661–2670.

(510) Trasatti, S. Work Function, Electronegativity, and Electrochemical Behavior of Metals. III. Electrolytic Hydrogen Evolution in Acid Solutions. J. Electroanal. Chem. Interfacial Electrochem. **1972**, 39, 163–184.

(511) Conway, B. E.; Jerkiewicz, G. Relation of Energies and Coverages of Underpotential and Overpotential Deposited H at Pt and other Metals to the "Volcano Curve" for Cathodic H_2 Evolution Kinetics. *Electrochim. Acta* **2000**, *45*, 4075–4083.

(512) Kiani, A.; Hatami, S. Fabrication of Platinum Coated Nanoporous Gold Film Electrode: A Nanostructured Ultra LowPlatinum Loading Electrocatalyst for Hydrogen Evolution Reaction. Int. J. Hydrog. Energy 2010, 35, 5202–5209.

(513) Sukeri, A.; Bertotti, M. Nanoporous Gold Surface: An Efficient Platform for Hydrogen Evolution Reaction at Very Low Overpotential. J. Braz. Chem. Soc. 2017, 29, 20170132.

(514) Perez, J.; Gonzalez, E. R.; Villullas, H. M. Hydrogen Evolution Reaction on Gold Single-Crystal Electrodes in Acid Solutions. *J. Phys. Chem. B* **1998**, *102*, 10931–10935.

(515) Kundu, M. K.; Bhowmik, T.; Barman, S. Gold Aerogel Supported on Graphitic Carbon Nitride: An Efficient Electrocatalyst for Oxygen Reduction Reaction and Hydrogen Evolution Reaction. *J. Mater. Chem. A* **2015**, *3*, 23120–23135.

(516) Shen, Y.; Träuble, M.; Wittstock, G. Detection of Hydrogen Peroxide Produced during Electrochemical Oxygen Reduction Using Scanning Electrochemical Microscopy. *Anal. Chem.* **2008**, *80*, 750– 759.

(517) Katsounaros, I.; Cherevko, S.; Zeradjanin, A. R.; Mayrhofer, K. J. J. Oxygen Electrochemistry as a Cornerstone for Sustainable Energy Conversion. *Angew. Chem., Int. Ed.* **2014**, *53*, 102–121.

(518) Zeis, R.; Mathur, A.; Fritz, G.; Lee, J.; Erlebacher, J. Platinum-Plated Nanoporous Gold: An Efficient, Low Pt Loading Electrocatalyst for PEM Fuel Cells. *J. Power Sources* **2007**, *165*, 65–72.

(519) Wang, Y.; Kim, J. Oxygen Evolution Reaction on Nanoporous Gold Modified with Ir and Pt: Synergistic Electrocatalysis between Structure and Composition. *Electroanalysis* **2019**, *31*, 1026–1033.

(520) Zhang, an; Wang, J.; Schützendübe, P.; Liang, H.; Huang, Y.; Wang, Z. Beyond Dealloying: Development of Nanoporous Gold via Metal-Induced Crystallization and its Electrochemical Properties. *Nanotechnology* **2019**, *30*, 375601.

(521) Wang, J.; Chen, F.; Jin, Y.; Johnston, R. L. Gold-Copper Aerogels with Intriguing Surface Electronic Modulation as Highly Active and Stable Electrocatalysts for Oxygen Reduction and Borohydride Oxidation. *ChemSusChem* **2018**, *11*, 1354–1364.

(522) Bełtowska-Brezezinska, M.; Łuczak, T.; Holze, R. Electrocatalytic Oxidation of Mono- and Polyhydric Alcohols on Gold and Platinum. J. Appl. Electrochem. **1997**, *27*, 999–1011.

(523) Kwon, Y.; Lai, S. C. S.; Rodriguez, P.; Koper, M. T. M. Electrocatalytic Oxidation of Alcohols on Gold in Alkaline Media: Base or Gold Catalysis? J. Am. Chem. Soc. 2011, 133, 6914–6917.

(524) Beyhan, S.; Uosaki, K.; Feliu, J. M.; Herrero, E. Electrochemical and in situ FTIR studies of ethanol adsorption and oxidation on gold single crystal electrodes in alkaline media. *J. Electroanal. Chem.* **2013**, 707, 89–94.

(525) Vega, A. A.; Newman, R. C. Methanol Electro-Oxidation on Nanoporous Metals Formed by Dealloying of Ag–Au–Pt Alloys. *J. Appl. Electrochem.* **2016**, *46*, 995–1010.

(526) Yang, Z.; Pedireddy, S.; Lee, H. K.; Liu, Y.; Tjiu, W. W.; Phang, I. Y.; Ling, X. Y. Manipulating the d-Band Electronic Structure of Platinum-Functionalized Nanoporous Gold Bowls: Synergistic Intermetallic Interactions Enhance Catalysis. *Chem. Mater.* **2016**, *28*, 5080–5086.

(527) Pedireddy, S.; Lee, H. K.; Koh, C. S. L.; Tan, J. M. R.; Tjiu, W. W.; Ling, X. Y. Nanoporous Gold Bowls: A Kinetic Approach to Control Open Shell Structures and Size-Tunable Lattice Strain for Electrocatalytic Applications. *Small* **2016**, *12*, 4531–4540.

(528) Borkowska, Z.; Tymosiak-Zielinska, A.; Shul, G. Electrooxidation of Methanol on Polycrystalline and Single Crystal Gold Electrodes. *Electrochim. Acta* **2004**, *49*, 1209–1220.

(529) Lee, K.-U.; Byun, J. Y.; Shin, H.-J.; Kim, S. H. Nanoporous Gold-Palladium: A Binary Alloy with High Catalytic Activity for the Electro-Oxidation of Ethanol. *J. Alloys Compd.* **2020**, *842*, 155847.

(530) Chen, L. Y.; Chen, N.; Hou, Y.; Wang, Z. C.; Lv, S. H.; Fujita, T.; Jiang, J. H.; Hirata, A.; Chen, M. W. Geometrically Controlled Nanoporous PdAu Bimetallic Catalysts with Tunable Pd/Au Ratio for Direct Ethanol Fuel Cells. *ACS Catal.* **2013**, *3*, 1220–1230.

(531) Rezaei, B.; Havakeshian, E.; Ensafi, A. A. Electrocatalytic Activity of Bimetallic Pd Au Nanostructure Supported on Nanoporous Stainless Steel Surface Using Galvanic Replacement Reaction Toward the Glycerol Oxidation in Alkaline Media. J. Electroanal. Chem. 2016, 782, 108–116.

(532) Lee, D.; Jang, H. Y.; Hong, S.; Park, S. Synthesis of Hollow and Nanoporous Gold/Platinum Alloy Nanoparticles and their Electrocatalytic Activity for Formic Acid Oxidation. *J. Colloid Interface Sci.* **2012**, *388*, 74–79.

(533) Wang, R.; Liu, J.; Liu, P.; Bi, X.; Yan, X.; Wang, W.; Meng, Y.; Ge, X.; Chen, M.; Ding, Y. Ultra-Thin Layer Structured Anodes for Highly Durable Low-Pt Direct Formic Acid Fuel Cells. *Nano Res.* **2014**, *7*, 1569–1580.

(534) Wang, H.; Ge, X. Facile Fabrication of Porous Pd-Au Bimetallic Nanostructures for Electrocatalysis. *Electroanalysis* **2012**, 24, 911–916.

(535) Ma, X.; Shen, Y.; Yao, S.; Shu, M.; Si, R.; An, C. Self-Supported Nanoporous Au₃Cu Electrode with Enriched Gold on Surface for Efficient Electrochemical Reduction of CO₂. *Chem.—Eur. J.* **2020**, *26*, 4143.

(536) Back, S.; Yeom, M. S.; Jung, Y. Understanding the Effects of Au Morphology on CO_2 Electrocatalysis. J. Phys. Chem. C 2018, 122, 4274–4280.

(537) Mariano, R. G.; McKelvey, K.; White, H. S.; Kanan, M. W. Selective Increase in CO_2 Electroreduction Activity at Grain-Boundary Surface Terminations. *Science* **2017**, 358, 1187–1192.

(538) Feng, X.; Jiang, K.; Fan, S.; Kanan, M. W. Grain-Boundary-Dependent CO₂ Electroreduction Activity. *J. Am. Chem. Soc.* 2015, 137, 4606–4609.

(539) Welch, A. J.; DuChene, J. S.; Tagliabue, G.; Davoyan, A.; Cheng, W.-H.; Atwater, H. A. Nanoporous Gold as a Highly Selective and Active Carbon Dioxide Reduction Catalyst. *ACS Appl. Energy Mater.* **2019**, *2*, 164–170.

(540) Mascaretti, L.; Niorettini, A.; Bricchi, B. R.; Ghidelli, M.; Naldoni, A.; Caramori, S.; Li Bassi, A.; Berardi, S. Syngas Evolution from CO_2 Electroreduction by Porous Au Nanostructures. *ACS Appl. Energy Mater.* **2020**, *3*, 4658–4668.

(541) Qiu, H.; Huang, X. Effects of Pt Decoration on the Electrocatalytic Activity of Nanoporous Gold Electrode Toward Glucose and its Potential Application for Constructing a Nonenzymatic Glucose Sensor. *J. Electroanal. Chem.* **2010**, *643*, 39–45.

(542) Tavakkoli, N.; Nasrollahi, S.; Vatankhah, G. Electrocatalytic Determination of Ascorbic Acid Using a Palladium Coated Nanoporous Gold Film Electrode. *Electroanalysis* **2012**, *24*, 368–375.

(543) Lee, E.; Sung, M.; Wang, Y.; Kim, J. Atomic Layer Electrodeposition of Pt on Nanoporous Au and its Application in pH Sensing. *Electroanalysis* **2018**, *30*, 2028–2034.

(544) Scaglione, F.; Xue, Y.; Celegato, F.; Rizzi, P.; Battezzati, L. Amorphous Molybdenum Sulphide @ Nanoporous Gold as Catalyst for Hydrogen Evolution Reaction in Acidic Environment. *J. Mater. Sci.* **2018**, *53*, 12388–12398.

(545) Phan-Quang, G. C.; Yang, Z.; Koh, C. S. L.; Sim, H. Y. F.; Leong, S. X.; Ling, X. Y. Plasmonic-Induced Overgrowth of Amorphous Molybdenum Sulfide on Nanoporous Gold: An Ambient Synthesis Method of Hybrid Nanoparticles with Enhanced Electrocatalytic Activity. *J. Chem. Phys.* **2019**, *151*, 244709.

(546) Xiao, X.; Engelbrekt, C.; Zhang, M.; Li, Z.; Ulstrup, J.; Zhang, J.; Si, P. A Straight Forward Approach to Electrodeposit Tungsten Disulfide/Poly(3,4-ethylenedioxythiophene) Composites onto Nanoporous Gold for the Hydrogen Evolution Reaction. *Appl. Surf. Sci.* **2017**, *410*, 308–314.

(547) Kulesza, P. J.; Galus, Z. Mixed-Valence Electron Hopping, Redox Conduction and Migration Effects in Solid-State Electrochemistry of Transition Metal Hexacyanoferrates. *J. Electroanal. Chem.* **1992**, 323, 261–274.

(548) Sun, L.; Campbell, M. G.; Dincă, M. Electrically Conductive Porous Metal-Organic Frameworks. *Angew. Chem., Int. Ed.* **2016**, *55*, 3566–3579.

(549) Yang, Y.; Wang, S.-Q.; Wen, H.; Ye, T.; Chen, J.; Li, C.-P.; Du, M. Nanoporous Gold Embedded ZIF Composite for Enhanced Electrochemical Nitrogen Fixation. *Angew. Chem., Int. Ed.* **2019**, *58*, 15362–15366.

(550) Ghaderi, S.; Mehrgardi, M. A. Prussian Blue-Modified Nanoporous Gold Film Electrode for Amperometric Determination of Hydrogen Peroxide. *Bioelectrochemistry* **2014**, *98*, 64–69.

(551) Yao, R.-Q.; Shi, H.; Wan, W.-B.; Wen, Z.; Lang, X.-Y.; Jiang, Q. Flexible Co-Mo-N/Au Electrodes with a Hierarchical Nanoporous Architecture as Highly Efficient Electrocatalysts for Oxygen Evolution Reaction. *Adv. Mater.* **2020**, *32*, No. 1907214.

(552) Ruffino, F.; Grimaldi, M. G. Nanoporous Gold-Based Sensing. *Coatings* **2020**, *10*, 899.

(553) Heller, A.; Feldman, B. Electrochemical Glucose Sensors and Their Applications in Diabetes Management. *Chem. Rev.* 2008, *108*, 2482–2505.

(554) Meng, F.; Yan, X.; Liu, J.; Gu, J.; Zou, Z. Nanoporous Gold as Non-Enzymatic Sensor for Hydrogen Peroxide. *Electrochim. Acta* 2011, 56, 4657–4662.

(555) Zhong, Y.; Liu, M.-M.; Chen, Y.; Yang, Y.-J.; Wu, L.-N.; Bai, F.-Q.; Lei, Y.; Gao, F.; Liu, A.-L. A High-Performance Amperometric Sensor Based on a Monodisperse Pt-Au Bimetallic Nanoporous Electrode for Determination of Hydrogen Peroxide released from living cells. *Mikrochim. Acta* **2020**, *187*, 499.

(556) Shekarriz, R.; Friedrichsen, D. M.; Brooks, B.; Silaski, G.; Rios, L.; Wiest, E.; Kanagy, N. L. Sensor of Transdermal Biomarkers for Blood Perfusion Monitoring. *Sensing and Bio-Sensing Research* **2020**, 28, No. 100328.

(557) Yan, X.; Meng, F.; Xie, Y.; Liu, J.; Ding, Y. Direct N_2H_4/H_2O_2 Fuel Cells Powered by Nanoporous Gold Leaves. *Sci. Rep.* **2012**, *2*, 941.

(558) Jin, W.; Liu, J.; Wang, Y.; Yao, Y.; Gu, J.; Zou, Z. Direct $NaBH_4-H_2O_2$ Fuel Cell Based on Nanoporous Gold Leaves. *Int. J. Hydrog. Energy* **2013**, *38*, 10992–10997.

(559) Johnson, D. C.; LaCourse, W. R. Liquid Chromatography with Pulsed Electrochemical Detection at Gold and Platinum Electrodes. *Anal. Chem.* **1990**, *62*, 589A–597A.

(560) Bindra, D. S.; Wilson, G. S. Pulsed Amperometric Detection of Glucose in Biological Fluids at a Surface-Modified Gold Electrode. *Anal. Chem.* **1989**, *61*, 2566–2570.

(561) Holade, Y.; Tuleushova, N.; Tingry, S.; Servat, K.; Napporn, T. W.; Guesmi, H.; Cornu, D.; Kokoh, K. B. Recent Advances in the Electrooxidation of Biomass-based Organic Molecules for Energy, Chemicals and Hydrogen Production. *Catal. Sci. Technol.* **2020**, *10*, 3071–3112.

(562) Vedovato, V.; Vanbroekhoven, K.; Pant, D.; Helsen, J. Electrosynthesis of Biobased Chemicals Using Carbohydrates as a Feedstock. *Molecules* **2020**, *25*, 3712.

(563) Lucas, F. W. S.; Grim, R. G.; Tacey, S. A.; Downes, C. A.; Hasse, J.; Roman, A. M.; Farberow, C. A.; Schaidle, J. A.; Holewinski, A. Electrochemical Routes for the Valorization of Biomass-Derived Feedstocks: From Chemistry to Application. *ACS Energy Lett.* **2021**, *67*, 1205–1270.

(564) Seo, B.; Kim, J. Electrooxidation of Glucose at Nanoporous Gold Surfaces: Structure Dependent Electrocatalysis and Its Application to Amperometric Detection. *Electroanalysis* **2010**, *22*, 939–945.

(565) Chen, L. Y.; Lang, X. Y.; Fujita, T.; Chen, M. W. Nanoporous Gold for Enzyme-Free Electrochemical Glucose Sensors. *Scr. Mater.* **2011**, *65*, 17–20.

(566) Jeong, H.; Kim, J. Electrochemical Oxidation of Glucose at Nanoporous Black Gold Surfaces in the Presence of High Concentration of Chloride Ions and Application to Amperometric Detection. *Electrochim. Acta* **2012**, *80*, 383–389.

(567) Jeong, H.; Kim, J. Fabrication of Nanoporous Au Films with Ultra-High Surface Area for Sensitive Electrochemical Detection of Glucose in the Presence of Cl⁻. *Appl. Surf. Sci.* **2014**, *297*, 84–88.

(568) Li, Q.; Cui, S.; Yan, X. Electrocatalytic Oxidation of Glucose on Nanoporous Gold Membranes. *J. Solid State Electrochem.* **2012**, *16*, 1099–1104.

(569) Wang, J.; Gao, H.; Sun, F.; Xu, C. Nanoporous PtAu Alloy as an Electrochemical Sensor for Glucose and Hydrogen Peroxide. *Sens. Actuators*, B **2014**, *191*, 612–618.

(570) Lang, X.-Y.; Fu, H.-Y.; Hou, C.; Han, G.-F.; Yang, P.; Liu, Y.-B.; Jiang, Q. Nanoporous Gold Supported Cobalt Oxide Microelectrodes as High-Performance Electrochemical Biosensors. *Nat. Commun.* **2013**, *4*, 2169.

(571) Kumar, A.; Selva, J. S.G.; Gonçalves, J. M.; Araki, K.; Bertotti, M. Nanoporous Gold-based Dopamine Sensor with Sensitivity Boosted by Interferant Ascorbic Acid. *Electrochim. Acta* **2019**, *322*, No. 134772.

(572) Kumar, A.; Furtado, V. L.; Gonçalves, J. M.; Bannitz-Fernandes, R.; Netto, L. E. S.; Araki, K.; Bertotti, M. Amperometric Microsensor Based on Nanoporous Gold for Ascorbic Acid Detection in Highly Acidic Biological Extracts. *Anal. Chim. Acta* **2020**, *1095*, 61–70.

(573) Silva, T. A.; Khan, M. R. K.; Fatibello-Filho, O.; Collinson, M. M. Simultaneous Electrochemical Sensing of Ascorbic Acid and Uric Acid under Biofouling Conditions Using Nanoporous Gold Electrodes. *J. Electroanal. Chem.* **2019**, *846*, 113160.

(574) Bae, J. H.; Kim, Y.-R.; Soyoung Kim, R.; Chung, T. D. Enhanced Electrochemical Reactions of 1,4-Benzoquinone at Nanoporous Electrodes. *Phys. Chem. Chem. Phys.* **2013**, *15*, 10645–10653.

(575) Rezaei, B.; Damiri, S. Fabrication of a Nanostructure Thin Film on the Gold Electrode using Continuous Pulsed-Potential Technique and its Application for the Electrocatalytic Determination of Metronidazole. *Electrochim. Acta* **2010**, *55*, 1801–1808.

(576) Wierzbicka, E.; Sulka, G. D. Fabrication of Highly Ordered Nanoporous Thin Au Films and Their Application for Electrochemical Determination of Epinephrine. *Sens. Actuators, B* **2016**, *222*, 270–279.

(577) Wierzbicka, E.; Sulka, G. D. Nanoporous Spongelike Au–Ag Films for Electrochemical Epinephrine Sensing. *J. Electroanal. Chem.* **2016**, *762*, 43–50.

(578) Gooding, J. J.; Mearns, F.; Yang, W.; Liu, J. Self-Assembled Monolayers into the 21st Century: Recent Advances and Applications. *Electroanalysis* **2003**, *15*, 81–96.

(579) Lu, L.; Dong, Y.; Wang, J.; Li, Q.; Wu, X. Direct Electrochemistry and Bioelectrocatalysis of Horseradish Peroxidase Entrapped in a Self-Supporting Nanoporous Gold Electrode: A New Strategy to Improve the Orientation of Immobilized Enzymes. *Anal. Methods* **2015**, *7*, 6686–6694.

(580) Yan, X.; Ma, S.; Tang, J.; Tanner, D.; Ulstrup, J.; Xiao, X.; Zhang, J. Direct Electron Transfer of Fructose Dehydrogenase Immobilized on Thiol-Gold Electrodes. *Electrochim. Acta* **2021**, *392*, No. 138946.

(581) Chen, L. Y.; Fujita, T.; Chen, M. W. Biofunctionalized Nanoporous Gold for Electrochemical Biosensors. *Electrochim. Acta* **2012**, *67*, 1–5.

(582) Hu, X.; Wang, R.; Ding, Y.; Zhang, X.; Jin, W. Electrochemiluminescence of CdTe Quantum Dots as Labels at Nanoporous Gold Leaf Electrodes for Ultrasensitive DNA Analysis. *Talanta* **2010**, *80*, 1737–1743.

(583) Veselinovic, J.; AlMashtoub, S.; Nagella, S.; Seker, E. Interplay of Effective Surface Area, Mass Transport, and Electrochemical Features in Nanoporous Nucleic Acid Sensors. *Anal. Chem.* **2020**, *92*, 10751–10758.

(584) Zhu, Y.; Zhou, C.; Yan, X.; Yan, Y.; Wang, Q. Aptamer-Functionalized Nanoporous Gold Film for High-Performance Direct Electrochemical Detection of Bisphenol A in Human Serum. *Anal. Chim. Acta* 2015, 883, 81–89.

(585) Xiao, X.; Siepenkoetter, T.; Conghaile, P. O.; Leech, D.; Magner, E. Nanoporous Gold-Based Biofuel Cells on Contact Lenses. *ACS Appl. Mater. Interfaces* **2018**, *10*, 7107–7116.

(586) Obrovac, M. N.; Chevrier, V. L. Alloy Negative Electrodes for Li-ion Batteries. *Chem. Rev.* **2014**, *114*, 11444–11502.

(587) Yu, Y.; Yan, C.; Gu, L.; Lang, X.; Tang, K.; Zhang, L.; Hou, Y.; Wang, Z.; Chen, M. W.; Schmidt, O. G.; Maier, J. Three-Dimensional (3D) Bicontinuous Au/Amorphous-Ge Thin Films as Fast and High-Capacity Anodes for Lithium-Ion Batteries. *Adv. Energy Mater.* **2013**, *3*, 281–285. (588) Yu, Y.; Gu, L.; Lang, X.; Zhu, C.; Fujita, T.; Chen, M.; Maier, J. Li Storage in 3D Nanoporous Au-Supported Nanocrystalline Tin. *Adv. Mater.* **2011**, *23*, 2443–2447.

(589) Ye, J.; Baumgaertel, A. C.; Wang, Y. M.; Biener, J.; Biener, M. M. Structural Optimization of 3D Porous Electrodes for High-Rate Performance Lithium Ion Batteries. *ACS Nano* **2015**, *9*, 2194–2202. (590) Guo, X.; Han, J.; Zhang, L.; Liu, P.; Hirata, A.; Chen, L.; Fujita, T.; Chen, M. A Nanoporous Metal Recuperated MnO₂ Anode

for Lithium Ion Batteries. Nanoscale 2015, 7, 15111-15116. (591) Chen, Y.; Freunberger, S. A.; Peng, Z.; Bardé, F.; Bruce, P. G.

(31) Chen, 1.; Freunberger, S. A.; Feng, Z.; Barte, F. G. Li-O₂ Battery with a Dimethylformamide Electrolyte. *J. Am. Chem. Soc.* **2012**, 134, 7952–7957.

(592) Chen, Y.; Freunberger, S. A.; Peng, Z.; Fontaine, O.; Bruce, P. G. Charging a Li- O_2 Battery Using a Redox Mediator. *Nat. Chem.* **2013**, *5*, 489–494.

(593) Hashimoto, T.; Hayashi, K. Aqueous and Nonaqueous Sodium-Air Cells with Nanoporous Gold Cathode. *Electrochim. Acta* **2015**, *182*, 809–814.

(594) Chen, L. Y.; Kang, J. L.; Hou, Y.; Liu, P.; Fujita, T.; Hirata, A.; Chen, M. W. High-Energy-Density Nonaqueous MnO₂@Nanoporous Gold Based Supercapacitors. *J. Mater. Chem. A* **2013**, *1*, 9202.

(595) Lang, X.; Hirata, A.; Fujita, T.; Chen, M. Nanoporous Metal/ Oxide Hybrid Electrodes for Electrochemical Supercapacitors. *Nat. Nanotechnol.* **2011**, *6*, 232–236.

(596) Béguin, F.; Presser, V.; Balducci, A.; Frackowiak, E. Carbons and electrolytes for advanced supercapacitors. *Adv. Mater.* **2014**, *26* (2219), 2219.

(597) Lang, X. Y.; Yuan, H. T.; Iwasa, Y.; Chen, M. W. Three-Dimensional Nanoporous Gold for Electrochemical Supercapacitors. *Scr. Mater.* **2011**, *64*, 923–926.

(598) Chen, L. Y.; Hou, Y.; Kang, J. L.; Hirata, A.; Fujita, T.; Chen, M. W. Toward the Theoretical Capacitance of RuO_2 Reinforced by Highly Conductive Nanoporous Gold. *Adv. Energy Mater.* **2013**, *3*, 851–856.

(599) Singh, B. K.; Shaikh, A.; Badrayyana, S.; Mohapatra, D.; Dusane, R. O.; Parida, S. Nanoporous Gold–Copper Oxide Based All-Solid-State Micro-Supercapacitors. *RSC Adv.* **2016**, *6*, 100467–100475.

(600) Kim, S.-I.; Kim, S.-W.; Jung, K.; Kim, J.-B.; Jang, J.-H. Ideal Nanoporous Gold Based Supercapacitors with Theoretical Capacitance and High Energy/Power Density. *Nano Energy* **2016**, *24*, 17–24.

(601) Hou, Y.; Chen, L.; Zhang, L.; Kang, J.; Fujita, T.; Jiang, J.; Chen, M. Ultrahigh Capacitance of Nanoporous Metal Enhanced Conductive Polymer Pseudocapacitors. *J. Power Sources* **2013**, 225, 304–310.

(602) Lang, X.; Zhang, L.; Fujita, T.; Ding, Y.; Chen, M. Three-Dimensional Bicontinuous Nanoporous Au/Polyaniline Hybrid Films for High-Performance Electrochemical Supercapacitors. *J. Power Sources* **2012**, *197*, 325–329.

(603) Lee, K.-U.; Byun, J. Y.; Shin, H.-J.; Kim, S. H. A High-Performance Supercapacitor Based on Polyaniline-Nanoporous Gold. *J. Alloys Compd.* **2019**, *779*, 74–80.

(604) Zhang, C.; Xiao, J.; Qian, L.; Yuan, S.; Wang, S.; Lei, P. Planar Integration of Flexible Micro-Supercapacitors with Ultrafast Charge and Discharge Based on Interdigital Nanoporous Gold Electrodes on a chip. J. Mater. Chem. A **2016**, *4*, 9502–9510.

(605) Purkait, T.; Singh, G.; Kamboj, N.; Das, M.; Dey, R. S. Allporous Heterostructure of Reduced Graphene Oxide–Polypyrrole– Nanoporous Gold for a Planar Flexible Supercapacitor Showing Outstanding Volumetric Capacitance and Energy Density. *J. Mater. Chem. A* **2018**, *6*, 22858–22869.

(606) Seker, E.; Reed, M.; Begley, M. Nanoporous Gold: Fabrication, Characterization, and Applications. *Materials* **2009**, *2*, 2188–2215.

(607) Wang, H.; Sapi, A.; Thompson, C. M.; Liu, F.; Zherebetskyy, D.; Krier, J. M.; Carl, L. M.; Cai, X.; Wang, L.-W.; Somorjai, G. A. Dramatically Different Kinetics and Mechanism at Solid/Liquid and Solid/Gas Interfaces for Catalytic Isopropanol Oxidation over Size-

Controlled Platinum Nanoparticles. J. Am. Chem. Soc. 2014, 136, 10515-10520.

(608) Clark, J. H.; Macquarrie, D. J. Heterogeneous Catalysis in Liquid Phase Transformations of Importance in the Industrial Preparation of Fine Chemicals. *Org. Process Res. Dev.* **1997**, *1*, 149–162.

(609) Zhang, X.; Ding, Y. Unsupported Nanoporous Gold for Heterogeneous Catalysis. *Catal. Sci. Technol.* **2013**, *3*, 2862.

(610) Yamamoto, Y. Perspectives on Organic Synthesis Using Nanoporous Metal Skeleton Catalysts. *Tetrahedron* **2014**, *70*, 2305–2317.

(611) Jin, T.; Terada, M.; Bao, M.; Yamamoto, Y. Catalytic Performance of Nanoporous Metal Skeleton Catalysts for Molecular Transformations. *ChemSusChem* **2019**, *12*, 2936–2954.

(612) Takale, B. S.; Bao, M.; Yamamoto, Y.; Almansour, A. I.; Arumugam, N.; Kumar, S. R. Applications of Metal Nanopore Catalysts in Organic Synthesis. *Synlett* **2015**, *26*, 2355–2380.

(613) Stratakis, M.; Lykakis, I. N. Nanogold(0)-Catalyzed Addition of Heteroelement σ Linkages to Functional Groups. *Synthesis* **2019**, *51*, 2435–2454.

(614) Steinebrunner, D.; Schnurpfeil, G.; Wichmann, A.; Wöhrle, D.; Wittstock, A. Synergistic Effect in Zinc Phthalocyanine— Nanoporous Gold Hybrid Materials for Enhanced Photocatalytic Oxidations. *Catalysts* **2019**, *9*, 555.

(615) Ciesielski, P. N.; Scott, A. M.; Faulkner, C. J.; Berron, B. J.; Cliffel, D. E.; Jennings, G. K. Functionalized Nanoporous Gold Leaf Electrode Films for the Immobilization of Photosystem I. *ACS Nano* **2008**, *2*, 2465–2472.

(616) Corberán, V. C.; González-Pérez, M. E.; Martínez-González, S.; Gómez-Avilés, A. Green Oxidation of Fatty Alcohols: Challenges and Opportunities. *Appl. Catal., A* **2014**, 474, 211–223.

(617) Tanaka, S.; Minato, T.; Ito, E.; Hara, M.; Kim, Y.; Yamamoto, Y.; Asao, N. Selective Aerobic Oxidation of Methanol in the Coexistence of Amines by Nanoporous Gold Catalysts: Highly Efficient Synthesis of Formamides. *Chem.—Eur. J.* **2013**, *19*, 11832–11836.

(618) Yudha S, S.; Kusuma, I.; Asao, N. Aerobic Oxidation of Hydroxylamines with Nanoporous Gold Catalyst as an Efficient Synthetic Method of Nitrones. *Tetrahedron* **2015**, *71*, 6459–6462.

(619) Alipour Najmi, A.; Jafariyeh-Yazdi, E.; Hadian, M.; Hermans, J.; Bischoff, R.; Yue, J.; Dömling, A.; Wittstock, A.; Permentier, H. P. Nanoporous Gold Catalyst for the Oxidative N-Dealkylation of Drug Molecules: A Method for Synthesis of N-Dealkylated Metabolites. *ChemMedChem.* **2022**, *17*, No. e202200040.

(620) Kavthe, R. D.; Ishikawa, Y.; Kusuma, I.; Asao, N. Chemoselective Aerobic Cross-Dehydrogenative Coupling of Terminal Alkynes with Hydrosilanes by a Nanoporous Gold Catalyst. *Chem.—Eur. J.* **2018**, *24*, 15777–15780.

(621) Zhang, J.-S.; Liu, L.; Chen, T.; Han, L.-B. Cross-Dehydrogenative Alkynylation: A Powerful Tool for the Synthesis of Internal Alkynes. *ChemSusChem* **2020**, *13*, 4776–4794.

(622) Dou, J.; Tang, Y.; Nguyen, L.; Tong, X.; Thapa, P. S.; Tao, F. F. Oxidation of Cyclohexene Catalyzed by Nanoporous Au(Ag) in Liquid Phase. *Catal. Lett.* **201**7, 147, 442–452.

(623) Asao, N.; Ishikawa, Y.; Hatakeyama, N.; Menggenbateer; Yamamoto, Y.; Chen, M.; Zhang, W.; Inoue, A. Nanostructured Materials as Catalysts: Nanoporous-Gold-Catalyzed Oxidation of Organosilanes with Water. *Angew. Chem., Int. Ed.* **2010**, *49*, 10093–10095.

(624) Li, H.; Guo, H.; Li, Z.; Wu, C.; Li, J.; Zhao, C.; Guo, S.; Ding, Y.; He, W.; Li, Y. Silylation Reactions on Nanoporous Gold via Homolytic Si-H Activation of Silanes. *Chem. Sci.* **2018**, *9*, 4808–4813. (625) Takale, B. S.; Wang, S.; Zhang, X.; Feng, X.; Yu, X.; Jin, T.; Bao, M.; Yamamoto, Y. Chemoselective Reduction of α,β -Unsaturated Aldehydes Using an Unsupported Nanoporous Gold Catalyst. *Chem. Commun.* **2014**, *50*, 14401–14404.

(626) Yan, M.; Jin, T.; Ishikawa, Y.; Minato, T.; Fujita, T.; Chen, L.-Y.; Bao, M.; Asao, N.; Chen, M.-W.; Yamamoto, Y. Nanoporous Gold Catalyst for Highly Selective Semihydrogenation of Alkynes: Remarkable Effect of Amine Additives. J. Am. Chem. Soc. 2012, 134, 17536–17542.

(627) Wagh, Y. S.; Asao, N. Selective Transfer Semihydrogenation of Alkynes with Nanoporous Gold Catalysts. *J. Org. Chem.* **2015**, *80*, 847–851.

(628) Yan, M.; Jin, T.; Chen, Q.; Ho, H. E.; Fujita, T.; Chen, L.-Y.; Bao, M.; Chen, M.-W.; Asao, N.; Yamamoto, Y. Unsupported Nanoporous Gold Catalyst for Highly Selective Hydrogenation of Quinolines. *Org. Lett.* **2013**, *15*, 1484–1487.

(629) Zhao, Y.; Zhang, S.; Yamamoto, Y.; Bao, M.; Jin, T.; Terada, M. Heterogeneous Catalytic Reduction of Tertiary Amides with Hydrosilanes Using Unsupported Nanoporous Gold Catalyst. *Adv. Synth. Catal.* **2019**, *361*, 4817–4824.

(630) Takale, B. S.; Tao, S. M.; Yu, X. Q.; Feng, X. J.; Jin, T.; Bao, M.; Yamamoto, Y. Exclusive Chemoselective Reduction of Imines in the Coexistence of Aldehydes Using AuNPore Catalyst. *Org. Lett.* **2014**, *16*, 2558–2561.

(631) Xiao, Z.; Yang, H.; Yin, S.; Zhang, J.; Yang, Z.; Yuan, K.; Ding, Y. Electrochemical Reduction of Functionalized Carbonyl Compounds: Enhanced Reactivity over Tailored Nanoporous Gold. *Nanoscale* **2020**, *12*, 4314–4319.

(632) Takale, B. S.; Feng, X.; Lu, Y.; Bao, M.; Jin, T.; Minato, T.; Yamamoto, Y. Unsupported Nanoporous Gold Catalyst for Chemoselective Hydrogenation Reactions under Low Pressure: Effect of Residual Silver on the Reaction. *J. Am. Chem. Soc.* **2016**, *138*, 10356– 10364.

(633) Zhao, Y.; Feng, X.; Zhang, S.; Yamamoto, Y.; Bao, M. Hydrodebromination of Aromatic Bromides Catalyzed by Unsupported Nanoporous Gold: Heterolytic Cleavage of Hydrogen Molecule. *ChemCatChem.* **2020**, *12*, 4951–4957.

(634) Della Pina, C.; Falletta, E.; Rossi, M. Selective Oxidation of Tertiary Amines on Gold Catalysts. *Top. Catal.* **2007**, *44*, 325–329.

(635) Ishikawa, Y.; Yamamoto, Y.; Asao, N. Selective Hydrosilylation of Alkynes with a Nanoporous Gold Catalyst. *Catal. Sci. Technol.* **2013**, *3*, 2902.

(636) Chen, Q.; Zhao, J.; Ishikawa, Y.; Asao, N.; Yamamoto, Y.; Jin, T. Remarkable Catalytic Property of Nanoporous Gold on Activation of Diborons for Direct Diboration of Alkynes. *Org. Lett.* **2013**, *15*, 5766–5769.

(637) Chen, Q.; Zhang, X.; Su, S.; Xu, Z.; Li, N.; Li, Y.; Zhou, H.; Bao, M.; Yamamoto, Y.; Jin, T. Nanoporous Gold-Catalyzed Diboration of Methylenecyclopropanes via a Distal Bond Cleavage. *ACS Catal.* **2018**, *8*, 5901–5906.

(638) Silva Olaya, A. R. Ph.D. Thesis. University of Oldenburg, 2022.